### THREE DIMENSIONAL ULTIMATE STRENGTH ANALYSIS OF BEAM - COLUMNS

ΒY

#### MOHAMED A. EL-KHENFAS

B.Sc. (LIBYA) 1974 M.Sc. USC (U.S.A.) 1980

#### Thesis submitted

#### for the

degree of Doctor of Philosophy at the University of Sheffield

September 1987



### **IMAGING SERVICES NORTH**

Boston Spa, Wetherby West Yorkshire, LS23 7BQ www.bl.uk

# BEST COPY AVAILABLE.

# VARIABLE PRINT QUALITY

## DEDICATED TO MY FAMILY

.

.

•

A rigorous formulation for the structural response of thinwalled members of arbitrary open cross-section acted upon by a general system of loads is developed based on energy principles and virtual work concepts. Full account is taken of three dimensional behaviour, including sectorial warping effects. The analysis incorporates the effect of initial geometrical deflections. Different patterns of residual stress, non-coincidence of the shear centre and centroid, a complete absence of symmetry in the section and the influence of higher order terms in the strain-displacement relationships including products of the derivatives of axial displacements are also incorporated.

A computer program based on finite element analysis suitable for application in both the elastic and inelastic ranges is developed. This is used to solve the differential equations governing the ultimate strength of beam-columns in space.

The program is written in the Fortran 77 Language. The main function of the program is to follow the loss of stiffness due to spread of yield and hence to trace the full load-deflection response up to collapse. It may be used in a wide variety of ways.

Three types of analysis have been conducted in this study. These are: Linear, Partial Non-linear and Full Non-linear. The Linear involves only the small deflection theory. Partial Non-linear analysis uses non-linear strains while the Full Non-linear analysis incorporates both non-linear strains and nonlinear stiffness matrices. Several illustrative examples, previously investigated either theoretically or experimentally, have been chosen to check the validity of both the

- i -

analytical approach and the computer program. These examples cover flexural, flexural-torsional, biaxial bending, and bending and torsional behaviour in the elastic and inelastic ranges. They contain a wide range of parameters e.g. different cross-section shapes, loading, boundary conditions and initial imperfections. Finally the program has been used to study the ultimate strength of steel members subjected to compression, bending and torsion in a more rigorous fashion than has previously been possible.

#### Acknowledgement

The work presented in this thesis has been supported by Scholarships made available by University of Al-Fateh, Tripoli, Libya (S.P.L.A.J).

The Author would like to express his sincere gratitude to Dr. D.A. Nethercot for his supervision, valuable guidance, encouragement and general interest throughout this study. The candidate also wishes to thank Professor T.H. Hanna (Head of the Department) and all his academic staff for their support.

The candidate is grateful for the facilities made available by the Department of Civil and Structural Engineering and the Computer Centre at the University of Sheffield.

Finally, the candidate wishes to express his deepest appreciation to his family for their support, encouragement and understanding throughout the course of his work. In particular, the candidate's parents have been a continued source of financial and moral support, understanding and patience through his many years of academic study.

- El-Khenfas, M.A. and Nethercoth, D.A., (1987), "Ultimate Strength Analysis of Steel Beam-Columns Subjected to Biaxial Bending and Torsion", Applied Solid Mechanics -2 Conference, University of Strathclyde, Glasgow, U.K., April 7/8.
- 2. El-Khenfas, M.A., and Nethercot, D.A., (1987a), " A General Formulation for Three Dimensional Analysis of Beam-columns", International Journal of Mechanical Sciences, (submitted for publication).
- 3. El-Khenfas, M.A. and Nethercot, D.A., (1987b), "Elastic Analysis of Beam-Columns in Space", International Journal of Mechanical Sciences, (submitted for publication).
- 4. El-Khenfas, M.A. and Nethercot, D.A., (1987c)"Inelastic Analysis of Beam-Columns in Space", International Journal of Mechanical Sciences, (submitted for publication).
- 5. Wang, Y.C., El-Khenfas, M.A., And Nethercot, D.A. (1987), "Lateral- Torsional Buckling of Ben-Restrained Beams", Journal of Construction Steel Research, (in press).
- El-Khenfas, M.A., (1987a), "TDFE-Computer Program for Analysis of Beams in Space in the Elastic and Inelastic ranges", Report, No. 1CE, Civil and Structural Engineering Department, Sheffield University.
- El-Khenfas, M.A., (1987b), "Elastic Analysis of Beam-Columns in Space", Report, No. 2CE, Civil and Structural engineering department, Sheffield University.
- El-Khenfas, M.A., (1987c), "Inelastic Analysis of Beam-Column in 3-Dimensions", Report, No. 3CE, Civil and Structural Engineering Department, Sheffield University.

-iii' -

#### CONTENTS

1

	Page	No.
Summary		i
Acknowledgments		iii
List of Publications		iii
Contents		iv
List of Tables		xii
List of Figures		xiv
Notation		
CHAPTER 1 General Introduction		
1.1 Introduction.		1
1.2 Domain of the Study		2
1.3 Outline of the Thesis		2
		5
CHAPTER 2 Review of Previous Work		
2.1 Introduction		6
2.2 Review of Previous Work		6
2.2 1 Historical & General		6
$2 \cdot 2 \cdot 7$ misconical a concrat		12
2.2.2 Liabut Denaviour 2.2.2.1 Flowwool & Lataral-Torsional Bucklin	- ·	13
2.2.2.1 Flexural & Lateral-ID(Sional Duckiin	5	15
		15
2.2.2.3 Bending and forsion		17
2.2.3 Inelastic Benaviour		17
2.2.3.1 Flexural & Lateral-Torsional Bucklin	g	17
2.2.3.2 Biaxial Bending		20
2.2.3.3 Bending and Torsion	é	21
2.2.4 Experimental Development	ļ	21
2.2.5 Finite Element Development		22
CHAPTER 3 General Formulation of Beam-Column Analysis		
in Three Dimensions		
3.1 Introduction		34
3.2 Assumptions	-	35
3.3 Theoretical Analysis		36
3.3.1 Kinematics of the Cross-Section		36
3.3.2 Stress-Strain Relationships		38
3.3.3 Strain-Displacement Relationships		39
3.3.4 Method of Analysis	, i	45
3.3.4.1 Principle of Virtual Work	1	45
3.3.4.2 Total Potential Energy	1	47
3.4 Imperfections	1	48
3.5 Equilibrium Equations	1	53
3.5.1 Virtual Work		53
3.5.2 Potential Energy		58
3.6 Comparison with Previous Formulations		72
3.7 Conclusions		74
		•
CHAPTER 4 Derivation of Stiffness Matrices		
4.1 Introduction	;	82
4.2 Interpolation Functions		83
4.2 Stiffness Matrices		85
4.3.1 Principle of Virtual Work		86
4.3.1.1 Linear Tangential Stiffness Matrix		86
J Prusal rankenorar portruess habilt		

— iv —

	4.4 4.5	<ul> <li>4.3.1.2 Linear Geometrical Stiffness Matrix</li> <li>4.3.1.3 Nonlinear Tangential Stiffness Matrix</li> <li>4.3.1.4 Nonlinear Geometrical Matrix</li> <li>4.3.2 By Total Potential Energy</li> <li>4.3.2.1 Linear Tangential Stiffness Matrix</li> <li>4.3.2.2 Linear Geometrical Stiffness Matrix</li> <li>4.3.2.3 Nonlinear Tangent Stiffness Matrix</li> <li>4.3.2.4 Nonlinear Geometrical Stiffness Matrix</li> <li>4.3.2.5 Initial Stiffness Matrix</li> <li>4.3.2.6 Assembly</li> <li>Transformation Matrix</li> </ul>	87 88 99 91 97 97 98 98 98 98 98
		4.5.1 Linear Strain Matrix	101
		4.5.2 Nonlinear Strain Matrix	104
	4.6	Conclusions	106
CHAPTER	5 Ana	alytical Procedure and Computer Program Structure	
	5.1	Introduction	120
	5.2	Finite Element Method	120
		5.2.1 Number of Elements	121
		5.2.2 Number of Segments in the Cross-Section	121
		5.2.3 Section and Sectorial Properties	122
		5.2.4 Internal Forces	127
	5.3	Assembly of Stiffness Matrices	128
	5.4	Method of Solution	130
	2.2	Convergence Criteria	100
		5.5.1 Out of balance force	121
	F 6	J.J.2 ILERALIVE Frocess	125
	2.0	5 6 1 Computer Program Description	126
	57	S. O. I Computer Program Structure	127
	2.1	Computational Steps	142
	2.0	001010310115	
CHAPTER	6. E1a	astic Analysis of Beam-Columns in Thee-Dimensional	
	6.1	Introduction	151
	6.2	Analysis	152
	6.3	Numerical Solutions	154
		6.3.1 Linear Bending and Torsion	154
		6.3.2 Flexural and Flexural Torsional Buckling	155
		6.3.3 Biaxial Bending	157
		6.3.4 Nonlinear Bending and Torsion	158
		6.3.5 Biaxial Bending and Torsion	159
	6.4	Conclusions	161
CHAPIER	7 1	Inelastic Analysis of Deam-Column in Space	175
	7 2	Assumptions	176
	7 2	Numerical Results	177
	1.0	7.3 1 Column with Initial Deflections	178
		7.3.2 Flexural and Flexural-Torsional Buckling	178
		7.3.3 Biaxial Bending	180
	7.4	Conclusions	181
CHAPTER	8 U1	timate Strength of Beams under Bending and Torsion	
	8.1	Introduction	191
	8.2	Analysis	192

8.3	Numerical Results	193
	8.3.1 Comparison With Previous Analysis	194
	8.3.2 Elastic Analysis	196
	8.3.2.1 Biaxial Bending and Torsion	196
	8.3.3 Inelastic Analysis	198
	8.3.3.1 Torsional Moment Applied at	
	Mid-Span	198
	8.3.3.2 Bending and Torsion	198
	8.3.3.3 Non-Uniform Bending Moment and Torsion	200
	8.3.3.3 Biaxial Bending and Torsion	201
8.4	Examination of the Failure State of Members	
	under Bending and Torsion	202
	8.4.1 Stability of I-Section under Flexural and	
	Torsional Loading	205
	8.4.1.1 Stability of Beam-Column of I-Section	
	under Bending and Torsion Based on	
	Pastor and DeWolf (1979) Suggestions	207
	8.4.1.2 Stability of Beam-Column of Thin	
	Walled Section Subjected to Combined	
	Bending and Torsion Based on Author	
	Suggestions	209
	8.4.1.3 Stability of Beam-Column under	
	Biaxial Bending and Torsion 211	
8.5	Prediction of Stability of Beam-Column under Flexura	1
	Bending and Torsion using a Regression Analysis	212
8.6	General Features of the Analysis	214
8.7	Conclusions	215
CHAPTER O CO	nolugions and Further Work	
9.1	Introduction	244
9.2	General Formulation	244
9,3	Derivation of Stiffness Matrices	245
9.4	Analysis Type Options	246
9.5	Development of Computer Program (TDCP)	246
9.6	Comparison with Previous Work	246
9.7	Ultimate Strength Behaviour of Members under	_
	Bending and Torsion	247
9.8	Future Work	248

#### REFERENCES

#### APPENDICES

,

#### LIST OF TABLES

TABLE	NO.	TITLE	PAGE
Table	3.1	Comparison Between the Author's Formulation and	
		Previous Studies	75
Table	3.2	Comparison of Author's Assumptions against Those	
		of Chen & Atsuta (1977), Rajaskaran (1970), and	
		Rajaskaran & Murray (1973)	76
Table	4.1	Linear Tangential Stiffness Matrix	108
Table	4.2	Linear Geometrical Stiffness Matrix	109
Table	4.3	Non-Linear Tangential Stiffness Matrix	110
Table	4.4	Non-Linear Geometrical Stiffness Matrix	111
Table	4.5	Linear Tangential Stiffness Matrix	112
Table	4.6	Linear Geometrical Stiffness Matrix	113
Table	4.7	Non-Linear Tangential Stiffness Matrix	114
Table	4.8	Non-Linear Geometrical Stiffness Matrix	115
Table	4.9	Initial Geometrical Matrix	116
Table	4.10	Linear strain matrix	117
Table	4.11	Non-Linear strain matrix	118
Table	6.1	Examples Selected for Study	163
Table	6.2	Comparison Between Author's Results and Those	
		by BSCA (1970)	164
Table	6.3	Comparison of Calculated Results with Anderson	
		and Trahair (1972) for Stability of Doubly	
		Symmetric Beams	165
Table	6.4	Comparison of Author's with Those of Anderson	
		and Trahair (1972) for Stability of	

.

- vi i -

Monosymmetric Beams

- Table 6.5 Comparison of Calculated Results with Exact Solution of Culver (1966a), for Biaxial Loading, no Initial Deflection 167
- Table 6.6Comparison of Calculated Results with ExactSolution of Culver (1966a) for Beam-Columnunder Biaxial Bending, no Initial Deflection167
- Table 6.7 Comparison of Calculated Results with Exact Solution of Culver (1966b) for Beam-Column Subjected to Biaxial Bending, Initial Deflection Included 168
- Table 6.8 Comparison of Calculated Results with Exact Solution of Culver (1966b) for Beam-Column Subjected to Biaxial Bending, Initial Deflection Included 168
- Table 6.9 Comparison of Author's Results with Moore (1986) Analysis and Tests for Beams of Z-Sections Subjected to Bending and Torsion 169
- Table 7.1Problems Chosen for Study181
- Table 7.2 Comparison Between Author's Results and Those Obtained by Fujita and Yoshida (1972) for Compressive Column with Initial Deflection 182
- Table 7.3Data for Problems Chosen for Purpose ofComparison with Lindner (1981)183
- Table 7.4Comparison of Selected Tests and AnalysisConducted by Lindner (1981) Against ThoseObtained by Author184
- Table 7.5 Data for Problems Chosen for Comparison,

-- Vin -

166

Anslijn (1983)

Table 7.6	Comparison of Author's Results with Experimental	
	Data of Anslijn (1983)	186
Table 7.7	Comparison of Author's results with Solution of	
	Matthey (1984)	187
Table 8.1	Comparison of Author's Result against Those	
	Obtained by Pastor & DeWolf (1979), for a Beam	
	Subjected to Constant Torsional Moment and	
	Equal End Moments (g=50 ksi,W12x14)	216
Table 8.2	Results Obtained for Three Analyses of	
	Beam-Column under Bending and Torsional Moment	
	at Mid-span	217
Table 8.3	Presentation of Calculation of Streses Caused	
	by Bending and Torsion of Case A	218
Table 8.4	Presentation of Calculation of Streses Caused	
	by Bending and Torsion of Case B	219
Table 8.5	Presentation of Calculation of Streses Caused	
	by Bending and Torsion of Case C	220
Table 8.6	Presentation of Calculation of Streses Caused	
	by Bending and Torsion of Case A	221
Table 8.7	Presentation of Calculation of Streses Caused	
	by Bending and Torsion of Case B	222
Table 8.8	Presentation of Calculation of Streses Caused	
	by Bending and Torsion of Case C	223
Table 8.9	Inetraction of Stresses of the Three cases	
	(A, B, and C)	224
Table 8.10	Ultimate Strength of Beam-column under Biaxial	
	Bending and Torsion	225
,		

— ix—

185

#### LIST OF FIGURES

FIGURE	No.	TITLE	PAGE
Figure	1.1	Relations Between Squash Load $(\overline{P}_y)$ , Euler',	
		Load and Failure Load of Imperfect Comlumns	5
Figure	2.1	Lateral Buckling Strength of Simply Supported	
·		I-Beam (Trahair, 1975)	30
Figure	2.2	Beam-Column Subjected to Biaxial Bending about as	kes 31
Figure	2.3a	Isolated H-Column under Biaxial Bending	32
Figure	2.3b	Decomposition of a Axial Loading (Perkoz and	
		Winter 1966)	32
Figure	2.4a	Idealization of Beams with Geometrical	
		Imperfections (Yoshida and Maegawa 1984)	33
Figure	2.4b	Residual stress Distributions (Yoshida and	
		Maegawa 1984)	33
Figure	3.1	Definition of Problem	77
Figure	3.2	Stress-Strain Relationship	78
Figure	3.3	Deformed and Undeformed Body	79
Figure	3.4	Sectorial Coordinates of Thin-Walled Section	80
Figure	3.5	General Three Dimensional Body	80
Figure	3.6	Typical Patterns of Residual Stress Adjusted to	
		Ensure Full 3-D Equilibrium	81
Figure	4.1	Transformation of Beam Element	119
Figure	5.1	Location of Nodes at Discontinuties	144
Figure	5.2	Number of Elements and Segments in the	
		Cross-Section	144

- x ---

x —

.

Figure 5.3	Cross-Section Illustrating the Process of	
	Evaluating Section Properties	145
Figure 5.4	Sectorial Coordinates of Thin-Walled Section	146
Figure 5.5	Newton-Raphson Method	147
Figure 5.6	Example Showing Number of Elements and Internal	
	and External Forces	148
Figure 5.7	Flow Char of TDCP (El-Khenfas 1987a)	149
Figure 6.1	Comparison Between Author's Results and Roberts	
	and Azizian (1983a) for I-Sections Subjected to	
	Axial Load and Vertical Deflection	170
Figure 6.2	Comparison Between the Results Obtained by The	
	Author and Those of Roberts and Azizian (1983b)	171
Figure 6.3	Comparison of Authors Results with Analysis of	
	Moore(1986) and Test data, Lateral Load Applied	
	Through Shear Centre	172
Figure 6.4	Comparison of Author's Results with Moore (1986)	
	and Vlasov (1961)	173
Figure 6.5	Load - Deflection Curve for Three Types of	
	Analyses of a Simply Supported I-Beam Subjected	
	to Biaxial Bending and Constant Torsion at	
	Mid-Span	174
Figure 7.1	Adjusted Residual Stress Distribution Presented	
	by Fukumoto et al. (1972)	188
Figure 7.2	Comparison Between Author's Results and	
	Experimental of Anslijn(1983) on Beam-Column	Ĺ
	under Biaxial Bending	189
Figure 7.3	Comparison of Authors Results with Anslijn	
	(1983) Tests - in-Plane and out-of-Plane	L

xi

Displacements Test 48

Figure 8.1	Comparison Between the Bending and Warping	
• .	Stresses Obtained by Pastor & DeWolf (1979)	
	Against Those By Author's Approach	226

- Figure 8.2 Comparison Between Total Stress and Rotation of I-Beam under Constant Torsional Moment at Mid-Span and Equal end Moments to Those obtained by Author and Pastor and DeWolf (1979) 227
- Figure 8.3 Interaction Curves of Three Analyses Obtained by Elastic Analysis 228
- Figure 8.4 Comparison Between Three Analyses of a Beam-Column under Biaxial Bending and Initial Deflection (L/1000) 229
- Figure 8.5 Interaction Curves of Rotations and Loads of Beam-Column Subjected to Biaxial Bending and Constant Torsion at Mid-Span 230
- Figure 8.6 Comparison Between the Three Analyses of a Beam-Column under Biaxial Bending and Constant Torsional Moment at Mid-Span 231
- Figure 8.7 Load Rotation Curve of I-Beam Under Torsional Moment at Mid-Span, with or without inclusion of residual stress 132
- Figure 8.8 Spread of Yield for Linear and Nonlinear Analysis of Beam-Column Subjected to Bending and Torsion 233
- Figure 8.9 Interaction Curves of Equal End Moments and Concentrated Torsion Applied at Mid-span 234

Figure 8.10 Interaction Between Bending and Torsion for a

- xii -

190

		Beam-Column Subjected to Uniform or Nonuniform	
		End Moments and Concentrated Torsion	235
Figure	8.11	Typical Spread of Yield at Mid-span of	
		Beam-Column under Unequal end Moments and	
		Torsion, No Residaul Stress Included	236
Figure	8.12	Typical Spread of Yield (Mid-span) of Beam-	
		Column Subjected to Equal Constant End Moments	
		and Torsional Moment at Mid-Span	237
Figure	8.13	Comparison of Authors Results for Linear and	
		Full Non-Linear Analysis of Beam-Column	
		Subjected to Compressive Load with Eccentricity	
		and Constant Torsion at Mid-Span	238
Figure	8.14	Spread of Yield for Example of Fig. 8.13	239
Figure	8.15	Comparison Between Linear and Full Non-Linear	
		Analysis of Beam-Column Under Biaxial Bending	
		and Constant Torsion at Mid-Span	240
Figure	8.16	Typical Spread of Yield of Beam-Column under	
		Biaxial Bending and Torsion	241
Figure	8.17	Torque Diagrams for Pin-Ended Beam (Salmon and	
		Johnson, 1980)	242
Figure	8.18	Interaction Curves for Beam-Column of I-Beam	
		Section (W12x14) under Flexural Bending and	
		Torsion at Mid-Span	243

-xiii-

•

#### Notation

ACross-sectional area
$$A_{fy}$$
 $M_y/M_{yy}+M_z/M_{yz}+M_t/M_{y\omega}$  $A_{fp}$  $M_y/M_{py}+M_z/M_{pz}+M_t/M_{p\omega}$ AA $P/P_{pl} + M_y/M_{py}+M_z/M_{pz}+M_t/M_{p\omega}$  $a_y, a_z$ Distance of transverse load (concentrated or distributed) below or above shear centreBFlange width (I-section)Ccentroid of the section

$$c_1 \qquad \qquad (\frac{M_y}{M_{yy}})^2 + (\frac{M_z}{M_{yz}})^2 + (\frac{M_t}{M_{y\omega}})^2$$

c2

 $\left(\frac{M_{y}}{M_{py}}\right)^{2} + \left(\frac{M_{z}}{M_{pz}}\right)^{2} + \left(\frac{M_{t}}{M_{pw}}\right)^{2}$ 

D Beam depth E Elastic modulus. E<sub>sh</sub> Strain hardening modulus E<sub>t</sub> Tangent modulus e<sub>y</sub>, e<sub>z</sub> Eccentricities of axial load

Concentrated force
Internal axial load
Surface traction force
Body force
Elastic shear modulus
Strain hardening shear modulus
Tangent shear modulus
D-T <sub>f</sub>
Polar moment of inertia
Moment of inertia about Y and Z axes
Product moment of inertia about Y and
Z axes
Warping product moment of inertia
about Y and Z axes
Warping moment of inertia.
Torsional constant
Beam length
Beam segment length

L

.

 $M_{cr} \qquad \qquad \frac{\Pi}{L} \underbrace{\text{EI}_{y}\text{GK}}_{y} \underbrace{1 - \frac{\Pi^{2}\text{EI}_{\omega}}{L^{2}\text{GK}}}$ 

 $\begin{array}{c} M_{p} & Plastic moment \\ M_{p} & \int_{A}^{f} \sigma_{p}^{2} dA \\ \overline{M}_{y} & \sigma_{y} \overline{Z} \end{array}$ 

- xv -

Mt	Applied constant torsional moment
$M_{xi}^{L}, M_{yi}^{L}, M_{zi}^{L}$	Internal moments to the left of node
	i about X, Y, and Z axes
$M_{xi}^{R}, M_{yi}^{R}, M_{zi}^{R}$	Internal moments to the right of node
	i about X, Y, and Z axes
<sup>m</sup> xi <sup>, m</sup> yi <sup>, m</sup> xi	Applied moments to node i about
	X, Y, and Z axes
M <sup>R</sup> wi, M <sup>R</sup> wi	Internal moments to the right of node
	to the right of node i
NEL	Number of elements
NELS	Number of elements in the cross-section
P	Internal axial load
Pt	Total force array
Py	σy <sup>A</sup>
Q <sub>y</sub> , Q <sub>z</sub>	Shear resultant
q <sub>x</sub> , q <sub>y</sub> , q <sub>z</sub>	Transverse loads applied about X,
	Y, and Z axes
R	Total displacements array
r <sub>E</sub>	Element displacement
r, r	Projection of $\rho$ on the
٢	tangent and perpendicular to the tangent
	at point A on the contour, Fig. 3.2a
r <sub>y</sub> , r <sub>z</sub>	Radius of gyration about Y and Z axes
s <sub>c</sub>	shear centre coordinate

- xvi -

•

.

.

-

<sup>S</sup> y' <sup>S</sup> Z	Plastic section modulii about
	Y and Z axes
T <sub>f</sub>	Flange thickness
tω	Web thickness
U, V, W	Displacements in the X, Y, and Z directions
<sup>U</sup> o, Vo, <sup>©</sup> o	Initial displacements
$U_{XZ} = \frac{\partial^2 U}{\partial X \partial z}$	Partial differentiation of U
·· · · · · · · · · · · · · · · · · · ·	with respect to y and z
Ū	Strain energy
ر <b>۷</b>	Potential energy due to applied load
Ŷ, Z	First moment of area about Y and Z axes
Y <sub>s</sub> , Z <sub>s</sub>	Coordinates of point s
<sup>z</sup> <sub>y</sub> , <sup>z</sup> <sub>z</sub>	ELastic section modulii about
	Y and Z axes
α	Angle between the tangent at the
	contour and vertical axis, Fig. 3.2a
β β β y'z' ω	Properties of cross-section defined below

 $\int_{A} Y(Y^2 + Z^2) dA - 2Z_s$ 

 $\int Z(Y^2 + Z^2) dA = 2Y_s$ 

β<sub>ω</sub>

ß У

β z

 $\int_{A}^{\omega} (Y^{2} + Z^{2}) dA$ 

- xvil -

$$\frac{1}{\beta_y} \frac{1}{y_A} \int Y(Y^2 + Z_{..}^2) dA$$

$$\frac{1}{\beta_z} \qquad \frac{1}{I_z} \int_A^Z Z(Y^2 + Z^2) dA$$

$$\overline{\beta}_{w} \qquad \frac{1}{I_{w}} \int_{A}^{w} (Y^{2} + Z^{2}) dA$$

$$Y \qquad Tolerance error$$

$$\varepsilon_{0} \qquad Axial strain$$

$$\varepsilon_{r} \qquad Residual strain$$

$$\varepsilon_{y} \qquad Yield strain$$

$$Linear strain$$

$$\Sigma_{0}^{L} \qquad Linear strain$$

$$\theta_{u}, \theta_{v} \qquad Slope of U and V with respect X axis$$

$$\sigma_{by}, \sigma_{bz} \qquad Bending stress about Y and Z axes$$

$$\sigma_{f} \qquad Flange tip residual stress$$

$$\sigma_{f} \qquad Residual stress$$

$$\sigma_{r} \qquad Residual stress$$

$$\sigma_{t} \qquad Total stress$$

.

.

	σ y	Yield stress
	σω	Warping residual stress
	$\overline{\sigma}_{\omega}$	Web residual stress
	ρ	Distance between a point on the cross
		-section and shear centre, Fig. 3.2a
	τ τ xz'yz	Shear stresses.
	Ф Ү У Z	Major and minor curvatures.
	Φω	Warping curvature.
	Ф с	Twisting calculated according to
		equation 8.7
	Ф <b>f.</b> e.	Twisting calculated according to
		finite element analysis
	or nj	Normalized Warping fuction.
	Π.'	Total potential energy
	<sup>o</sup> u <sub>i</sub>	Virtual displacements
	<sup>6</sup> w	Virtual work
	$\diamond$	Row
	{}	Column
	[]	Matrix
	[]	Transpose matrix
	[K <sub>D</sub> ]	Diagonal matrix
	[K <sup>L</sup> <sub>E</sub> ], [K <sup>NL</sup> ]	Linear and nonlinear tangential matrix
	[K <sup>L</sup> ], [K <sup>NL</sup> ] <b>G</b>	Linear and nonlinear gemetrical matrix
- '		

-xix -

•

[K <sub>0</sub> ]	Initial geometrical stiffness matrix
[mmnj] MNJ]	Coefficient matrices as defined i
	equation 4.7
[ K <sub>t</sub> ]	Total stiffness matrix
[K <sub>ww</sub> ]	Axial stiffness matrix.
[K <sub>uu</sub> ]	Transverse stiffness matrix about Z-Z
[K <sub>vv</sub> ]	Transverse stiffness matrix about Y-Y
[K <sub>uv</sub> ]	Coupling stiffness matrix about Z-Y
[K <sub>wv</sub> ]	Coupling stiffness matrix about W-Y
[K <sub>wu</sub> ]	Coupling stiffness matrix about W-Z

#### Chapter 1

#### Introduction

#### 1.1 General

The order of complexity of the response of a structural member in three-dimensions depends on a number of factors. These include the nature of the loading, material properties, and the kind of assumptions made in deriving the governing equations. The most general type of member, which combines both axial and flexural loading, is generally termed a 'beam-column'.

The behaviour of beam-columns depends on their slenderness and the load conditions as shown in Fig. 1.1. Failure can occur in either the elastic or inelastic range, in the form of flexural or flexural-torsional buckling, or, more generally, biaxial bending. When a member is bent about its weaker principal axis, or when it is prevented from deflecting laterally while being bent about its stronger principal axis, then only an in-plane flexural response is possible. Flexural-torsional buckling occurs when a member is bent about its stronger axis but is not restrained laterally so that it may buckle out of the plane of bending by deflecting laterally and twisting. If the member is bent about both axes and twisted it will respond in a full three dimensional manner.

- 1 -

#### 1.2 Aim of the study

The purpose of this study is to provide a general formulation suitable for many kinds of cross section such as channel, tee, L, Z, U, I, (mono or doubly symmetric), etc. for structural members acted upon by any form of loading and provided with very general support conditions. The validity and accuracy of this approach is demonstrated by several illustrative examples, the results of which are compared with those obtained previously from either theoretical or experimental investigations.

The domain of this study is summarized in the following:

- 1- Developing a theoretical analysis for a beam-column having an arbitrary open cross section, which is applicable to both elastic and inelastic analysis.
- 2- Derivation of linear and nonlinear tangential and geometrical stiffness matrices and strain-displacement matrices for a beam-column in space suitable for many kinds of cross-section subjected to a wide range of loading.
- 3- Developing a general computer program based on finite element analysis. This program is capable of implementing the formulation to provide numerical solutions.
- 4- Checking the validity and accuracy of both the derived equations and the computer program by comparing the results against those previously obtained by experimental and theoretical considerations.
- 5- Investigating problems of bending and torsion not previously fully solved in both the elastic and the inelastic ranges.

- 2 -

#### 1.3 Out-line of the thesis

This thesis contains nine chapters setting out the formulation and implementation of an ultimate strength study of the behaviour of steel beam-columns.

Chapter 1 provides a general introduction to the problem to be investigated. This is followed in Chapter 2 by a selective review of the previous work (theoretical and experimental), within the general area of the structural response of beam-columns.

Chapter 3 presents a pair of general three dimensional formulations, based on the concept of virtual work or the use of energy principles, in which the influence of higher order terms, the effect of initial imperfections (such as residual stresses, initial crockedness, etc.) have been included. Comparison between previous more restricted formulations and this general one are explained in detail. Chapter 4 presents the full stiffness matrices (linear/ nonlinear tangent and geometric stiffnesses and linear/ nonlinear strain matrix together with the interpolation functions and the transformation matrix) required for the implementation of this approach.

Chapter 5 describes both the analytical procedure and the computer program structure. The analytical process is used to generate the section and sectorial properties, internal forces, curvatures, tracing spread of yield through the entire cross-section, etc. The program TDCP (Three Dimensional Computer Program) is based on finite element computer concepts, is written in the Fortran-77 Language, contains a wide variety of options, (in-plane ,out of plane, uniform loads ,distributed loads, initial geometrical imperfections, different boundary conditions, etc.) and may be used to investigate the elastic and inelastic behaviour of members of thin-walled open cross-section,

- 3 -

under different load and support arrangements.

Chapters 6 and 7 contain comparisons between the results of this program and those derived previously by other techniques. These illustrate the advantage of both the modified formulation and the more advanced computer program which is capable of correctly accounting for factors such as, absence of symmetry, any form of loading, any pattern of residual stress, any set of initial deformations as well as varying degrees of sophistication in the assumed strain-displacement relations and general geometrical aspects of the problem.

In chapter 8, new problems involving the determination of ultimate strength under bending and torsion are presented in both the elastic and inelastic ranges for beam-columns of I-section.

Chapter 9, presents general conclusions and makes some suggestions for further work.



FIG. 1.1 RELATIONS BETWEEN SQUASH LOAD (P<sub>y</sub>), EULER'S LOAD AND FAILURE LOAD .OF IMPERFCT COLUMNS

#### Chapter 2

#### Review of Previous Work

#### 2.1 Introduction

The behaviour of thin-walled beams and beam-columns of open cross-section is a subject of importance to those concerned with the design of metallic structures. This was initially due to the growth of their use in aircraft followed by increases in their use as members in civil engineering structures. A great deal of research -both theoretical and experimental- has been carried out to provide comprehensive data on which safe and economical designs can be based.

In this thesis a general three-dimensional formulation for the structural response of steel beam-columns having an arbitrary open cross-section is derived taking into account almost all the known factors affecting their behaviour, such as initial crockedness, different patterns of residual stresses, and a wide variety of loads and boundary conditions. The ultimate strength behaviour of beamcolumns in both the elastic and inelastic ranges has been investigated by using a computer program based on the resulting finite element analysis. Before undertaking this contribution a review of previous work was conducted.

#### 2.2 Review of Previous Work

#### 2.2.1 Historical & General

The behaviour of a beam-column depends principally on its

- 6 -

slenderness ratio. Failure of slender members is often governed by buckling in the elastic range, for which the effects of small geometric imperfections are comparatively unimportant, and the presence of residual stress irrelevant. Thus elastic critical loads may be used to approximate the ultimate strength. However, members of intermediate slenderness normally fail by inelastic buckling, for which the residual stresses are important because they cause early yielding within the cross-section, resulting in a reduction in the effective stiffness of the member, and a lowering of the resistance to buckling. Stocky beamcolumns may attain loads determined largely from material strength considerations. Fig. 2.1 represents a typical moment-slenderness relationship for beams. It relates to both idealised members with no initial deflections or residual strains, and a real member for which the effects of these imperfections have been incorporated; failure may be either elastic or inelastic.

A great deal of research work (experimental and theoretical) has been conducted into the behaviour of steel members of thin-walled cross-section subjected to a variety of loading conditions. Some of these studies have included the effects of initial geometrical imperfections. Investigations have covered both the elastic and the inelastic ranges and have considered both two and three dimensional response.

When a laterally unsupported member is subjected to biaxial bending, it will usually deflect in both principal planes and twist at any load level as illustrated in Fig. 2.2. The importance of twisting lies in the fact that the ultimate load carrying capacity of an open cross-section, for which the torsional rigidity is small, will be crucially affected by the torsional aspect of the deformations.

The review given below covers only a selection of contributions to the general subject area, concentrating on some of the more significant developments.

Timoshenko (1910) developed the fundamental differential equations for flexure and torsion of doubly symmetric simply supported I-beams. He solved them for the elastic critical loads by using energy theorems. A particular study was made of the effect of the point of load application when it was remote from the shear centre axis. Wagner (1929) studied the torsional buckling of a thin-walled column; Kappus (1937) modified Wagner's equation and generalized it to deal with any thin-walled open cross section. Bleich (1933,1936) derived an equilibrium equation for a member subjected to axial compression and equal end moments having an I-cross section; in his study the critical loads for flexural-torsional buckling were determined.

A number of investigations have been carried out to determine the critical loads for I-beams subjected to a variety of different load cases. Tabulated results are given by Clark and Hill (1960), Timoshenko and Gere (1961), Vlasov (1961), Galambos (1968), Nethercot and Rockey (1971), and Nethercot (1972). Winter (1941) derived an approximate formula to determine the buckling loads of monosymmetric I-sections under equal end moments. Other load cases have been considered by Petterson (1952), Vlasov (1961), and Anderson and Trahair (1972).

General design methods based on the extensive research on the elastic flexural-torsional buckling of beams have been proposed by Clark and Hill (1960), Trahair (1966), Nethercot and Rockey (1971), and SSRC (1976).

Pekoz and Winter (1966) have noted that the twisting of a beam-column subjected to axial load with eccentricities  $e_{a}$  and  $e_{a}$ , can

λĮ

- 8 **-**

be explained by considering the physical model illustrated in Fig. 2.3, in which the applied load can be decomposed into four components viz:

1- equivalent to pure axial load P.

2- equivalent to bending moment about weak axis.

3- equivalent to bending moment about strong axis.

4- equivalent to bimoment which causes the bar to warp. (this system produces zero axial load and bending moments on the section.)

A survey of the elastic-plastic behaviour of columns of thinwalled section under biaxial loading has been prepared by Chen and Santathadaporn (1968). A complete account of the developments in this area must necessarily deal with two important aspects: solutions emerging from analytical studies and experimental results obtained by laboratory testing.

Experimental and theoretical investigations have been conducted by Black (1967a) who studied the non-linear elastic behaviour of an unsymmetrical thin-walled beam of open section. The theoretical relationships developed include the effect of torque components arising from the displacements of the beam axes and the point of application of the loads. Experiments were carried out to test the validity of this formulation and he reported that good agreement was obtained

SSRC (1976) reported research results obtained from several theoretical and experimental investigations on beam-columns, including members under uniaxial or biaxial bending, or transverse loads, or combinations of loads. These covered:

1- the evaluation of critical loads in perfect members.

2- the behaviour of beams with initial imperfections.

3- the importance of residual stresses in steel members.

- 9 -

- 4- developments relating the strain hardening modulus of steel to inelastic buckling behaviour.
- 5- studies of the combination of the effects of a)initial residual stress, b) geometrical shape imperfections and/ or uncertainties of load location.

A full review of research on beam-columns in steel structures conducted during the last forty years was carried out by Massonnet (1976). This covers the behaviour of beams and beam-columns in the elastic and inelastic range subjected to different load patterns. Since that date extensive additional studies (theoretical and experimental) on the behaviour and design of beam-column have been made. In (1976, 1977) Chen and Atsuta provided a two volume text on the behaviour of beam-columns in two and three dimensions; Volume 1 helps the reader to develop an understanding of in-plane behaviour, while the second volume provides a comprehensive source of information on biaxially loaded beam-columns as well as an explanation of their space behaviour under various load conditions. The two volumes taken together comprise the first single reference book to discuss the complete theory of beamcolumns systematically from the most elementary to the most advanced stage of development. They also covered some publications which provide background and design rules for beam-columns, specifically:

- 1- "Guide to Stability Design Criteria for Metal Structures", by SSRC.
- 2- "Stability of Steel Structures" by ECCS,
- 3- "Handbook of Structural Stability" by CRC of Japan,

Chen (1977) has provided a review of the theory and design rules for beam-columns under different load patterns and boundary conditions. The basic theoretical principles and methods of analysis in

- 10 -

two and three dimensions in the elastic and inelastic ranges have been included together with an assessment of the validity of the proposed interaction approach to the design of biaxially loaded members. Vinnakota (1977) has derived governing differential equations for an arbitrary open cross-section, without making use of the notions of centre of gravity, principal axes and shear centre. The finite difference method has been used to solve these equations for a number of problems having different load conditions.

Chen and Cheong-Siat-Moy (1980) have presented a review of the philosophy behind the various interaction formulas for a beamcolumn that have been proposed and are under consideration by various specification writing bodies such as the American Institute of Steel Construction. The general validity of these proposed interaction formulas has been demonstrated by comparison of computed loads with test results.

A survey of recent achievements in the analysis (experimental and numerical solutions) and design of steel members in the USA has been produced by Chen (1981), who investigated the behaviour of an isolated beam-column under the influence of the initial deflections, residual stresses, and various loading and boundary conditions. His proposed interaction formulas have been checked against both computed loads and the available test results.

Kennedy and Madugula (1982) made a comprehensive review of both theoretical and experimental work on the buckling of angles, covering single or built-up angles, equal-leg or unequal-leg angles subjected to axial (either concentric or eccentric) load, transverse load, or a combination of loads. Through their study they found that, depending upon the cross-section, effective length and applied load

- 11 -

configuration, members comprising of angle shapes can fail by any of the following :

- 1- Flexural buckling about the minor axis.
- 2- Torsional buckling about the shear centre.
- 3- Torsional-flexural buckling.
- 4- Local plate buckling.
- 5- Combination of torsional-flexural buckling and local buckling.

Cescotto et al. (1983) developed design rules for determining the buckling strength of beam-columns of monosymmetric sections (Tee and Triangle). The ultimate buckling loads obtained from experiments gave quite satisfactory agreement with numerical simulations for both cases. The suggested design rules appear as a useful complement to the E.C.C.S. Recommendations. Numerical and experimental analyses have been considered by Nakashima, et al.(1983) to investigate the buckling and post buckling behaviour of steel beams having an H-shape, subjected to a constant axial thrust and monotonically increasing end moments.

Interaction equations of beam-columns in the design specification of Western Europe have been investigated by Nethercot (1983). He presented some quantitative evaluation of these proposals. He also provided a tabulated comparison of these interaction formulae. His investigation covered the following aspects:

1- "Uniaxial bending leading to in-plane failure".

2- "Uniaxial bending producing lateral-torsional buckling".

3- "Biaxial bending".

A full review covering the theoretical and experimental analysis in both the elastic and inelastic ranges for columns, beams, and beam-columns covering the years 1744 to 1984 has been presented by Cuk (1984). Nethercot (1986) reviewed comprehensively the lateral

- 12 -
buckling of beams dealing with theoretical approaches in both the elastic and inelastic ranges. It was found that the original theoretical developments could be traced through to the most recent ultimate strength approaches.

This introductory review is intended to provide an indication of both the historical development of the subject and the wide range of research already conducted. In the remainder of this chapter attention will be focussed on specific aspects of the subject.

#### 2.2.2 Elastic Behaviour

#### 2.2.2.1 Flexural & Lateral Torsional Buckling

The elastic flexural and lateral-torsional buckling of beams of different cross-section subjected to a wide variety of loads and boundary conditions have been studied by many investigators.

Anderson and Trahair (1972) presented tabulated results for simply supported monosymmetric I-beams and cantilevers with concentrated and distributed loads, and investigated the influence of load height on the elastic buckling moment. Their results compared very well with test data.

Epstein and Murray (1976) developed a three-dimensional large deflection theory for the analysis of thin walled beams. Numerical examples are presented to illustrate the application of their theory to the solution of elastic torsional post buckling behaviour of I-beams. They reported that their solutions compared well with results obtained from experiments.

Kitipornchai and Trahair (1980) developed a simple method for determining section properties for a wide range of monosymmetric

- 13 -

I-beams, including sections with lipped flanges and also they presented a method to calculate the elastic critical loads of monosymmetric beams. Their rule for the calculated elastic critical load for both monosymmetric and doubly symmetric I-beams has been compared with AS 1250, BS 449, and the AISC specification. More accurate and consistent results were obtained.

A second order differential equation has been derived by Warnick and Walston (1980) using a coordinate system whose orientation remains fixed in space for symmetrical members under different loading conditions. Several examples were examined to investigate the lateral buckling of I-beams. Results of their method were comparable with test data.

The behaviour of nonprismatic structural members (simply supported or cantilever) under transverse concentrated loads has been studied by Brown (1981) using the finite difference method to determine the critical loads of simply supported and cantilever beams. He found that the effect of loads placed either below or above the centroid was significant in all types of beams but leads to an increase, with decreasing free end depth for the cantilever.

Cuk (1984) investigated theoretically isolated/ continuous beam-columns subjected to transverse loads and end moments to determine the elastic flexural-torsional buckling. He reported that his results compared favourably with those obtained experimentally.

Kitipornchai et al (1985) proposed an alternative approximation formula to evaluate the elastic lateral buckling of simply supported monosymmetric I-beams under moment gradient. They found three factors affecting the buckling of monosymmetric sections, which were

- 14 -

1- Wagner effect  $\beta_{v}$ .

2- End moment ratio  $\beta$ .

3- Degree of monosymmetry  $\rho$  ( $\rho = I_{2t} / (I_{2t} + I_{2t})$ ).

Their results were compared with those furnished by the design rule, which employed the moment modification factor  $m=1.75+1.03^{\beta}+.3^{\beta^2} \leq 2.56$ . It was found that the application of their approach gave reasonable results for beams of nearly equal flanges, but for higher degrees of monosymmetric (P<.3 and P>.7) unsafe results were obtained when compared with the previous formula.

#### 2.2.2.2 Biaxial Bending

Culver (1966a,b) developed an exact numerical method to solve the differential equations governing biaxial bending and torsion established by Timoshenko and Vlasov (1961). His analysis covers two cases viz :

i- Biaxial bending without initial imperfections

ii- Inclusion of the initial imperfections.

The results of both cases were compared with experimental data; satisfactory agreement was obtained.

A governing differential equation for members of thin-walled section subjected to biaxial bending has been derived by Soltis and Christiano (1972). The effects of large deformations and higher order terms were included. Several illustrative examples have been solved by small and large deflection analysis. The results obtained by these analyses yielded similar results up to 80% of the critical load. For higher loads both out-of-plane displacements and the twist were overestimated by the small deformation approach.

- 15 -

#### 2.2.2.3 Bending and Torsion

A two-volume treatise on the behaviour of beams and girders subjected to transverse loading causing torsion has been prepared by BCSA (1968,1970). The first of these presents general theory and formulae together with graphs used to display solutions, while the latter presents worked examples (beams and griders subjected to bending and torsion) to calculate stresses (normal stress, bending stresses about Y and Z axes, warping stress, and shearing stresses) and deflections at any point along the member and around the cross-section.

A theoretical approach has been developed by Kitipornchai and Trahair (1975) to study the strength behaviour of tapered monosymmetric I-beams of constant depth subjected to bending and torsion. Also they carried out experiments on small-scale aluminium I-beams to confirm the validity of their theory. They reported that excellent agreement between the two analyses was obtained.

Pastor and DeWolf (1979) investigated theoretically the behaviour of wide-flange beams under equal end moments and a constant torque applied at mid-span. They considered only small deflections and ignored the coupling effects. The results obtained for three beams having sections W12x120, W12x36, and W 12x14 respectively under  $M_{\rm cr}$  / 100 at mid-span and monotonically increasing end moments were tabulated. They suggested the design of members subjected to flexural bending and torsion should involve two checks:

1- The total stress should be compared with the yield stress".

2- The applied moments should be compared with critical moments based on lateral-torsional buckling, with safety considerations.

#### 2.2.3 Inelastic

#### 2.2.3.1 Flexural and Lateral Torsional Buckling

Kennedy and Murty (1972) have conducted an experimental investigation aimed at verifying the design approaches for angle and tee struts of the AISC (1969) and CSA S16-1969 (1969) standards covering inelastic flexural, torsional-flexural, and plate buckling. As a result they recommended that design be based on the lowest of the calculated values.

For angle struts

$$\sigma = \frac{5625}{1.67 \left(\frac{B}{t}\right)^2} \text{ if } \frac{B}{t} \gg \frac{75}{\sqrt{\sigma}}$$

$$\sigma = 0.6\sigma \text{ if } \frac{B}{t} \gg \frac{75}{\sqrt{\sigma}}$$

For tee struts

$$\sigma = \frac{16900}{1.6} \left(\frac{D_{\omega}}{t^{\omega}}\right)^2 \quad \text{if} \quad \frac{D_{\omega}}{t_{\omega}} > \frac{130}{\sqrt{\sigma}}$$

$$\sigma = 0.6\sigma$$
 if  $\frac{D_{\omega}}{t_{\omega}} \ge \frac{130}{\sqrt{\sigma}}$ 

The inelastic flexural-torsional buckling of simply supported I-beams under uniform moment and different patterns and magnitudes of residual stresses has been studied theoretically by Trahair and Kitipornchai (1972). They deduced that the changes in the residual stress system led to variations in the yielded regions in the cross section, and consequent variation in the section rigidities. These

- 17 -

variations cause very significant changes in the inelastic critical moment.

Abdel-Sayed and Aglan (1973) studied the lateral torsional buckling of wide flange beam-columns subjected to axial force and equal end moments about the major axis. Initial imperfections were considered together with strain hardening effects. They come out with these general conclusions; the lateral torsional buckling reduces the strength of beam-column in the inelastic range, while the residual stresses have negligible effect on the buckling in the elastic range but a significant effect in the inelastic range.

Nethercot (1973a) presented a theoretical solution for monosymmetric I-beams loaded by equal end moments acting in the plane of the web to cause inelastic flexural-torsional instability. He employed the expression for the critical moments deduced by Galambos (1968), which was;

$$M_{cr} = \frac{\pi}{L} EI_{z}GK(1 + \frac{M_{\rho}}{GK})[1 + \frac{\pi^{2}EI_{\omega}}{L^{2}GK(1 + \frac{M_{\rho}}{EI_{\omega}})}]$$

in which  $EI_z$  is the minor flexural rigidity; GK is the torsional rigidity;  $EI_{\omega}$  is the warping rigidity;  $M_{\rho} = \int \sigma \rho^2 dA$ ; and  $\rho$  is the distance from the shear centre to the point where the stress,  $\sigma$ , acts. His conclusion was that the Galambos equation was valid for inelastic buckling providing the stiffness terms,  $EI_z$ , GK,  $EI_{\omega}$ , and  $M_{\rho}$  are correctly reduced to allow for the presence of yielded material.

Yoshida and Maegawa (1984) examined I-beams subjected to the influence of residual stresses, various loading conditions and geometrical imperfections to determine the lateral-torsional buckling

- 18 -

strength, the load-deformation behaviour and the spread of yielded portions in the beam. They also examined the relation between the ultimate strength and the buckling strength for the four theoretical models of Fig. 2.4, which are:

Model-I A beam with out-of-plane deflection.

- Model-II Straight beam subjected to a concentrated load at the top flange with eccentricity  $e_y$  ( $e_y$  is the distance from centre line of web to the loading point on the top flange).
  - Model-III Beam subjected to a vertical concentrated load  $P_z$  on the top flange and  $P_z$  applied horizontally at the same point, where the ratio  $\frac{P^y}{P}$  is kept constant.
  - Model-IV A beam with out-of-plane displacement under eccentric loading.

Matthey (1984) studied the ultimate strength behaviour of beam-columns of I-section subjected to axial force, and bending moments about the  $\chi$  and  $\gamma$  axes. Residual stresses and initial deflections were included. Those variables were arranged in a systematic fashion to form the framework for a study of more than 2500 cases. For each case the compressive load was applied up to a predetermined limit followed by end moment loading to failure. He used these results to calculate the Performance Factor which is defined as the ratio between the ultimate load for every case obtained from his calculations and that given by the design rules of EC3 (1983), SIA (1979), SIAC (1961), and Chen (1979).

Kitipornchai and Lee (1986a) investigated theoretically the inelastic flexural and flexural-torsional buckling of single-angle, tee and double-angle cross-sections used as simply supported columns subjected to axial load. They found that the flexural buckling mode is

- 19 -

the dominant failure mode for most shapes, except for single unequal angles, for tees, and for double angles whose radii of gyration give  $\frac{r_y}{r_z}$  greater than 1.0. These theoretical results were checked against other analyses and were in reasonable agreement.

#### 2.2.2.3 Biaxial Bending

An approximate formulation has been provided by Syal and Sharma (1971) for the solution of the generalized problem of biaxially loaded columns with equal or unequal load eccentricities. The effects of the residual stresses, cross-section shape B/D, and warping at the ends being either permitted or restrained have been included. They reported that their results match those previously obtained. Linder (1972) conducted a theoretical investigation to determine the ultimate load of columns of bisymmetrical section under biaxial loading. He employed polynomial expressions for the displacements (U, V, and  $\Phi$ ) in order to obtain a general solution. He provided examples to demonstrate his approach, which incorporates different slenderness, eccentricities, and residual stress.

Epstein et al. (1978) extended the work developed by Epstein and Murray (1976) to deal with nontrivial inelastic stability problems for the prediction of the maximum load-carrying capacity of thin-walled beam-columns of open cross-section under biaxial bending. Results for inelastic biaxial bending and for instability of laterally unsupported beams were compared with the experimental results obtained by Birnstiel (1968) and Lee and Galambos (1963) respectively and satisfactory agreement was obtained.

- 20 -

#### Bending and Torsion

Theoretical studies on biaxially loaded thin-walled beamcolumns of open cross-section with and without incorporating torsional effects have been investigated by Razzaq (1974). He undertook some experimental work to verify the validity of the theoretical predictions of his analysis. Twenty beam specimens were tested up to collapse at two different slenderness ratios for the following loading conditions:

1- Subjected to equal end moments about a principal axis.

- 2- Subjected first to concentrated torque at midspan, and subsequently to equal end-moments about minor axis.
- 3- Reverse of case 2 .

It was found that good agreement was obtained between theory and experiment.

Kollbrunner, et al. (1978) have examined theoretically the ultimate strength behaviour of a cantilever member of I-section subjected to bending and warping torsion. They reported that the comparison of the analytical results with those obtained from experiments was good. Kollbrunner, et al. (1979) have carried out theoretical investigations on the elastic-plastic behaviour of thinwalled fixed ended I-beams under bending and torsion. The results compared well with those obtained from experiments in terms of ultimate loads, internal forces, and twisting angles.

#### 2.2.3 Experimental Studies

The behaviour of laterally unsupported angles of equal and unequal leg lengths for a variety of  $\frac{L}{t}$  have been investigated experimentally by Thomas et al. (1972). Uniform moments were applied about an axis parallel to an angle leg. Their conclusion was that the

- 21 -

angle of twist ( $\Phi$ ) causes a reduction in the maximum section stress and has a significant influence on the maximum loads.

Kitipornchai and Trahair (1975) have conducted an experimental investigation of inelastic flexural-torsional buckling of full-scale simply supported I-beams under concentrated loads. The end supports allow for rotation about both minor and major axes, transverse displacements and twisting about the longitudinal axis were restrained. but warping was free. The beams were loaded to failure and all but one failure was inelastic. They found that the effects of residual stresses was not important, and they confirmed that by theoretical predictions, while the geometrical imperfections were more significant in reducing the strength of beams below their theoretical buckling loads. They also reported that their results were consistent when compared against theoretical and test data together with the Australian Code for the elastic and inelastic ranges.

82 experiments have been conduct in Leige (1983) on beamcolumns of I-cross-section (HEA200 and steel grade Fe 360), subjected to biaxially eccentric loading. The following variables were accounted for:

1- The slenderness ratio in both planes  $(\frac{L}{r_v} \text{ and } \frac{L}{r_v})$ .

2- The bending moments applied in both planes.

3- The axial load applied.

4- The moment gradient in both planes.

An extensive survey of the experimental investigations preformed at various institutions on beams and girders which failed by lateral instability has been prepared by Fukumoto and Kubo (1977). A total of 275 tests have been included in their review; 159 for beams and 119 for welded beams and girders. They employed statistical

- 22 -

characteristics (Mean values (m) and mean minus twice standard deviation) in order to have a comparison between test results and those obtained from recommended design formulae.

The variation of inelastic beam capacity with changing moment gradient of simply supported laterally continuous I-beams has been investigated experimentally by Dux and Kitipornchai (1983). Beam slendernesses were chosen such that buckling should occur in the loading range between the first yielding and the attainment of the plastic moment. The points of load application were prevented from moving laterally and twisting. The test results were compared with theoretical predictions. An experimental investigation to determine the inelastic flexural-torsional buckling of continuous beams of I-section in a sub-assemblage of a three dimensional structural framework has been carried out by Cuk (1984). He reported good agreement with the results obtained from his parallel theoretical study.

A series of experiments on tees and angles in compression has been conducted by Wilhoite et al. (1984a,b) to study their behavior and strength. The tested bars were made of high strength, low-alloy steel with improved formability to match the requirements of ASTM-A-718-81, grade 60. They reported that the results obtained for both sections match fairly well the theoretical predictions.

Kitipornchai and Lee (1986b) investigated experimentally the inelastic flexural and flexural-torsional buckling of single-angle, tee and double-angle cross-sections subjected to axial load. They found that flexural buckling is the dominant failure mode for most shapes, except for single unequal angles, for tees, and double angles whose radii of gyration give  $\frac{r_y}{r_z}$  greater than 1.0. Their experimental results were carried out on 54 struts with modified slendernesses ranging from

- 23 -

0.33 to 1.08 for simply supported members prevented from twisting about their longitudinal axis. These experimental results are in reasonable agreement with theoretical predictions.

#### 2.2.4 Finite Element Development

Based on the finite element concept much research has been carried out to obtain formulations for beams and beam-columns in three dimensions, using the equilibrium condition, the virtual work principle or energy principles. Some of these have considered only uniform torsion, whilst others have studied both uniform and nonuniform torsion where the loads applied may be axial, flexural (uniaxial & biaxial) or combinations.

The designation 'finite element concepts', as employed by Barsoum and Gallagher (1970) ,was intended to characterize the formulation of a relationship between the forces and displacements of a single member via simplified assumptions as to the behaviour of the element in terms of stress or displacements. Energy theorems were used to develop the governing differential equation to study the torsion and combined flexural-torsional instability of one dimensional members of a constant cross-section in the elastic range. Their stiffness equations gave excellent agreement (as the number of elements was increased) with existing theory when compared with the following cases:

- 1- Torsional buckling (pure torsion).
- 2- Lateral buckling of simply supported beam under applied moments.
- 3- Lateral buckling of a cantilever beam acted upon by concentrated load at the shear centre. Rajasekaran (1971) presented a finite element analysis, based

- 24 -

on the principle of virtual work, for thin-walled members of open section made from material having a trilinear stress-strain curve. The effects of initial imperfections, residual stresses and different patterns of loads have been included. The validity of this formulation has been demonstrated by Rajasekaran and Murray (1973) for several problems in both the elastic and the inelastic ranges by comparison with existing results.

Nethercot (1973b) presented solutions for the inelastic lateral buckling of I-beams loaded with either uniform or nonuniform moments, with or without the inclusion of the effect of residual stresses obtained by the finite element method. His analysis was applied to several illustrative examples and the results compared with existing theoretical and experimental data, good agreement being obtained.

Epstein and Murray (1976) developed a formulation for the analysis of thin -walled beams of arbitrary open cross-section subjected to arbitrary large displacements in three-dimensions based on a set of kinematic assumptions. They reported that numerical solutions obtained for elastic lateral-torsional buckling for several problems employing their model were consistent with experimental results. The approach opened the way for predicting the real behaviour of structural elements in the large deflection range.

Roberts and Azizian (1981) derived expressions for the second order strains in a thin walled bar of open cross section subjected to flexural, torsional and axial displacements based on energy methods. These expressions could be used for nonlinear analysis. Such analysis would be difficult so it was necessary to employ numerical solution techniques. Roberts and Azizian (1983a) used these expressions to derive

- 25 -

the equilibrium differential equations by assuming linear elastic material behaviour. They supported their theory by providing several illustrative examples.

Chaudhary (1982) used the differential equation derived by Vlasov (1961) to develop a general stiffness matrix for a structural member (monosymmetric) of thin walled open cross section subjected to a concentric axial force. The analysis built upon the hypothesis that the presence of bimoment leads to coupling of rotational displacements. He found that bimoment has an important effect with a reduction in wall thickness of the cross section of the structure. A three-dimensional formulation for beams of an arbitrary open section based on large deflection assumptions has been derived by Ramm and Osterrieder (1983). Several illustrative examples are compared with previous work.

Opperman (1983) presented a mathematical model to study the spatial behaviour of thin walled open cross-sections using the finite element method. He reported that good agreement had been achieved through comparisons with experimental results for several load cases.

Attard (1985) presented a nonlinear theory of nonuniform torsion for straight prismatic bars having an open section under conservative loads, where the nonlinear effects of changes in the geometry are ignored in the linear elastic theory of nonuniform torsion. Yeong Bin Yang and Mc.guire (1985) adopted the equilibrium equations of thin-walled beams based on the principle of virtual displacements and an updated Lagrangian procedure.

ι

Hasegawa, et al. (1985) presented an analysis scheme for the problem of out-of-plane instability of thin-walled beams and frames. Based on the second order strain-displacement relationships, and the theorem of virtual work, they derived the general stiffness equation of

- 26 -

linearized finite displacements for a thin-walled member. Numerical examples, which covered the out-of-plane instability for both straight and curved beams, were compared with existing results. Their conclusion was that the results obtained by them were accurate, efficient and versatile for wide applications. Nishida and Fukumoto (1985) derived an exact expression for the fundamental equations of a member with initial imperfections subjected to the action of bending and torsional moments, and also to investigate the strength of a beam under various load and support conditions.

Wekezer (1985) developed an analysis for the nonlinear torsion of bars of variable cross-section. The stiffness matrix is expressed as a function of the coordinates of a discrete set of points selected from the mid-surface of the bar. The geometrical description of the mid-surface of a bar and its strain were considered. Stresses were obtained from the linear stress-strain relations. He reported good agreement with previously obtained results, either theoretical or experimental.

An incremental equilibrium equation has been derived by Sakimoto, et al. (1985) for a beam-column with arbitrary open crosssection. Features provided in the analysis are:

i- Inelastic warping torsion of a member.

- ii- Yield of the material is judged as a bi-axial stress problem associated with both normal and shear stresses.
- iii- The stress distribution development of the plastic zones in the cross-section can be easily displayed at each incremental load step.

- 27 -

vi- The effect of arbitrarily distributed residual stresses can be considered.

They reported that their approach shows good agreement when compared with available theoretical and experimental results.

Attard (1986) developed two finite element formulations, the first one ignores initial bending curvature and the second takes into consideration the first order initial curvature. The lateral buckling loads for straight elastic prismatic beams of thin-walled section under conservative loads were investigated. Close agreement was obtained with experimental data.

Ohga and Hara (1986) developed a finite element-transfer matrix method that can be applied to linear and nonlinear problems of thin-walled members under various loading conditions. They employed the Newton-Raphson method to achieve convergence of each iteration step. The section was divided into small layers in order to trace the spread of yield. The von-Mises yield criterion was used. The accuracy of their method was demonstrated by the results obtained by experimental evidence.

Kanok-Nukulchai et al. (1986) presented a formulation for the large deflection of members using a Lagrangian mode of description for the structural elements. Their assumptions were based on

- 1- Appropriate selection of element geometry, nodes as well as nodal variables
- 2- Implementation of an element shape function which incorporate all the kinematic characteristics of the applied class of structure.

- 28 -

3- Reduction of the three dimensional model into a suitable form. To establish an element model, several problems have been solved which were found to be in close agreement with experimental results.

Total potential energy has been used by Chan and Kitipornchai (1986) to provide a formulation for a general thin walled beam-column incorporating member geometrical nonlinearity. The proposed finite element formulations were demonstrated on a number of buckling problems ,including flexural-torsional buckling of rectangular beams, tee beams under moment gradient and angle beam-columns. Good agreement has been achieved when compared with independent numerical solutions. The efficiency of his method has been compared with the experimental results obtained by Fukumoto and Nishida (1981).





- 30-



# Y-axis

Z - axis

# BEAM-COLUMNS SUBJECTED TO BIAXIAL BENDING FIG. 2.2 AXES ABOUT BOTH



















FIG. 2.4 a IDEALIZATION OF BEAMS WITH GEOMETRICAL IMPERFECTIONS [ YOSHIDA AND MAEGAWA 1984]



FIG. 2.4 b <u>RESIDUAL STRESS DISTRIBUTIONS</u> [YOSHIDA AND MAEGAWA 1984]

#### Chapter 3

#### 3.1 Introduction

Much previous research has been conducted to obtain the governing differential equations for beams and beam-columns in three dimensions based on considerations of equilibrium, virtual work, or total potential energy. Some studies considered only uniform torsion, whilst others studied both uniform and non-uniform torsion.

A total potential energy approach has been used by many researchers. In 1970 Barsoum and Gallagher developed a stiffness for torsional and combined flexural-torsional elastic equation instability of one-dimensional members of uniform cross section of doubly symmetric shape for which the shear centre and centroid coincide. Rajasekaran and Murray (1973) derived the differential equations for an arbitrary cross section including inelastic material behaviour. as well as the effect of initial imperfections such as residual stresses. Roberts and Azizian (1981) derived expressions for the second order strains of thin-walled bars of open cross section subjected to flexural, torsional and axial displacements. Roberts and Azizian (1983a) used these expressions to derive the equilibrium differential equations by assuming linear elastic material behaviour, Attard (1986) presented a nonlinear theory of non-uniform torsion for straight prismatic bars having an open section under the action of conservative loads and, Chan and Kitipornchai (1986) applied it to a general formulation for thin walled beam-columns, obtain

L

incorporating member geometrical nonlinearity. The proposed finite element formulations were applied to a number of buckling problems, including flexural-torsional buckling of rectangular beams, lateral buckling of tee beams under moment gradient and angle section beamcolumns.

Nishida and Fukumoto (1985) derived an exact expression for the fundamental equations of a member with initial imperfections subjected to the action of bending and torsional moments using the principle of virtual work. The same approach with the second order strain displacement relationships has been considered by Hasegawa et al. (1985) to develop the equilibrium equation for a member having an open cross-section under flexural loading. Yang and McGuire (1986) employed the equilibrium equations of a thin walled beam based on the principle of virtual displacements and an updated Lagrangian procedure to derive the stiffness matrices.

A complete formulation of the equilibrium of a thin walled member of arbitrary cross section in space is presented herein. The equilibrium equations have been derived by two methods, the first is based on the principle of virtual work, whilst the second is based on the energy theorems. Imperfections are included in both formulations. The validity and accuracy of this formulation has been tested on a number of elastic (chapter 6) and inelastic (chapter 7) applications.

#### 3.2 Assumptions

The following assumptions have been used in the analysis:-I- The beam-column has a general open cross-section.

- 35 -

- II- Transverse displacements are much larger than the longitudinal ones.
- III- The member length is assumed very large compared with its cross-sectional dimensions.
- IV- No distortion of the cross-section occurs apart from warping, (Vlasov 1961).
- V- The shearing strain in the middle surface for open crosssections and in the planes normal to the individual plate elements may be either neglected or included.
- VI- Yielding is governed by normal stresses only.
- VII- Applied loads are conservative.

#### 3.3 Theoretical Analysis

The development of the equilibrium equations for a threedimensional beam-column of thin walled open cross-section requires that attention be given to:

- 3.3.1- Kinematics of a Section.
  - 3.3.2- Stress-Strain Relationship.
  - 3.3.3- Strain-Displacement Relationship.
  - 3.3.4- Analysis Methods.
  - 3.3.4.1- Virtual work.
  - 3.3.4.2- Total Potential Energy.

## 3.3.1 Kinematics of the Section

A thin walled open cross-section is shown in Fig. 3.1 having Cartesian coordinates X, Y, and, Z, curvilinear coordinates n and s, where s is measured along the centre line of the section, n is perpendicular to s, and t(s) is the thickness of the cross-section at a

- 36 -

distance s.  $S_c$  and C are the shear centre and the section's centroid. Let U , V , W , $\zeta$  ,and n denote the displacements in the X,Y ,and Z directions,  $\Phi$  be the angle of twist and  $U_s$  and  $V_s$  the displacements of point S from the initial position in which  $Y_s$  and  $Z_s$  represent its coordinates.

The transverse displacement of any point on the cross section can be described by the transverse displacement of a reference point and the rotation of the cross section in its own plane.  $U_0$ ,  $V_0$ , and  $\phi_0$ are the initial deformations from the original position. By taking an arbitrary point A on the cross-section the transverse displacements  $U_A$ and  $V_A$  can be expressed as:

$$U_{A} = (U_{S} + U_{O}) + (Z - Z_{S})SIN(\phi + \phi_{O}) - (Y - Y_{S})[1 - COS(\phi + \phi_{O})]$$
(3.1)

$$V_{A} = (V_{s} + V_{o}) + (Y - Y_{s})SIN(\phi + \phi_{o}) - (Z - Z_{s})[1 - COS(\phi + \phi_{o})] \quad (3.2)$$

From Fig 3.1 the direction cosines are

$$SIN_{x} = -\frac{dy}{ds}$$
(3.3a)

$$\cos\alpha = \frac{dz}{ds}$$
(3.3b)

where  $\alpha$  is the angle between the tangent at any point on the contour to the Z-axis.

If  $n_{\rm A}$  and  $\zeta_{\rm A}$  are the displacements of point A in the tangential and normal directions respectively.

 $n_{A} = V_{A} COS\alpha - U_{A} SIN\alpha$  (3.4)

 $\zeta_{A} = V_{A} SIN\alpha + U_{A} COS\alpha$ (3.5)

- 37 -

Substituting equations 3.1 and 3.2 into equations 3.4 and 3.5 yields

$$n_{A} = (V_{s}+V_{o})COS_{a} - (U_{s}+U_{o})SIN_{a} - r_{n} \{1 - COS(\phi + \phi_{0})\} + rSIN(\phi + \phi_{0})$$
(3.6)

$$\zeta_{A} = (V_{s} + V_{o})SIN_{a} + (U_{s} + U_{o})COS_{a} - r_{n}SIN(\phi + \phi) - r[i - COS(\phi + \phi_{o})]$$
(3.7)

where  

$$r = (Y - Y_S)COS_a + (Z - Z_S)SIN_a$$
 (3.8a)

$$r_n = (Z - Z_s) COS_a - (Y - Y_s) SIn_a$$
 (3.8b)

#### 3.3.2 Stress-Strain Relationships

The stress-strain relationship must either be given in analytic form or may be approximated by one or more polynomials or by piecewise linearization; the use of both methods is summarized in the following section.

#### 1- Analytical Method

Fig. 3.2a represents a  $\sigma - \epsilon$  relationship. It shows that after initial yielding this relation is not linear and ,in particular, the slope of the stress-strain curve is not constant. The tangent of the slope d $\sigma$  / d $\epsilon$  is denoted by the tangent modulus E<sub>1</sub>.

$$E_{t} = \frac{d\sigma}{d\varepsilon} \langle E \text{ for } \varepsilon \rangle \varepsilon_{y}$$
 (3.9)

# 2- Idealization of Stress-Strain Curve

An elastic-plastic strain-hardening stress-strain curve is a commonly used idealization of the true  $\sigma - \epsilon$  curve for steel in which three straight lines, as illustrated in Fig. 3.2b, are used with the stress being defined by ;

$$E\varepsilon \qquad |\varepsilon| \le \varepsilon_{y}$$

$$\sigma = \sigma_{y} \qquad \varepsilon_{Y} \le |\varepsilon| \le \varepsilon_{st} \qquad (3.10)$$

$$\sigma_{y} + E_{s} (\varepsilon - \varepsilon_{st}) \qquad |\varepsilon| \ge \varepsilon_{st}$$

# 3.3.3 Strain-Displacement Relationships

The strain configuration is defined by displacements U, V and W which are continuous functions of position within the solid. In general this may involve rigid body displacements and deformation of the body as shown in Fig. 3.3

The general strain-displacement relationships are:

$$\varepsilon_{ij} = \frac{1}{2} [U_{j,i} + U_{i,j} + U_{i,k} U_{j,k}]$$
(3.11)

where i,j,and k=1,2,3

In unbridged notation the nine equations are :

$$\varepsilon_{xx} = W_{,x} + \frac{1}{2} \{W_{,x}^2 + V_{,x}^2 + U_{,x}^2\}$$
 (3.12a)

$$\varepsilon_{yy} = W_{yy} + \frac{1}{2} \{W_{yx}^2 + V_{yx}^2 + U_{yx}^2\}$$
 (3.12b)

$$\varepsilon_{zz} = W_{yz} + \frac{1}{2} \{W_{yx}^2 + V_{yx}^2 + U_{yx}^2\}$$
 (3.12c)

$$\epsilon_{xy} = \epsilon_{yx} = \frac{1}{2} \{ W_{,y} + U_{,x} + U_{,x} W_{,y} + U_{,x} U_{,y} + V_{,x} V_{,y} \}$$
(3.12d)

- 39 -

$$\varepsilon_{xz} = \varepsilon_{zx} = \frac{1}{2} \{ W_{,z} + V_{,x} + W_{,x} W_{,z} + U_{,x} U_{,z} + V_{,x} V_{,z} \}$$
(3.12e)

$$\epsilon_{zy} = \epsilon_{yz} = \frac{1}{2} \{ U_{,z} + V_{,y} + W_{,y}W_{,z} + U_{,y}U_{,z} + V_{,y}V_{,z} \}$$
 (3.121)

in which subscripts x, y, z denote differentiation with respect to that parameter.

Equation 3.12 allows for shear deformation due to St. Venant torsion based on the hypothesis that the twisting of any cross-section will produce only shear stresses. This shear stress varies from zero at the centre of the section to its maximum value at an external surface. It is assumed that twisting of such sections does not produce any longitudinal stresses (Zbirohowski-Koscia 1967).

Using assumption V leads to

$$\epsilon_{yy} = \epsilon_{zz} = \epsilon_{yz} = 0. \tag{3.13}$$

so equation 3.12 can be reduced to

$$\varepsilon_{xx} = W_{,x} + \frac{1}{2} \{W_{,x}^2 + V_{,x}^2 + U_{,x}^2\}$$
(3.14a)

$$\varepsilon_{yx} = \varepsilon_{xy} = \frac{1}{2} \{ W_{,y} + U_{,x} + W_{,x} W_{,y} + U_{,x} U_{,y} + V_{,x} V_{,y} \}$$
(3.14b)

$$\varepsilon_{zx} = \varepsilon_{xz} = \frac{1}{2} \{ W_{,z} + V_{,x} + W_{,x} W_{,z} + U_{,x} U_{,z} + V_{,x} V_{,z} \}$$
(3.14c)

Equation 3.11 can be written in matrix form by using equations 3.13 and 3.14 as

$$\begin{bmatrix} \varepsilon_{ij} \end{bmatrix} = \begin{bmatrix} \varepsilon_{xx} & \varepsilon_{yx} & \varepsilon_{y'z} \\ \varepsilon_{zx} & 0 & 0 \\ \varepsilon_{xz} & 0 & 0 \end{bmatrix}$$
(3.15)

The curvilinear shear strain can be obtained from equation 3.12 using assumption 2.

$$Y_{sx} = 2\varepsilon_{sx} = \frac{d\gamma_A}{dx} + \frac{dW}{ds} + \frac{dU_A}{ds} \frac{dU_A}{dx} + \frac{dV_A}{ds} \frac{dV_A}{dx}$$
(3.16)

$$Y_{nx} = 2\varepsilon_{nx} = \frac{dY_A}{dx} + \frac{dW}{dn} + \frac{dU_A}{dn} \frac{dU_A}{dx} + \frac{dV_A}{dn} \frac{dV_A}{dx}$$
(3.17)

Apply assumption 5 to equation 3.16.

,

$$\frac{dV_A}{dx} + \frac{dW}{ds} + \frac{dU_A}{ds} \frac{dU_A}{dx} + \frac{dV_A}{ds} \frac{dV_A}{dx} = 0$$
(3.18)

Substituting equations 3.1, 3.2, and 3.7 in equation 3.18 yields

$$\frac{dW}{ds} = -\frac{dy}{ds} [V_{,x} SIN(\Phi + \Phi_{o}) + U_{,x} COS(\Phi + \Phi_{o})]$$
$$-\frac{dz}{ds} [V_{,x} COS(\Phi + \Phi_{o}) - U_{,x} SIN(\Phi + \Phi_{o})] - r^{2}, x \qquad (3.19)$$

Integrating equation 3.19 as

$$W = W_{c} - Y[V_{x}SIN(\Phi + \Phi_{o}) + U_{x}COS(\Phi + \Phi_{o})]$$

$$- Z[V, COS(\Phi + \Phi_{o}) - U, SIN(\Phi + \Phi_{o})] - \int \{r \Phi, x\} ds \qquad (3.20)$$

where

$$W_{c} = W_{A} + Y_{A}[V_{x}SIN(\Phi + \Phi_{o}) + U_{x}COS(\Phi + \Phi_{o})]$$

ş

$$= Z_{A}[V_{,x}COS(\phi + \phi_{o}) + U_{,x}SIN(\phi + \phi_{o})]$$
(3.21)

where the subscript A is the integral limits e.g. Y at  $Y=Y_A$  and  $Z=Z_A$ .

The last term of equation 3.20 may be determined from the sectional coordinates of a beam having a cross section with a curved outline asgiven in Fig. 3.4 and is calculated by integration of this equation.

$$\delta\omega_{\mathrm{Ds}_{1}s_{2}} = \delta S_{1} \times r_{s_{1}s_{2}}$$
(3.22a)

where  $Ds_1s_2$  is equal to the incremental area,  $r_{s_1s_2}$  is equal to the distance of pole D to a line passing through points  $S_1$  and  $S_2$ and  $dS_{12}$  is the distance between  $S_1$  and  $S_2$ . Integrating the above equation between O and  $S_n$  (the length of the centerline of a segment of the section measured from point O) yields.

$$\omega_{\text{DOS}} = \int_{0}^{S_{n}} r(s) ds \qquad (3.22b)$$

Further details are given in Zbirohowski-Koscia (1967). Denoting the normalized function by  $\overline{\omega}_n$ 

$$\overline{\omega}_{n} = \int_{0}^{S_{n}} r ds \qquad (3.23)$$

Substituting equation 3.22b into equation 3.20 yields

$$W = W_{c} - Y[V_{,x}SIN(\Phi + \Phi_{o}) + U_{,x}COS(\Phi + \Phi_{o})]$$
  
- Z[V\_{,x}COS(\Phi + \Phi\_{o}) - U\_{,x}SIN(\Phi + \Phi\_{o})] + \overline{\omega}\_{n}\Phi\_{,x} (3.24)

- 42 -

Equation 3.14 in terms of stress-displacements can be rewritten as:

$$\sigma_{xx} = E \left[ W_{,x} + \frac{1}{2} \{ V_{,x}^2 + U_{,x}^2 + W_{,x}^2 \} \right] + \sigma_r \qquad (3.25a)$$

$$\sigma_{xy} = \frac{E}{2} [W_{,y} + U_{,x} + W_{,x}W_{,y} + U_{,x}U_{,y} + V_{,x}V_{,y}] \qquad (3.25b)$$

$$\sigma_{xz} = \frac{E}{2} [W_{,z} + V_{,x} + W_{,x}W_{,z} + U_{,x}U_{,z} + V_{,x}V_{,z}] \qquad (3.25c)$$

Substituting equations 3.1, 3.2 and 3.20 into equation 3.25a yields a longitudinal strain which can be written as .

$$\epsilon_{xx} = [W_{,x} + \frac{1}{2}[U_{,x}^{2} + V_{,x}^{2} + \rho^{2}\phi_{,x}^{2}]]$$

$$-Y \{U_{,xx} - V_{,xx}\phi - U_{,xx}\frac{\phi^{2}}{2} + V_{,xx}\phi_{0} + U_{,x}\phi_{,x}\phi_{0}\}$$

$$-Z \{V_{,xx} - V_{,xx}\frac{\phi^{2}}{2} - U_{,xx}\phi - U_{,xx}\phi_{0} + V_{,x}\phi_{,x}\phi_{0}\}$$

$$-Y_{s}\phi_{,x}\{V_{,x} - U_{,x}\phi\} + Z_{s}\phi_{,x}\{U_{,x} + V_{,x}\phi\} - \overline{\omega}_{n}\phi_{,xx}] + \epsilon_{r} \qquad (3.26)$$

Equation 3.26 may be written in another form as

$$\varepsilon = \varepsilon_0 - \Phi_z Y - \Phi_y Z + \Phi_\omega \overline{\omega}_n + \varepsilon_r$$
 (3.27)

where  $\epsilon_0$ ,  $\Phi_{,x}$ ,  $\Phi_{,y}$  and  $\Phi_{\omega}$  are the representative axial strain , biaxial curvatures about the  $\gamma$ , and Z axes and warping curvature respectively. They are given by

$$\varepsilon_{0} = W_{,x} + \frac{1}{2} \{ U_{,x}^{2} + V_{,x}^{2} + \rho^{2} \phi_{,x}^{2} \}$$
  
-  $Y_{s} \phi_{,x} \{ V_{,x} - U_{,x} \phi \} + Z_{s} \phi_{,x} \{ U_{,x} + V_{,x} \phi \} + \varepsilon_{r}$  (3.28)

1

- 43 -

$$\dot{\phi}_{y} = V_{,xx} - V_{,xx2} - U_{,xx} - U_{,xx} - U_{,xx} - V_{,x} + V_{,x} + V_{,x}$$
(3.29a)

$$\Phi_{z} = U_{,xx} - U_{,xx}\frac{\Phi^{2}}{2} - V_{,xx}\Phi + V_{,xx}\Phi - U_{,x}\Phi_{,x}\Phi$$
(3.29b)

$$\Phi_{\omega} = \Phi_{,xx} \tag{3.29c}$$

Equation 3.27 can be written in matrix form as

.

$$\sigma = E \begin{bmatrix} 1 & -Z & -Y & -\overline{\omega}_n \end{bmatrix} \begin{bmatrix} \varepsilon_0 \\ \phi_y \\ \phi_z \\ \phi_w \end{bmatrix}$$
(3.30)

The stress resultants for a thin-walled member in space are

$$P = \int_{A}^{\sigma} \sigma_{xx} dA \qquad (3.31a)$$

$$Q_{y} = \int_{A}^{\sigma} \sigma_{xy} dA \qquad (3.31b)$$

$$Q_{z} = \int_{A}^{\sigma} \sigma_{xz} dA \qquad (3.31c)$$

$$M_{y} = \int_{A}^{\sigma} \sigma_{xx} Z dA \qquad (3.31d)$$

$$M_{z} = \int_{A}^{\sigma} \sigma_{xx} Y dA \qquad (3.31e)$$

$$W_{\omega} = \int_{A}^{\sigma} \sigma_{xx} \overline{\omega}_{n} dA \qquad (3.31f)$$

. = 44 =

.

$$M_{x} = \int [\sigma_{xy} (Y-Y_{s}) - \sigma_{xz} (Z-Z_{s})] dA \qquad (3.31g)$$

$$A$$

$$M_{\rho} = \int \sigma_{xx} \left[ (Y - Y_{s})^{2} + (Z - Z_{s})^{2} \right] dA \qquad (3.31h)$$

The generalized stress and generalized strain relations are then obtained by combining equations 3.30, 3.31, and 3.32.

$$\begin{bmatrix} P \\ M_{y} \\ M_{z} \\ M_{w} \end{bmatrix} = \begin{bmatrix} EA & EA\overline{Z} & EA\overline{Y} & -ES_{w} \\ & EI_{y} & -EI_{yz} & -EI_{wz} \\ & & EI_{z} & -EI_{wy} \\ & & & & EI_{w} \end{bmatrix} \begin{bmatrix} \varepsilon_{0} \\ \varphi_{y} \\ \varphi_{z} \\ \varphi_{w} \end{bmatrix}$$
(3.32)

#### 3.3.4 Method of Analysis

Consider the equilibrium of a general three dimensional body of the type illustrated in Fig. 3.5 The external forces acting on the body are surface traction  $f^{s}$ , body forces  $f^{b}$  and concentrated forces  $F_{i}$ . These forces include all externally applied forces and reactions and have in general three components corresponding to the three coordinate axes.

$$\mathbf{f}^{b} = \begin{bmatrix} \mathbf{f}_{x}^{b} \\ \mathbf{f}_{y}^{b} \\ \mathbf{f}_{z}^{b} \end{bmatrix}, \qquad \mathbf{f}^{s} = \begin{bmatrix} \mathbf{f}_{x}^{s} \\ \mathbf{f}_{y}^{s} \\ \mathbf{f}_{z}^{s} \end{bmatrix}, \qquad \mathbf{F}_{i} = \begin{bmatrix} \mathbf{F}_{xi} \\ \mathbf{F}_{yi} \\ \mathbf{F}_{zi} \end{bmatrix}$$
(3.33)

# 3.3.4.1- Principle of Virtual Work (displacements)

When a particle begins to move from a state of rest, it is gaining kinetic energy. Therefore, according to the law of kinetic energy, the forces  $F_i$  which act on the particle are performing net positive work. Hence, the particle does not move unless it can undergo some arbitrary small displacement, say  $\delta r_i$  (virtual displacement), for which the corresponding increment of work  $\delta w$  (virtual work) of forces is positive.

An equivalent approach which may be used to obtain equations that express the equilibrium of the body is through the use of the principle of virtual displacements. This principle states that for the body to be in equilibrium for any compatible set of "small" virtual displacements imposed on the body the total internal virtual work must be equal to the total external virtual work i.e.

$$\int_{V}^{\sigma_{ij} \delta \epsilon_{ij} dv} = \int_{V}^{f^{b} \delta r dv}$$

+ 
$$\int f^{s} \delta r ds + \int F_{i} \delta r ds$$
 (3.34)

The external work will be given by the product of the actual forces  $f^b$ ,  $f^s$ , and  $F_i$  acting through the virtual displacements  $\delta r$ , where

 $r = [U V \Phi W]$  (3.35)

The internal virtual work is equal to the actual stresses  $\sigma_{ij}$ acting through the virtual strains  $\delta \epsilon_{ij}$  corresponding to the imposed virtual displacements.

$$\delta \varepsilon_{ij} = \delta \left[ \varepsilon_{xx} \varepsilon_{yy} \varepsilon_{zz} \varepsilon_{xy} \varepsilon_{yz} \varepsilon_{zx} \right]$$
(3.36)

- 46 -

The superscript S denotes that surface displacements are considered and the superscript i denotes the displacements at the point where the concentrated force  $F_i$  acts (Bathe and Wilson 1976). If the body forces and traction forces are neglected equation 3.34 can be written as

$$\int_{V}^{\sigma_{ij}\delta\epsilon_{ij}dv} \int_{F_{i}\delta r ds} (3.37)$$

Substituting equations 3.25a,b, and c into equation 3.37 yields and assuming the initial strains are neglected herein by this approach.

$$\int_{V}^{*} \frac{E_{E_{xx}} \delta W_{,x} dV}{V} + \frac{1}{2} \int_{V}^{\sigma} \sigma_{,xx} \delta [W_{,x}^{2} + U_{,x}^{2} + V_{,x}^{2}] dV$$

$$+ \int_{V}^{d} \frac{G[\{W_{,y} + U_{,x}\} \delta\{W_{,y} + U_{,x}\} + \{W_{,z} + V_{,x}\} \delta\{W_{,z} + V_{,x}\}] dV$$

$$\frac{1}{2} \int_{V}^{\sigma} \sigma_{xz} \delta\{W_{,x}W_{,z} + U_{,x}U_{,z} + V_{,x}V_{,z}\} dV$$

$$\frac{1}{2} \int_{V}^{\sigma} \sigma_{xy} \delta\{W_{,x}W_{,y} + U_{,x}U_{,y} + V_{,x}V_{,y}\} dV - \int_{S}^{F_{1}} \delta r ds = 0 \quad (3.38)$$

# 3.3.4.2 \_\_\_\_\_ Total Potential Energy

Several references such as Bleich (1952), Kerensky et al. (1956), Timoshenko and Gere (1961), Galambos (1968), and Zienkiewicz (1977) have covered full details of the energy theorems, these can be summarized as: the total potential energy is equal to the summation of both strain energy  $(\overline{U})$  and potential energy due to applied loads  $(\overline{V})$ .

$$\Pi = \overline{U} + \overline{V}$$
(3.39)

- 47 -

where

$$\overline{U} = \frac{1}{2} \int [\sigma_{ji}] [D] [\Sigma_{ij} dV]$$
(3.40a)

where D is Hookean's constant relating to the material properties. The above equation including the initial strains can be written in an other form as:

$$\overline{U} = \frac{1}{2} \int_{V} [\Sigma^{L} + \Sigma^{NL}] [D] [\Sigma^{L} + \Sigma^{NL}]$$
(3.40b)

where  $\Sigma^{\rm L} and \ \Sigma^{\rm NL}$  are linear and nonlinear strains.

$$\overline{V} = \int \vec{P} \delta r dA \qquad (3.40c)$$

where

$$P^{=} = [F_{xk} Q_{yk} M_{zk} Q_{zk} M_{yk} M_{xk} M_{\omega k}]^{T}$$
(3.40d)

where k is equal to 1 or 2 and the superscipt T is the transpose of a matrix.

# 3.4 Imperfections

The following factors severely weaken the strength of the structure:

i- Initial deflections.

ii- Residual stresses.
The initial deflection  $(U_0, V_0, and \Phi_0)$  quantities may be completely arbitrary but for convenience the following basic arrangements are automatically included:

i- Sine wave.

$$U_{o} = SIN \Pi_{L}^{X} \qquad (X = 0 \text{ to } L) \qquad (3.41)$$

ii- Polynomial Function.

$$U_0 = a_0 X + a_1 X^2 + a_2 X^3 + \dots + a_n X^n$$
 (3.42)

Different arrangements are provided for each of the main deflection components i.e.  $U_{o}$ ,  $V_{o}$ ,  $\Phi_{o}$ .

# ii- Residual Stresses

The presence of residual stresses, exerts a significant influence on the way in which yield spreads through the cross-section; it must therefore be included in any analysis which attempts to model real beam-column behaviour. These stresses arise from non-uniform temperature distributions during manufacture (Nethercot 1974), so that the parts which cool slowly, such as the web to flange junction of an I-section will normally be in residual tension (Young 1971, Nethercot and Trahair(1983)) with balancing residual compression elsewhere in the section.

These longitudinal residual stresses must produce a system of forces in the cross-section which must itself satisfy the following three equilibrium equations:

$$\int \sigma_r dA = 0 \qquad (3.43)$$

- 49 -

$$\int_{A}^{\sigma} \sigma_{r} Y dA = 0$$
 (3.44)

 $\int \sigma_r^Z dA = 0 \qquad (3.45)$ 

If the measured residual stress distributions do not satisfy the above conditions, then these stresses must be adjusted until the necessary balance is obtained. For I and H sections this can best be achieved by modifying the maximum web residual stress until equilibrium results.

Several different forms of standard residual stress distribution have been proposed for use in analytical work (O'conner (1955), Young 1975, Nethercot (1974), Nethercot and Trahair (1983), Kitipornchai and Wang-Chung (1985)). Some of these distributions are considered herein although because the present study addresses the full 3-dimensional response of a beam-column some limitations in the proposals not identified in the context of 2-dimensional behaviour have been corrected.

1- Lehigh distribution.

2- Parabolic distribution.

3- Triangle distribution.

#### 1- Lehigh Pattern

Fig. 3.6a illustrates the residual stress distributions used in much of the work conducted at Lehigh University. A linear function is chosen to represent the variations of the stresses along the web and the flanges. It can be written as:

- 50 -

$$\sigma_{i}=a_{0}X + a_{1}$$
 (3.46)

Application of the boundary conditions to the above equation yields i)- For the flange

at 
$$X = 0$$
  $\sigma_{fw} = a_1$  (3.47a)

at X = B  $\sigma_{f} = a_0 B + \sigma_{fw}$  (3.47b)

from equation 3.47a and 3.47b, we obtain

$$a_0 = \frac{\sigma_f - \sigma_{fW}}{B}$$
(3.48)

Substituting equations 3.47a and 3.48 into equation 3.46 yields equation 3.49, which can be used to evaluate the residual stress at any point along the flanges.

$$\sigma_1 = \frac{\sigma_f - \sigma_{fW}}{B} X + \sigma_{fW}$$
(3.49)

ii)- for the web

The residual stress is assumed constant. So equation 3.47 after applying the boundary condition becomes

$$\sigma = \sigma_{W} \tag{3.50}$$

where

 $\sigma_{\rm f}$  is the flange tip residual stress.  $\sigma_{\rm W}$  is the web residual stress  $\sigma_{\rm fw}$  is the web to flange junction residual stress

#### 2- Parabolic Distribution

Recent measurements have suggested that the Lehigh model is too simple to accurately model the true patterns found in many shapes. An improvement results if simple parabolic functions are assumed to represent the residual stress distributions for both web and flange at any point along the flange or the web. It can be written as:

$$\sigma_1 = a_0 X^2 + a_1$$
 (3.51)

Where  $a_0$ , and  $a_1$  are arbitrary constants to be determined from boundary conditions for the web and flanges.

i)- For the web

at X=0 
$$\sigma_{\rm u}$$
 = a<sub>1</sub> (3.52a)

at 
$$\frac{X=D}{2} \sigma_{fw} = a_0 \frac{D^2}{2} + \sigma_w$$
 (3.52b)

$$a_0 = \frac{\sigma_{fw} - \sigma_w}{\left(\frac{D}{2}\right)^2}$$
(3.53)

Substituting equations 3.52a and 3.53 into equation 3.51 yield a general equation which represents distributions of the residual stresses at any point along the web.

$$\sigma = \frac{\sigma_{fw} - \sigma_{w}}{\left(\frac{D}{2}\right)^{2}} X^{2} + \sigma_{w}$$
(3.54)

ii)- For the flange

Equation 3.55 can be obtained similar to equation 3.54, by substituting for the boundary conditions in equation 3.51

$$\sigma_{1} = \frac{\sigma_{f} - \sigma_{fw}}{\left(\frac{B}{2}\right)^{2}} X^{2} + \sigma_{f}$$

(3.55)

This distribution is illustrated in Fig. 3.6b

# 3- Triangular Distribution

This is an alternative simple distribution illustrated in Fig. 2c. Equation 3.47 can be implemented to calculate the residual stresses at any point along the web or the flange noting the change in boundary condition required. More sophisticated models (El-Khenfas and Nethercot 1987a), including any arbitrary pattern which satisfies equations 3.43, 3.44, and 3.45, may readily be incorporated.

# 3.5 Equilibrium Equations

Two methods are employed herein in order to obtain general equilibrium equations for different kinds of open thin-walled sections under a variety of loading and boundary conditions, which are:

1- Virtual Work

2- Total Potential Energy

# 3.5.1 By Virtual Work

Using Appendix A, assuming the terms containing the products of  $W_{,x}$  in the shear strain i.e.  $W_{,x}W_{,y}$  and  $W_{,x}W_{,z}$ , to have significant magnitude so that they cannot be assumed equal to zero, in equation 3.38 the general equilibrium equation of an arbitrary thin-walled open cross-section can be written as:

- 53 -

$$\int_{0}^{L} \left[ EAE[\{W_{,x} + \frac{1}{2}(U_{,x}^{2} + V_{,x}^{2} + \rho^{2}\Phi_{,x}^{2}) \right] \\ - \overline{Y}(V_{,xx}\Phi + U_{,xx}) - \overline{Z}(V_{,xx}-U_{,xx}\Phi) \\ - Y_{S}\Phi_{,x}U_{,x} + Z_{S}V_{,x}\Phi_{,x} - \frac{ES_{\omega}}{EA}\Phi_{,x})\deltaW_{,x}] \\ - EA\overline{Y}\left[ W_{,x}\delta(V_{,xx}\Phi+U_{,xx}) + (\frac{1}{2}(U_{,x}^{2} + V_{,x}^{2}) + Y_{,x}\Phi_{,xx}\Phi_{,xx}) + (\frac{1}{2}(U_{,x}^{2} + V_{,x}^{2}) + Y_{,x}\Phi_{,xx}\Phi_{,xx}) + (\frac{1}{2}(U_{,x}^{2} + V_{,x}^{2}) + Y_{,x}\Phi_{,xx}\Phi_{,xx}) + (\frac{1}{2}EA\overline{Y}(Y_{,x}^{2} + Z_{,x}^{2})\Phi_{,x}^{2}\deltaU_{,xx}) \\ + EAW_{,x}\delta(V_{,xx}-U_{,xx}\Phi) + (\frac{1}{2}(U_{,x}^{2} + V_{,x}^{2}) + Y_{,x}\Phi_{,xx}\Phi_{,xx}) \\ + Y_{,x}\Phi_{,x}\delta(V_{,xx}-U_{,xx}\Phi) + (\frac{1}{2}(U_{,xx}^{2} + V_{,x}^{2}) + Y_{,xx}\Phi_{,xx}) \\ + Y_{,x}\Phi_{,xx}\delta\Phi_{,xx} + EI_{,x}V_{,xx}\delta(V_{,xx}-U_{,x}\Phi_{,xx}) \\ - \delta(U_{,xx}V_{,xx}\Phi) - Z_{,x}\Phi_{,xx}\delta(V_{,xx}-U_{,x}\Phi_{,xx}) \\ - \delta(U_{,xx}V_{,xx}\Phi) - Z_{,x}\Phi_{,xx}\Phi_{,xx} + V_{,xx}\Phi_{,xx}) + V_{,xx}\delta(\Phi_{,x}V_{,x}) \\ + EI_{,x}C\delta(U_{,xx}V_{,xx} - U_{,xx}\Phi_{,xx} + V_{,xx}\Phi_{,xx}) + V_{,xx}\delta(\Phi_{,x}V_{,x}) \\ - U_{,xx}\delta(\Phi_{,x}U_{,x}) - \Phi_{,x}^{2}(Z_{,x}G_{,xx} + Y_{,x}\Phi_{,xx}) ] \\ + EI_{,xy}C\delta(\Phi_{,x}U_{,xx} + U_{,xx}\Phi_{,xx}) + \Phi_{,xx}\delta(\Phi_{,x}U_{,x}) - Y_{,x}\Phi_{,xx}^{2} \\ - U_{,xx}\delta(\Phi_{,x}U_{,xx} - U_{,xx}\Phi_{,xx}) + \Phi_{,xx}\delta(\Phi_{,x}U_{,x}) - Y_{,x}\Phi_{,x}^{2} \\ - ES_{,x}W_{,xx} + \frac{1}{2}\Phi_{,x}(U_{,xx} + V_{,xx}\Phi_{,x}) + Y_{,xx}\Phi_{,xx} - Z_{,x}\Phi_{,xx}] \\ + EI_{,xy}C\delta(\Phi_{,xx}(V_{,xx} - U_{,xx}\Phi_{,xx}) + Y_{,xy}\Phi_{,xx} - Z_{,x}\Phi_{,xx}] \\ + EI_{,xy}C\delta(\Phi_{,xx}(V_{,xx} - U_{,xx}\Phi_{,xx})) + \Phi_{,xx}\delta(\Phi_{,x}U_{,x}) - Z_{,x}\Phi_{,x}^{2} \\ + EI_{,xy}C\delta(\Phi_{,xx}(V_{,xx} - U_{,xx}\Phi_{,xx}) + Y_{,xx}\Phi_{,x}A_{,x}) - Z_{,x}\Phi_{,x}^{2} \\ + EI_{,xy}C\delta(\Phi_{,xx}(V_{,xx} - U_{,xx}\Phi_{,xx})) + \Phi_{,xx}\delta(\Phi_{,x}U_{,x}) - Z_{,x}\Phi_{,x}^{2} \\ + EI_{,xy}C\delta(\Phi_{,xx}(V_{,xx} - U_{,xx}\Phi_{,x})) + \Phi_{,xx}\delta(\Phi_{,x}U_{,x}) - Z_{,x}\Phi_{,x}^{2} \\ + EI_{,xy}C\delta(\Phi_{,xx}(V_{,xx} - U_{,xx}\Phi_{,x}) + Y_{,xy}\Phi_{,x} - Z_{,x}\Phi_{,x}) \\ + EI_{,xy}C\delta(\Phi_{,xx}(V_{,xx} - U_{,xx}\Phi_{,x}) + Y_{,xy}\Phi_{,x} - Z_{,x}\Phi_{,x}) \\ + EI_{,xy}C\delta(\Phi_{,xx}(V_{,xx} - U_{,xx}\Phi_{,x})) + \Phi_{,xy}C\delta(\Phi_{,x}U_{,x}) - Z_{,x}\Phi_{,x}\Phi_{,x}) \\ + E$$

~

- 54 -

$$+ \frac{1}{2} (Y_{s}^{2} + Z_{s}^{2}) j \delta \phi, xx$$

$$+ \beta_{w} \phi_{xx}^{2} \delta \phi_{xx} + \beta_{y} \delta (\phi_{xx}^{2} U_{yxx}) + \beta_{z} \delta (\phi_{xx}^{2} V_{yxx}) + KG \phi_{xx} \delta \phi_{yx}$$

$$+ P[V_{xx} \delta V_{yx} + U_{yx} \delta U_{yx} + Z_{s} \delta (\phi_{yx} (V_{xx} \phi + U_{yx}))$$

$$- Y_{s} \delta \{\phi_{xx} (V_{yx} - U_{yx} \phi)\}]$$

$$+ M_{p} \phi_{xx} \delta \phi_{yx} + M_{z} \delta \{\phi_{yx} (V_{yx} - U_{yx} \phi)\} - M_{y} \delta \{\phi_{yx} (V_{yx} \phi + U_{yx})\}$$

$$+ Q_{y} [\delta (V_{yx} \phi - W_{yx} (V_{yx} \phi + U_{yx}))$$

$$- Z_{s} \delta [V_{yx} (U_{yxx} \phi + U_{yx} \phi_{yx}) + V_{yx} (V_{yxx} \phi_{yx} + U_{yxx} \phi_{yx})]$$

$$- Q_{z} [\delta (U_{yx} \phi + W_{yx} (V_{yx} + U_{yx} \phi))$$

$$+ Y_{s} \delta [V_{yxx} (V_{yxx} \phi + V_{yx} \phi_{yx}) + U_{yx} (U_{yxx} \phi_{yx} + V_{yx} \phi_{yx})]$$

$$M_{x} \delta \{V_{yxx} (V_{yx} \phi + V_{yx} \phi_{yx}) - U_{yxx} (U_{yx} \phi + V_{yx})] + [F_{ij}] [T_{ij}] dx$$

$$(3.56)$$

$$If the higher order terms such as  $V_{yx} U_{yxx} \phi_{yx}, V_{yxx} \phi_{yx},$ 

$$W_{yx} U_{yx} \phi_{yx}, \text{etc. are neglected from the above equation, then it can be written as:$$

$$L_{ij}$$$$

$$\int_{0}^{\infty} \left[ EA\{W, x = \overline{Y}U, xx = \overline{Z}V, xx = \frac{ES}{EA}\phi, xx\}\delta W, x \right]$$
  

$$= EA\overline{Y}W, x\delta U, xx = EA\overline{Z}W, x\delta V, xx + EI_{z}U, xx\delta U, xx + EI_{y}V, xx\delta V, xx$$
  

$$+EI_{\omega}\phi, xx\delta\phi, xx + EI_{yz}\delta U, xxV, xx + EI_{\omega y}\delta\phi, xxU, xx + EI_{\omega z}\delta(\phi, xxV, xx)$$
  

$$= -ES_{\omega}W, x\delta\phi, xx + KG\phi, x\delta\phi, x$$
  

$$+ P[V, x\delta V, x + U, x\delta U, x + Z_{s}\delta(\phi, xU, x) - Y_{s}\delta(\phi, xV, x)]$$

- 55 -

$$+ M_{p} \Phi_{x} \delta \Phi_{x} + M_{z} \delta (\Phi_{x} V_{x}) - M_{y} \delta (\Phi_{x} V_{x}) + Q_{y} \delta (V_{x} \Phi)$$

$$+ Q_{z} \delta (U_{x} \Phi) + M_{x} \delta (V_{x} V_{x}) - U_{x} V_{x} V_{x}) + LF_{i} J\{r\} dx \qquad (3.57)$$

For a doubly symmetric section, the above equation reduces to  $\int_{0}^{L} \frac{1}{E^{A(W,x^{\delta W},x^{\prime})} + E^{I}z^{U}x^{\delta U}x^{\delta U}x^{\prime} + E^{I}y^{V}x^{\delta V}x^{\prime} + E^{I}w^{\Phi}x^{\delta \Phi}x^{\delta \Phi}x^{\epsilon}}{e^{X^{\delta \Phi}x^{\epsilon}} + P^{V}x^{\delta V}x^{\epsilon} + U^{A}x^{\delta U}x^{\epsilon}}$   $+ KG^{\Phi}x^{\delta \Phi}x^{\epsilon} + P^{V}x^{\delta V}x^{\epsilon} + U^{A}x^{\delta U}x^{\epsilon}$   $+ M_{\rho}\Phi,x^{\delta \Phi}x^{\epsilon} + M_{z}^{\delta(\Phi}x^{V}x^{\prime}) - M_{y}^{\delta(\Phi}x^{U}x^{\prime}) + Q_{y}^{\delta(V}x^{\Phi})$   $+ Q_{z}^{\delta(U,x\Phi)} + M_{x}^{\delta(V}x^{U}x^{U}x^{\epsilon} - U^{A}x^{V}x^{\epsilon}) + [F_{i}]\{r\} dx \qquad (3.58)$ The general equation obtained from Appendix B , where the

product terms in the shear strains i.e. ( $W_{,x}W_{,y}$  and  $W_{,x}W_{,z}$ ) are not included can be written as:-

$$\int_{0}^{L} \left[ EAL\{W, x + \frac{1}{2}(U_{,x}^{2} + V_{,x}^{2} + \rho^{2}\Phi_{,x}^{2}) - \overline{Y}(V_{,xx}\Phi + U_{,xx}) - \overline{Z}(V_{,xx}-U_{,xx}\Phi) - Y_{x}V_{,xx}\Phi + U_{,xx}) - \overline{Z}(V_{,xx}-U_{,xx}\Phi) - Y_{x}V_{,xx}\Phi + Z_{x}U_{,x}\Phi_{,x} - \frac{ES_{\omega}}{EA}\Phi_{x}\}\delta W_{,x} \right]$$

$$- EA\overline{Y}[W_{,x}+\{\frac{1}{2}(U_{,x}^{2} + V_{,x}^{2}) - Y_{x}\Phi_{,x}V_{,x} + Z_{x}\Phi_{,x}U_{,x} - \frac{1}{2}(Y_{x}^{2} + Z_{x}^{2})\Phi_{,x}^{2}]\delta U_{,xx}$$

$$- EA\overline{Z}[W_{,x}+\{\frac{1}{2}(U_{,x}^{2} + V_{,x}^{2}) - Y_{x}\Phi_{,x}V_{,x} + Z_{x}\Phi_{,x}U_{,x}] - Y_{x}\Phi_{,x}V_{,x}$$

- 56 -

+ 
$$E_{i}^{i} \bigoplus_{x,x} \delta_{0,xx} - E_{i}^{j} (U_{,xx}^{i} + V_{,xx}^{i} - Z_{s}^{\frac{2}{2}}) \delta_{v,xx}$$
  
+  $E_{i}^{j} (V_{,xx}^{i} + U_{,xx}^{i} + Y_{s}^{\frac{2}{2}}) \delta_{v,xx}$   
-  $E_{i}^{j} (U_{,xx}^{i} - V_{,xx}^{i} + Z_{s}^{i} + Z_{s}^{i} + U_{,xx}^{i}) \delta_{v,xx}$   
-  $(V_{,xx}^{i} + U_{,xx}^{i} + Y_{s}^{\frac{2}{2}}) \delta_{v,xx}^{i} + (V_{,xx}^{i} + U_{,xx}) \delta_{0,xx}]$   
+  $E_{i}^{i} (\Phi_{,xx}^{i} - Y_{s}^{\frac{2}{2}}) \delta_{v,xx}^{i} + (V_{,xx}^{i} + U_{,xx}) \delta_{0,xx}]$   
-  $E_{i}^{i} (\Phi_{,xx}^{i} - Z_{s}^{\frac{2}{2}}) \delta_{v,xx}^{i} - (U_{,xx}^{i} - V_{,xx}) \delta_{0,xx}]$   
-  $E_{i}^{i} (\Phi_{,xx}^{i} - Z_{s}^{\frac{2}{2}}) \delta_{v,xx}^{i} + (U_{,xx}^{i} - V_{,xx}) \delta_{0,xx}]$   
-  $E_{i}^{i} (\Phi_{,xx}^{i} + Z_{s}^{i}) \delta_{v,xx}^{i} + S_{s}^{i} (\Phi_{,x}^{i} V_{,xx}^{i} + Z_{s}^{i} \Phi_{,xx}^{i})$   
+  $B_{i}^{i} \Phi_{,x}^{i} \delta_{,xx}^{i} + B_{y}^{i} \delta_{v}^{i} (\Phi_{,x}^{i} U_{,xx}^{i}) + B_{z}^{i} (\Phi_{,x}^{i} V_{,xx}^{i}) + G_{v}^{i} \delta_{,x}^{i} \delta_{,x}$   
+  $B_{i}^{i} \Phi_{,x}^{i} \delta_{,xx}^{i} + B_{y}^{i} \delta_{v}^{i} (\Phi_{,x}^{i} U_{,xx}^{i}) + B_{z}^{i} (\Phi_{,x}^{i} V_{,xx}^{i}) + G_{v}^{i} \delta_{,x}^{i} \delta_{,x}^{i}$   
+  $P_{i}^{i} V_{,x}^{i} \delta_{,xx}^{i} + B_{y}^{i} \delta_{v}^{i} (\Phi_{,x}^{i} U_{,xx}^{i}) + B_{z}^{i} (\Phi_{,x}^{i} V_{,xx}^{i})$   
-  $Y_{s}^{i} \delta_{v}^{i} (\Psi_{,x}^{i} + U_{,x}^{i} \delta_{,x}^{i}) + M_{y}^{i} (U_{,xx}^{i} + V_{,xx}^{\frac{2}{2}})$   
+  $M_{0}^{i} \delta_{,x}^{i} \delta_{,x}^{i} + M_{z}^{i} \delta_{v}^{i} (\Psi_{,xx}^{i} - U_{,xx}^{\frac{2}{2}}) + F_{1}^{i} (r) d_{x}^{i} d_{x}^{i} (J_{,xy}^{i}) + \delta_{x}^{i} (J_{,xy}^{i})$   
Neglecting the higher order terms  $(U_{,x}^{i} V_{,xx}^{i} + \Phi_{,x}^{i} V_{,xx}^{i})$   
etc.) in the above equation, then it can be written as:  

$$\int_{0}^{i} \left[ E_{A}^{i} (W_{,xx}^{i} - \overline{V}_{,xx}^{i} - \overline{V}_{,xx}^{i} - \frac{E_{i}^{i} \omega}{E_{A}^{i} \delta_{,xx}^{i}} - E_{i} V_{,xx}^{i} \delta_{,xx}^{i} + E_{i} U_{,xx}^{i} \delta_{,xx}^{i} + E_{i} V_{,xx}^{i} \delta_{,xx}^{i} + E_{i} V_{,xx}^{i} \delta_{,xx}^{i} + E_{i} U_{,xx}^{i} \delta_{,xx}^{i} + E_{i} V_{,xx}^{i} \delta_{,xx}^{i} +$$

- 57 -

•

•

$$+ EI_{\omega y}^{(\phi}, xx^{\delta U}, xx^{+}, U, xx^{)\delta \Phi}, xx^{]}$$

$$+ EI_{\omega z}^{(\phi}, xx^{\delta V}, xx^{+}, - V, xx^{\delta \Phi}, xx^{)}$$

$$-ES_{\omega}^{(W}, x^{-}, Y_{S}^{\phi}, x^{V}, x^{+}, Z_{S}^{\phi}, xU, x^{)\delta \Phi}, xx$$

$$+ GK_{\phi}, x^{\delta \Phi}, x^{+} P[V, x^{\delta V}, x^{+}, U, x^{\delta U}, x^{+}, Z_{S}^{\delta}(\Phi, x^{U}, x)]$$

$$- Y_{S}^{\delta}(\Phi, x^{V}, x) + M_{y}^{\delta}(U, xx^{\Phi}) + M_{\rho}^{\phi}, x^{\delta \Phi}, x$$

$$(3.60)$$

$$+ M_{z}^{\delta}(V, xx^{\Phi}) + F_{i}^{[r]} dx$$

for doubly symmetric I-sections  $Y_s$ ,  $Z_s$ ,  $\overline{Y}$ , and  $\overline{Z}$  are equal to zero, if we substitute these in the above equation it yields equation 3.61

$$\int_{0}^{L} \left[ EA[W_{,x} + EI_{\omega}\Phi_{,xx}\delta\Phi_{,xx} - EI_{y}V_{,xx}\delta V_{,xx} + EI_{z}U_{,xx}\delta U_{,xx} + GK\Phi_{,x}\delta\Phi_{,x} + P[V_{,x}\delta V_{,x} + U_{,x}\delta U_{,x} + Z_{s}\delta(\Phi_{,x}U_{,x}) + M_{y}\delta(U_{,xx}\Phi) + M_{\rho}\Phi_{,x}\delta\Phi_{,x} + M_{z}\delta(V_{,xx}\Phi) + F_{i}\{r\} \right] dx$$
(3.61)

# 3.5.2 Total Potential Energy

Substituting equations (3.1-3.27) into equation 3.40 yields a general equilibrium equation incorporating the product of the derivatives of longitudinal displacements with respect to the Y and Z directions.

$$\Pi = \frac{1}{2} \int_{0}^{L} \left[ EA \{ W_{,x}^{2} + \frac{1}{2} W_{,x} (U_{,x}^{2} + V_{,x}^{2} + \frac{\rho^{2}}{EA} \Phi_{,x}^{2}) \right]$$

- 58 -

- $2\overline{Y}W_{,x}(V_{,xx} + U_{,xx} + V_{,x} + V_{,x})$
- <sup>2</sup>2W,x(V,xx U,xx<sup>¢</sup> U,x<sup>¢</sup>,x)
- 2W, X's &, X', X + 2W, X<sup>2</sup> S', X<sup>¢</sup>, X 2<sup>ES</sup><sup>W</sup>W, X<sup>¢</sup>, XX
  - 2EAW,x{V,xx\* + U,xx + V,x\*,x
- +  $V_{,XX} U_{,XX} U_{,X} + V_{,XX} \frac{ES}{EA}_{,XX}$
- xx. +  $2EA\overline{Y} \{Y_{S} \phi, Y, x^{-2} S_{S} \phi, U, x^{-1} = \frac{1}{2} (U_{x}^{2} + V_{x}^{2} + (Y_{S}^{2} + Z_{S}^{2}) \phi_{x}^{2}) \} U, xx$ +  $2EA\overline{Z} \{Y_{s} \oplus Y, x^{-2} \oplus Y, x^{-2}$
- xx. +  $2ES_{\mu}\{Y_{s}\phi, Y, x^{-2}s^{\phi}, x^{\mu}, x^{-1}\frac{1}{2}(u_{x}^{2}+v_{y}^{2}x^{+}(Y_{s}^{2}+z_{s}^{2})\phi_{x}^{2})\}\phi$
- +  $EI_{\omega} \{ \Phi^2, x + \overline{B}_{\omega} \Phi^2, \Phi, xx \}$  +  $GK\Phi^2, x$
- xx, +  $EI_{y}\{(v_{,xx}^{2} - 2U_{,xx}^{2}v_{,xx}^{2}) - 2V_{,xx}(V_{,x}^{2}v_{,x}^{2} + U_{,x}^{2}v_{,x}^{2}) + \overline{B}_{y}^{2}v_{,xx}^{2}\}$ +  $EI_{z}$ {( $2V_{x}^{*}xx^{\Phi}U_{xx} + U_{xx}^{2}$ ) +  $2U_{xx}(V_{x}^{*}x^{*} - U_{x}^{*}x^{*}) + \overline{B}_{z}\phi_{xx}^{2}V$
- +  $2EI_{yz}^{f}(U_{,xx}^{V}V_{,xx} + (V_{,xx}^{\Phi} + V_{,x}^{\Phi}V_{,x} U_{,x}^{\Phi}V_{,xx})$
- =  $(U_{xx} + V_{xx} + U_{xx} + U_{xx} + U_{xx})$
- +  $2EI_{y\omega}$  { (V,xx $\phi$  + V,x $\phi$ ,x) $\phi$ ,xx Ys $\phi^2$ ,x\*
- +  $(\phi_{xx}V_{x} U_{xx}\phi_{xx})\phi_{xx} + U_{xx}\phi_{xx}^{+}$
- +  $2EI_{zw}^{\{}$  = (U, x x + U, x  $\phi$ , x)  $\phi$ , xx  $Z_{s} \phi^{2}_{,x} \phi$ , xx
- $(\phi, x_x U_x + V, x^{\phi}, x_x)^{\phi}, x + V, x^{\phi}, x^x)^{\phi}$
- +  $P\{U_{x}^{2} + V_{x}^{2} + \rho_{0}^{2}\phi_{x}^{2} + 2(-2_{s}U_{s}x + Y_{s}V_{s}x)\phi_{x}$

+ 
$$2(Z_{s}V_{,x} + Y_{s}U_{,x})\phi_{,x}\phi_{,x}\phi_{,x}\phi_{,xx}$$
  
+  $Q_{y}\{W_{,x}(V_{,x} - U_{,x}\phi) + 2(\psi U_{,x} + \frac{1}{2}e^{2}V_{,x})$   
+  $Y_{s}(c\phi_{,x}c + \frac{1}{2}e^{2}e_{,x}) - Z_{s}c\phi_{,x}\}$   
-  $Q_{z}\{W_{,x}(V_{,x}^{+} + U_{,x}) - 2(\psi V_{,x} + \frac{1}{2}b^{2}U_{,x})$   
+  $Z_{s}(-b\phi_{,x}\phi + \frac{1}{2}\phi^{2}\phi_{,x}) - Y_{s}\phi\phi_{,x}\}$   
+  $2M_{y}\{(U_{,x} - V_{,x}\phi)\phi_{,x} - W_{,x}(V_{,xx} - U_{,xx}\phi - U_{,x}\phi_{,x}))$   
+  $2M_{z}\{(V_{,x} - U_{,x}\phi)\phi_{,x} - W_{,x}(U_{,xx} + V_{,xx}\phi + V_{,x}\phi_{,x}))$   
+  $M_{x}\{\phi_{,x}\phi^{2} + U_{,x}(U_{,x}\phi_{,x} + U_{,xx}\phi) - (U_{,x} - V_{,x}\phi)V_{,xx}$   
-  $V_{,x}(V_{,x}\phi_{,x} + V_{,xx}\phi) + (V_{,x} - U_{,x}\phi)U_{,xx}\}$   
-  $M_{z}\phi_{0}[V_{,xx} - U_{,x}\phi_{,x}) + M_{z}\phi_{0}[U_{,xx} + V_{,x}\phi_{,x})$   
+  $\frac{1}{2}P[Z_{s}V_{,x}\phi_{,x}\phi_{,x} + Y_{s}U_{,x}\phi_{,x}\phi_{,x}]$   
-  $Q_{y}\phi_{0}[V_{,x} + Y_{s}\phi_{,x} + Z_{s}(\frac{\phi_{0}}{2}+2\phi]$   
+  $\langle P\rangle(r)\frac{1}{2}dx + BB + CC$  (3.62)  
where CC is defined in appendix C and BB as

$$BB = [\sigma \frac{d\overline{\omega}_n}{xz dZ} + \sigma \frac{d\overline{\omega}_n}{xy dY}]\overline{X}$$
(3.63)

.

•

where

$$\overline{X} = \{ \Phi_{,x} + Y(V_{,x} \Phi_{,x} + V_{,xx} \Phi + U_{,xx}) + Z(U_{,x} \Phi_{,x} + U_{,xx} \Phi - V_{,xx}) \} \Phi_{,x}$$
(3.64)

- 60 -

Equation 3.63 is equal to zero and full details are given in appendix A.

The cross section properties  $\overline{\beta}_y$ ,  $\overline{\beta}_z$ , and  $\overline{\beta}_\omega$  are

$$\overline{\beta}_{y} = \frac{1}{1_{y}} \int_{A} Y(Y^{2} + Z^{2}) dA - 2Y_{s}$$
(3.65a)

$$\overline{\beta}_{z} = \frac{1}{I_{z}} \int_{A}^{Z} (Y^{2} + Z^{2}) dA - 2Z_{s}$$
(3.65b)

$$\overline{\beta}_{\omega} = \frac{1}{I_{\omega}} \int_{A}^{\omega} (Y^2 + Z^2) aA \qquad (3.65c)$$

$$\rho_0^2 = \int [(Y - Y_s)^2 + (Z - Z_s)^2 dA \qquad (3.65d)$$
  
A
  
The variation to the equation 3.62 after substituting for

The variation to the equation 3.62 after substituting for the terms CC from appendix CC yields

$$\delta \Pi = \int_{0}^{1} \left[ EALW_{,x} + \frac{1}{2} (U_{,x}^{2} + V_{,x}^{2} + \frac{\rho^{2}}{EA} + \frac{\rho^{2}}{EA}$$

- 61 -

$$- \Sigma(6V_{,xx} - U_{,xx}60 - 66U_{,xx} - U_{,x}60_{,x} - V_{,x}60_{,x}) + Z_{,x}60_{,x}) + Z_{,x}60_{,x} + V_{,x}60_{,x}) + Z_{,x}60_{,x} + V_{,x}60_{,x}) + Z_{,x}60_{,x}) + Z_{,x}60_{,x} + V_{,x}60_{,x}) + Z_{,x}60_{,x}) + Z_{,x}60_{,x} + V_{,x}60_{,x}) + Z_{,x}60_{,x}) + Z_{,x}60_{,x} + V_{,x}60_{,x}) + Z_{,x}60_{,x} + V_{,x}60_{,x}) + Z_{,x}60_{,x}) + Z_{,x}60_{,x} + V_{,x}70_{,x}60_{,x} + V_{,x}70_{,x}60_{,x}) + Z_{,x}60_{,x}) + Z_{,x}60_{,x} + V_{,x}70_{,x}60_{,x}) + Z_{,x}60_{,x}) + Z_{,x}70_{,x}60_{,x}) + Z_{,x}70_{,x}70_{,x}0_{,x}) + Z_{,x}70_{,x}70_{,x}0_{,x}) + Z_{,x}70_{,x}70_{,x}0_{,x}) + Z_{,x}70_{,x}0_{,x}) + Z_{,x}70$$

$$- V_{xx}((u_{xx}66 + 66u_{xx})) + V_{xx}66e_{xx} + v_{xx}66e_{xx}) + V_{xx}66e_{xx} + V_{xx}66e_{xx}) + V_{xx}66e_{xx}) + V_{xx}66e_{xx} + V_{xx}66e_{xx}) + V_{xx}66e_{xx}) + V_{xx}66e_{xx} + V_{xx}66e_{xx}) + V_{xx}66e_{xx} + V_{xx}66e_{xx} + V_{xx}66e_{xx}) + V_{xx}66e_{xx} + V_{xx}66e_{xx} + V_{xx}66e_{xx}) + V_{xx}66e_{xx} + V_{xx}66e_{xx} + V_{xx}66e_{xx}) + V_{xx}66e_{xx} + V_{$$

$$+ EI_{uz}[(-U_{,xx}\phi - U_{,x}\phi_{,x} - Z_{s}\phi_{,x}^{2})\delta\phi_{,xx}$$

$$= \phi_{,xx}\{U_{,xx}\delta\phi + \phi\deltaU_{,xx}+U_{,x}\delta\phi_{,x}+\phi_{,x}\deltaU_{,x}-2Z_{s}\phi_{,xx}\phi_{,x}\delta\phi_{,x}\}]$$

$$= (\phi_{,xx}U_{,x} + V_{,x}\phi_{,xx})\delta\phi_{,x} - \phi_{,x}\{\phi_{,xx}(\deltaU_{,x}+\deltaV_{,x}) + (V_{,x}+U_{,x})\delta\phi_{,xx}\}]$$

$$+ P[U_{,x}\deltaU_{,x} + V_{,x}\deltaV_{,x} + \rho_{2}^{2}\phi_{,x}\delta\phi_{,x} + (Z_{s}U_{,x} - Y_{s}V_{,x})\delta\phi_{,x}] + (Z_{s}U_{,x} - Y_{s}V_{,x})\delta\phi_{,x} + \phi_{,x}(Z_{s}\deltaU_{,x} - Y_{,s}\deltaV_{,x})$$

$$+ \phi_{,x}\phi(Z_{s}\deltaV_{,x} + Y_{s}\deltaU_{,x})$$

$$+ (Z_{s}V_{,x}+Y_{,s}U_{,x})\phi_{,x}\delta\phi + (Z_{s}V_{,x}+Y_{s}U_{,x})\phi\delta\phi_{,x}]$$

$$+ \frac{1}{2}Q_{y}[(V_{,x}-\delta U_{,x})\deltaw_{,x}+W_{,x}(\delta V_{,x}-U_{,x}\delta\phi-\phi\delta U_{,x})$$

$$+ 2(\phi\delta U_{,x}+U_{,x}\delta\phi+V_{,x}\phi\delta\phi+\frac{1}{2}\phi^{2}\delta V_{,x})$$

$$+ Z_{s}(\frac{1}{2}\phi^{2}\delta\phi_{,x} + \phi_{,x}\delta\phi)]$$

$$- Y_{s}(\phi\delta\phi_{,x} + \phi_{,x}\delta\phi)$$

$$- 2(\phi\delta U_{,x}+U_{,x}\delta\phi+U_{,x}\phi\delta\phi+\frac{1}{2}\phi^{2}\delta U_{,x})$$

$$+ Z_{s}(\frac{1}{2}\phi^{2}\delta\phi_{,x} + \phi_{,x}\delta\phi)$$

$$- 2(\phi\delta U_{,x}+U_{,x}\delta\phi+U_{,x}\phi\delta\phi+\frac{1}{2}\phi^{2}\delta U_{,x})$$

$$+ Z_{s}(\frac{1}{2}\phi^{2}\delta\phi_{,x} + \phi_{,x}\delta\phi)$$

$$- Y_{s}(\phi\delta\phi_{,x} + \phi_{,x}\delta\phi)$$

$$+ Y_{s}(\phi\delta\phi_{,x} + \phi_{,x}\delta\phi)$$

$$+ Y_{s}(\phi\delta\phi_{,x} + \phi_{,x}\delta\phi)$$

$$+ W_{y}(U_{,x}\delta\phi_{,x}+\phi_{,x}\delta\psi)$$

$$+ W_{y}(U_{,x}\delta\phi_{,x}+\phi_{,x}\delta\phi)$$

$$+ W_{y}(U_{,x}\delta\phi_{,x}+\phi_{,x}\delta\phi)$$

$$+ W_{y}(U_{,x}\delta\phi_{,x}+\phi_{,x}\delta\phi)$$

$$+ W_{y}(\delta\psi_{,x}-U_{,x}\phi\phi_{,x})$$

$$+ W_{y}(\delta\psi_{,x}-U_{,x}\phi\phi_{,x})$$

$$+ W_{x}(\delta\psi_{,x}-U_{,x}\phi\phi_{,x})$$

$$+ W_{x}(\delta\psi_{,x}-U_{,x}\phi\phi_{,x})$$

$$+ W_{x}(\delta\psi_{,x}-U_{,x}\phi\phi_{,x})$$

- 64 -

•

+ 
$$M_{z}[V_{,x}\delta\phi_{,x}+\phi_{,x}\delta^{V}_{,x}-U_{,x}\phi_{,x}\delta\phi_{,x}-\phi\phi_{,x}\delta^{U}_{,x}]$$
  
=  $(V_{,xx}\phi_{}+U_{,xx}+V_{,x}\phi_{,x})\delta^{W}_{,x}$   
=  $W_{,x}[\phi\delta V_{,xx}+V_{,xx}\delta\phi_{}+\delta^{U}_{,xx}+V_{,x}\delta\phi_{,x}+\phi_{,x}\delta^{V}_{,x}]]$   
=  $M_{w}[\phi_{,xx}\delta^{W}_{,x}+W_{,x}\delta\phi_{,x}-U_{,x}\delta^{V}_{,xx}-V_{,xx}\delta^{U}_{,x}$   
+  $\frac{M_{x}}{2}(V_{,x}\delta^{U}_{,xx}+U_{,xx}\delta^{V}_{,x}-U_{,x}\delta^{V}_{,xx}-V_{,xx}\delta^{U}_{,x}$   
+  $(U_{,x}^{2}-V_{,x}^{2}+\phi^{2})\delta\phi_{,x}+2\phi_{,x}(U_{,x}\delta^{U}_{,x}-V_{,x}\delta^{V}_{,x}+\phi\delta\phi$   
=  $2(U_{,x}U_{,xx}\delta\phi_{}+U_{,x}\phi\delta^{U}_{,xx}+U_{,xx}\phi\delta^{U}_{,x})$   
=  $2(U_{,x}U_{,xx}\delta\phi_{}+U_{,x}\phi\delta^{U}_{,xx}+U_{,xx}\phi\delta^{U}_{,x})$   
=  $2(U_{,x}U_{,xx}\delta\phi_{}+U_{,x}\phi\delta^{U}_{,xx}+U_{,xx}\phi\delta^{U}_{,x})$   
=  $M_{z}\phi_{0}[\delta^{U}_{,xx}-U_{,x}\delta\phi_{,x}+\phi_{,x}\delta^{U}_{,x}]$   
+  $M_{z}\phi_{0}[\delta^{U}_{,xx}+V_{,x}\delta\phi_{,x}+\phi_{,x}\delta^{U}_{,x}]$   
+  $\frac{1}{2}P\phi_{0}[Z_{y}(V_{,x}\delta\phi_{,x}+\phi_{,x}\delta^{V}_{,x})+Y_{y}(U_{,x}\delta\phi_{,x}+\phi_{,x}\delta^{U}_{,x}))$   
=  $Q_{y}\phi_{0}[\delta^{V}_{,x}+Y_{y}\delta\phi_{,x}+2Z_{y}\delta\phi]$   
+  $(D_{1}U_{,xx}+D_{5}V_{,xx}+D_{6}\phi_{,xx})\delta^{U}_{,xx}$   
+  $(D_{2}V_{,xx}+D_{4}U_{,xx}+D_{7}\phi_{,xx})\delta\phi_{,xx}+\langle P \rangle \{\delta r\} ]dx$  (3.66)

If the higher order terms such as  $U_{,x}\phi_{,x}V_{,xx}$ ,  $\phi^{2}U_{,xx}$ ,  $\phi_{,xx}U_{,xx}\phi$ , etc. are neglected in the above equation it yields

$$\delta I = \int_{0}^{1} \left[ EA[W_{,x} - \overline{Y}U_{,xx} - \overline{Z}V_{,xx} - \frac{Es_{\omega}}{EA} + xx + U_{,xx} + V_{,xx}] \delta W_{,x} \right]$$

$$= EAW_{,x} \left[ \overline{Z} \delta V_{,xx} + \overline{Y} \delta U_{,xx} + \delta V_{,xx} + \delta U_{,xx} + \frac{ES_{\omega}}{EA} \delta + xx) \right]$$

$$EI_{\omega} + xx^{\delta} + xx^{+GK} + x^{\delta} + x^{\delta} + x^{\delta} + xx^{\delta} + xx^{\delta}$$

$$- M_{2} e_{0} [\delta V_{,xx} - U_{,x} \delta e_{,x} - e_{,x} \delta U_{,x}]$$

$$+ M_{2} e_{0} [\delta U_{,xx} + V_{,x} \delta e_{,x} + e_{,x} \delta V_{,x}]$$

$$+ \frac{1}{2} P e_{0} [Z_{5} (V_{,x} \delta e_{,x} + e_{,x} \delta V_{,x}) + Y_{5} (U_{,x} \delta e_{,x} + e_{,x} \delta U_{,x})$$

$$- Q_{y} e_{0} [\delta V_{,x} + Y_{5} \delta e_{,x} - 2Y_{5} e_{}]$$

$$+ (D_{3} e_{,xx} + D_{6} U_{,xx} + D_{7} V_{,xx}) \delta e_{,xx} + CP [\delta r] ] dx$$

$$+ (D_{3} e_{,xx} + D_{6} U_{,xx} + D_{7} V_{,xx}) \delta e_{,xx} + CP [\delta r] ] dx$$

$$+ C = 0$$

$$+ D = 0$$

$$+ E = 0$$

$$+ D = 0$$

•

$$+ (D_{1}U_{,xx} + D_{5}V_{,xx} + D_{6}\Phi_{,xx})\delta U_{,xx}$$

$$+ (D_{2}V_{,xx} + D_{4}U_{,xx} + D_{7}\Phi_{,xx})\delta V_{,xx}$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

$$+ (D_{3}\Phi_{,xx} + D_{6}U_{,xx} + D_{7}V_{,xx})\delta \Phi_{,xx} + \langle P \rangle \{\delta r\} ] dx$$

Four equations for longitudinal, in-plane, out-of-plane, and torsional actions have been obtained from equation 3.66. the longitudinal equilibrium equation can be obtained by minimizing equation 3.66 as  $\frac{\partial \Pi}{\partial W} = 0$ , for the in-plane and out-of-plane conditions the equilibrium can be written as  $\frac{\partial \Pi}{\partial U} = 0$ ,  $\frac{\partial \Pi}{\partial V} = 0$ , and in the same manner the torsional equation  $\frac{\partial \Pi}{\partial \Phi} = 0$ . Integration by parts and use of the calculus of variations (Erwin (1976)) are incorporated in the analysis; these equations can be represented as:

$$\frac{\text{Longitudinal equation}}{\text{EA[W}, x^{+}\frac{1}{2}(V_{,x}^{2}+U_{,x}^{2}+\rho_{0}^{2}\phi_{,x}^{2}) - Y(V_{,xx}\phi + U_{,xx} + V_{,x}\phi_{,x})}{+ \overline{Z}(V_{,xx}-U_{,xx}\phi - U_{,x}\phi_{,x}) - Y_{s}\phi_{,x}V_{,x} + Z_{s}\phi_{,x}U_{,x}] - ES_{\omega}\phi_{,xx}}{- (V_{,xx}\phi + U_{,xx} + V_{,x}\phi_{,x}) - Y_{s}\phi_{,xx} - U_{,xx}\phi - U_{,x}\phi_{,x})}$$

$$- M_{\omega}\phi_{,xx} - M_{y}(V_{,xx}-U_{,xx}\phi - U_{,x}\phi_{,x})$$

$$- M_{z}(V_{,xx}\phi + U_{,xx} + V_{,x}\phi_{,x})$$

+ 
$$\frac{Q_y}{2}(V_{,x}-\Phi U_{,x}) - \frac{Q_y}{2}(U_{,x}+\Phi V_{,x}) + \overline{P}=0$$
 (3.69)

.

$$\frac{\text{In-plane equation}}{\text{EAW}_{,x} \{ (V_{,x} + Z_{s}^{\phi}, x)^{-\overline{Z}}Y_{s}^{\nabla}V_{,x}^{\phi}, xx + Z_{s}^{(U}, xx^{\phi}, x^{+\phi}, xx^{U}, x) \} \\ + EA\{ (U_{,xx}U_{,x} + (Y_{s}^{2} + Z_{s}^{2})^{\phi}, x^{\phi}, xx)^{+U}, xx^{\nabla}V_{,x}^{+\overline{Y}}Y_{s}^{\nabla}V_{,xx}^{\phi}, x\} \\ + ES_{\omega}^{(\phi}, xx^{\nabla}V_{,x} + Y_{s}^{\phi}, xx^{\phi}) \\ - EI_{y}^{(U_{,xxx}^{\phi} + 2\overline{B}_{z}^{\phi}, x^{\phi}, xx)} \\ + EI_{y}^{(-V_{,xxx}^{+U}, xxx^{\phi+2U}, xx^{\phi+U}, x^{\phi}, xx^{+U}, x^{\phi}, xx]} \\ - EI_{yz}^{(U_{,xxx}^{+2\phi V}, xxx^{+2V}, xx^{\phi}, x^{+V}, x^{\phi}, xx^{+U}, x^{\phi}, xx]} \\ - EI_{z\omega}^{(\phi}, xx^{\phi}, x^{+\phi}, x^{+\phi}, xxx] - EI_{\omega y}^{(\phi}, xxx^{\phi} - \phi, x^{\phi}, xx] \\ + P[V_{,x}^{--Y_{s}^{\phi}}, x^{+Z_{s}^{\phi}}, x^{\phi}] \\ - \frac{1}{2}Q_{z}^{(W_{,x}^{\phi+2W_{,x}^{\phi})} + \frac{1}{2}Q_{y}^{(W_{,x}^{+\phi^{2}})} \\ + M_{z}^{\phi}, x^{-M_{y}^{\phi}\phi}, x^{-M_{,x}^{(\phi}, x^{V}, x^{-U}, xx]} + \overline{M}_{z}^{=0}$$
(3.70)

$$\frac{\text{Out-of-plane equation}}{\text{EAW}, x^{\{(U,x^{-Y}s^{\Phi},x^{)-\overline{YY}s^{U},x^{\Phi},xx^{-Y}s^{(V,xx^{\Phi},x^{+\Phi},xx^{V,x})\}}$$

$$= \text{EA}\{(V,xx^{V},x^{+}(Y_{s}^{2}+Z_{s}^{2})^{\Phi},x^{\Phi},xx^{)+V},xx^{U},x^{+\overline{ZZ}}s^{V},xx^{\Phi},x^{\}}$$

$$+ \text{ES}_{\omega}(\Phi,xx^{U},x^{-}Z_{s}^{\Phi},xx^{\Phi})$$

$$+ \text{EI}_{y}\{V,xxx^{\Phi} - 2\overline{B}_{y}\Phi,x^{\Phi},xx^{\}}$$

$$= \text{EI}_{z}[U,xxx^{+V},xxx^{\Phi+2V},xx^{\Phi+V},x^{\Phi},xx^{-U},x^{\Phi},xx^{]}$$

.

- 69 -

$$= EI_{yz}[V_{,xxx}^{-2\phiU_{,xxx}^{-2U_{,xx}^{,x}}, x^{-V_{,x}^{,x}}, x^{-U_{,x}^{,xx}}]$$

$$= EI_{yu}[\phi_{,xx}^{,x}, x^{+}, \phi_{,xxx}^{,x}] + EI_{uz}[\phi_{,xxx}^{,x} - \phi_{,x}^{,xx}]$$

$$+ P[U_{,x}^{,x}, z^{s}, x^{+}, Y^{s}\phi_{,x}^{,x}]$$

$$= \frac{1}{2}Q_{y}(W_{,x}^{,s+W_{,x}^{,x}}) - \frac{1}{2}Q_{z}(W_{,x}^{,s+\Phi^{2}})$$

Torsional equation

- EAW, x [p<sup>2</sup>-YV, x + ZU, x + Y<sub>S</sub>U, x + Z<sub>S</sub>V, x V, x + U, x] + EAY[Y<sub>S</sub>U, x x V, x - Z<sub>S</sub>U, x x U, x + U, x x <sup>¢</sup>, x (Y<sup>2</sup><sub>S</sub> + Z<sup>2</sup>)]
- + EAZ[Y<sub>s</sub>V<sub>,xx</sub>V<sub>,x</sub>-Z<sub>s</sub>V<sub>,xx</sub>U<sub>,x</sub>+V<sub>,xx</sub>, (Y<sub>s</sub>+2<sub>s</sub>)]
- + Esu[-Y,xx<sup>\*</sup>,x+<sup>2</sup>s<sup>0</sup>,xx<sup>\*</sup>,x<sup>-U</sup>,xx<sup>U</sup>,x<sup>-V</sup>,xx<sup>V</sup>,x
- +  $\phi_{,x}\phi_{,xx}(Y_{S}^{2}+Z_{S}^{2})$ ]  $EI_{\omega}\phi_{,xxx}$  +  $GK\phi_{,x}$
- EIy[V, x x x V, x + V, x x V, x 2<sup>B</sup>y U, x x <sup>Φ</sup>, x<sup>1</sup>
- EI<sub>z</sub>[U, xx<sub>x</sub>V, x-U, xxV, x+U, xx<sup>U</sup>, x-2<sup>B</sup>zV, xx<sup>Φ</sup>, x<sup>1</sup>
- EI<sub>wy</sub>[U,xxx+V,xxx<sup>¢+¢</sup>,xxx(V,x-V)+¢,xV,xx+U,xx<sup>¢</sup>,x<sup>]</sup>
- + EI<sub>wz</sub>[-V,xxx+U,xxx¢-¢,xxx(U,x-U)+¢,xU,xx+V,xx¢,x]
- + P[p<sup>2</sup>¢,x-(Y<sub>s</sub>V,x-Z<sub>s</sub>U,x)
- $+(Z_{s}^{\phi}, x_{x}^{V} + Y_{s}^{\phi}, x_{x}^{U}) + (Z_{s}^{V}, x_{x}^{+} Y_{s}^{U}, x^{)\phi}]$ -  $Q_{y}[V_{y}^{V}, x_{\phi}^{+} Z_{s}^{\frac{\phi^{2}}{2}}] - Q_{z}[U_{y}^{U}, x_{\phi}^{0} - Y_{s}^{\frac{\phi^{2}}{2}}]$
- 10

(3.71)

+ 
$$M_y[U_{,x}-V_{,x}\Phi+W_{,x}U_{,x}-V_{,xx}\Phi]$$
  
+  $M_z[V_{,x}-U_{,x}\Phi-W_{,x}V_{,x}-U_{,xx}\Phi]$   
+  $\frac{M_x}{2}[U_{,x}^2+V_{,x}^2-\Phi^2-\Phi_{,x}\Phi] + \overline{M}_t = 0$  (3.72)  
Neglecting higher order terms  $U_{,xx}\Phi_{,x}$ ,  $V_{,xx}V_{,x}$ ,  $U_{,xx}U_{,x}$ , etc.

from equations 3.70 3.71, 3.72 yields the following equations

Longitudinal equation

$$EAW_{,x} + \overline{P} = 0 \tag{3.73}$$

# In-plane equation

- +  $EI_zU_{,xxx} EI_yV_{,xxx} EI_{yz}U_{,xxx} EI_{z\omega}\Phi_{,xxx}$
- +  $P[V_{,x} Y_{s}\phi_{,x}] Q_{z}W_{,x} + Q_{y}W_{,x}$
- +  $M_z \Phi_{,x}$  +  $M_{,x} U_{,xx}$  +  $\overline{M}_z = 0$  (3.74)

Out-of-plane equation

- +  $EI_{z}U_{,xxx} EI_{y}U_{,xxx} EI_{yz}V_{,xxx} EI_{y\omega}\Phi_{,xxx}$
- +  $P[U_{,x} + Z_{s}\phi_{,x}] Q_{z}W_{,x} + Q_{y}W_{,x}$
- $+ M_z \Phi_{,x} + M_{,x} U_{,xx} + \overline{M}_z = 0$  (3.75)

Torsional equation

$$= EI_{\omega} \phi_{xxx} + GK_{\phi} \phi_{x} - EI_{\omega}yU_{xxx} - EI_{\omega}zV_{xxx}$$

+ 
$$P[\rho_0^2 \phi_x - Q_y V - Q_z U$$

$$+ M_v U_{,x} + M_z V_{,x} + \overline{M}_t = 0$$
 (3.76)

Equations 3.73 - 3.76 represent the equilibrium equations of a beam-column in 3-dimensions. Those equations may be shown to reduce to those obtained by Culver (1966), Vlasov (1961), Timoshenko and Gere (1961), etc. if certain limitations are imposed.

#### 3.6 Comparison with Previous Formulations

The result given by equation 3.60 is identical to that obtained by Barsoum and Gallagher (1970) for a doubly symmetric I-beam ,where the effects of the initial imperfections and higher order terms were ignored. If the nonlinear terms and the product of the derivatives of the axial displacements with respect to x, y and x, z ( $W_{,x}W_{,y}$  and  $W_{,x}W_{,z}$ ) are neglected and the effect of residual strains is included ,then equation 3.60 gives virtually the same results as that derived previously by Rajasekaran (1971)

The effect of higher order terms, using a principal axes basis ,but neglecting the effect of initial imperfections and the  $W_{,x}W_{,y}$  and  $W_{,x}W_{,z}$  coupling, has been included in the nonlinear equilibrium equation derived by Attard (1986). His results agree with equation 3.60. Similarly the results of equation 3.63 agree with the solution by Roberts and Azizian (1983), (whose basic assumption were subsequently used by Attard) when the loads are limited to those included by Roberts and Azizian.

- 72 -

The stiffness matrices obtained by Yang and McGuire (1986) for symmetric sections, using the principal axes, neglecting higher order terms and residual stresses and including the products of derivatives of the axial displacements with respect to x, y and x, z respectively i.e ( $W_{,x}W_{,y}$  and  $W_{,x}W_{,z}$  not equal to zero) are similar to equation 3.57 when the higher order terms such as ( $\Phi^2_{,x}U_{,xx}$ ,  $V_{,xx}\Phi\Phi_{,xx}$ , etc.) are assumed equal to zero.

The present formulation enables the limitations of earlier studies (Barsoum and Gallagher (1970), Rajasekaran (1971), Rajasekaran and Murray (1973), Vinnakota (1977), Nishida and Fukumoto (1985), and Attard (1986)) to be identified. Table 3.1 summarises the findings. Table 3.2 presents a comparison between the assumptions that have employed by the Author and those used previously [Rajasekaran (1971), Rajasekaran and Murray (1973), and Chen and Atsuta (1976)] to develop a formulation for beam-columns in space. Complete derivations of the linear and nonlinear tangential, geometrical, and strain-displacements matrices are presented in Chapter 4. In this context it is particularly important to note that further use of these formulations as the basis for inelastic analysis of biaxial bending leads to effective crosssections for which several assumptions that might have been valid within the elastic range e.g. doubly symmetric, shear centre and centroid coincide etc., are no longer correct. The importance of not resorting to such simplifications is clearly illustrated by the use of the present formulation in a variety of example problems (El-Khenfas and Nethercot (1987b,c), and chapters 6 and 7 respectively. Specific limitations in previous studies include:

- 73 -

- 1- Neglect of higher order terms and/ or the effects of residual stress .
- 2- Using the principal axes for an arbitrary cross-section and neglecting the shift of the shear centre and the centroid even in the elastic range.

# 3.7- Conclusions

A method has been used to derive the general equilibrium condition for thin walled beams of arbitrary cross-section under general loading. The effects of higher order terms and initial imperfections have been included. The method is based on the principle of virtual work, where the product of the derivatives of axial displacements with respect to x, y and x, z ( $W_{,x}W_{,y}$  and  $W_{,x}W_{,z}$ ) are accounted for in equation 3.60 or alternatively are neglected in equation 3.57. An alternative derivation is also possible, based on the use of energy theorems leading to equation 3.63.

Both methods compare very well with the available theories ,permitting their simplifications and limitations to be clearly identified. The validity and accuracy of this theory has been demonstrated in several applications (El-Khenfas and Nethercot (1987b,c)) and Chapters 6/7 respectively, ,which cover both elastic and inelastic analysis.

- 74 -

Table 3.1 Comparison between the author's formulation and previous studies

	Author	Method of analysis	Analysis type	Imperfections	Axis Bystem	H.O.T. of strain disp. relations	Product of W, XW, YEW, XW, Z	ŀ			
								Kg	ĸz	of strains	Notes
1970	Barsoum and Gallagher	En	E	N	Principal	N	N	N	N	N	1
1971	Rajasckaran	V.W.	E, I	Y	General	N	N	N	N	N	2
1983	Roberts and Azizian	En	E	Y	Principal	Y	N	-	-	-	3
1986	Attard	En	E	-	Principal	Y	Y	N	N	N	4
1977	Vinnokuta	Finite of difference	E, I	Y	General	N	N	N	N	N	-
1985	Pukumoto & Nishida	V.W.	E,I	Y	General	N	N	N	N	N	
1986	Yang and McGuir	V.W.	E,I	Y	Principal	N	Y	N	N	N	-
-	Authors' formulation	V.W.,En	E, I	Y	General	Y	Y	Y	Y	Y	5

1. Barsoum and Gallagher's work is included since it is regarded as the first coverage of torsional-flexural effects by the finite element method.

2. More details are given in Table 3.2

 This work was performed for general types of cross-sections, but the Wagner effect was not included; it is not clear exactly how non-linear strains were incorporated.

4. Imperfections were ignored (not mentioned).

5. Even for elastic analysis inclusion of higher order terms in the solution has a significant effect on the results, see Chapter 6

and Chan & Atsuta (1977) of deriving the general differential equilibrium equation

Item		Descriptions Refs. (above ) Authors formulations							
1	$\sin\phi = \phi$ , $\cos\phi = 1$			$\sin\phi = \phi - \frac{\phi^3}{3}, \ \cos\phi = 1 - \frac{\phi^2}{2}$					
2		V <sub>3</sub> , U <sub>3</sub> shear centre displacements		V <sub>3</sub> , U <sub>3</sub> displacements of point S V = V <sub>3</sub> + V <sub>0</sub> V <sub>0</sub> , U <sub>0</sub> initial displacements U = U <sub>3</sub> + U <sub>0</sub> $\phi = \phi + \phi_0$					
3	In-plane displacement	$V_A = V_3 + (Y - Y_0)\phi$		$V_A = V + (Y - Y_3)Sin\phi + (Z - Z_3)[1 - Cos\phi]$					
4	Out-of-plane displacement	$U_{A} = U_{3} + (Z - Z_{5})\phi$		$U_A = U - (Y - Y_S)[1 - Cos\phi] + (Z - Z_S)Sin\phi$					
5	Shear strain	$\frac{\partial \eta}{\partial x} + \frac{\partial w}{\partial s} = 0$	2	$\frac{\partial \eta}{\partial x} + \frac{\partial w}{\partial s} + \frac{\partial u}{\partial x} \frac{\partial u}{\partial x} + \frac{\partial v}{\partial s} \frac{\partial v}{\partial x} = 0$	2				
6	Longitudinal displacement	$W = W_{\rm C} - 2V_{\rm X} - YU_{\rm X} + W_{\rm B} \Phi_{\rm X}$	3	$W = W_{c} - Y[V, XSin\phi + U, XCos\phi] - Z(V, XCos\phi - U, XSin\phi) + W_{n}\phi$	3				
7	Axial strain	$\varepsilon = v_{1X} - ZV_{1XX} - YU_{1XX} + v_n \Phi_{1XX}$	4	$e = W_{,X} + \frac{1}{2} \begin{bmatrix} 2 \\ y_{,X} + v_{,X}^{2} + \rho_{,X}^{2} \end{bmatrix} - \frac{1}{2} Y_{5} \phi_{,X} (V_{,X} - U_{,X} \phi) - Z_{5} \phi_{,X} (U_{,X} + V_{,X} \phi) \\ - \frac{2}{2} Y \begin{bmatrix} 0 \\ y_{,XX} + V_{,XX} \phi - U_{,XX} \frac{\phi^{2}}{2} \end{bmatrix} - Z \begin{bmatrix} v_{,XX} - v_{,XX} \frac{\phi^{2}}{2} - U_{,XX} \phi \end{bmatrix} + \overline{w}_{0} \phi_{,XX}$					
8	Product of derivatives of axial dis- placement	$W_{,\chi} W_{,\gamma}$ and $W_{,\chi} W_{,Z} = 0$	5	$W_{X} W_{Y}$ and $W_{X} W_{Z} \neq 0$	5				
9	Linear K <sub>G</sub>	+		Torsional moment and shear forces are included	6				
10	Non-linear stiffness matrices	Neglected		Considered	7				

- 1. Substitution for SIN¢ and COS¢ takes place after item No. 7.
- 2. Because of the bending and torsional flexibility of thin walled open cross-sections, the effects of shearing strain along the mid surface of the plate segment are extremely small and can be neglected.
- 3. The longitudinal displacement obtained from integrating the equation presented in item No. 4.
- 4. The higher order strain terms such as V,xx4, U,xx4, Y,44, Y,46, X,40, Z,46, XU,X, etc. have significant values and are incorporated in the present approach, but are not considered in Refs. (Rajasekaran (1970), Rajasekaran & Murray (1973), and Chen & Atsuta 1977)
- 5. The product of the derivative of axial displacements with respect to X/Y and X/Z ( $W, _{X}W, _{Y}$  and  $W, _{X}W, _{Z}$ ) are accounted for but neglected by other authors as higher order terms, while  $W, _{X}^{2}$  is ignored in all cases.
- 6. Inclusion of the previous step affects even the linear geometrical stiffness matrix through inclusion of the terms presented in Chapler 4 which are not given by Ref. (Rajasekaran (1970), Rajasekaran & Murray (1973), and Chen & Atsuta 1977)
- 7. The non-linear matrices (tangential, geometrical, strain-displacement) have an important effect because in thin-walled sections the coupling between the displacements and rotations are significant. Neglecting this coupling leads to overestimates of load carrying



(a) Cross section of thin walled open section



 (b) Thin walled beam of open section - A general system of loads

FIG. 3.1 DEFINITION OF PROBLEM







**b** IDEALIZATION OF STRESS - STRAIN CURVE

FIG.3.2 STRESS-STRAIN RELATIONSHIP







FIG 3.4 SECTIONAL COORDINATES OF THIN WALLED SECTION



# FIG.3.5 GENERAL THREE DIMENSIONAL BODY



 $\sigma = \frac{\sigma_F - \sigma_{FW}}{B} X + \sigma_{FW} (For flanges)$  $\sigma = \sigma_W \qquad (For web)$ 

(a) Lehigh distribution



 $\sigma_{FW} = \frac{\sigma_{FW}\sigma_{W}}{(D/2)^{2}} \qquad (Web)$   $\sigma = \frac{\sigma_{FW}\sigma_{W}}{(D/2)^{2}} \qquad (Web)$   $\sigma = \frac{\sigma_{F}-\sigma_{FW}}{(B/2)^{2}} \qquad (Flange)$ 

(b) Parabolic distribution



(c) Triangular distribution

FIG. 3.6 TYPICAL PATTERNS OF RESIDUAL STRESS ADJUSTED TO ENSURE FULL 3-D EQUILIBRIUM

#### Chapter 4

#### Derivation of Stiffness Matrices

# 4.1 Introduction

The most important feature of the matrix stiffness analysis of structures is the formulation of the stiffness matrix for a discrete element. This is required in order that the real structure may be represented by a system with a finite number of degrees of freedom upon which the actual analysis can be performed.

These stiffness matrices are usually formulated by assuming that the displacements within an element vary in a suitable fashion; normally polynomials, whose coefficients are equal to the number of element nodal point displacements, are selected. It is then necessary to introduce these interpolation functions to the equilibrium equations in order to evaluate the element stiffness matrices.

Many previous investigators have derived stiffness matrices for a beam-column element in three dimensions in the form of a 12x12 stiffness matrix for a member having 6 degrees of freedom at each node. Tecan and Mahapatra (1969), produced a tangent stiffness matrix for a space frame member using a Taylor expansion (series). Przemieniecki (1968) gave two methods to obtain the linear tangential and geometrical stiffnesses. The first was based on the displacement method, while the second employed the force method. Based on considerations of nonlinear displacements and the second variation of the strains, Roberts and Azizian (1983) developed both the geometrical and the tangential stiffnesses.

- 82 -

A 14x14 element stiffness matrix for a beam-column has been developed by Tebedge and Tall (1973), Barsoum and Gallagher (1970), Rajasekaran and Murry (1973), Yang and Mcguire (1986), and Kitiporncahi and Chan (1987) by using the strain displacement relationships. The element nodal displacements were taken to be adequately represented by a linear polynomial for the axial displacements, by a cubic polynomial for lateral displacements and twists, and by a quadratic polynomial for the rotations and warping.

#### 4.2- Interpolation Function

In the finite element approach the displacements are approximated by forming shape functions. The functions chosen to represent the solution within the element are most commonly taken as polynomial series for the following reasons :

- a- It is straightforward to perform differentiation or integration and to computerise the processes.
- b- It is straightforward to increase the accuracy of the results by increasing the order of the function.

The approximate function chosen for the displacement model must be capable of reproducing certain features of the true displacements if convergence to the true solution is to be obtained Zienkiewicz (1977), Bathe and Wilson (1976), Richards (1977), Rao (1982). It must be continuous within the element and there must be compatibility between adjacent elements. In general the number of terms in the chosen series must be directly related to the total number of degrees of freedom for the element.

In a one dimensional situation (extensional or flexural behaviour of bars), the required functions for the longitudinal and

- 83 -

flexural displacements may be approximated respectively as

$$\phi_{i}(x) = a_{0} + a_{1}\frac{X}{L}$$
 (4.1)

$$\phi_j(x) = a_0 + a_1 \frac{X}{L} + a_2 \frac{X^2}{L} + a_3 \frac{X^3}{L}$$
 (4.2)

where i=1,2 and j=1,2,3,and 4

Applying the boundary conditions enables the arbitrary constants  $a_0$ ,  $a_1$ ,  $a_2$ , and  $a_3$  to be obtained leading to the following equations, which are illustrated by Richards (1977), Zienkiewciez (1977):

$$\Phi_1(X) = 1 - 3\frac{X^2}{L^2} + 2\frac{X^3}{L^3}$$
 (4.3a)

$$\Phi_2(X) = X - 2\frac{X^2}{L} + \frac{X^3}{L^2}$$
 (4.3b)

$$\Phi_3(X) = -2\frac{X^3}{L^3} + 3\frac{X^2}{L^2}$$
(4.3c)

$$\Phi_{4}(X) = \frac{X^{3}}{L^{2}} - \frac{X^{2}}{L}$$
(4.3d)

Axial displacement can be represented in matrix form as

$$\begin{bmatrix} W \end{bmatrix} = \begin{bmatrix} (1 \frac{X}{L}) & \frac{X}{L} \end{bmatrix} \begin{bmatrix} W_1 \\ W_2 \end{bmatrix}$$
(4.4)

Equation 4.3 represents the flexural displacements ( about the X, Y, and Z axes), it can be re-arranged in matrix form as:

[V], [U], and 
$$[\phi] = [\phi_1(X) \phi_2(X) \phi_3(X) \phi_4(X)] \begin{bmatrix} \delta_1 \\ \delta_2 \\ \delta_3 \\ \delta_4 \end{bmatrix}$$
 (4.5)

- 84 -
## 4.3- Stiffness Matrices

The element stiffness matrices play an important part in the analysis of beam-columns having an arbitrary open cross-section. These stiffnesses can be explained as; tangential stiffness matrix  $[K_E^L]$  is the combination of axial stiffness  $K_{WW}$ , flexural stiffnesses  $K_{UU}$  and  $K_{VV}$  in the U and V directions respectively, torsional stiffness  $K_{\phi\phi}$ , stiffness caused by coupling between W and V  $K_{WV}$ , and similarly for  $K_{WV}$ ,  $K_{W\phi}$ ,  $K_{UV}$ ,  $K_{U\phi}$ , and  $K_{V\phi}$ . The element geometrical stiffness matrix  $[K_G^L]$ , also represents the combination of the effects of the element forces such as : axial forces  $F_x$ , shear forces  $Q_y$  and  $Q_z$ , bending moments  $M_v$  and  $M_z$  and torsion  $M_v$ , etc..

The nonlinear element stiffness and geometrical matrices  $[K_E^{\rm NL}]$  and  $[K_G^{\rm NL}]$  incorporate the higher terms and the coupling between the displacements. Full details are covered later in this section.

The employment of these matrices in assembling the element stiffnesses, imposing boundary conditions, solving the equilibrium equations for nodal displacements, and evaluating the element stresses is explained in full in chapter 5. This section concentrates on the development of the stiffness matrices (geometrical and tangential) of a beam-column in space. These stiffnesses are developed by either the principle of virtual work or by using total potential energy.

The following notation will used in determining the total linear, and nonlinear stiffnesses:

$$[n_{N,I}^{i}] = [n_{1,I}^{i} n_{2,I}^{i} n_{3,I}^{i} n_{4,I}^{i}]$$

$$i = \text{the derivative degree with respect to X}$$

$$i = 1 \text{ to } 4$$

$$(4.6)$$

I = kind of function (O linear and 1 cubic )

Example

$$\begin{bmatrix} K_{VU\phi}^{ijk} \end{bmatrix} = \int_{0}^{L} n_{N,I}^{i} \cdot n_{N,J}^{j} \cdot n_{N,K}^{k} dx$$
(4.7)  

$$k, j, i = \text{the derivative degree with respect to X}$$

$$K, J, \text{ and } I = 1 \qquad \text{cubic function}$$

$$0 \qquad \text{linear function}$$

$$N = 1, 2, 3, 4$$

## 4.3.1- Principle of Virtual Work

The stiffness matrices explained herein are obtained by substituting equation 4.5 into equation 3.56 and then carrying out the integration by assuming that parameters such as  $EI_y$ ,  $EI_z$ , GK,  $M_y$ , M ,..., etc. are calculated at each node. The averages of the values evaluated at the nodes of the element are used in the stiffnesses. Some rearrangement needs to be introduced into equation 3.56 in order to: obtain both linear and nonlinear geometrical and tangential stiffness matrices which are classified as:

- 1- Linear tangential stiffness matrix  $[K_r^1]$
- 2- Linear geometrical stiffness matrix  $[K_G^1]$
- 3- Nonlinear tangential stiffness matrix  $[K_E^{N1}]$
- 4- Nonlinear geometrical stiffness matrix  $[K_{C}^{NL}]$

## 4.3.1.1 Linear Tangential Stiffness Matrix

The Linear tangential stiffness matrix  $[K_E^L]$  incorporates terms such as GK,  $EI_Y$ ,  $EI_z$ , warping EI which represent torsional, flexural and warping rigidities for the member. It is defined by equation 4.8. This equation is expressed in Table 4.1.

- 86 -

$$[K_{E}^{L}] = \begin{bmatrix} a & b & c & d \\ & e & f & g \\ & & h & j \\ & & & i \end{bmatrix}$$
(4.8)  
where

$$a=K_{WW}^{11}$$
,  $b=\overline{Y}K_{UW}^{12}$ ,  $e=K_{UU}^{22}$ ,  $c=\overline{Z}K_{UW}^{12}$ 

$$f = K_{UV}^{22}$$
,  $d = S_{\omega} K_{\phi W}^{12}$ ,  $g = K_{\phi U}^{22}$ 

 $j=K_{\phi\phi}^{22}$ ,  $h=K_{VV}^{22}$ ,  $i=K_{\phi\phi}^{22}$ 

## 4.3.1.2 Linear Geometrical Stiffness Matrix

The Linear geometrical matrix [ $K_{G}^{L}$ ]) includes parameters for the shear forces ( $Q_{y}, Q_{z}$ ), bending moments ( $M_{y}, M_{z}$ ), and torsional moment ( $M_{x}$ ), etc.. The full definition of the terms of equation 4.9 is presented in Table 4.2.

$$[K_{G}^{L}] = \begin{bmatrix} A & B \\ C & D & E \\ F & G \\ H \end{bmatrix}$$
(4.9)

where

A= 
$$Q_y K_{UW}^{11}$$
, B= $Q_z K_{VW}^{11}$ , C= $F_x K_{UU}^{11}$ 

 $D=M_x(K_{VW}^{11} - K_{VU}^{12})$ ,  $F=F_xK_{VV}^{11}$ ,  $H=M_pK_{V\Phi}^{10}$ 

$$E = (Z_sF_x + M_y)K_{U\Phi}^{11} + Q_zK_{U\Phi}^{10}$$

$$\mathbf{u} = (\mathbf{Y}_{\mathbf{S}}\mathbf{F}_{\mathbf{X}} + \mathbf{M}_{\mathbf{Z}})\mathbf{K}]_{\Phi}^{11} + \mathbf{Q}_{\mathbf{Y}}\mathbf{K}]_{\Phi}^{10}$$

## 4.3.1.3 Monlinear Tangential Stiffness Matrix

The nonlinear tangential stiffness matrix  $[K_E^{NL}]$  is given in equation 4.10. It incorporates the influence of the higher order terms and their coupling. Details of the terms are presented in Table 4.3.

$$[K_{E}^{NL}] = \begin{bmatrix} \overline{A} & \overline{B} & \overline{C} \\ \overline{D} & \overline{E} & \overline{F} \\ & \overline{G} & \overline{H} \\ & & \overline{I} \end{bmatrix}$$
(4.10)

where

$$\overline{A} = EAK_{WVU}^{111}, B_{1} = EAK_{WUV}^{111}$$

$$\overline{C} = EA\{-\frac{Y}{2}K_{VV\phi}^{120} + \overline{Z}K_{WV\phi}^{120} - K_{WU\phi}^{111}$$

$$\overline{D} = \frac{EAY}{2}K_{UUU}^{112} - EI_{YZ}K_{U\phiU}^{202} + EI_{YZ}K_{U\phiU}^{202}$$

$$\overline{E} = \frac{EAY}{2}K_{VVU}^{112} - \frac{EA\overline{Z}}{2}K_{UUV}^{112} + EI_{YZ}K_{U\phiV}^{202}$$

$$\overline{F} = EAY[Z_{S}K_{VVU}^{111} + Y_{S}K_{\phi\phiU}^{112}] + \frac{1}{2}EA(Y_{S}^{2} + Z_{S}^{2})K_{\phi\phiU}^{112} +$$

$$ES_{W}K_{UU\phi}^{112} - EI_{YZ}[K_{UV\phi}^{211} + K_{\phi\phiU}^{112} + K_{UU\phi}^{220}] +$$

$$EI_{Z}[K_{UV\phi}^{211} - K_{UV\phi}^{220} - Y_{S}K_{\phi\phiU}^{112} + EI_{Y}[K_{UV\phi}^{121} + Z_{S}K_{\phi\phiU}^{112} - K_{UV\phi}^{220}]$$

$$\overline{H} = EA\overline{Z}[Z_{S}K_{\phiUV}^{111} + Y_{S}K_{\phiUV}^{112}] + \frac{1}{2}EA(Y_{S}^{2} + Z_{S}^{2})K_{\phi\phiV}^{112}$$

+  $ES_{uvv\phi}^{K_{112}} - EI_{yz}[K_{vv\phi}^{211} + Z_sK_{\phi\phiv}^{112} + K_{uv\phi}^{220}]$ 

+ 
$$EI_{z}[K_{vu\phi}^{211} - K_{vu\phi}^{220} - Y_{s}K_{\phi\phiv}^{112}]$$
+  $EI_{y}[K_{vu\phi}^{211} + Z_{s}K_{\phi\phiv}^{112} - K_{vu\phi}^{220}]$   
 $\overline{C} = EA_{\overline{2}}K_{vvv}^{112} - EI_{y}K_{v\phiv}^{211} + EI_{yz}K_{v\phiv}^{202}$   
 $\overline{T} = ES_{\omega}[Y_{s}K_{\phiv\phi}^{112} - Z_{s}K_{\phiu\phi}^{112} + \frac{1}{2}(Y_{s}^{2} + Z_{s}^{2})K_{\phi\phi\phi}^{202}]$   
+  $EI_{y\omega}[K_{\phiv\phi}^{220} + K_{\phiv\phi}^{211} + Y_{s}K_{\phi\phi\phi}^{112}]$ 

+  $EI_{\omega z}[K_{\phi u \phi}^{220} + K_{\phi u \phi}^{211} + Z_{\phi \phi \phi}^{112}]$ 

## 4.3.1.4 Nonlinear Geometrical Stiffness Matrix

The nonlinear geometrical stiffness matrix  $[K_G^{N1}]$  includes the full higher order terms which are described in equation 4.11. The expressions for the terms of equation 4.11 are fully explained in Table 4.4.

$$[K_{G}^{NL}] = \begin{bmatrix} A_{1} & B_{1} & C_{1} \\ D_{1} & E_{1} & F_{1} \\ & G_{1} & H_{1} \\ & & I_{1} \end{bmatrix}$$
 (4.11)

where

$$A_1 = -Q_z K_{w\phi v}^{101}$$
,  $B_1 = -Q_y K_{w\phi v}^{101}$ ,  $C_1 = -Q_y K_{wv\phi}^{110} - Q_z K_{wu\phi}^{110}$ 

$$D_{1} = -Q_{y} Z_{s} [K_{u\phi u}^{102} + K_{u\phi u}^{211} + K_{u\phi u}^{101}]$$

$$\begin{aligned} -Q_{z}Y_{s}\{K_{u}^{111} + K_{u}^{211}\} + M_{x}\{K_{u}^{201} + K_{u}^{120} - k_{u}^{111}\} \\ C_{1} = -Q_{z}Y_{s}[K_{v\phiv}^{102} + K_{v\phiv}^{211} + K_{v\phiv}^{101}] \\ -Q_{z}Y_{s}\{K_{v\phiv}^{111} + K_{v\phiv}^{211}\} + M_{x}\{K_{v\phiv}^{201} + K_{v\phiv}^{120} - k_{v\phiv}^{111}\} \\ E = -Q_{y}Z_{s}K_{u\phiv}^{211} - Q_{z}Y_{s}K_{u\phiv}^{211} \\ F = [PY_{s} - M_{z}]K_{\phi\phiu}^{101} + Q_{z}Y_{s}[K_{uu\phi}^{120} + K_{uv\phi}^{111}] \\ -Q_{y}Z_{s}\{K_{uv\phi}^{121} + K_{uu\phi}^{121} + K_{uv\phi}^{211}] + M_{x}\{K_{uu\phi}^{210} - K_{uu\phi}^{120} + K_{uu\phi}^{111}\} \\ H = [PZ_{s} - M_{y}]K_{\phi\phiv}^{101} + Q_{z}Y_{s}[K_{vv\phi}^{120} + K_{vu\phi}^{121}] \\ -Q_{y}Z_{s}\{K_{vu\phi}^{121} + K_{uu\phi}^{121} - K_{uv\phi}^{120}\} + M_{x}(K_{vv\phi}^{210} - K_{uv\phi}^{120} + K_{uu\phi}^{111}] \end{aligned}$$

The subassemblage of the element stiffnesses stated in equations 4.8 - 4.11 is the total stiffness matrix which is employed in the analysis of beam-columns under different load patterns together with various boundary conditions in the elastic and inelastic ranges as presented in equation 4.12.

$$[K_{t}] = [K_{E}^{L}] + [K_{G}^{L}] + [K_{E}^{NL}] + [K_{G}^{NL}]$$
(4.12)

Further details of the assembly of the full geometrical linear and nonlinear stiffness matrices and the element tangent stiffnesses are explained in chapter 5 and the validity of their implementation in the analysis is demonstrated in chapters 6 and 7 respectively.

- 90 -

## 4.3.2 Total Potential Energy

Recall equation 3.62 from chapter 3 which represents the equilibrium equation of a beam-column of thin-walled open cross-section in three dimensions. Substituting equations 4.5 and 4.7 into 3.62, and carrying out the integrations the equilibrium condition can be obtained as a stationary point of the total potential energy as:

$$\delta \Pi = \frac{\partial \Pi}{\partial W} dW + \frac{\partial \Pi}{\partial U} dU + \frac{\partial \Pi}{\partial U} dV + \frac{\partial \Pi}{\partial \Phi} d\Phi \qquad (4.13)$$

or

$$\delta \Pi = \frac{\partial \Pi}{\partial r} \tag{4.14}$$

where

$$r = [W_1 \quad U_1 \quad \theta_{u1} \quad V_1 \quad \theta_{v1} \quad \Phi_1 \quad \chi_1 \quad W_2 \quad U_2 \quad \theta_{u2} \quad V_2 \quad \theta_{v2} \quad \Phi_2 \quad \chi_2]^T$$
  
After some rearrangement the general form of the equation can be  
written as:

$$\delta \Pi = \int_{0}^{1} \left[ \left[ EA[\langle W \rangle [K_{WW}^{11}] - Y(\langle V \rangle [K_{V\phi W}^{201}] + \langle U \rangle [K_{UW}^{21}] + \langle V \rangle [K_{V\phi W}^{111}] \right) \right] \right]$$

+ 
$$\frac{1}{2}(\langle v \rangle ([K_{vvw}^{111}] + \langle v \rangle [K_{uuw}^{111}] + \frac{\rho_o^2}{EA} \langle \phi \rangle [K_{\phi\phiw}^{111}])$$

$$- \mathbb{Z}(\langle \mathbf{V} \rangle [\mathbf{K}_{\mathbf{V}\mathbf{W}}^{21}] - \langle \mathbf{U} \rangle [\mathbf{K}_{\mathbf{U}\boldsymbol{\phi}\mathbf{W}}^{201}] - \langle \mathbf{U} \rangle [\mathbf{K}_{\mathbf{U}\boldsymbol{\phi}\mathbf{W}}^{111}]) - \frac{\mathbf{ES}_{\omega}}{\mathbf{EA}} \langle \boldsymbol{\Phi} \rangle [\mathbf{K}_{\boldsymbol{\phi}\mathbf{W}}^{21}]$$

$$- Y_{s} < \phi > [K_{\phi uw}^{111}] + Z_{s} < \phi > [K_{\phi vw}^{111}] - < V > [[K_{v\phi w}^{201}] - [K_{vw}^{21}] - [K_{v\phi w}^{111}] \}$$

$$= \langle U \rangle \{ [K_{2}^{0}0_{1}^{1}] - [K_{1}^{0}1_{1}] + [K_{1}^{1}1_{1}^{1}] \} = M_{\omega} \langle \phi \rangle [K_{\phi}^{2}1_{\omega}] \}$$

$$= M_{y}(\langle V \rangle [K_{VW}^{21}] = \langle U \rangle [K_{U\phi W}^{201}] = \langle U \rangle [K_{U\phi W}^{111}] \rangle$$

- +  $EI_{wz} \langle \phi \rangle \{ [K_{\phi v}^{22}] + [K_{\phi \phi v}^{121}] \}$ +  $P \langle V \rangle ([K_{vv}^{11}] - Y_s \langle \phi \rangle [K_{\phi v}^{11}] + Z_s \langle \phi \rangle [K_{\phi \phi v}^{101}] )$ -  $\frac{1}{2} Q_y \{ \langle W \rangle [K_{wv}^{11}] + 2(\langle \phi \rangle [K_{\phi \phi v}^{001}] + \langle \phi \rangle [K_{\phi v}^{01}] ) \} + \frac{1}{2} Q_z \langle W \rangle [K_{w \phi v}^{101}] \}$ -  $M_y \{ \langle W \rangle [K_{wv}^{12}] + \langle \phi \rangle [K_{\phi \phi v}^{011}] \}$
- +  $EI_{\omega y} \langle \phi \rangle \{ [K_{\phi \phi v}^{202}] [K_{\phi \phi v}^{211}] + [K_{\phi \phi v}^{121}] \}$
- +  $\langle V \rangle [K_{V\varphi v}^{112}]$  +  $\langle V \rangle [K_{U\varphi v}^{202}]$  +  $\langle V \rangle [K_{V\varphi v}^{211}]$  +  $\langle U \rangle [K_{U\varphi v}^{211}]$
- +  $EI_{yz} \{ \langle U \rangle [K_{uv}^{22}] + \langle V \rangle [K_{v\phi v}^{202}] \langle U \rangle [K_{u\phi v}^{112}] \}$
- +  $\operatorname{EI}_{z}\{\langle U \rangle [K_{u\phi v}^{202}] + \langle U \rangle [K_{u\phi v}^{211}] + \overline{\beta}_{z} \langle \phi \rangle [K_{\phi \phi v}^{112}] \}$
- $\langle v \rangle [K_{v\phi v}^{112}] \langle v \rangle [K_{v\phi v}^{211}] \}$
- +  $EI_{y}\{\langle V \rangle [K_{vv}^{22}] \langle U \rangle [K_{u\phi v}^{202}] \langle U \rangle [K_{u\phi v}^{112}]$
- +  $\langle V \rangle [K_{VVV}^{211}] \}$  +  $ES_{\omega} \{\langle \Phi \rangle [K_{\Phi \Phi V}^{211}] + \langle \Phi \rangle [K_{\Phi VV}^{211}] \}$
- +  $\frac{1}{2}(\langle U \rangle [K_{uuv}^{112}] + \langle V \rangle [K_{vvv}^{112}]$
- +  $EA\{\overline{Z}Y_{s} < V > ([K_{v\phi v}^{112} + [K_{v\phi v}^{211}]) Z_{s} < U > [K_{u\phi v}^{112}]$
- +  $EA\overline{Y}\{Y_{s} < U > ([K_{u\phi v}^{211} + [K_{uvv}^{211}])\}$
- $\overline{Z}[K_{wv}^{12}] + Z_{s}[K_{w\phiv}^{111}] [K_{w\phiv}^{102}] Y_{s}[K_{w\phiv}^{111}] + [K_{wv}^{12}] \}$
- + EA<W>{[K<sub> $WVV</sub></sub>] <math>\overline{Y}$ ([K<sub> $W\phiv</sub><sup>102</sup>] + [K<sub><math>W\phiv</sub><sup>111</sup>])</sub>$ </sub></sub>
- $+ \frac{Q_{y}}{2} (\langle V \rangle [K_{vw}^{11}] \langle \phi \rangle [K_{\phi vw}^{021}] \frac{Q_{z}}{2} (\langle U \rangle [K_{uw}^{11}] + \langle \phi \rangle [K_{\phi vw}^{021}] + \overline{P}] \{W\}$
- $M_{z}(\langle U \rangle [K_{uw}^{21}] + \langle V \rangle [K_{v\phi w}^{201}] + \langle V \rangle [K_{v\phi w}^{111}])$

- +  $P < V > \{[K_{uu}^{11}] + Z_{s} < \Phi > [K_{\phi u}^{11}] + Y_{s} < \Phi > [K_{\phi \phi u}^{101}] \}$ +  $\frac{1}{2}Q_{z} \{ < W > [K_{wu}^{11}] - 2(< \Phi > [K_{\phi \phi u}^{001}] - < \Phi > [K_{\phi u}^{01}] \} \} = \frac{1}{2}Q_{y} < W > [K_{w\phi u}^{101}] \}$
- +  $EI_{\omega y} \langle \phi \rangle \{ [K_{\phi u}^{22}] [K_{\phi \phi u}^{121} \} \}$
- +  $EI_{\omega z} \langle \phi \rangle \{ [K_{\phi \phi u}^{202}] + [K_{\phi \phi u}^{211}] [K_{\phi \phi u}^{121}] \}$
- $\langle U \rangle [K_{u \varphi u}^{112}] \langle V \rangle [K_{v \varphi u}^{202}] \langle V \rangle [K_{v \varphi u}^{211}] + \langle U \rangle [K_{u \varphi u}^{211}] \}$
- +  $EI_{y_2} \{\langle V \rangle [K_{v_u}^{22}] + \langle V \rangle [K_{v_{\psi_u}}^{202}] \langle U \rangle [K_{u_{\psi_u}}^{112}] \}$
- $\operatorname{EI}_{\mathbf{y}}\{\langle \mathbf{v} \rangle [\mathsf{K}_{\mathbf{v}\phi\mathbf{u}}^{202}] + \langle \mathbf{v} \rangle [\mathsf{K}_{\mathbf{v}\phi\mathbf{u}}^{211}] \overline{\boldsymbol{\beta}}_{\mathbf{y}} \langle \boldsymbol{\varphi} \rangle [\mathsf{K}_{\boldsymbol{\phi}\phi\mathbf{v}}^{112}] \}$
- +  $\langle U \rangle [K_{u\phi u}^{112}] + \langle U \rangle [K_{u\phi u}^{211}] \}$
- +  $EI_{z} \{ \langle U \rangle [K_{uu}^{22}] \langle V \rangle [K_{v\phi u}^{202}] + \langle V \rangle [K_{v\phi u}^{112}] \}$
- +  $\langle U \rangle [K_{uuu}^{211}] \}$  +  $ES_{\omega} \{-Z_{s} \langle \Phi \rangle [K_{\phi \phi u}^{211}] + \langle \Phi \rangle [K_{\phi uu}^{211}] \}$
- +  $\frac{1}{2}(\langle U \rangle [K_{uuu}^{112}] + \langle V \rangle [K_{vvu}^{112}] (Y_s^2 + Z_s^2) \langle \Phi \rangle [K_{\phi \phi u}^{112}]) \}$
- $= EAY{Z_{s}(U)([K_{u\phi u}^{112} + [K_{u\phi u}^{211}]) Y_{s}(V)[K_{v\phi u}^{112}]}$
- +  $EAZ \{-Z_{s} < V > ([K_{v\phi u}^{211} + [K_{vuu}^{211}])\}$
- $\Upsilon[K_{wu}^{12}] \Upsilon_{s}[K_{w\phi u}^{111}] [K_{w\phi u}^{102}] Z_{s}[K_{w\phi u}^{111}] + [K_{wu}^{12}] \}$
- + EA<W>{[K<sub>wuu</sub>] $-\overline{Z}$ ([K<sub>w</sub><sup>102</sup>] + [K<sup>111</sup><sub>w</sub><sup>11</sup>)
- +  $2(\langle v \rangle [K_{v\phi v}^{102}] + \langle v \rangle [K_{v\phi v}^{201}]) \}] \{v\}$
- +  $\frac{1}{2}M_{x}\{\langle U \rangle [K_{uv}^{21}] \langle U \rangle [K_{uv}^{12}] + 2\langle \phi \rangle [K_{\phi vv}^{111}]$
- +  $M_z \{ \langle \varphi \rangle [K_{\varphi \vee}^{11}] \langle W \rangle [K_{W \varphi \vee}^{102}] \langle W \rangle [K_{W \varphi \vee}^{111}] \}$

- +  $EI_{yz} \{ \langle U \rangle ([K_{uu\phi}^{220}] + [K_{uv\phi}^{211}]) + \langle V \rangle ([K_{vv\phi}^{220}] + [K_{vu\phi}^{211}] \}$
- +  $EI_{y}\{\langle U \rangle ([K_{uv\phi}^{220}] + [K_{uu\phi}^{211}] + [K_{uv\phi}^{211}]) + 2\overline{B}_{y}\langle V \rangle [K_{v\phi\phi}^{211}]\}$
- +  $EI_z \{\langle V \rangle ([K_{vu\phi}^{220}] + [K_{vv\phi}^{211}] + [K_{vu\phi}^{211}]) + 2\overline{\beta}_y \langle U \rangle [K_{u\phi\phi}^{211}] \}$
- +  $2[K_{\phi\phi\phi}^{211}]$  +  $GK < \phi > [\frac{11}{\phi\phi}]$
- +  $EI_{\omega} < \phi > \{ [K_{\phi\phi}^{22}] + \overline{\beta}_{\omega} \{ [K_{\phi\phi\phi}^{112}] \} \}$
- +  $\frac{1}{2}$ ( $Y_s^2$ + $Z_s^2$ ) < $\phi$ >[ $\frac{112}{\phi\phi\phi}$ ] + ( $Y_s^2$ + $Z_s^2$ ) < $\phi$ >[ $\frac{211}{\phi\phi\phi}$ ]}
- $+ \mathbb{ES}_{\omega} \{ \mathbb{Y}_{s}(\langle \mathbb{V} \rangle [\mathbb{K}^{112}_{\mathfrak{v}\phi\phi}] + \langle \phi \rangle [\mathbb{K}^{211}_{\phi\mathfrak{v}\phi}]) \mathbb{Z}_{s}(\langle \mathbb{U} \rangle [\mathbb{K}^{112}_{\mathfrak{u}\phi\phi}] + \langle \phi \rangle [\mathbb{K}^{211}_{\phi\mathfrak{u}\phi}])$
- +  $EA\overline{Z} \{Y_{s} < U > [K_{vv\phi}^{211} Z_{s} < V > [K_{vu\phi}^{211}] + (Y_{s}^{2} + Z_{s}^{s}) < V > [K_{v\phi\phi}^{211}] \}$
- +  $EA\overline{Y}{Y_{s} < U > [K_{uv\phi}^{211} Z_{s} < U > [K_{uu\phi}^{211}] + (Y_{s}^{2} + Z_{s}^{s}) < U > [K_{u\phi\phi}^{211}]}$
- +  $ES_{w}\{\langle W \rangle (Z_{s}[K_{wu\phi}^{111}] Y_{s}[K_{wv\phi}^{111}])\}$
- $-[K_{wu\phi}^{120}] + K_{wv\phi}^{120}] + \frac{ES}{EA}[K_{w\phi}^{12}]\}$
- $\Upsilon_{s}[K_{wu\phi}^{111}] + Z_{s}[K_{wv\phi}^{111}] [K_{wu\phi}^{111}] [K_{wv\phi}^{111}]$
- + EA<W> $\left\{-\overline{Y}\left(\left[K_{WV\phi}^{120}\right]-\left[K_{WV\phi}^{111}\right]\right) + \overline{Z}\left(\left[K_{WU\phi}^{120}\right]-\left[K_{WU\phi}^{111}\right]\right)\right\}$
- $2(\langle U \rangle [K_{U \phi U}^{102}] + \langle V \rangle [K_{U \phi U}^{201}] \rangle ] \{U\}$
- +  $\frac{1}{2}M_{\chi} \{\langle V \rangle [K_{Vu}^{21}] \langle V \rangle [K_{Vu}^{12}] + 2 \langle \phi \rangle [K_{\phi uu}^{111}] \}$
- +  $M_{y} \{ \langle \phi \rangle [K_{\phi u}^{11}] = \langle W \rangle [K_{W \phi u}^{102} + \langle W \rangle [K_{W \phi u}^{111}] \}$
- $M_{z} \{ \langle W \rangle [K_{wu}^{12}] + \langle \phi \rangle [K_{\phi \phi u}^{011}] \}$

$$-\frac{1}{2}Q_{z}\{(-2 < U > [K_{u\phi}^{10}] + < W > [K_{wv\phi}^{110}] + 2 < U > ([K_{u\phi\phi}^{100}] + Y_{s} < \phi > (\frac{1}{2}[K_{\phi\phi\phi}^{001}] + [K_{\phi\phi\phi}^{100}] - 2[K_{\phi\phi\phi}^{010}] + Y_{s} < \phi > (\frac{1}{2}[K_{\phi\phi\phi}^{011}] + [K_{\phi\phi\phi}^{101}] - 2[K_{\phi\phi\phi}^{010}] + Z_{s} < \phi > ([K_{u\phi}^{011}] + [K_{\phi\phi\phi}^{101}]) \}$$

$$+ M_{y}\{ < U > [K_{u\phi}^{111}] - < V > [K_{u\phi\phi}^{110}] + M_{z}\{ < U > [K_{u\phi\phi}^{110}] + M_{z}\{ < V > [K_{v\phi\phi}^{110}] + M_{z}\{ < V > [K_{v\phi\phi}^{110}] + M_{z}\{ < U > [K_{u\phi\phi}^{110}] + M_{z}\{ < U > [K_{u\phi\phi}^{110}] + M_{z}\{ < U > [K_{u\phi\phi}^{110}] + M_{z}\{ < U > [K_{u\phi\phi}^{111}] + M_{z}\{ < U > [K_{v\phi\phi}^{111}] \} + M_{z}\{ < U > [K_{u\phi\phi}^{111}] \} + M_{z}\{ < U > [K_{u\phi\phi}^{111}] + M_{z}\{ < U > [K_{u\phi\phi}^{111}] \} + M_{z}\{ < U >$$

+ 
$$Z_{s} < \phi > (\frac{1}{2} [K_{\phi \phi \phi}^{001}] + [K_{\phi \phi \phi}^{100}] + 2 [K_{\phi \phi \phi}^{010}]$$

 $- \Upsilon_{\mathbf{S}}^{\langle \phi \rangle} ([K^{01}_{\phi \phi}] + [K^{10}_{\phi \phi}]) \}$ 

+ 
$$\frac{1}{2}Q_{y}\{(-2 < v > [K_{v\phi}^{10}] - < w > [K_{wu\phi}^{110}] - 2 < v > ([K_{v\phi\phi}^{100}]$$

+ 
$$Z_{s} \langle V \rangle [K_{v\phi\phi}^{110}] + Y_{s} \langle U \rangle [K_{u\phi\phi}^{110}] + Z_{s} \langle V \rangle [K_{v\phi\phi}^{101}] + Y_{s} \langle U \rangle [K_{u\phi\phi}^{101}]$$

+ 
$$M_{\rho} < \phi > [K_{\phi\phi}^{11}] + P\{Z_{s} < U > [K_{u\phi}^{11}] - Y_{s} < V > [K_{v\phi}^{11}]$$

$$- \langle \Phi \rangle (Z_{s}[K_{\varphi\varphi\varphi}^{112}] - [K_{\varphi u\varphi}^{220}] + 2Z_{s}[K_{\varphi\varphi\varphi}^{211}] + [K_{\varphi u\varphi}^{112}]) \}$$

$$= \langle U \rangle ([K_{u\phi\phi}^{202}] + [K_{u\phi\phi}^{112}] - [K_{u\phi\phi}^{121}]) + \frac{1}{2}[K_{u\phi\phi}^{112}])$$

+ 
$$EI_{\omega z} \{ \langle V \rangle ([K_{V\phi}^{22}] + [K_{V\phi\phi}^{121}]) \}$$

+ 
$$\langle \phi \rangle (-Y_{s}[K_{\phi \phi \phi}^{112}] + [K_{\phi v \phi}^{220}] - 2Y_{s}[K_{\phi \phi \phi}^{211}] + [K_{\phi v \phi}^{112}])$$

+ 
$$\langle V \rangle ([K_{V\phi\phi}^{202}] - [K_{V\phi\phi}^{112}] + [K_{V\phi\phi}^{121}]) + \frac{1}{2} \langle V \rangle [K_{V\phi\phi}^{112}])$$

٠

+ 
$$\langle \phi \rangle [K_{\psi \psi \psi}^{001}] + 2 \langle \phi \rangle [K_{\psi \psi \psi}^{1001}] - 2 \langle U \rangle [K_{UU\psi}^{1201}] + 2 \langle V \rangle [K_{VV\psi}^{1201}] \} [\delta \phi \}$$
  
=  $M_{y} \phi_{0} [\{V''\} - \langle U \rangle [K_{U\psi}^{11}] \{\phi \} - \langle \phi \rangle [K_{\psi U}^{11}] \{U \} ]$   
+  $M_{z} \phi_{0} [\{U''\} + \langle V \rangle [K_{V\psi}^{11}] \{\phi \} + \langle \phi \rangle [K_{\psi}^{11}] \{V \} \rangle ]$   
+  $\frac{1}{2} P \phi_{0} [Z_{s} (\langle V \rangle [K_{U\psi}^{11}] \{\phi \} + \langle \phi \rangle [K_{\psi}^{11}] \{U \} \rangle ]$   
+  $\frac{1}{2} P \phi_{0} [Y_{s} (\langle U \rangle [K_{U\psi}^{11}] \{\phi \} + \langle \phi \rangle [K_{\psi}^{11}] \{U \} \rangle ]$   
=  $Q_{y} \phi_{0} [\{U'\} + Y_{s} \{\phi'\} + 2Z_{s} \{\phi \} ]$   
+  $Q_{z} \phi_{0} [\{U'\} + Y_{s} \{\phi'\} + 2Y_{s} \{\phi \} ] dx$   
+  $\int_{0}^{1} [[D_{1} \langle U \rangle [K_{UU}^{22}] + D_{5} \langle V \rangle [K_{VU}^{22}] + D_{6} \langle \phi \rangle [K_{\psi U}^{22}] ] [\delta U ]$   
+  $[D_{2} \langle V \rangle [K_{\psi U}^{22}] + D_{4} \langle U \rangle [K_{U\psi}^{22}] + D_{7} \langle \phi \rangle [K_{\psi \psi}^{22}] ] [\delta V ]$   
+  $[D_{3} \langle V \rangle [K_{\psi \psi}^{22}] + D_{6} \langle U \rangle [K_{U\psi}^{22}] + D_{7} \langle \phi \rangle [K_{\psi \psi}^{22}] ] [\delta \phi ] ] dx$  (4.16)  
where  $D_{1}, D_{2}, \dots, D_{7}$  were defined in Appendix C  
Equation 4.16 can be written in a simple form as:

$$\delta \Pi = \langle r \rangle [K_{T2}] \{r\} + \{\overline{P}\} + \{r_0 \overline{P}\}$$
(4.17)

where

•

 $\{r_0\} = [W_{01} \ U_{01} \ \theta_{01} \ V_{01} \ \Phi_{01} \ \chi_1 \ W_{02} \ U_{02} \ \theta_{02} \ V_{02} \ \Phi_{02} \ \chi_{02}]^T$  (4.18) are the initial deflections.

$$\{\overline{P}\} = [F_{x1} Q_{z1} M_{y1} Q_{y1} M_{x1} M_{w1} F_{x2} Q_{z2} M_{y2} Q_{y2} M_{x2} M_{w2}]^{T} (4.19)$$

$$[K_{T2}] = [K_E^L + K_G^L + K_E^{NL} + K_G^{NL} + K_o]$$
(4.20)

Carry out the integration of equation 4.16 by assuming the terms  $\text{EI}_{v}$ ,  $\text{EI}_{\tau}$ ,  $\overline{Y}$ ,  $M_{v}$ ,  $M_{\omega}$ , etc. to vary linearly in the form

$$EI_{y} = EI_{y1} + (EI_{y2} - EI_{y1})\frac{x}{1}$$
(4.20a)

or

$$M_{y} = M_{y1} + (M_{y2} - M_{y1})\frac{\chi}{1}$$
(4.20b)

where 1 and 2 are nodal member numbers of the element

The terms of equation 4.20 can be described as: 1- Linear tangential stiffness matrix  $[K_E^1]$ 2- Linear geometrical stiffness matrix  $[K_G^1]$ 3- Nonlinear tangential stiffness matrix  $[K_E^{N1}]$ 4- Nonlinear geometrical stiffness matrix  $[K_G^{NL}]$ 5- Initial stiffness matrix  $[K_G]$ 

## 4.3.2.1 Linear Tangential Stiffness Matrix

The linear tangential stiffness matrix [ $K_E^L$ ] contains terms such as EI<sub>y</sub>, ES<sub>w</sub>, EI<sub>wy</sub>, etc. These expressions are fully presented in Table 4.5.

## 4.3.2.2 Linear Geometrical Stiffness Matrix

Table 4.6 gives the terms of the linear geometrical stiffness matrix  $[K_G^L]$ . The Wagner effect together with internal forces such as  $M_v$ ,  $Q_z$ , etc are included.

## 4.3.2.3 Nonlinear Tangential Stiffness Matrix

The higher order terms such as  $EI_{y',x}V_{,xx}$ ,  $EI_{\omega y}\Phi_{,x}V_{,xx}$ . are accounted for, and are given Table 4.7. These expressions are detailed in Appendices G and J.

## 4.3.2.4 Nonlinear Geometrical Stiffness Matrix

The nonlinear geometrical stiffness matrix  $[K_G^{NL}]$  for a beamelement in three dimensions is presented in Table 4.8. The terms of that Table are given in full detail in Appendices H and J.

## 4.3.2.5 Initial Stiffness Matrix

The inclusion of the initial deflections in the stiffness matrix is explained in Table 4.9.

## 4.3.2.6 Assembly

The assembly of the submatrices presented in Tables 4.5 to 4.9 corresponding to the linear geometrical and tangential stiffness matrix and the nonlinear terms together with the initial stiffness matrix is demonstrated by several illustrative examples in Chapters 6 & 7 respectively.

## 4.4- Transformation Matrix

The transformation matrix has been explained in many references such as Weaver (1967), Poresi and Lynn (1977), and Chen and Atsuta (1977). The transformation presented herein is based on using vector notation. The element displacements with respect to the local coordinates can be related to those in the global coordinate system by the transformation [T].

- 98 -

$${r_E}_L = [T]{r_E}_G$$
 (4.21)

where [ T ] is the element transformation matrix whic can be obtained in the following using vector notation.

Let us consider a beam element which is defined by joint numbers II and JJ, a third point KK is used to locate the principal axes as shown in Fig 4.1. The transformation from local to global coordinates is most conveniently developed using vector notation.

The unit vectors in the  $\overline{s1}$  and  $\overline{g1}$  directions are given by

$$\overline{s1} = S1X \overline{I} + S1Y \overline{j} + S1Z \overline{k}$$
 (4.22)

$$\overline{g1} = GX \overline{1} + GY \overline{j} + GZ \overline{k}$$
(4.23)

where  $\overline{I}$ ,  $\overline{J}$ , and  $\overline{k}$  are unit vectors in the  $\overline{X}$ ,  $\overline{Y}$ , and  $\overline{Z}$  directions respectively, the direction cosines are

$$S1X = \frac{X(JJ) - X(II)}{1}$$
,  $S1Y = \frac{Y(JJ) - Y(II)}{1}$ ,  $S1Z = \frac{Z(JJ) - Z(II)}{1}$ 

$$GX = \frac{X(KK) - X(II)}{1}$$
,  $GY = \frac{Y(KK) - Y(II)}{1}$ ,  $GZ = \frac{Z(KK) - Z(II)}{1}$ 

where

$$1 = \sqrt{[X(JJ)-X(II)]^{2} + [Y(JJ)-Y(II)]^{2} + [Z(JJ)-Z(II)]^{2}}$$

# $G = \sqrt{[X(KK)-X(II)]^{2} + [Y(KK)-Y(II)]^{2} + [Z(KK)-Z(II)]^{2}}$

The unit vectors in the  $\overline{s3}$  and  $\overline{s2}$  directions are given respectively by

$$\overline{\mathbf{s3}} = \frac{\overline{\mathbf{s1xg}}}{|\overline{\mathbf{s3}}|} = S3X \,\overline{\mathbf{i}} + S3Y \,\overline{\mathbf{j}} + S3Z \,\overline{\mathbf{k}}$$
(4.24)

$$\overline{s2} = \overline{s3}\overline{xs1} = S2X \ \overline{i} + S2Y \ \overline{j} + S2Z \ \overline{k}$$
 (4.25)

Equation 4.22, 3.24, and 4.25 can be written in matrix form as:

$$\begin{bmatrix} \overline{s1} \\ \overline{s2} \\ \overline{s3} \end{bmatrix} = \begin{bmatrix} S_A \end{bmatrix} \begin{bmatrix} \overline{1} \\ \overline{j} \\ \overline{k} \end{bmatrix}$$
(4.26)

where

$$\begin{bmatrix} S_{A} \end{bmatrix}_{3\times 3} = \begin{bmatrix} S 1X & S 1Y & S 1Z \\ S 2X & S 2Y & S 2Z \\ S 3X & S 3Y & S 3Z \end{bmatrix}$$
(4.27)

The transformation matrix of a beam-column having 7 degrees of free dom at each node can be written as:

$$[S] = \begin{bmatrix} [S_A] \\ [S_A] \\ I \end{bmatrix}$$
(4.28)

The assembly of the total transformation matrix in 3-dimensions for a beam element is performed in equation 4.29, the diagonal terms corresponding to the warping are set to unity. So the transformation stiffness to a beam element can be written as:

$$[T] = \begin{bmatrix} [S] \\ [S] \end{bmatrix}$$
(4.29)

## 4.5 Strain Displacement Matrices

In this section a full explanation of the expressions for the strain displacement matrices are presented. The linear and higher order

- 100 -

terms are included. The validity and the accuracy of the inclusion of both linear and nonlinear strains are demonstrated for several problems in the examples given in chapters 6 and 7.

## 4.5.1 Linear Strain Matrix

The axial strain due to displacement in the X-direction is  $W_{,x}$ .  $\overline{\Phi}_{z}$ ,  $\overline{\Phi}_{y}$ , and  $\overline{\Phi}_{\omega}$  are the curvatures produced from the displacements U, V, and  $\Phi$ , which are known as the strains due to curvature. They produce additional axial strains in the beam whose values vary linearly with the depth of the beam for the biaxial and warping curvature. The total linear axial strain  $\Sigma^{1}_{xx}$  due to the deformations is given by

$$\Sigma_{xx}^{1} = W_{,x} - \overline{\Phi}_{y} - \overline{\Phi}_{z} + \overline{\Phi}_{\omega}$$
(4.30)

where

$$\overline{\Phi}_{\mathbf{y}} = U_{\mathbf{y}\mathbf{x}\mathbf{x}}^{\mathbf{Z}}$$
(4.31a)

$$\overline{\Phi}_{z} = U_{,xx}Y \tag{4.31b}$$

$$\overline{\Phi}_{\omega} = \Phi_{,xx}\overline{\omega}_{n} \tag{4.31c}$$

in which Y and Z are the coordinates of the observed point on the cross section (position where the strain needs to be evaluated).  $\overline{w}_n$  is the normalized function explained in the previous chapter. The curvatures  $V_{,xx}$ ,  $U_{,xx}$ , and  $\Phi_{,xx}$  are the second derivatives of the in-plane and out-of-plane displacements, and rotations with respect to X respectively.

Equation 4.30 can be written in anther form as:

- 101 -

$$\Sigma_{xx}^{1} = \Sigma_{x} - \phi_{y}^{Z} - \phi_{z}^{Y} - \phi_{\omega}^{-} \omega_{n}$$
(4.32)

where

 $\Phi_{y} = U_{,xx} \tag{4.33a}$ 

$$\Phi_z = V_{,xx} \tag{4.33b}$$

$$\Phi_{\omega} = \Phi_{xx} \tag{4.33d}$$

Substituting equation 4.4 into equation 4.33a and applying the two boundary conditions which are; X=0 and X=1 (at the two nodes) yields

$$\begin{bmatrix} \Sigma_{\mathbf{X}}^{1} \\ \Sigma_{\mathbf{X}}^{2} \end{bmatrix} = \begin{bmatrix} \frac{1}{1} & -\frac{1}{1} \\ -\frac{1}{1} & \frac{1}{1} \end{bmatrix} \begin{bmatrix} W_{1} \\ W_{2} \end{bmatrix}$$
(4.34)

$$\begin{bmatrix} \Sigma_{\mathbf{x}} \end{bmatrix} = \begin{bmatrix} \Sigma_{\mathbf{x}}^{1} \\ \Sigma_{\mathbf{x}}^{2} \\ \mathbf{x} \end{bmatrix}$$
(4.35)

where  $\Sigma_X^1$ ,  $\Sigma_X^2$  are the axial strains at node 1 and 2, 1 is the segment beam length,  $W_1$  and  $W_2$  are longitudinal displacements at node 1 and 2.

The value of the Curvature  $\Phi_y$  can be obtained by substituting equation 4.3 into equation 4.33a and applying the same boundary conditions described and stated above to obtain

$$\begin{bmatrix} \Phi_{y1} \\ \Phi_{y2} \end{bmatrix} = \begin{bmatrix} \frac{6}{1^2} & \frac{4}{1} & \frac{6}{1^2} & \frac{2}{1} \\ \frac{6}{1^2} & \frac{2}{1} & \frac{6}{1^2} & \frac{4}{1} \end{bmatrix} \begin{bmatrix} U_1 \\ \theta_{u1} \\ U_2 \\ \theta_{u2} \end{bmatrix}$$
(4.36)

where

-----

$$\begin{bmatrix} \phi_{\mathbf{y}} \end{bmatrix} = \begin{bmatrix} \phi_{\mathbf{y}1} \\ \phi_{\mathbf{y}2} \end{bmatrix}$$
(4.37)

where  $\phi_{y1}$  and  $\phi_{y2}$  are the curvatures at nodes 1 and 2 respectively. Similar procedures may be applied for  $\phi_z$  and  $\phi_{\omega}$ , which can be written as:

$$\begin{bmatrix} \Phi_{z} \end{bmatrix} = \begin{bmatrix} \Phi_{z1} \\ \Phi_{z2} \end{bmatrix}$$
(4.38)

$$\begin{bmatrix} \Phi_{\omega} \end{bmatrix} = \begin{bmatrix} \Phi_{\omega} \\ \Phi_{\omega} 2 \end{bmatrix}$$
(4.39)

Substituting for equations 4.35, 4.37, 4.38, and 4.39 enables the complete result to be written in matrix from as:

$$\begin{bmatrix} \Sigma_{\mathbf{x}} \\ \Phi_{\mathbf{y}} \\ \Phi_{\mathbf{z}} \\ \Phi_{\mathbf{z}} \end{bmatrix} = \begin{bmatrix} \Sigma_{\mathbf{x}1} & \Sigma_{\mathbf{x}2} & \Phi_{\mathbf{y}1} & \Phi_{\mathbf{y}2} & \Phi_{\mathbf{z}1} & \Phi_{\mathbf{z}2} & \Phi\omega^{1} & \Phi_{\omega^{2}} \end{bmatrix}^{\mathrm{T}}$$
(4.40)

Equation 4.40 after some arrangements can be expressed as

$$\begin{bmatrix} \Sigma_{X1} \\ \Phi_{y1} \\ \Phi_{z1} \\ \Phi_{w1} \\ \Xi_{x2} \\ \Phi_{y2} \\ \Phi_{y2} \\ \Phi_{z2} \\ \Phi_{w2} \end{bmatrix}$$
(4.41)

where

 $\overline{X}$  is the element linear strain matrix, which is given in Table 4.10.

4.5.2 Nonlinear Strain Matrix Nonlinear strains occur due to coupling between the displacements (U, V, W, and  $\phi$ ). Those strains are, axial nonlinear strains ( $\overline{\Sigma}$ ) and the nonlinear strain due to the curvatures ( $\phi_y$  and  $\phi_z$ ) about the Y and Z axes. The total nonlinear strain can be expressed as:

$$\Sigma_{\mathbf{x}}^{\mathbf{n}} = \overline{\Sigma} + \Phi_{\mathbf{y}}^{\mathbf{n}} \mathbf{\hat{Y}} + \Phi_{\mathbf{z}}^{\mathbf{n}} \mathbf{Z}$$
(4.42)

where

$$\overline{\Sigma} = \frac{1}{2} [V_{,x}^{2} + U_{,x}^{2} + \rho \phi_{,x}^{2}] + Z_{s} \phi_{,x} [U_{,x} + V_{,x} \phi] - Y_{s} \phi_{,x} [V_{,x} - U_{,x} \phi] \quad (4.43)$$

$$\Phi_{y}^{n} = U_{yxx}\Phi + \frac{V_{yxx}\Phi^{2}}{2}$$
(4.44)

$$\Phi_{z}^{n} = -V_{,xx}\Phi + \frac{U_{,xx}\Phi^{2}}{2}$$
(3.45)

Equations 4.43 to 4.45 can be written in matrix form as:

$$\begin{bmatrix} \overline{\Sigma} \\ \Phi_{y}^{n} \\ \Phi_{z}^{n} \end{bmatrix} = \begin{bmatrix} \overline{\Sigma}_{1} & \overline{\Sigma}_{2} \Phi_{y1}^{n} & \Phi_{y2}^{n} & \Phi_{z1}^{n} & \Phi_{z2}^{n} & \Phi_{\omega1}^{n} & \Phi_{\omega2}^{n} \end{bmatrix}^{\underline{\Gamma}}$$
(4.46)

Substituting equation 4.3 into equation 4.46 by employing some mathematical calculations yields



Where  $\overline{\Sigma}_1$  and  $\overline{\Sigma}_2$  are nonlinear axial curvatures at node 1 and 2. Additional axial strains are induced by the curvatures at the two nodes, which are  $\Phi_{y1}^n$ ,  $\Phi_{y2}^n$ ,  $\Phi_{z1}^n$ , etc. The terms of the matrix  $X^n$  of the nonlinear axial strains are presented in Table 4.11.

(4.47)

The total strain matrix  $[\overline{B}]$  for a beam-column in three dimensions is the summation of the linear and nonlinear strain matrices, which have been given in equations 4.41 and 4.47 respectively. B<sup>=</sup> is the total axial strain. Both  $\overline{B}$  and  $\overline{B}^{=}$  are illustrated as:

 $[B^{=}] = [\overline{B}]\{r\}$  (4.48)

where

$$\{B^{=}\} = \begin{bmatrix} \overline{\Sigma}_{x1} + \overline{\Sigma}_{1} \\ \Phi_{y1} + \Phi_{y1}^{n} \\ \Phi_{z1} + \Phi_{z1}^{n} \\ \Phi_{\omega1} + \Phi_{\omega1}^{n} \\ \Sigma_{x2} + \overline{\Sigma}_{2} \\ \Phi_{y2} + \Phi_{y2}^{n} \\ \Phi_{z2} + \Phi_{z2}^{n} \\ \Phi_{\omega2} + \Phi_{\omega2}^{n} \end{bmatrix}$$

and

$$[\overline{B}] = [\overline{X} = x^{n}]_{4x14}$$
 (4.50)

## 4.6 Conclusions

The element stiffness matrices of a beam-column of thinwalled open cross-section in three dimensions were derived. The interpolation functions employed were linear polynomials for the axial displacements, cubic polynomials for lateral displacements and twists, and quadratic polynomials for the rotations and warping. Linear and nonlinear geometrical and tangential stiffnesses have been obtained by two methods which are the principle of virtual work and total potential energy. In the latter method the initial deformation stiffness matrix was included. The terms of both linear and nonlinear stiffness matrices are calculated based on the average of the values of the two nodes i.e.  $(M_{y1} + M_{y2})$  when using virtual work, while for the potential energy the evaluation was based on a linear function  $(a + a_1 \frac{X}{L})$  between the two nodes. The transformation matrix for the element displacements with respect to the local coordinate and the global coordinate systems was developed together with linear and nonlinear strain displacement

(4.49)

matrices. The important concepts of these stiffnesses are:

- 1- The sections have arbitrary shapes such [, I, T, L, etc.
- 2- The effect of non-coincident shear centre and centroid are accounted for i.e. Wagner effect is incorporated.
- 3- Different load patterns are included.
- 4- The higher order terms are involved.
- 5- Strains vary linearly along the cross-section.
- 6- The section and sectorial properties and forces were evaluated for an arbitrary axis system.

Those matrices can deal with the analysis of a member in space under eccentric load, uniaxial and biaxial bending, and bending and torsion or any combination of these loads. These matrices can be arranged to perform several analyses such as Linear, partial Nonlinear, and Full nonlinear analysis. In "Linear" the basic stiffness matrices (tangent and geometrical which includes flexural rigidities, torsional moments and/ or shear forces) and second order strains are used. In "Partial non-linear" the the linear stiffnesses matrices and higher order terms of the strain-displacement relations are included. In "Full non-linear" incorporation of both the contributions of the higher order terms to the stiffness matrices and strain-displacement relations is included. Further details of these analyses can be found in the papers by El-Khenfas and Nethercot (1987b,c) and in Chapters 6 and 8.

0													
	λ	[	_										
- o <u>v</u>	В	с	Ţ										
	aı	bı	Ā			1							
- ož	bı	C1	Ē	ē	]	_							
	ā	Б	a	b	E	]							
- x	b	ē	b	с	F	G	T	_					
- 0		OŸ		ož		x	- 0		_				
	- A	- B	- a1	- b1	ā	- Б		л		_			
- oī	в	D	+ b1	c1/2	ษี	<b>c</b> 1/2	- oī	- B	- 2C				
	- a1	- b1	- Ã	- B	- a	- b	·	a	bi	Ā	]	_	
οž	- b1	c1/2	B	D	b	c/2	- ož	- b1	C]	- B	- 2ē		
	- ā	- b	- a	- b	- E	- F		ā	- Đ	a	- b	Е	
x	b	īc/2	b	c/2	F	Т	- x	- Đ	Ē	b	c	- F	м
X = ES	<u>w</u>	$B = \frac{6E}{l}$		a1 =	12EIy L <sup>3</sup>	Ζ.	E = <u>1</u>	2EI w 4	6 G 52	к	a = <u>1</u>	2EIw Q 3	2
$O = \frac{E\Lambda}{\ell}$	-	$C = \frac{4E}{l}$	<u>I z</u>	b1 =	GEI <sub>YZ</sub> L		$\mathbf{F} = \mathbf{G}$	$\frac{EI_W}{\ell^2}$ +	<u>1</u> GК 10	·	b = <u>(</u>	l <sup>2</sup>	-
$A = \frac{12}{g}$	EIZ	$D = \frac{2E}{s}$		ć1 = _	4FI yz L		T = 2	EIW -	£ GK		c = 4	EIwz Q2	-



-108-

### a b QY/2 c - T а - QZ/L - T b с • - Ā - Ē C Α aı - B D в D bı C1

		- QY/2		QZ/ L										
	- a	- b		- т	- A	- B		a						
- QY/2	b	d	- T	м	в	F	QY/2	- b	с					
			- a	- b	Ā	B			- т	a				
QZ/£	-	- M	b	d	- B	- F	QZ/2	т		- b	c			
	- A	- E	Ā	Ē	- a1	- b1						aı	- b1	
_	Е	F	- Ē	- F	bı	dı						- b1	- c1	

$$a = \frac{6}{5L} P \qquad a_1 = \frac{6}{5L} n_p \qquad T = \frac{M_x}{L} \qquad C = \frac{1}{10} \frac{M_y}{y} - \frac{1}{10} \frac{PZ}{s} - \frac{1}{10} \frac{Q}{y}$$

$$b = \frac{1}{10} P \qquad b_1 = \frac{1}{10} n_p \qquad M = \frac{M_x}{2} \qquad D = \frac{1}{10} \frac{M_y}{y} - \frac{1}{10} \frac{PZ}{s} + \frac{1}{10} \frac{LQ}{y}$$

$$c = \frac{2}{15} L P \qquad c_1 = \frac{2LM_p}{15} \qquad A = -\frac{6}{5L} \frac{M_y}{y} - \frac{6}{5L} \frac{PZ}{s} - \frac{1}{2} \frac{Q}{y} \qquad E = -\frac{1}{10} \frac{M_y}{y} + \frac{1}{10} \frac{PZ}{s} + \frac{1}{10} \frac{LQ}{y}$$

$$d = -\frac{L}{30} P \qquad d_1 = -\frac{L}{30} \frac{M_p}{p} \qquad B = -\frac{2}{15} \frac{LM_y}{15} + \frac{2}{15} \frac{LZ}{s} \qquad F = -\frac{1}{30} \frac{M_y}{y} - \frac{L}{30} \frac{LZ}{s} - \frac{L^2Q}{60} \frac{Q}{y}$$

For  $\overline{\lambda}$ ,  $\overline{B}$  etc. Change Y to Z and vice versa in the expressions for  $\lambda$ , 3 etc

												n1 - E1 (1051 + BOIS) - E1 (3015 + 1015)
	1											41 - EI (1022 + SUZZ) - EI (5022 + FOZZ)
		_ Fo	or dei	Einiti	lons of	E term	s see	Appe	endi	хE		H - EL (FUJI + BO13) - EL (SU13 + 1013)
υ												0 + EI (1032 + 6023) - EI (5023 + 1023) y
v			]									Q = E1 (1041 + 5014) - E1 <sub>7</sub> (3014 + F014) 7
ū	aı	C1										N - EL (1042 + 3024) - EL (5024 + 1024) y
v	bı	d,			]							•1 - EI (1013 + 0013) - EI (1013 + 1031
						1						ri - El (1014 + 5014) - Fl (5014 + F041
Xı	a	c	a	С			-					61 - EI (LOS) + E(SOS) - EI <sup>A</sup> (205) + 1.035
Z 1	b	ā	ь	d								h1 - EI (1024 + 5024) - EI (5024 + 1042
	- U	- v	- Ū	- v	- X1	- Z1		7				1, • EI <sub>E</sub> (1033 • 5033) - EI <sub>J</sub> (5033 • 1033
Ŵ			N	0	J	к	- W	1				1, - El (1034 + 5034) - El (5034 + 1943
Ŧ			Q	R	L	м	- Ÿ			]		h <sub>1</sub> = EI (1043 + 5054) = EI (3054 + 1034
Ŵ	e1	g1			Ĵ	ĸ	- W	11	k <sub>1</sub>			m <sub>1</sub> = 21 <sub>9</sub> (1044 + 5044) - E1 <sub>9</sub> (5044 + F044
Ŷ	f1	hı			ī	M	- Y	11	101			$U = -\frac{EA}{21}AU1 \qquad Y = -\frac{EA}{21}AU2$
X 2	ē	ģ	е	g			- X2	ī	k	i	k	$W = -\frac{EA}{21}AU3 \qquad Y = -\frac{EA}{21}AU4$
Z 2	Ē	Б	£	h			- X <sub>2</sub>	ī	m	1	m	$U = \frac{EA}{21}AY1  \overline{V} = -\frac{EA}{21}AY2$

m - E1 (FUHA + SOHA) - E1 (SUHA + FOHA) For a, b, c,...etc. change x, Yull, TYIL, T HULL etc. to Z, VOIL, Tull, Z\_ FOIL etc.  $U = -\frac{EA}{21}AU1 \qquad V = -\frac{EA}{21}AU2$  $x_2 = -\frac{EA}{21}(Y_2^2 + Z_3^2)AO_3 - \frac{1}{21}(EI_y + EI_z)AO_5$  $W = -\frac{EA}{21}AU3 \qquad Y = -\frac{EA}{21}AU4 \qquad + Y \frac{EA}{91}AV3 = Z \frac{EA}{91}AU3$  $U = -\frac{EA}{21}AV1 \qquad \overline{Y} = -\frac{EA}{21}AV2 \qquad Z_1 = -\frac{EA}{21}(Y_{\overline{g}}^2 + Z_{\overline{g}}^2)AO2 - \frac{1}{21}(E1_y + E1_z)AO2$  $\overline{W} = -\frac{EA}{21}AV3 \quad \overline{Y} = -\frac{EA}{21}AV4 \qquad + Ynybn\frac{EA}{1}AV2 - 2\frac{EA}{91}AU2$  $x_1 = -\frac{EA}{21}(y_0^2 + z_0^2)A01 - \frac{1}{21}(EI_y + EI_z)A01$   $Z_2 = -\frac{EA}{21}(y_0^2 + z_0^2)A04 - \frac{1}{21}(EI_y + EI_z)A04$ +  $Y = \frac{EA}{1}AVI = Z = \frac{EA}{1}AUI$ + Y  $\frac{EA}{1}$  AV4 - Z  $\frac{EA}{0}$  AV4

5 - EI (2112 + 1112) + T EL 1012 - - EI (2721 + 1721) + 1 EI 1021 - FI (2422 + 1452) + 1 51 1022 - EF (2413 + 1413) + T ET 1013 7 - 81 (2414 + 1414) + T EI 1014 E - EI (2423 + 1423) + T EI 1023 h + EI (2424 + 1424) + T EI 1024 T - EI (2439 + 1453) + T EI 1033  $T = EI_{y}(2V34 + TV34)(+ T_{EI_{y}}E034)$ E . El (244) + 1443) + T EL TUAS а - БГ, (2744 + 1744) + Т EL 1044 J--EI (1431+Z413) + + E1 FOI3 K--EI (1941+2914) + # KI 1014 H--EI (1442+7424) + # EI 1024

 $\mathbf{e}_1 = \mathbf{n}_1(1011 + \mathbf{n}011) = \mathbf{n}_1(1011 + \mathbf{n}011)$ 

b1 - E1 (1015 + E015) - E1 (2015 + 1021)

## Table 4.3 Non-Linear Stiffness Hatrix



For definitions of terms see Appendix F

For a, b atc change y to s and vice verse in espression for a, b, stc.

P -- H [ CX24 + BO12 ] + -- 10 (42-41)

1 111-

<sup>a</sup> 1													
<sup>B</sup> 1	Az												
C <sub>1</sub>	B <sub>z</sub>	C <sub>z</sub>											
<b>b</b> <sub>1</sub>	Ayz	Byz	Ay										
ē,	Byz	Cyz	<sup>B</sup> y	с <sub>у</sub>									
*e1	Awy	B wy	Awz	Bwz	m								
f <sub>1</sub>	B <sub>wy</sub>	Cwy	B <sub>wz</sub>	C wz	n	0							
-a <sub>1</sub>	-b <sub>1</sub>	-c <sub>1</sub>	-6 <sub>1</sub>	- <sup>c</sup> 1	- e <sub>1</sub>	f <sub>1</sub>							
-b <sub>1</sub>	-A <sub>z</sub>	-B <sub>z</sub>	-A <sub>yz</sub>	-B <sub>yz</sub>	-A <sub>wy</sub>	-B <sub>wy</sub>	<sup>ь</sup> 1	A z					
-d <sub>1</sub>	Fz	Dz	Fyz	Dyz	+F <sub>wy</sub>	-D <sub>wy</sub>	+d1	-Fz	Е <sub>z</sub>				
-51	-A <sub>yz</sub>	-B <sub>yz</sub>	-A <sub>y</sub>	-B y	-A z	-B <sub>wz</sub>	Б <sub>1</sub>	^yz	-F <sub>yz</sub>	Ay			
-ā <sub>1</sub>	Fyz	Dyz	Fy	Dy	f <sub>wz</sub>	-D <sub>wz</sub>	dı	-F <sub>zy</sub>	Eyz	-Fy	Ey		
-e <sub>1</sub>	-A wy	-В	-A <sub>wz</sub>	-B <sub>wz</sub>	-m	-n	e <sub>1</sub>	Awy	-F <sub>wy</sub>	Awz	-Fwz	m	
<sup>2</sup> 1	Fwy	Dwy	Fwz	D <sub>wz</sub>	Q	R	-e <sub>1</sub>	-F <sub>wy</sub>	Ewy	-F <sub>wz</sub>	Ewz	-Q	N

$\mathbf{T} + \mathbf{T}_2 = \mathbf{T}_1$	$k = Ek_2 - Ek_1$	$T_{s} = T_{s2} - T_{s1}$	
A <sub>2</sub> + EI <sub>22</sub> - EI	$A_{u} EI_{u2} - EI_{u1}$	$A_{GK} = GK_2 = GK_1$	
E5 = E	<sup>3</sup> u1		
$\bullet_1 \bullet \frac{EA_1}{1} \bullet \frac{A}{21}$	$b_1 = \frac{A}{L^2} (\tilde{Y} = Y_g)$		
$a_1 = \frac{EA_1}{1}(T_{s1}$	$= \mathbf{T}_{\mathbf{L}}$ $\mathbf{d}_{\mathbf{L}} + = \frac{\mathbf{E}_{\mathbf{L}}}{2}$	$\frac{A_1}{1}(\Upsilon_{g1},-\Upsilon_1) = \frac{A}{1}(\Upsilon_g,-\Upsilon)$	
$e_1 = -\frac{ES_{col}}{L^2}$	$F_1 = \frac{ES_2}{1} = \frac{ES_2}{1} = \frac{ES_3}{1} = \frac{ES_3}{1}$	$\frac{1}{1} = \frac{E_{\text{m}}^2}{1}$ , and $\overline{d}_1$ obsage Y to Z or vice ve	·rss.

 $A_{z} = 12\frac{EI_{z}}{L^{3}} + 6\frac{A_{z}}{1^{3}} = B_{z} = 6\frac{EI_{z}}{L^{2}} + 2\frac{A_{z}}{1^{2}}$   $C_{z} = 8\frac{EI_{z}}{L} + 6\frac{A_{z}}{1} = D_{z} = 2\frac{EI_{z}}{L} + \frac{A_{z}}{1}$   $E_{z} = 8\frac{EI_{z}}{L} + 3\frac{A_{z}}{1} = D_{z} = 2\frac{EI_{z}}{L} + 8\frac{A_{z}}{1}$ Change the subscript of the following terms;  $\overline{A}_{z}$ ,  $\overline{B}_{z}$ ,  $\overline{C}_{z}$ ,  $\overline{D}_{z}$ ,  $E_{z}, \overline{F}_{z} = ad = EI_{z1}, A_{z} \text{ to } y, yz, yw, and wz \text{ respectively in the form of}$   $\overline{A}_{y}, \dots, and EI_{y1}, and \overline{A}_{y}$   $m_{z} = 12\frac{EI_{w1}}{L} + 6\frac{A_{w}}{1^{2}} + \frac{6}{51}GK_{1} + .6\frac{GK}{1} = n + 6\frac{EI_{w1}}{L^{2}} + 2\frac{A_{w}}{1^{2}} + \frac{1}{10}GK_{1} - \frac{1}{10}\frac{A_{z}}{CK}$   $u = 8\frac{EI_{w1}}{1} + A_{w} + \frac{2}{15}IGK_{1} + \frac{1}{30}A_{GK}$   $Q = 6\frac{EI_{w1}}{L^{2}} + 8\frac{A_{w}}{L^{2}} + \frac{1}{10}GK_{1}$   $R + 2\frac{EI_{w1}}{2} + \frac{A_{w}}{1} - \frac{1}{30}GK_{1} - \frac{1}{60}A_{GK}$   $N = 8\frac{EI_{w1}}{1} + 3\frac{A_{w}}{1} - \frac{31}{30}(K_{1} + \frac{1}{10}A_{GK})$ 

m-41a	4.6	Lin	ear Ge	ometri	cal S	stiffne	ess Ma	trix						
18016	4.0							• • 51	F1+-10F	ь.	10"1+10"	F		$I = \frac{H_{w1}}{1} = \frac{AQ_{x}}{2u} = \frac{H_{x1}}{2u} + \frac{AH_{x}}{2u} = C = \frac{1}{2u} = -\frac{31}{210}H_{x} = D = \frac{Q_{x1}}{21} + \frac{AQ_{x}}{21} + \frac{BQ_{x1}}{21} + \frac{AQ_{x1}}{21} + A$
								a + 2 15	$1F_1 + 1/30$ $\frac{1}{5}1F_1 - \frac{1}{60}1$	DIAF du	10 <sup>F</sup> 1	LAF		$H = \frac{\Delta Q_y}{2^{2q}} - \frac{H_{y1}}{1} + \frac{\Delta H_y}{1} = H_{y1} - \frac{H_{w1}}{1} + \frac{\Delta H_{w}}{1} = $
								E = 6.	р1 • <u>-6</u> ан	h =	H + 1	۵M p		$\mathbf{u} = \frac{6}{51} \frac{M}{y_1} \frac{-6}{10} \frac{M}{y_2} - 510 \frac{13}{21} \frac{100}{70} \frac{-6}{10} \frac{7}{2} \frac{100}{10} \frac{100}$
								1 + 2 15	H <sub>p1</sub> + 1/301	Δ <sup>M</sup> ρ 0 = 1	1M + 1	-1AH 0 p		$b_1 = \frac{1}{10} \frac{M_y}{y_1} \frac{1}{10} \frac{1}{10} \frac{1}{z_1} \frac{3}{70} \frac{10}{z_1} \frac{1}{10} \frac{1}{z_1} \frac{1}{10} \frac{1}{z_1} \frac{1}{10} \frac{1}{z_1} $
								= = <u>1</u> H	n	• <u>1</u> 10 p1	14H	$r = -\frac{1}{30}$ 1	M <sub>0</sub> 1 60 <sup>16H</sup>	$e_1 = \frac{1}{10} \frac{M_y}{r_1} \frac{1}{10} \frac{10}{r_1} \frac{10}{10} \frac{10}{r_1} \frac{10}{r_1} \frac{10}{r_1} \frac{10}{r_2} \frac{1}{r_1} \frac{1}{r_2} \frac{1}{r_1} \frac{1}{r_1} \frac{1}{r_1} \frac{1}{r_2} \frac{1}{r_1} \frac{1}{r_1} \frac{1}{r_2} \frac{1}{r_1} \frac{1}{r_1} \frac{1}{r_2} \frac{1}{r_1} \frac{1}{r_1} \frac{1}{r_2} \frac{1}{r_1} 1$
D	a		1					A =	1 _ 1 <sup>6H</sup> x	8 . Hat	36Mz			d1 + 2 14 1 + 10 14 H + . 5L 20 + 17 12 40 + 2 17 12 + 10 14 Faz
ė	ь	с								1	8 4 1			$e_1 = \frac{6}{51} H_0 + \frac{6}{110} H_0 + \frac{1}{100} H_0 + \frac{6}{1100} H_0 + \frac{6}{1000} H_0 + \frac{6}{1000} H_0 + \frac{6}{1000} H_0 + \frac{6}{1000} H_0 + \frac{1}{1000} H_0 + $
F		A	а					"F + 1	F <sup>2</sup> -F <sup>1</sup>	<sup>А</sup> Н <sub>х</sub> т )	12 <sup>-H</sup> 21	^ <b>H</b>	,* My 2-My	<sup>10</sup> $f_1 = \frac{1}{2} \frac$
G	-A		ъ	с				40 <sub>y</sub> =	0,2 - 0,1	40 <sub>2</sub> =	012 - 011	AY, -	1 <sub>52</sub> - 1 <sub>5</sub>	10 y1 70 <sup>-1</sup> z1 420 <sup>-11</sup> z <sup>2</sup> 10 <sup>14</sup> s1
н	a <sub>1</sub>	ь <sub>1</sub>		-6 <sub>1</sub>	g			۵M2 -	Mzz-Mz4	H <sub>4</sub> + 2	u2 <sup>-M</sup> v1	-	<sup>H</sup> 2 <sup>-H</sup> 1	$h_1 = \frac{1}{10} H_{y_1} \frac{4^{10}y_1}{10} \frac{1}{10} \frac{1}{10^2} \frac{1}{170^{10}z_1} \frac{1}{10^2} F_1 \frac{1}{10} F_{0} F_{0} \frac{1}{10} F_{0} F_{0} \frac{1}{10} F_{0} F_{0} \frac{1}{10} F_{0} F_{0} \frac{1}{10} F_{0} \frac{1}{10} F_{0} F_{0} \frac{1}{10} F_{0} \frac$
1	°1	dı	-ē1	-d <sub>1</sub>	h	i		A7			٢.			$r_1 = \frac{6}{51} H_y + \frac{6}{10} M_y510_{z_1} \frac{26}{70} 140_{z_1} + \frac{6}{51} Z_{z_1} + \Gamma_1 + \frac{6}{10} FAz_y$
	-D	-E	-F	-G	-1	1-		•	*92 * 1	1	p = j = p = r	44		$m_1 = \frac{1}{10} y_1 \frac{1}{10} 10 z_1 \frac{23}{210} 100 z \frac{1}{10} z_1 z_{x1}$
-D	-a	-b		Α	t <sub>1</sub>	11	D	a						$\mathbf{m}_{1} = \frac{1}{10} \frac{1}{y_{1}} \frac{1}{10} \frac{10}{z_{1}} \frac{2}{35} \frac{100}{z_{1}} \frac{1}{10} \frac{1}{z_{1}} \frac{1}{z_{1}} \frac{1}{z_{1}}$
J	b	e	-B	-C	<b>v</b> <sub>1</sub>	×1	-J	-d	ſ					$\sigma_1 = \frac{2}{15} \frac{14}{y_1} \frac{12}{210} \alpha_2 \frac{2}{15} \frac{1F_1Z_1}{15}$
-F		-A	-a	-b	-ī,	-1	F		Ð	а		,		$t_1 = -\frac{6}{6}M_{\psi_1} - \frac{6}{6}M_{\psi_1} - 510_{\psi_1} + \frac{1}{2}140_{\psi_1} + \frac{6}{6}\xi_1 + \frac{6}{6}\xi_$
м	в	с	d	e	-w <sub>1</sub>	-×1	- M	-В		-d	ſ		4	517.1 7 272 251 31.1 9
-н	e	J	-ē,	-j <sub>1</sub>	-g	-h	н	r <sub>1</sub>	<sup>m</sup> 1	-r <sub>1</sub>	-m <sub>1</sub>	g		$m = \frac{1}{10} \frac{M}{y_1} \frac{1}{10} \frac{10}{z_1} \frac{2}{15} \frac{100}{z_1} \frac{1}{10} \frac{1}{z_3} F_1$
N	f <sub>1</sub>	h	-ī,	-ħ <sub>1</sub>	m	г	0	n <sub>1</sub>	<sup>a</sup> 1	-n <sub>1</sub>	-ō <sub>1</sub>	-m	0	$\mathbf{z}_{1} = \frac{1}{30} \mathbf{w}_{1} \mathbf{v}_{-\frac{1}{60}}^{-\frac{1}{2}} \mathbf{z}_{1}^{+\frac{1}{8q}} 1^{2} \Delta 0_{\frac{1}{2},\frac{1}{30}} \mathbf{f}_{1} \mathbf{z}_{\frac{1}{2}},$
L	· · ·		I		I	·	L	L			l	J		and vice versa.

-113-

a	a <sub>1</sub>						Detai	ls to th	nese tei	rms are	given	in app	endices	G, I	and J
b	a2	a <sub>3</sub>		-			reape.	cervery							
с	d <sub>1</sub>	d <sub>3</sub>	b <sub>1</sub>		-										
d	<sup>d</sup> 2	d <sub>4</sub>	<sup>ь</sup> 2	b <sub>3</sub>											
е	e <sub>1</sub>	e2	f <sub>1</sub>	f <sub>2</sub>	c <sub>1</sub>										
f	e <sub>3</sub>	e4	f <sub>3</sub>	fA	°2	°3									
	-a	-b	-c	-d	-е	-f									
J	a <sub>4</sub>	<sup>a</sup> 6	d <sub>5</sub>	d <sub>7</sub>	е <sub>5</sub>	e <sub>6</sub>	-J	<sup>a</sup> 8							
g	<sup>a</sup> 5	a <sub>7</sub>	d <sub>6</sub>	d <sub>8</sub>	<sup>е</sup> 7	e <sub>8</sub>	-g	a <sub>g</sub>	<sup>a</sup> 10						
h	d <sub>5</sub>	d <sub>7</sub>	<sup>b</sup> 4	b <sub>6</sub>	f <sub>5</sub>	f <sub>6</sub>	-h	d <sub>g</sub>	d <sub>11</sub>	<sup>b</sup> 8					
1°	d <sub>6</sub>	d <sup>8</sup>	b <sub>5</sub>	<sup>b</sup> 7	f <sub>7</sub>	f <sub>8</sub>	-1°	<sup>d</sup> 10	<sup>d</sup> 12	<sup>b</sup> 9	<sup>b</sup> 10				
0	е <sub>5</sub>	e <sub>6</sub>	f <sub>5</sub>	f <sub>6</sub>	°4	°6	-0	e <sub>0</sub>	e <sub>11</sub>	f <sub>9</sub>	f <sub>11</sub>	°8			
11	<sup>е</sup> 7	е <sub>8</sub>	f <sub>7</sub>	f <sub>8</sub>	с <sub>5</sub>	с <sub>7</sub>	-N	е <sub>10</sub>	e <sub>12</sub>	f <sub>10</sub>	f <sub>12</sub>	с <sub>9</sub>	° <sub>10</sub>		

Table 4.7 Nonlinear Tangential Stiffness Matrix

E <sub>1</sub>	A <sub>1</sub>													
E2	A <sub>2</sub>	۸ <sub>3</sub>												
E <sub>3</sub>			-A <sub>1</sub>											
E4			-^2	-^ <sub>3</sub>			Full de	efinitio	on of te	erms se	e appen	dices	G, I an	d.
Е <sub>5</sub>	<sup>B</sup> 1	<sup>B</sup> 3	C <sub>1</sub>	с <sub>3</sub>	D <sub>1</sub>		respect	tively						
E <sub>6</sub>	<sup>B</sup> 2	<sup>B</sup> 4	c <sub>2</sub>	C <sub>4</sub>	D <sub>2</sub>	D3								
	-E <sub>1</sub>	-E <sub>2</sub>	-E3	-E <sub>4</sub>	-E <sub>5</sub>	-E <sub>6</sub>								
E <sub>7</sub>	A4	А <sub>6</sub>			<sup>B</sup> 13	<sup>B</sup> 14	-Е <sub>7</sub>	А <sub>8</sub>						
<sup>Е</sup> 8	A <sub>5</sub>	۸ <sub>7</sub>			<sup>B</sup> 15	<sup>B</sup> 16	-E <sub>8</sub>	А <sub>9</sub>	A <sub>10</sub>					
E <sub>9</sub>			-^44	-A <sub>6</sub>	C <sub>13</sub>	C <sub>14</sub>	-Е <sub>9</sub>			-^8				
E <sub>10</sub>			-^ <sub>5</sub>	-A <sub>7</sub>	с <sub>15</sub>	с <sub>16</sub>	-E <sub>10</sub>			-^9	-A <sub>10</sub>			
E <sub>11</sub>	<sup>B</sup> 5	<sup>B</sup> 7	с <sub>5</sub>	с <sub>7</sub>	D <sub>4</sub>	<sup>D</sup> 5	-E <sub>11</sub>	в <sub>9</sub>	B <sub>11</sub>	с <sub>9</sub>	C <sub>11</sub>	D <sub>8</sub>	]	-
E <sub>12</sub>	<sup>B</sup> 6	<sup>B</sup> 8	с <sub>6</sub>	с <sub>8</sub>	D <sub>6</sub>	D <sub>7</sub>	<sup>-E</sup> 12	<sup>B</sup> 10	<sup>B</sup> 12	с <sub>10</sub>	C <sub>12</sub>	D <sub>9</sub>	<sup>0</sup> 10	

Table 4.8 Nonlinear Geometrical Matrices

Table 4.9 Initial Stiffness Matrix

$$A_{1} = [\frac{6}{11}(H_{y1} + P_{1}Z_{y}) + \frac{16}{1}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{6} = [-\frac{1}{30}(H_{y1} + P_{1}Z_{y}) - \frac{1}{30}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{2} = [\frac{1}{30}(A_{Hy} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{2} = [\frac{1}{30}(A_{Hy} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{2} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) + \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{3} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) + \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{3} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) + \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{3} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) + \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{3} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) + \frac{1}{30}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{4} = [\frac{21}{15}(H_{y1} + P_{1}Z_{y}) + \frac{1}{30}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) + \frac{1}{30}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) + \frac{1}{30}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) + \frac{1}{30}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) + \frac{1}{30}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) - \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) - \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) - \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) - \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) - \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) - \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) - \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}) - \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}] - \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}] - \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y})]_{\phi_{0}} + A_{1} = [\frac{1}{10}(H_{y1} + P_{1}Z_{y}] - \frac{1}{10}(\Delta H_{y} + \Delta P \Delta Z_{y}] + \frac{1}{10}(\Delta H_{y}$$

$-\frac{1}{\ell}$							$\frac{1}{\ell}$						
	$-\frac{6}{\ell^2}$	- 4						$\frac{6}{\ell^2}$	$-\frac{2}{2}$				
			$-\frac{6}{\ell^2}$	- <u>4</u>						$\frac{6}{\ell^2}$	- <u>2</u> <u>R</u>		
					$-\frac{6}{\ell^2}$	- <u>4</u> <u>R</u>						$\frac{6}{2}^{2}$	$-\frac{2}{\ell}$
- <u>1</u> £							$\frac{1}{\ell}$						-
	6 g <sup>2</sup>	4 2						$-\frac{6}{\ell^2}$	- <u>4</u> <u>e</u>				
			$\frac{6}{\ell^2}$	4 2						$-\frac{6}{\ell^2}$	$\frac{1}{2}$		
					6 e <sup>2</sup>	<u>4</u> 2						$-\frac{6}{l^2}$	$\frac{4}{2}$

## Table 4.10 Linear Strain Matrix

Linear strain can be written as

 $\varepsilon^{\ell} = \varepsilon^{\ell}_{xxo} - \gamma \phi_z - Z \phi_z - \overline{w} \phi_w$ 

where

$$\varepsilon_{xxo}^{\ell} = W_{,x}, \quad \varphi = V_{,xx}$$
$$\Phi_{z} = U_{,xx}, \text{ and } \quad \Phi_{w} = \Phi_{,xx}$$

## Table 4.11 Non-Linear Strain Matrix

001/2 0,1/2 - A/2 avo - 1/2 auo  $\theta_{v2}/2$ θ<sub>112</sub>/2  $-\overline{B}/2$ - c/2 av1 - C/2 - B/2- a<sub>u1</sub>  $\overline{C} = -\frac{11}{12} \phi_1 U_1 + \frac{118}{12} \phi_1 U_2$  $A = -\frac{6}{12} \phi_1 V_1 - \frac{4}{10} \phi_1 \theta_{V1} + \frac{6}{12} \phi_1 V_2 - \frac{2}{10} \phi_1 \theta_{V2}$ Nonlinear strain  $\Sigma^{\rm NL} = \frac{1}{2} (U_{,x}^2 + V_{,x}^2) - Y_{,x}^{\phi} X_{,x}^{V} + Z_{,x}^{\phi} X_{,x}^{U}$  $a_{vo} = -\frac{6}{1^2}V_1 - \frac{4}{1}v_1 + \frac{6}{1^2}V_2 - \frac{2}{1}v_2$  $B = \frac{6}{12} \phi_2 V_1 + \frac{2}{1} \phi_2 \theta_{V1} - \frac{6}{12} \phi_2 V_2 + \frac{4}{1} \phi_2 \theta_{V2}$ +  $V_{xx}\phi$  +  $U_{xx}\frac{\phi^2}{2}$  -  $U_{xx}\phi$  -  $V_{xx}\frac{\phi^2}{2}$  $\overline{A} = \frac{6}{1^2} \phi_1 V_1 - \frac{4}{1} \phi_1 \theta_{v1} + \frac{6}{1^2} \phi_1 V_2 - \frac{2}{1} \phi_1 \theta_{v2}$  $a_{v1} = \frac{6}{12}V_1 + \frac{2}{10}v_1 - \frac{6}{12}V_2 + \frac{4}{10}v_2$  $C = -\frac{4}{1^2} \phi_1 v_1 + \frac{48}{1^2} \phi_1 v_2$  $a_{uo} = -\frac{6}{12}U_1 - \frac{4}{10}U_1 + \frac{6}{12}U_2 - \frac{2}{10}U_2$  $\mathbb{B} = \frac{6}{1^2} \phi_2 U_1 + \frac{2}{1} \phi_2 \theta_{u1} - \frac{6}{1^2} \phi_2 U_2 + \frac{4}{1} \phi_2 \theta_{u2}$  $a_{u1} = \frac{6}{12}U_1 + \frac{2}{1}u_1 - \frac{6}{12}U_2 + \frac{4}{1}u_2$ 



FIG. 4.1 TRANSFORMATION OF BEAM-ELEMENT

## Chapter 5

## Analytical Procedure and Computer Program Structure

## 5.1 Introduction

Using finite element concepts a comprehensive analytical procedure and computer implementation of the general formulation for a beam-column of thin-walled open cross-section in space is presented. The program TDA (Three Dimensional Analysis) can be used in both the elastic and inelastic resulting analytical processes are systematically arranged in for numerical evaluation.

The sectional and sectorial properties for I-section beamcolumns have been mentioned in several references such as Chan and Kitipornchai (1987), Zhibrroski-Koscia (1967), and Zienkiewicz (1977). The finite element concepts are fully explained by Zienkiewicz (1977), Bathe and Wilson (1976), and Rao (1982) for the analysis of structural members.

A full coverage of the analytical procedure and general descriptions for the program TDCP is presented herein. Details of the results obtained by applying TDCP to selected problems covering different aspects of elastic and inelastic behaviour are contained in chapters 6, 7, and 8 and in references (El-Khenfas 1987c,b)

## 5.2 Finite Element Method

The first step in the finite element analysis is equivalent to replacing the domain having an infinite number of degrees of freedom by a system having a finite number of degrees of freedom.
#### 5.2.1 Number of Elements (NEL)

If the beam has no change in geometry, material properties and external conditions ( like applied loads ), then it can be divided into equal lengths as shown in Fig. 5.1a. If discontinuities are present Figs (5.1b and 5.1c), nodes have to be introduced at these discontinuities. The number of elements to be chosen for idealization is related to the accuracy desired with an increase in the number of elements generally providing more accurate results, but at the cost of greater computation.

#### 5.2.2 Number of Segments over the Cross-Section (NELS)

In order to evaluate the cross-sectional properties appearing in the various element stiffness matrices (El-Khenfas and Nethercot 1987a), it is convenient to divide each plate element of the crosssection into a number of small segments. The size of these segments must be chosen with care for both the elastic and inelastic analysis. The number required for reasonable accuracy depends on the pattern of applied loads (which affects the type of response) and the residual stress patterns (which affects the spread of yield). Any strain distribution over the cross-section leading to a mix of elastic and yielded regions may be conveniently allowed for by considering the member's cross-section to be divided as shown in Fig. 5.2a. The flanges and the web are divided into NY segments along each plate length and NZ through its thickness.  $\Delta A_{i,j}$  is the segment area, where i=1,NY and j=1,NZ and its centroidal coordinate are  $Y_{i,j}$  and  $Z_{i,j}$  as given in Fig. 5.2b.

For elastic analysis the number of segments over the crosssection can simply reflect the natural division i.e. web, half flange

- 121 -

etc. For the elastic-plastic analysis, however, the sub-division must be selected with care, with the number of segments depending on the loads and initial geometrical imperfections, e.g. if the load applied is uniaxial and residual stresses are absent then the flange thickness may be divided into a small number of segments (NZ equal to 10 and NY taken between 20 and 30). Other load patterns which cause the spread of yield to be unsymmetrical as illustrated in Fig. 5.2b, require a much finer subdivision e.g. NY and NZ taken as 80 and 10.

#### 5.2.3 Section Properties

The accurate determination of section properties is one of the most important aspects of the whole analytical process. It depends principally on locating the instantaneous positions of the centroid and the shear centre for the partially yielded cross-section.

#### 1- Centroid

Fig 5.3 shows the cross-section of a thin walled steel beam. Every element is divided into a number of plate segments. The overall cross-section centroid coordinates  $Y_{c}$  and  $Z_{c}$  with respect to reference axes Y and Z can be evaluated as:

$$Y_{c} = \frac{\sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, J=NZ} Y_{i,j} A_{i,j}}{\sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, J=NZ} A_{i,j}}$$

- 122 -

$$Z_{c} = \frac{\sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, J=NZ} Z_{i,j} A_{i,j}}{\sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, J=NZ} \Delta A_{i,j}}$$

The moments of inertia about the Y and Z axes and the product moment of inertia about these axes can be expressed as

$$I_{Y} = \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, j=NZ} z_{i,j}^{2} A_{i,j}$$

$$I_{Z} = \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY,j=NZ} Y_{i,j}^{A} A_{i,j}$$

$$I_{YZ} = \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, j=NZ} Z_{i,j}Y_{i,j}A_{i,j}$$

The above properties with respect to the y and z axes can be rewritten as

$$I_y = I_Y + \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, j=NZ} (Z_{i,j} - Z_c)^{2} \Delta A_{i,j}$$

$$I_z = I_Z + \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, j=NZ} (Y_{i,j} - Y_c)^{2} \Delta A_{i,j}$$

$$I_{yz} = I_{YZ} + \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, j=NZ} (Z_{i,j} - Z_c) (Y_{i,j} - Y_c) A_{i,j}$$

.

.- 123 -

$$TAN\beta = \frac{I_{yz}}{I_z - I_y}$$

where  $\beta$  is the inclination of the principal axes (  $\eta$  and  $\zeta$ ) to the y and z axes and is positive if anticlockwise from the y-axis). The principal moments of inertia about nand  $\zeta$  can be written as:

$$I_{z} = I_{y} COS^{2}\beta + I_{z} SIN^{2}\beta - I_{yz} SIN^{2}\beta$$

$$I_{\eta} = I_{z} COS^{2}\beta + I_{y} SIN^{2}\beta - I_{yz} SIN 2\beta$$

#### ii- Shear Centre

Sectorial coordinates ( $\omega$ ) of points  $s_1$  and  $s_2$  can be written as  $\omega_{Ds_1s_2}$ , where D denotes the ' pole',  $s_1$  and  $s_2$  are points of intersection of the radius,  $ds_{12}$  is the distance between the two points, and  $h_{s_1s_2}$  is the distance of pole D from a line passing through points  $s_1$  and  $s_2$ . To determine the sectorial coordinates of a beam having an arbitrary thin-walled open cross-section as illustrated in Fig. 5.4, by definition  $\omega_{Ds_1s_2}$  is equal to the algebraic sum of twice  $\delta\omega Ds_1s_2$  The increment of this value is equal to the product of

 $\delta \omega_{\text{Ds}_1 \text{s}_2} = \delta d s_{12} x h_{s_1 s_2}$ 

if  $\delta \omega_{DS_1S_2}$  is equal to  $\delta \omega$ , and  $\delta ds_{12}$  is equal to ds. Substituting for these in the above equation and integrating from 0 to  $s_n$ 

$$\omega = \int_{-1}^{n} s_1 s_2 ds$$

where  $s_n$  is the length of the path between  $s_1$  and  $s_2$  for any

\_ 124 -

The sectorial properties with respect to an arbitrary pole D are  $(S_{\omega})$  sectorial moment of a section from pole D, product warping moment of inertia about Y and Z axes  $(I_{\omega y})$  and  $(I_{\omega z})$ , and warping moment of inertia  $(I_{\omega})$ . They can be obtained from

$$s_{\omega} = \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY,j=NZ} \omega_{i,j} A_{i,j}$$

$$I_{\omega y} = \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, j=NZ} Z_{i,j} \omega_{i,j} A_{i,j}$$

$$I_{\omega z} = \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, j=NZ} Y_{i,j} A_{i,j}$$

$$I_{\omega z} = \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, j=NZ} \omega_{i,j}^{2} A_{i,j}$$

Those properties can used to determine the location of shear centre. Further details are given in several references such Chen and Atsuta (1977) and Zhibrorski-Koscia (1967)

$$Y_{s} = \frac{I_{zz}I_{y\omega} - I_{yz}I_{\omega z}}{I_{yy}I_{zz} - I_{yz}^{2}}$$

$$Z_{s} = \frac{I_{yy}I_{z\omega} - I_{yz}I_{\omega y}}{I_{yy}I_{zz} - I_{yz}^{2}}$$

Once the shear centre coordinates  $(Y_s \text{ and } Z_s)$  have been determined, then it may be used as a ' pole' to recalculate the

previous sectorial properties which can be written in the following form:

$$\omega_{i,j} = \sum_{i,j=1}^{i=NZ, j=NY} \rho_{i,j}^{B_{i,j}}$$

where  $\rho_{i,j}$  is the distance between the centroid of the segment and the shear centre and  $B_{i,j}$  is the segment width.

$$(\omega_{o})_{i,j} = \sum_{i,j=1}^{i=NZ, j=NY} \rho_{oi,j}B_{i,j}$$

$$(\omega_{n})_{i,j} = \frac{1}{A} \sum_{i,j=1}^{i=NZ, j=NY} [(\omega_{i,j}^{t})_{i,j}^{t}]^{t}$$

$$I_{\omega y} = \sum_{n=1}^{n=NELS} \sum_{i,j=1}^{j=NY,i=NZ} \omega_{i,j}Y_{i,j}t_{i,j}B_{i,j}$$

$$I_{\omega z} = \sum_{n=1}^{n=NELS} \sum_{i,j=1}^{j=NY,iNZ} \omega_{i,j} Z_{i,j} t_{i,j} B_{i,j}$$

$$\widetilde{I}_{\omega} = \sum_{n=1}^{n=NELS} \sum_{i,j=1}^{j=NY,iNZ} \omega_{i,j}^{2} t_{i,j}^{B} t_{i,j}$$

In which A is the cross-section area and  $t_{i,j}$  is the segment thickness.  $(\omega_n)_{i,j}$  is the segment normalized function,  $I_{\omega}y$ ,  $I_{\omega}z$  are the product warping moment of inertia with respect to y and z axes. The warping moment of inertia is denoted by  $I_{\omega}$ . whilst  $(\rho_0)_{i,j}$  is the distance between the centroid of the segment and sectional centroid. At any stage in the analysis it is assumed that the full cross-sectional strain distributions will be known (from the deformation). Increments of the internal forces corresponding to changes of the strain distribution due to a change in the applied load parameter may then be determined from the stresses calculated in each plate segment at its centroid. The stress resultantsmay be evaluated numerically as:

$$P = \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY,j=NZ} \sigma_{i,j} A_{i,j}$$

$$M_{y} = \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, j=NZ} \sigma_{i,j}^{Z_{i,j}, A_{i,j}}$$

$$M_{z} = \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, j=NZ} \sigma_{i,j}Y_{i,j}A_{i,j}$$

$$M_{\omega} = \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, j=NZ} \sigma_{i,j} \omega_{i,j} A_{i,j}$$

$$M_{\rho} = \sum_{n=1}^{NELS} \sum_{i,j=1}^{i=NY, j=NZ} \sigma_{i,j} \rho_{i,j}^{2\Delta A_{i,j}}$$

where  $\Delta A_{i,j} = B_{i,j}t_{i,j}$ 

#### 5.3 Assembly of the Stiffness Matrices

The element stiffness matrices of equation 4.17 were evaluated with respect to nodal displacements in the local coordinates. For an illustration equation 4.14 can be written as:

$$\{r\}_{L}^{T}[K_{T2}^{E}]\{r\}_{L} + \{r\}_{L}\{P^{t}\} = 0$$
(5.1)

where

$$\{\overline{P}_{F}\} = \{\overline{P} + r_{O}\overline{P}\}$$

The transformation of the element stiffness matrices from local principal generalized coordinates to the global coordinates can take place as

$$\{r_{E}\}_{L}^{T}[K_{T}^{E}]\{r_{E}\}_{L} + \{r_{E}\}_{L}\{P^{E}\} = 0$$
(5.2)

where

$$\{r_{\rm E}\}_{\rm L} = [T]\{r_{\rm L}\}_{\rm G}$$
(5.3)

where the transformation matrix [T] is given and explained in chapter 4 equation 4.29.  $\{r\}_L$  and  $\{r\}_G$  are the element displacements with respect to local and global respectively.  $P_E$  is the element applied loads and [ $K_T^E$ ] is the assembly to the element stiffness matrices. Equation 5.1 can be written in the local system as

$$\delta \Pi = \{r\}^{T} [K_{T}^{E}] \{r\}_{L} + \{\overline{P}_{E}\}^{T} \{r_{E}\} L$$
(5.4)

Substituting equation 5.3 into equation 5.4 yields

$$\Xi(\mathbf{x}) = [T]^{T}[K_{T}^{E}][T]\{r\}_{G} + \{\overline{P}_{E}\}_{G}$$
(5.5)

or

$$[K_T^E]_G]\{r\}_G + \{\overline{P}_E\} = 0$$
(5.6)

For equilibrium equation 5.6 equals zero as

$$\Xi(\mathbf{x}) = \frac{\delta \Pi}{\delta \{r\}_{G}} = 0$$
 (5.7)

If the beam is divided into a number of elements then the assembly of these members can be presented in the following form

$$[K_{T}]{R} + {P} = 0$$
 (5.8)

where  $\{R\}$ ,  $\{P\}$ , and  $[K_T]$  are the assembly to the elements displacements, applied loads, and to the stiffness matrices (tangential, geometrical linear and nonlinear and initial displacement matrix respectively). Their assembly arrangement are given in equations 5.9 to 5.11.



equation 5.8 represents the finite element form of the total equilibrium equation.

### 5.4 Method of Solution

The complexity of the analysis is related to a number of factors. It depends upon the nature of the loading, the material properties, and the kinds of assumptions made in deriving the stiffness matrices (Chapter 4). A solution can be obtained by using the Newton-Raphson technique, which has been employed by many investigators. Fig. 5.5 illustrates this method.

The discretised system of algebraic equation takes the form

$$\Xi(R) = [K_T]{R} + {P} = 0$$
 (5.12)

If  $R=R_n$  is a solution to the above equation, it can be written, by using Taylor's expression as

$$E(R_{n+1}) = E(R_n) + (\frac{dE}{dR})_n \Delta R_n = 0$$
 (5.13)

with

$$R_{n+1} = R_n + \Delta R^n \tag{5.14}$$

From equation No.542 we obtain

$$\frac{dE}{dR} = \frac{dP}{dR} = K_{T}$$

The steps of the Newton-Raphson procedure employed are:

1- Calculate the out of balance forces

$$\{\Delta P\}_{n+1} = \{P\} - [K_T]_n \{R\}_n$$

.<mark>– 130 –</mark>

2- Compute the increments in displacements to equilibrate the unbalanced forces.

$$\{\Delta R\}_{n} = [K_{T}]^{-1} \{\Delta P\}_{n}$$

3- Update the displacements

 ${R}_{n} = {R}_{n-1} + {R}_{n}$ 

4- Repeat 1 to 3 until the imbalance of forces is sufficienttl small.

5- Apply another load increment and repeat steps 1 to 4.

#### 5.5 Convergence Criteria

The numerical solution scheme outlined above is based on a step by step approach which involves the combination of an incremental and an iterative process until the desired convergence limit is achieved. Possible convergence criteria can be classified under:

1- Force criteria

2- Displacement criteria

3- Stress criteria

Bergan and Clough (1972) suggested three alternative norms for measuring the tolerance  $\gamma$  in order to achieve convergence which were

1- " Modified Absolute Norm :;

$$|\Sigma| = \left[ \begin{bmatrix} \frac{1}{N} \sum \frac{\Delta R_i}{R_i, ref} \end{bmatrix} \right]$$

- 131 -

$$|\Sigma| = \left[ \begin{bmatrix} \frac{1}{N} \sum \frac{\Delta R_i}{R_i, ref} \end{bmatrix}^{\frac{1}{2}} \right]$$

3- " Maximum Norm "

$$|\Sigma| = \max_{i} \frac{\Delta R_{i}}{R_{i,ref}}$$

They employed the following criterion to define those norms as

Σ < Υ

where  $\gamma$  ranges between  $10^{-2}$  to  $10^{-6}$  depending on the desired accuracy, N is the total number of unknown components, and  $\Delta R_i$  is the change in displacement component i. Every component is scaled by a reference displacement quantity  $R_{i,ref}$ .

Crisfield (1981) modified Riks's approach to be suitable for the finite element method and to produce a fast incremental solution. The procedure was applied in conjunction with the modified Newton-Raphson technique. He reported that his approach not only allowed limit points to be passed but also improved the convergence characteristics. He provided illustrative examples which covered large deflection analysis of shallow elastic shells and the collapse analysis of a stiffened steel diaphragm from a box-girder. His criterion may be stated as:

$$\frac{\left|\overline{B}\right|}{\max \left(\lambda \left|P\right|, \left|\overline{r}\right|\right)} < 10^{-4}$$

 $|\overline{g}| = g^T \kappa_D^{-1} g$ 

Where  $|\overline{g}|$  is proposed by Peano and Riccioni (1978).  $\overline{g}$  is the out-of-balance force, P the total applied force,  $\overline{r}$  the reaction vector and  $K_{D}$  is a diagonal matrix containing the leading diagonal elements of the tangent stiffness matrix.

To ensure convergence to the correct result certain simple requirements have to be satisfied, the most obvious of which is clearly that the displacement function should be able to represent the displacement distribution as closely as possible. In general the larger and more complex the numerical problem the more difficult it will be to achieve satisfactory convergence.

In the present analysis two methods were applied together to meet the necessary convergence limit; these can be explained as:

i- Out of balance forcesii- Iterative process

#### 5.5.1 Out of Balance Forces

If the sum of the total forces at each node is not in equilibrium i.e. the summation of the internal forces does not balance the external forces, the difference between them is termed the out of balance force. These residual forces should be re-applied until both balance each other. To explain this phenomenon, the illustrative example of Fig. 5.6 shows the load position, beam segment length, and support conditions. The out-of-balance forces are

- 133 -

$$\Delta Q_{yi} = \left[\frac{M_{i}^{L} - M_{i}^{R}}{L_{i}} + \frac{M_{i+1}^{L} - M_{i+1}^{R}}{L_{i+1}}\right] - Q$$

 $\Delta M_{\omega i} = (M_{\omega i}^{L} - M_{\omega i}^{R} + M_{\omega}$ 

where  $M_{yi}^L / M_{yi}^R$  and  $M_{\omega i}^R / M_{\omega i}^R$  are the bending moments about the Y-axis and bimoment at node i from left and right.  $\Delta Q$  sub i and  $\Delta$ .  $M_{\omega i}$  are out-of-balance forces for the shear and bimoment respectively,  $L_i$  is the element segment length, and the subscript i varies from 1 to NJ (NJ= number of joints). If the residual forces  $\{\Delta P_i\}$  obtained from the equilibrium of node i do not have small values.  $\{\Delta P_i\}$  should be reapplied as an increment of force and this process repeated until a balance is achieved. A full description is given in Fig. 5.6.

where

$$\{\Delta P_i\} = \begin{bmatrix} \Delta P_i & \Delta M_{yi} & \Delta Q_{zi} & \Delta M_{zi} & \Delta Q_i & \Delta M_{ti} & \Delta M_{\omega}i \end{bmatrix}$$

where  $\{\Delta P_i\}$  is an array to the out-of-balance forces at node i.

#### 5.5.2 Iterative Process

In this method both the unbalanced or residual forces within the structure nodes and the increment of displacements are implemented at the same time. The out of balance force and the variation of the deformations are described as:

i- Out of balance force

$$\Sigma = \left[ \frac{\left[ \Delta P_{I} \right]^{T} \left[ K_{D} \right] \left[ \Delta P_{I} \right]}{\left[ P_{T} \right]^{T} \left[ K_{D} \right] \left[ P_{T} \right]} \right]^{2}$$

ii- Increment of Displacements

$$\Sigma = \left[ \frac{\left[ \Delta R_{I} \right]^{T} \left[ K_{D} \right] \left[ \Delta R_{I} \right]}{\left[ R_{T} \right]^{T} \left[ K_{D} \right] \left[ R_{T} \right]} \right]^{2} < \gamma$$

where  $\Delta R_{I}$  are the incremental displacements at the iteration,  $R_{t}$  are the updated total displacements,  $\Delta P_{I}$  is the out-of-balance force and  $P_{T}$ are the total applied forces.  $K_{D}$  is the diagonal stiffness of the whole stiffness matrix. Based on the accuracy desired the tolerance  $\gamma$  is chosen in this analysis to be equal to 0.001, for both case i and case ii.

#### 5.6 Computer Program Description

A very general finite element program (TDCP) has been written in the Fortran 77 Language. It contains many subroutines, the main functions of which are to follow the loss of stiffness due to spread of yield and hence to trace the three dimensional load-deflection response up to collapse of a beam-column having almost any open cross-section composed of a series of flat plates. The Newton-Raphson technique has been used for both the the elastic and the inelastic ranges. The power and versatility of the program has been demonstrated by its application to a range of selected problems; these are presented in chapters 6, 7,

- 135 -

and 8 respectively.

#### 5.6.1 Program Structure

The full TDCP details together with its listing are given in ref. (El-Khenfas 1987a). This program uses many subroutines to provide a solution for the elastic and inelastic response of beam-columns. The more important routines and their functions are briefly detailed below.

MAIN Reads the input data, controls the calling sequences.

- OUT1,2,3 Presents the output of loading, displacements, stresses, internal forces, etc.
- STIFF1,2 Forms the linear and nonlinear tangential stiffness matrices.
- GEOMET1,2 Forms the linear and nonlinear geometrical stiffness matrices.
- TRANS Transforms the elements stiffness and loads, and displacements from local to global. (or to the references axis).
- ASSEM Controls the assembly of the total stiffness matrices or the applied loads.
- ITLOAD Controls the load or displacement increments.
- STRAN1,2 Evaluates linear and nonlinear curvatures
- BOUND Modifies the total stiffness matrix of the structure according to the boundary conditions.
- PROPER Calls subroutines to calculate the section and sectorial properties, trace spread of yield, and internal forces.

- 136 -

- ELASPL Perform the elastic-plastic analysis using Newton-Raphson technique.
- RESD Calls several routines to generate different residual stress distributions.
- INDEF Forms the initial crookedness by sinusoidal or polynomial functions.
- GAUSS Used to solve the equilibrium equation by Gauss elimination.
- START Restart facility from a previous equilibrium solution.
- STORE Store the whole of the information obtained from the previous iteration.

A basic flow chart is given in Fig. 5.7.

### 5.7 Computational Steps

The computational processes of TDCP can be explained in the following steps

#### STEP 0 Initialization

In this step initialisations has to be made to the whole arrays

STEP 1 Input Data

a) Beam Details

N J	number	of	nodes		
NEL	number	of	element	S	
NDF	number	of	degree	of	freedom

SLEND slenderness ratio  $(\frac{L}{r_y})$ 

# b) ( Cross-Section Details

NES .	number of elements in the cross-section
NJS	number of joints in the section
T(M)	plate thickness
NOSI(M)	node I number
NOSJ(M)	node J number
SEY1(M)	coordinates of node 1 with respect to Y-axis
SEY2(M)	coordinates of node 2 with respect to Y-axis
SEZ1(M)	coordinates of node 1 with respect to Z-axis
SEZ2(M)	coordinates of node 2 with respect to Z-axis
м	is the number of elements in the section
c) Applied Load	ds
ECY, ECZ, ECR	load eccentricities
APL(7xN-6)	Axial load (F <sub>x1</sub> )
APL(7xN-5)	Shear force Q <sub>z1</sub>
APL(7xN-4)	Bending moment M y1
APL(7xN-3)	Shear force Q <sub>y1</sub>
APL (7 xN-2)	Bending moment Mz1
APL(7xN-1)	Torsion M <sub>x1</sub>
APL(7xN)	Bimoment M 1
APL(7xN+1)	Axial load (F <sub>x2</sub> )
APL(7xN+2)	Shear force Q <sub>z2</sub>
APL(7xN+3)	Bending moment My1
APL(7xN+4)	Shear force Q <sub>y1</sub>

APL(7xN+5)	Bending moment M <sub>z1</sub>
APL(7xN+6)	Torsion M <sub>x1</sub>
APL(7xN+7)	Bimoment M w1
N	Number of elements (1,2,,NEL)
INCL	Type of load increment
1	Axial load
2	Bending moment about Y-axis
3.	Bending moment about Z-axis
12	Axial + Bending about Y-axis
13	Axial + Bending about z-axis
123	Biaxial bending

Different combination can be performed

٠

d) .	Initial D	Deflection (R)
IND	EF	O No initial deflection
		1 Initial deflection present
U <sub>0</sub> ,	V <sub>0</sub> , Φ <sub>0</sub>	Out-of-plane and in-plane deflections and twisting
		respectively (they are given in TDCP by ACY, ACZ,
		ACR with respect to L)
IND	EFT	0 Sine wave
IND	EFT	1 polynomial ( you have to insert to the program
		the wanted function)
e) _	Type of A	analysis
IAN	AL	O Elastic
IAN	AL	1 Inelastic

Analysis options

•

L	i	ne	Ş	а	r
	L:	Li	Line	Line	Linea

2 Non-linear

3 Full non-linear

f) Stresses

ŧ.

IRES	O No residual stress
	) Residual stress present
KIND	1 Leigh distribution
	2 Parabolic distribution
	3 Triangular Distribution
	4 ~ 6 Other types
σ <sub>f</sub>	Flange tip residual stress
<sup>σ</sup> fw	Web-flange junction residual stress
σ <sub>w</sub>	Web residual stress
NMAT	Number of material type
ESHD	Strain hardening (values)
E	Young's modulus
G	Shear Modulus
g) Boundary Co	nditions (B.C.)
IRR	Number of restrained nodes (B.C)
K KM	O Degree of freedom restrained
	1 Degree of freedom unrestrained
JJM	Number of the degree of freedom restrained

STEP 2 Load Increment

PAX = AP1 + AFX1

Where PAX is the total load, AP1 is the load increment (usually given as ratio e.g. with respect the squash load or Euler load in the case of a compressive column load), and AFX1 is equal to the

- 140 -

value of the internal load of the previous step.

PAMY, Z = AP1Y, Z + PAMY3

Where PAMY or Z is the total bending moment, AP1Y and AP1Z are either the increment of bending moment about the Y or Z-axis, and PAMY3 is the internal moment of the previous step.

#### STEP 3 Displacements

Displacements have to be updated after each load increment as

 $R = R + \Delta R$ 

#### STEP 4 Strains

Strains have to be calculated at each load increment ,and then used to calculated the internal forces of to check spread of yield.

#### STEP 5 Section and Sectorial Properties

Section and sectorial properties have to be calculated each time of calculating curvatures (  $\phi_y$ ,  $\phi_z$ ,  $\phi_\omega$ ,  $\Sigma_o$ ).

STEP 6 Store

The complete results obtained have to be saved by storing them

#### STEP 7 Start

In this step the program can be restarted from where it stopped

Step 8 Control Cards

IYYY 0 Start th program from the beginning 1 Restart from where it stopped

MAXIT Number of cycles of loading required

ISTEP=ISTEP+1 Number of steps required

ISTEP=MAXIT When the number of steps (ISTEP) is equal to number of cycles of loading required, the program store the complete data and stop executions.

For other cycles of loads repeat step 2 up to step 8 again and again until failure (collapse occured).

#### 5.8 Conclusions

A very general TDCP (Three Dimensional Computer Program) based on finite element concepts has been developed. It was written in the Fortran 77 Language. The main function of its subroutines is to follow the loss of stiffness due to spread of yield and hence to trace the full load -deflection response of beam-column of an arbitrary open thin-walled cross-section subjected to different loading and boundary conditions up to collapse. This program can deal with three types of analysis which are Linear, Partial non-linear, or Full non-linear. These methods are fully explained in Chapter 4.

The section and sectorial properties were evaluated at each load increment together with the internal forces based of the strains which in turn depend upon the type of analysis required. The convergence to the correct result employed two methods at the same time. The first one used the Out-of-balance forces which represent the differences between the applied loads and the internal loads, while the second is based on the increments of the displacements with respect to the total displacements.

- 142 -

The main points drawn from this study may be summarised as:

- 1- A TDCP has been developed for the ultimate strength analysis of members to incorporating several types of open thin-walled sections (], T, L, I,etc.).
- 2- The linear and non-linear strains are used to generate the stress-results and to trace the spread of yield.
- 3- The section and sectorial properties can be evaluated in the elastic and inelastic ranges with respect to instantaneous shear centre and centroid.
- 4- The convergence to a correct result is based on the incremental displacements and the out-of-balance forces.





a. Beam divided to equal elements

- b. Nodes to be introduced at changing loads
- c. Node to be introduced at changing thickness

FIG.5.1 LOCATION OF NODES AT DISCONTINUITIES



a. Number of elements in the cross-section



b Number of plates in the cross-sectional element

FIG.5.2 NUMBER OF ELEMENTS AND SEGMENTS IN THE CROSS-SECTION



# FIG.5.3 CROSS-SECTION ILLUSTRATING THE PROCESS OF EVALUATING SECTION PROPERTIES



FIG. 5.4 SECTORIAL COORDINATES OF THIN WALLED SECTION



FIG.55 NEWTON RAPHSON METHOD

147-



a. Illustration of the number of elements(NEL), number of joints(NJ) and applied loads



b. Distribution of internal and external forces

# FIG.5.6 EXAMPLE SHOWING NUMBER OF ELEMENTS AND INTERNAL AND EXTERNAL FORCES



-149-



#### Chapter 6

## Elastic Analysis of Beam-Column in Three-Dimensional

#### 6.1- Introduction

Beam-columns are defined as structural members which are subjected to combined axial compression and bending. The bending may result from eccentric loading, transverse loads, applied moments, or combinations of loads.

The classical linear buckling theories for elastic beams and columns were investigated by Bleich(1933) who considered I-beams with unequal flanges subjected to axial load and equal and opposite end moments, then he applied his theory to the problem of beams of I-section with the tension flange restrained against lateral displacements. Johnston (1941), Massonnet (1947), Horne (1954) and Salvadori (1955,1956) studied doubly symmetric sections under combined axial thrusts and equal end moments.

Differential equations governing the flexural torsional buckling of thin-walled member subjected to eccentric thrusts have been used by Goodier (1942), Timoshenko and Gere (1961) and Vlasov (1961). Their results were confirmed by tests done by Hill and Clark (1951a,b). Goodier (1942,1956) examined the stability of a general open section under axial load and end moments; Anderson and Trahair (1972) concentrated on lateral torsional buckling of monosymmetric sections.

Flexural-torsional behaviour has been studied by, Vlasov (1961), and Trahair (1966,1969). They investigated the effect of elastic end restraints, while the effect of continuous diaphram

\_151 \_

restrained has been considered by Hancock and Trahair (1978,1979). A BSCA publication (1968,1970) investigated members of different cross sections such as (Tee, Channel, Z, L, etc. sections) under bending and torsion. Moore (1986) used the finite difference method to solve his developed differential equation in order to study nonuniform torsional buckling of Z-sections under lateral loads.

The behaviour of beam-columns depends principally on: their slenderness, the shape and dimensions of the cross-section, the exact form of the applied loading, and the conditions of support provided. These will control not only the load carrying capacity but also the form of structural response e.g. in-plane flexure, biaxial bending etc. (Chen and Atsuta(1977)). For slender members elastic analysis will provide a close approximation to their true behaviour; more stocky members, however, failing by inelastic action, will need to be analysed in a manner which accounts for the gradual development of plasticity. Even for such cases the basis for an adequate analytical approach must be developed from an elastic treatment of the problem. Such a formulation has been presented in chapter 3. In this chapter the application of this approach to a series of elastic problems, which cover a range of structural phenomena, is reported.

#### 6.2 Analysis

The analysis used herein employs the set of assumptions normally associated with analytical studies of the behaviour of thinwalled beam-columns in the elastic range. These are:

-152 -

- 1 The member may have any arbitrary open cross-section with each plate element having a thickness significantly less than its width.
- 2- The member's cross-section is constant along the span.
- 3- Both shear deformation and distortion of the cross-section (apart from warping) are ignored.
- 4- Different patterns of initial deflections (U<sub>0</sub>, V<sub>0</sub>, and  $\Phi_0$ ), may be present.

Full details of the problem formulation, leading to a series of stiffness matrices for use in a nonlinear finite element approach, are available in chapter 3. A feature of this formulation is the inclusion of several levels of so-called higher order terms in the geometric strain-displacement relationships resulting in a hierarchy of stiffness matrices. This permits different types of combined bending and torsion problem to be treated with different levels of sophistication. A computer program based on the material of chapter 3 has been prepared (El-Khenfas 1987a). using a Newton Raphson solution scheme.

The numerical solution scheme is based on a step by step approach which involves the combination of incremental and iterative processes until the desired convergence is achieved; details were given in chapter 5. Some judgement is necessary when deciding on:

i- Number of elements along member length

If the beam has no change in the geometry, material properties and external load and restraint conditions, then it can be divided into equal lengths. If discontinuities are present, nodes have

- 153 -

to be introduced at these points.

ii- Number of segments in the cross-section.

The cross-section is divided into a number of plate elements, each element containing several small segments. These segments are then used to evaluate the cross-sectional and sectorial properties such as A,  $I_y$ ,  $I_z$ ,  $I_\omega$ , K, etc.. Both the number of plate elements and the number of segments used will affect the accuracy with an increase generally providing more accurate results, but at the cost of greater computation.

#### 6.3 Numerical Results

Several cases have been chosen to cover the range of problem types outlined in the Introduction; these are listed in Table 6.1

#### 6.3.1 Linear Bending and Torsion

The first class of problem considered is the linear combination of bending and torsion of the type covered in design guides by Terrington (1968,1970). Because direct comparison is made against the results of Terrington, which are themselves based on the solution of the governing differential equations, the example is quoted in the original imperial system of units.

A simply supported beam of 30 ft. span subjected to lateral loads  $Q_z=1$  ton and  $Q_y = 0.1 x Q_z$  applied at midspan on the top of a compound section, consisting of a  $18x7\frac{1}{2}x$  1501b RSJ with a 10x3x19 lb channel welded to the top flange has been selected. Table 6.2 compares longitudinal stresses due to vertical and horizontal bending with those given by Terrington. In determining these values the contributions to

\_154 \_

the member stiffness matrices arising from the nonlinear strains included in the formulation (chapter 3) have been suppressed. It is clear from the results of Table 6.2 that the analysis is capable of dealing adequately with this type of behaviour, the stress values obtained being almost indistinguishable from those of Terrington 1970.

#### 6.3.2 Flexural and Flexural-Torsional Buckling

The analysis may also be applied to bifurcation type problems, involving the determination of elastic critical loads. In this case a slightly different solution technique, in which the lowest eigenvalue of the stiffness matrix including destabilising effects is determined, must be employed.

The example selected for study is a simply supported beam of monosymmetric I-section previously analysed by Anderson and Trahair (1972). This type of problem is inherently more complex than the more usual doubly symmetrical section due to the need to include the Wagner effect in the torsional aspects of the behaviour, a feature that is accentuated if loading is applied remote from the shear centre axis.

Anderson and Trahair(1972) checked their theory against experimental evidence, and they confirmed that the Wagner effect results in either an increase or decrease in the effective torsional stiffness for monosymmetric open cross sections. Because the smaller flange is further from the shear centre, the stresses in this flange have a greater lever arm and predominate in the Wagner effect. Thus when the smaller flange is in tension the Wagner effect provides an increase in torsional stiffness of the beam. The reverse is true when the smaller flange is in compression.

Two examples illustrating the accuracy of the author's

\_155 \_ -

approach have been selected for comparison. For both central load (cases A). and distributed load (Cases B) the transverse loads were applied at distances of up to  $7.6(\frac{D}{2})$  above or below the shear centre axis.

Tables 6.3 and 6.4 compare the results obtained by calculation for both doubly symmetrical and monosymmetric I-sections with those of Anderson and Trahair. Generally good agreement has been achieved with the largest differences being associated with loads applied below the level of the bottom flange. These are thought to be a result of the approximate form for displacements used by Anderson and Trahair; the present study makes no prior assumptions about the buckled shape.

A general equilibrium equation has been developed by Roberts and Azizian (1983a) governing the geometrically nonlinear behaviour of thin-walled open cross-sections by assuming small displacements and linear elastic material behaviour. The solution to this equation, when applied to beam-columns under equal end moments and axial load in the presence of in-plane initial deflection, has been checked against existing test data.

A simply supported I-section column subjected to a compressive load, possessing an initial deflection in the form of a half sine wave of maximum amplitude U<sub>o</sub> at mid-height has been chosen for comparison. Boundary conditions for in-plane and out-of-plane bending and twisting are fully restrained at both ends, while warping is permitted at both ends. (It should be pointed out, however, that since a pure in-plane response was expected the out-of-plane and torsional conditions will have no direct effect on the result.). Numerical results have been obtained, which allow for the effect of in-

-156 -
plane deflection on flexural buckling. These require suppression of some aspects of the full analysis corresponding to the equilibrium equation stated in chapter 3 in order to match Azizian's assumptions. Fig. 6.1 illustrates the load-deflection behaviour obtained from both analyses in which the difference between the failure loads given by the two analyses is approximately 1 per cent. A further comparison is presented in Fig. 6.2, which deals with the lateral- torsional response of an initially curved and twisted beam. In this case a full threedimensional response is obtained as the in-plane moments act through the initial out-of-plane deformations. Failure loads for the three levels of imperfection considered differ from those of Roberts and Azizian by a maximum of 4 per cent, the percentage differences being 4, 2.5, and 0.5 for the three cases respectively.

Results for the elastic critical moment of a simply supported T-section under equal end moments applied about the strong axis have been compared against those reported by Roberts and Azizian (1983b). For the case when the flange is in tension the differences between the two results are not more than 3%, while when the flange is in compression large differences between the two solutions were obtained. This case was therefore checked against the separate solution given by Kitipornchai and Chan (1987) which agreed with the authors to within 1 percent. They previously identified an error in certain aspects of the work of Robert and Azizian.

### 6.3.3 Biaxial bending

Tables 6.5, 6.6, 6.7, and 6.8 compare results for the biaxial bending of doubly symmetrical I-sections with the 'exact' theoretical solution of Culver (1966a,b). In both cases the member is assumed free

- 157 -

from residual stresses. For the results of Tables 6.5 and 6.6 no initial lack of straightness has been assumed; in the case of Tables 6.7 and 6.8 initial bow (both planes) and initial twist have been included as indicated in the first entry of each table. The comparison between the results obtained by the author) and those of Culver show excellent agreement, the percentage differences being not more than 2% for lateral and transverse displacements and a maximum 6.8% for the twists. The larger differences in rotations occured because of the inclusion of torsional moment and shear forces in the linear geometrical stiffness matrix as explained in chapter 4. All of the results of Tables 6.5 - 6.8 were obtained after applying some limitations ( neglect of higher order terms, and using the second order strain in determining the internal forces ), to the full formulation of chapter 3 in order to provide an analysis that was similar in concept to that employed by Culver.

#### 6.3.4 Nonlinear Bending and Torsion

Sections for which the principal axes are oblique e.g. a Z-section, will, even if loaded vertically, deflect both vertically and horizontally. Because of deflection in the horizontal direction, the applied load will also move with the beam so that it no longer acts in the same plane as the vertical reactions at both ends. As a result the section will also twist. The additional stresses caused by this twisting will reduce the load-carrying capacity of the member.

The problem of nonlinear bending and torsion has been extensively studied by Moore (1986), who concentrated on the effect of movement of the point of application of the load for Z-sections. He

- 158 -

extended the original theory of Vlasov (1961) and Timoshenko (1945) by solving the augmented differential equations using a finite difference method. The validity and accuracy of his theory were assessed by comparing its results with those obtained from experiments.

The example selected for study comprises a beam subjected to a pair of lateral point loads applied to the top flange at its 1/3 and points, either passing through the shear centre or with 2/3 eccentricities  $e_v = -25 \text{ mm}$ , or  $e_v = +25 \text{ mm}$ . Fig. 6.3 and Table 6.9 provide details of the load-deflection (in-plane and out-of-plane) full relations. The vertical displacement was evaluated at point 1 and point 2 and the vertical displacements were calculated at point 2. These values were obtained for a simply supported Z-section beam under two point loads placed at the 1/3 and 2/3 positions on the top flange through the shear centre. The values of the lateral passing displacements at the top and bottom flanges which are indicated by points 1 and 2 are closer to the experimental value than those of Moore, whilst for the in-plane behaviour deflections are closer to those conducted by Moore with little difference from the experimental data. Figure 6.4 represents the load-rotation curves of the results of Moore's tests, Vlasov ,and Moore's theory and those obtained by calculations which are based on the approach presented in chapter 3.

The author's values are generally rather closer to the experimental data reported by Moore than are the predictions of Moore's own theory.

#### 6.3.5 Biaxial Bending and Torsion

It is possible to implement the theoretical formulation of chapter 3 at a variety of levels by suppressing various of the higher

- 159 -

order contributions to the stiffness expressions. In particular, three classes of approach may be employed:

- i- Linear using only the basic stiffness matrix ( tangent and geometrical which includes torsional moments and/ or shear forces) and second order strains.
- 11- Nonlinear using only the same stiffness matrix as i but including the higher order strain terms such as  $\frac{1}{2}(v_{,x}^{2}+v_{,x}^{2}+v_{,x}^{2}+v_{,x}^{2})$ ,  $v_{,xx}^{\Phi}$  +  $v_{,xx}\frac{\Phi^{2}}{2}$ , and  $v_{,xx}^{\Phi}$  +  $v_{,xx}\frac{\Phi^{2}}{2}$  to  $v_{,xx}$ to the axial strain, bending strain about the Y and Z axis. Further details are given in chapters 4 and 5.
- iii- Full nonlinear using all higher order contributions to the stiffness matrices and strain-displacement relations as described in chapter 4.

Some quantitative indication of the effect of adopting these different degrees of sophistication may be obtained by considering specific examples. A simply supported beam-column of W12x14 section, free to warp at both ends with an initial lack of straightness  $U_0 = \frac{L}{1000}$  at the mid-span subjected to an axial load applied with eccentricities at both ends  $\frac{e_{y1}}{B} = \frac{e_{y2}}{B} = 0.025$  and  $\frac{e_{z1}}{D} = \frac{e_{z2}}{D} = 0.25$  has been considered. Several analyses have been performed in which this loading is assumed to act in conjunction with a constant applied mid-span torsional moment  $M_t$ . The load path assumed is full torsional loading (up to a selected value) followed by end loading to failure. Results have been obtained for a series of values of  $M_t$  using each of the 3 solution schemes outlined above.

For the case of  $\frac{M_t}{M_y} = 0.023$ ; the difference between the 3 approaches is clearly illustrated by the set of load deflection curves

-160 -

given in Fig. 6.5 ' Linear' analysis overestimates the maximum load by 24.2 per cent, whilst partial nonlinear analysis overestimates the failure load by 11.4 per cent as compared with the full nonlinear analysis. In the early stages of loading the differences between the three analyses are negligable, because the degree of coupling between the displacements is smaller. In other word the contributions of the nonlinear stiffness matrices which depend on the products of derivatives of displacements and rotations become insignificant. As the loads increase followed by change in displacements, then the differences between the analyses are noticeable and become more and more significant. Finally the findings for the full range of  $M_{\mu}/M_{\mu}$ values considered in the form of an interaction diagram of combinations of P and  $M_t$  at failure are described in detail in chapter 8.

## 6.4 Conclusions

An elastic analysis for thin-walled open cross-sections of general shape subjected to any combination of compression, biaxial bending and torsion has been presented. The approach utilises a nonlinear finite element formulation, which incorporates any form of initial deformations. A selection of illustrative studies, designed to demonstrate something of the range of problem types that may be treated has been reported. In each case excellent agreement with previously obtained theoretical results and / or experimental data has been obtained. Because of its general nature, the method can deal with

1- Arbitrary cross-sections, including those with no axes of symmetry.

-161 -

2 - Any arbitrary set of applied loads.

3- Any arrangement of flexural and torsional support conditions.

It therefore represents a powerful tool for studying the response of beam-columns in a very general manner.

#### Table 6.1 Examples Selected for Study

Reference	Section	Load Arrangement	2	Warping	In Def	itial lectio	n
	Туре		Λ	at the ends	U/B	V/D	¢ rad
BSCA (1970)		$\begin{array}{c} \underline{a_{y}} \\ \hline \\$	88	R	0	0	0.
Anderson & Trahair	+ 1.241 2.972 + 0.862 + + .4+	CASE B $\downarrow Q$ $\downarrow 1/2 + L/2 +$	368	R	o	0	o
(1972)	2.972		241				
Roberts &	+ 100 +			-	0	0.01	0.01
Azizian	200 - 6		490	R .	0	0.02	0.02
(1983)	(mm)	L=4000 mm			0	0.04	0.04
Poberts & Azizian (1983) Kitiporchai & Chan (1987)	+ 40 + 40 + (mm)	(( ∰ y H,y )) ↓ L = 4000 mm ↓	78	R	. 0	0.	0.
Culver		+ + ey	140260	U,R	ο.	0.	ο.
(1966 a,b )	14WF43	at both ends	60 & 140	U,R	0.015	0.089	.0173
Moore(1986)	+ 50 + 2 140 + (mm)		134	R	0.	ο.	0.
	W 12x.14	$\begin{array}{c} e_{y} \\ \hline \\ e_{z} \\ \hline \\ \hline \\ \hline \\ e_{z} \\ \hline \\ $	157	R	1000	0.	0.

R= Warping restrained U= Warping unrestrained B,D = Width of flange and Depth of the beam  $\lambda$  = Slenderness ratio

Load and section Points description		Co-ordinates (in.)		σ <sub>y</sub> (T/in <sup>2</sup> )		σ <sub>z</sub> (T/in <sup>2</sup> )		$\sigma_{T} = \sigma_{y} + \sigma_{z} (T/in^{2})$	
		Y	Z٠	BSCA	Authors	BCSA	Authors	BCSA	Authors
Q = 1 Ton	0'	-5.0	3.96	0.3230	0.3231	-0.3810	-0.3812	-0.058	-0.0581
10x3[19 2 016	0	-5.0	6.96	0.5680	0.5679	-0.3810	-0.3812	0.1870	0.1867
	1	0.0	6.96	0.5680	0.5679	0.0	0.0	0.5680	0.5679
c	2	5.0	3.96	0.3230	0.3231	0.3810	0.3812	0.7040	0.7042
18.74 ISC	2'	5.0	6.96	0.5680	0.5679	0.3810	0.3812	0.9490	0.9491
	3	0.0	3.38	0.2760	0.2758	0.0	0.0	0.2760	0.2758
	4	3.75	-11.36	-0.9270	-0.9269	0.2860	0.2859	-0.6410	-0.6411
	5	0.0	-11.36	-0.9270	-0.0269	0.0	0.0	-0.927	-0.9269
	6	-3.75	-11.36	-0.9270	-0.9269	-0.2860	-0.2859	-1.2130	-1.218

Table 6.2 Comparison between Authors results and those by BSCA (1970)

	K=0.3								
Type of loading	δ	ε	a <sub>2</sub>	γ	Failu	re load	% error		
					Anderson and Trahair (1971) x10 <sup>-3</sup>	Calculated			
		0.6	8.923	13.97	0.7853	0.7581	1.0		
в		0.3	4.49125	19.84	1.12	1.10 <sup>1</sup>	0.1		
CASE		0.0	0	29.63	1.67	1.6901	0.1		
		-0.3	-4.491	44.09	2.48	2.401	1.6		
		-0.6	-8.983	61.09	3.49	3.61	5.4		
		0.6	8.983	7.0	39.12	37.0 <sup>2</sup>	2.5		
		0.3	4.491	10.67	59.63	58.0 <sup>2</sup>	2.3		
SE A		0.0	0	17.78	99.36	99.0 <sup>2</sup>	0.3		
CAS		-0.3	-4.491	28.62	159.94	154.02	3.1		
		-0.6	-8.983	39.67	221.69	220.02	4.5		

Table 6.3 Comparison of calculated results with Anderson and Trahair (1971) for stability of double symmetric beams

 $\varepsilon = 0.068 a_z$  $\gamma_c = 184.16 Q_z$ 

 $\gamma_{0} = 18305.5 q_{z}$ 

		K=0.3								
					Failu	Failure load				
Type of loading	δ	3	a <sub>z</sub>	Y	Anderson and Trahair	Calculated	% error			
					(1971) x10 <sup>-3</sup>	×10 <sup>-3</sup>				
		0.6	8,9825	15.855	3.61	3.57 <sup>1</sup>	1.8			
		0.3	4.4913	23.06	5.25	5.10 <sup>1</sup>	1.5			
ase		0.0	0.0	34.763	7.92	7.80 <sup>1</sup>	1.2			
Ö		-0.3	-4.4913	51.19	11.67	11.88 <sup>1</sup>	2.1			
	4	-0.6	- <b>8.</b> 9825	70.84	16.45	16.50 <sup>1</sup>	3.1			
	0.20	0.6	8.9825	7.78	93.63	100.02	2.8			
		0.3	4.4913	12.16	146.44	143.5 <sup>2</sup>	2.1			
se A		0.0	0.0	20.24	243.72	243.0 <sup>2</sup>	1.3			
Cas		-0.3	-4.4913	31.53	379.796	363.0 <sup>2</sup>	6.0			
1		-0.6	-8.9825	43.12	519.28	475.02	7.2			

Table 6.4Comparison of Authors results with those of Anderson and<br/>Trahair (1971) for stability of monosymmetric beams

$$\varepsilon = 0.07 a_z$$
  
 $\gamma_A = 76.46 Q$   
 $\gamma_B = 4015.4 q_z$   
<sup>1</sup> concentrated load Q

<sup>2</sup> distributed load q kip in

kips

$\frac{L}{r_y} = 60$	$\frac{L}{r_y} = 60  \frac{e_y}{B} = 0.625,  \frac{e_z}{D} = 0.366  14WF43  Warping unrestrained$									
P P y	U/B x 10 <sup>-2</sup>		V/1	D x 10 <sup>-2</sup>	$\phi$ - Rotation (rad) x 10 <sup>-3</sup>					
	Culver	Calculated	Culver	Calculated	Culver	Calculated				
0.0	0.0	0.0	0.0	0.0	0.0	0.0				
0.048	1.515	1.590	0.092	0.092	0.05	0.058				
0.096	3.093	3.095	0.184	0.184	0.20	0.22				
0.144	4.738	4.993	0.276	0.276	0.40	0.48				
0.194	6.454	6.466	0.368	0.368	0.80	0.87				
0.246	8.273	8.248	0.461	0.460	1.30	1.39				

Table 6.5 Comparison of calculated results with exact solution of Culver (1966a), for biaxial loading, no initial deflection

Table 6.6 Comparison of calculated results with exact solution of Culver (1966a) for beam-column under biaxial bending, no initial deflection

$\frac{L}{r_y} = 140$	$\frac{L}{r_y} = 140 \qquad \frac{e_y}{B} = 0.0625, \frac{e_z}{D} = 0.366 \qquad 14WF43 \text{ Warping unrestrained}$									
P P y	U/E	x 10 <sup>-2</sup>	V/E	x10 <sup>-3</sup>	$\phi$ -Rotation (rad) x10 <sup>-2</sup>					
	Culver	Calculated	Culver	Calculated	Culver	Calculated				
0.0	0.0	0.0	0.0	0.0	0.0	0.0				
0.048	9.06	9.06	5.03	5.03	0.12	0.12				
0.096	20.68	20.66	10.15	10.15	0.57	0.57				
0.144	36.11	36.04	15.27	15.30	1.50	1.55				
0.194	57.80	57.33	20.33	20.40	3.43	3.25				
0.246	89.73	90.63	25.04	25.25	6.96	6.48				

Table 6.7 Comparison of calculated results with exact solution of Culver (1966b) for beam-column subjected to biaxial bending, initial deflection included

$\frac{L}{r_z} = 60 \qquad \frac{e_y}{B} = 0.625,  \frac{e_z}{D} = 0.366 \qquad 14WF43  Warping unrestrained$									
P P y	U/B × 10 <sup>-2</sup>		v/r	x 10 <sup>-2</sup>	$\phi$ - Rotation (rad) ×10 <sup>-3</sup>				
	Culver	Calculated	Culver	Calculated	Culver	Calculated			
0.0	1.475	1.475	0.864	0.864	0.173	0.173			
0.048	1.566	1.565	0.922	0.921	0.43	0.33			
0.096	3.199	3.196	1.845	1.844	1.31	1.10			
0.144	4.903	4.899	2.767	2.769	2,67	2.70			
0.192	6.688	6.680	3.689	3.697	4.59	3.88			

Table 6.8 Comparison of calculated results with exact solution of Culver (1966b) for beam-column subjected to biaxial bending, initial deflection included

$\frac{L}{r_z} = 140 \qquad \frac{e_y}{B} = 0.625, \qquad \frac{e_z}{D} = 0.366 \qquad 14WF43 \qquad \text{Warping Unrestrained}$									
P P y	U/E	3 x 10 <sup>-2</sup>	V/I	$0 \times 10^{-2}$	$\phi$ - Rotation (rad) x10 <sup>-3</sup>				
	Culver	Calculated	Culver	Calculated	Culver	Calculated			
0.0	3.45	3.45	2.02	2.02	0.403	0.403			
0.048	9.80	9.84	5.07	5.08	0.51	0.40			
0.096	22.51	22.40	10.14	10.22	2.05	1.54			
0.144	39.99	39.52	14.97	15.15	5.36	4.05			
0.192	66.15	64.45	19.07	19.65	12.01	13.80			

P P	U <sub>2</sub> -Displacement (mm)			U - Displacement (mm)			V - Displacement (mm)		
ey	Moore	Expt.	Authors	Moore	Expt.	Authors	Moore	Expt.	Authors
0	0	0	0	0	0	0	0	0	0
0.045	1.056	1.060	1.136	1.074	1.06	1.127	0.704	0.693	0.700
0.089	1.972	2.071	2.205	2.183	1.914	2.309	1.309	1.32	1.403
0.134	2.958	2.817	3.261	3.380	2.535	3.523	1.97	2.754	2.108
0.178	3.803	3.83	4.293	4.578	4.085	4.782	2.747	3.804	1.817
0.223	4.648	4.859	5.30	5.915	5.634	6.091	3.521	4.859	3.535
0.267	5.211	5.915	6.288	7.324	6.901	7.454	4.155	5.915	4.261
0.357	5,958	6.972	7.258	8.873	8.521	8.876	4.93	7.042	4.999
0.356	6.761	8.169	8.214	10.592	10.282	10.363	5.775	8.169	5.75
0.401	7.324	9.437	9.157	12.535	12.092	11.921	6.549	9.437	6.518
0.446	7.845	10.985	10.092	14.788	13.944	13.558	7.324	10.989	7.307

Table 6.9 Comparison of Author's results with Moore (1986) analysis and tests for beams of Z-section subjected to bearing and torsion



FIG.6.1 <u>COMPARISON BETWEEN AUTHOR'S RESULTS AND</u> <u>ROBERTS & AZIZIAN (1983) FOR I SECTIONS</u> <u>SUBJECTED TO AXIAL LOAD AND VERTICAL DEFLECTION</u>



FIG. 6.2 COMPARISON BETWEEN THE RESULTS OBTAINED BY THE AUTHOR AND THOSE OF ROBERTS AND AZIZIAN (1983).

-171-





FIG. 6.4 COMPARISON OF AUTHOR'S RESULTS WITH MOORE (1986) AND VLASOV (1961)

-173-



FIG. 6-5 LOAD - DEFLECTION CURVE FOR THREE TYPES OF ANALYSIS OF A SIMPLY SUPPORTED I - BEAM SUBJECTED TO BIAXIAL BENDING AND CONSTANT TORSION AT MID - SPAN

174-

#### Chapter 7

# Inelastic Analysis of Beam-Column in Space

### 7.1 Introduction

In order to study the response of a beam-column in the inelastic range it is necessary to allow correctly for the losses in stiffness that occur due to spread of yield through the cross-section, with due account being taken of variations along the length of the member. The presence of initial geometrical imperfections and residual stresses will affect both the initiation and propagation of yielding. Accurate monitoring of the full three-dimensional response of such members therefore requires that very careful consideration be given to the way in which gradual plastification is treated.

The inelastic behaviour of beam-columns of thin-walled open cross-section in three dimensions under various types of applied loads has been studied experimentally and analytically by several investigators such as Nethercot(1973a,b),Trahair and Kitipornchai (1972), Fujita and Yoshida (1972), Matthey (1984), Anslijn (1983), and Lindner (1981). The influence of initial geometrical imperfections has been incorporated by some of them and ignored by others. Reviews of the subject are available in many references such as Chen and Santathadaporn (1968), Chen (1977), Johnston (1976), Massonet (1976), and Chen and Atsuta (1977)

The application of the general finite element formulation presented in chapter 3 to assess the inelastic behaviour of thin-walled

\_ 175 \_

steel members deforming in space is described herein. This approach is applicable to various types of open cross-section, including those with no degree of symmetry.

In this chapter the accuracy and versatility of the theoretical approach which has been presented in chapter 3 is demonstrated by application to a wide range of selected problems covering flexural, flexural-torsional buckling, and bending and torsion.

## 7.2 Assumptions

In addition to the basic assumptions of Vlasov(1961) and those presented in chapter 3, the following- are taken into consideration:

- I- The distribution of elastic and yielded regions within the cross section is determined from the strain distribution and the material stress-strain curve.
- II- The moduli of the material are E and G in the elastic regions while in the yielded regions  $E_t$  and  $G_t$  are used, with  $E_t$  and  $G_t$  being dependant on the type of stress-strain curve employed. Any monotonically increasing curve may be assumed.
- III- Any reasonable pattern of residual stresses and virtually any distribution of initial crookedness may be included.
- IV- Control of the nonlinear numerical solution is as described in section 4.6 of chapter 4 with accuracy being based on a combined displacement change/ out of balance load check.
- V- Yield is controlled by normal stresses only ( although the inclusion of more sophisticated yield criteria is possible. with little extra work).

-176 -

VI- The sectional and sectorial properties were evaluated after each load increment in the inelastic range.

Various possible load types are given in Table 7.1., which lists the example problems considered herein. In the case of transverse loads, these may be either distributed or concentrated and may be applied at any eccentricity to the shear centre axis. Allowance for the effect of spread of yield in any region on cross-sectional properties makes no assumptions concerning the location of the shear centre, existence of symmetry etc.. Equilibrium of internal and external forces is maintained at all stages of the solution, see section 5.5 of chapter 5.

The sectorial properties such as warping moment of inertia  $(I_{\omega})$ , warping second moments of inertia  $(I_{y\omega})$  and  $(I_{z\omega})$ , and normalized function  $(\overline{w}_n)$  are not taken as the elastic values in the inelastic range as was the case in most previous work (Rajasekaran and Murray (1973), Yang and McGuire (1986), and Kitipornchai and Trahair (1975)) but are evaluated accounting for the loss of stiffness due to spread of yield in a similar fashion to the other sectional properties such as  $I_y$ ,  $\overline{Y}$ ,  $I_{yz}$ , etc.

## 7.3 Numerical Results

Several illustrative examples have been selected for study. Different aspects of behaviour such as flexural buckling, flexuraltorsional buckling, and biaxial bending are covered by these cases. The influence of initial deflection and residual stresses are incorporated. The particular features of each type of problem are listed in Table 7.1.

\_177 \_

In each case the number of elements and the number of plate segments within the cross-section were chosen so as to achieve accurate solutions without increasing the computational effect unnecessarily as described in Chapter 5. The convergence criterion of section 5.5 was employed.

### 7.3.1 Column with Initial Deflections

A simply supported I-section under axial load assuming various values of initial deformations, and including the effect of material strain-hardening ( $E_{sh} = E/10$ ) has been considered. The initial crookedness is represented by a sinusoidal function with its maximum value at midspan. Residual stress was not included so as to permit comparison with the previous solution of Fujita and Yoshida (1972).

Some limitations are imposed on the suggested formulation of chapter 3 in order to achieve comparable results; these are the influence of the higher order terms of the strain-displacement relationship and the nonlinear tangential and geometrical stiffness matrices which incorporate the derivative of the displacements such as  $PU_{,x}^{\phi}$ ,  $M_y V_{,\infty}^{\phi}$ ,  $EI_y U_{,xx} \frac{\phi^2}{2}$ ,  $Y_s EI_z U_{,xx} \phi_{,x}$  etc.. Four examples are chosen from those studied both theoretically and experimentally by Fujita and Yoshida (1972) and the results obtained are given in Table 7.2. The percentage differences between the present solution and the earlier results is in no case greater than 4 per cent.

#### 7.3.2 Flexural & Flexural Torsional Buckling

Fig. 7.1 shows a monosymmetric I- section together with the residual stresses produced by welding; these have adjusted so as to meet the equilibrium condition of equations 3.43, 3.44, and 3.45. This

-178 -

pattern has been implemented by Fukuomto et al. (1971) in the inelastic analysis of a simply supported beam under equal end moments applied about the strong axis containing initial out-of-plane displacements  $(U_0)$ . Warping was assumed to be unrestrained at both ends, while the in-plane  $(V_0)$  and out-of-plane deflection together with the twist  $(\Phi)$ were fixed at the two ends. The out-of-plane displacement is represented by a sine wave with its maximum amplitude at mid-span is equal to 0.5 inch. The residual stress at the flange tips  $(\sigma_f)$  was equal to the yield stress  $(\sigma_y)$ . From assumption V of section 7.2 this means that the spread of yield begins as soon as any load is applied. The correlation between the two results is good, the differences between them being of the order of 7 per cent, despite the different type of solution procedure employed.

Lindner (1981) presented a numerical simulation of some Berlin tests on beam-columns. Table 7.3 contains 5 of the test specimens selected on the basis of an attempt to cover a reasonably wide range of different variables. Different values of the initial deflections as represented by a sine wave are listed in Table 7.3. The values of end eccentricity used to generate moments in accordance with Lindner's investigation are also listed. In each case calculated solutions were obtained by applying the axial load with small increments until collapse occured. Comparison of the ultimate loads of the Berlin tests with those obtained by the Author are listed in Table 7.4. Generally very good agreement has been achieved, in which the percentage errors range between -4 to 6 % and -3 to 8% with respect to test data and analytical solution respectively.

- 179 -

# 7.3.3 Biaxial Bending

comparisons have been made with the tests on Several I-sections subjected to biaxially eccentric compression conducted by Anslijn (1983). Table 7.5 presents the input data for four tests chosen the purposes of comparison, whilst Table 7.6 compares the for calculated ultimate loads with the experimental values. Differences of less than 5% were obtained in each case. The test displacements were measured at the tips of the two flanges for vertical deflections and the top and the bottom of the web for horizontal deflections, the average being taken for both the in-plane V and out-of-plane U displacements. A typical comparison between these average values and those calculated by the program is given in Figs 7.2 and 7.3 for test No. 75 and No. 48. This shows excellent agreement for both in-plane and out-of-plane deflections. The differences found in the latter stages is thought to be due to cross-sectional distortion, leading to changes in flange tip movement that occur near the collapse.

Many cases from the comprehensive parametric study of biaxially loaded I-section beam-columns presented by Matthey (1984) have been selected for examination. For each analysis axial load was applied up to a predetermined limit followed by end moment loading to failure. Thus the actual discrepancies between the two solution given in Table 7.7 are of course, smaller than the numerical values would suggest since they relate to only a part of the applied loading. Nonetheless agreement is considered excellent with the 3 examples selected representing very different sets of parameters.

-180 -

#### 7.4 Conclusions

The inelastic three-dimensional behaviour of beam-columns under different loading conditions has been studied. The presence of initial imperfections such as residual stresses and initial deflections together with the effects of strain hardening were incorporated in the analysis. Results from several previous theoretical and experimental studies were used to check the Author's formulation and computer program which is capable of dealing with such problems. The validity and the accuracy of both the approach and the computer program shows an excellent agreement with those problems previously investigated, thereby confirming that the analysis represents a powerful tool for the study of this class of problem. Table 7.1 Problems Chosen for Study.

Reference	λ	loading Arrangement	Cross- Section	₹y	⊽r	Initial Deflection		
Fujita & Yashida (1972)	60	P		28.6 Kg/mm		$\begin{array}{c ccccccccccccccccccccccccccccccccccc$		
Fukumoto & et al (1971)	103	↓ 400 ↓		3666 kn/mm²		0.00125		
Lindner (1981)	19,28 42,52	The eccentric load applied with various				Several values of initial		
Anslijn (1993)	60,90	eccentricit- ies ,Table 3 and Table 5	HEA200		7	deflections are employed see Table7.5 and Table17		
Mathey (1984)	141, 47. 94	( <del>{})</del> )	HEA200	235.0 N/mm²		L 1000 100 .001		

ł

Table 7.2 Comparison between Author's result and those obtained by Fujita and Yashida (1972) for compressive column with initial deflection

		ŀ/P y	% error between			
$\delta = \frac{V_{O}}{L}$	Fujita and	d Yoshida	Authon	1 3	0 3 0	
	Experimental 1	Theoretical 2	Author 3	I and J	2 and 3	
0.0067	0.603	0.607	0.616	1.6	2.1	
0.0112	0.507	0.525	0.517	1.9	1.6	
0.0157	0.442	0.432	0.432	2.3	3.7	
0.0215	0.367	0.362	0.366	0	1.0	

.

Test No.	Yield stress (kN/cm <sup>2</sup> )	Initia	al crocke	Eccenti (d	ricities cm)	
		U (cm)	V o (cm)	φ o "rad"	Тор	Bottom
					e <sub>z1</sub>	e <sub>z2</sub>
1-3-13	65.0	0.030	0.041	0	-14.0	-9.5
2-1-11	65.0	-0.004	0.0035	0.003	2.5	2.5
3-1-33	69.0	-0.046	0.003	-0.006	-2.0	+ 2.0
3-2-21	68.0	0.042	0.010	-0.003	+6.5	0.0
5-3-22	68.0	-0.089	0.049	0.003	1.0	15.0

Table 7.3 Data for problems chosen for purpose of comparison with Lindner (1981)

Table 7.4 Comparison of selected tests and analysis conducted by Lindner (1981) against those obtained by Author

Test No	$\frac{L}{Y_y}$	Colla	pse load	% error cal.		
		Author	Test	Lindner	Test	Lindner
1-3-13	19	1124.0	1209.0	1178.0	7	4.5
2-1-11	28	2524.0	2492.0	2461.0	1	2.5
3-1-33	42	2322.0	2469.0	2254.0	6	-3
3-2-21	42	1715.0	1644.0	1860.0	-4	8
5-3-22	52	1120.8	1128.0	1170.0	1	4

Test No. (8)	Yield Stress [N/mm <sup>2</sup> ]	Initial deflections (m)		Residual Stresses [N/mm <sup>2</sup> ]			Eccentricities [mm]			Moment at er			
		U <sub>O</sub> V <sub>O</sub> φ 'cm' 'cm'	v	$\phi \times 10^{-3}$	Tip Flange	Flange	Web	Тор	Boti		Om	(Appiled)	
			'rad'		web		ey	ez	ey	ez	Му	M	
22	278.0	0.7	0.5	-1.11	100.0	-55.5	62.1	21.63	-1.24	21.63	1.24	7	
24	249.0	2.55	1.5	.50	101.0	-55.5	62.1	3.61	7.44	3.605	-7.44		X
48	249.0	0.8	0.9	1.11	100.0	-55.5	62.1	7.21	0	0	-2.48	$\overline{A}$	7
75	223.0	1.5	1.0	1.11	100.0	-55.5	62.1	21.63	7.44	-21.4	7.44	8	

# Table 7.5 Data for Problems Chosen for Comparison, Anslijn(1981)

Table 7.6 Comparison of Author's Result with Experimental Data of Anslijn (1981)

.

Test	Ľ	Collaps	e Load	% difference		
No.	r y	PT/Ps	Pc/Ps			
22	60	0.229	0.243	5.48		
24	60	0.409	0.428	4.44		
118	60	0.516	0.526	1.84		
75	96	0.220	0.224	1.82		

¶y = 2	$\nabla_{y} = 235.0 \text{ N/mm}^2 \text{ U}_0 = \text{V}_0 = \frac{l}{1000} \& \phi_0 = .001 \text{ red}$ , Residual stress Parabolic distribution									
Case No.	<u>و</u> ۲ <sub>۲</sub>	Bending moments		$m_{y} = \frac{M_{y}}{M_{py}}$		$m_{z} = \frac{M_Z}{M_{p_Z}}$		s age		
		м <sub>у</sub>	MZ	Matthey	Authors	Matthey	Calculated	differences		
11-324	141	X=0 X=L		0.343	0.351	0.479	0.490	2.2		
14-324	47	.5му		0.441	0.466	0.618	0.653	5.4		
36-242	94	Δ	$\Box$	0.103	0.110	0.481	0.514	6.4		

Table 7.7 Comparison of Author's Result with Solution of Matthey (1984)



FIG. 71 ADJUSTED RESIDUAL STRESS DISTRIBUTION PRESENTED BY FUKUMOTO ET AL (1972)



FIG.7.2 COMPARISON BETWEEN AUTHOR'S RESULTS AND EXPERIMENTS (ANSLIJN, 1981) ON BEAM-COLUMN UNDER BIAXIAL BENDING

-189-



#### CHAPTER 8

#### Ultimate Strength of Beams under Bending and Torsion

#### 8.1 Introduction

Beams and beam-columns of various types of cross section under flexural and torsional loads are important in the design of steel members in both the elastic and inelastic ranges. Many researchers have made an effort to investigate this behaviour although most studies have concentrated on elastic analysis based on simplified calculations.

Sourochnikoff (1951) presented a theoretical analysis of I-beams under combined bending and torsion in order to obtain the allowable stresses which can be used in the design of structural steel members. Terrington (1968,1970) provided a two-volume treatise on the behaviour of beams subjected to transverse loading causing torsion. The first of these presents general theory and formulae together with graphs used to display solutions, while the latter presents worked examples to calculate stresses in beams under flexural and torsional loads.

Laterally unsupported beams of I or WF sections subjected to bending and torsion have been studied by Chu and Johnston (1974). Based on the interaction between these two effects they presented a series of curves for use by design engineers. Pastor and DeWolf (1979) investigated wide flange I-beams subjected to uniform flexure and concentrated torsion. They provided tabulated results for the stresses which can be employed to check the ultimate strength behaviour based on an elastic limit condition but allowing for the interactive effects of deformation. Kollbrunner, et al. (1979) have examined theoretically and

\_191 \_

experimentally the elastic-plastic behaviour of thin-walled fixed ended I-beams under bending and torsion.

The design of beams having W-shapes under combined bending and torsion has been investgated by Johnston <sup>-</sup>(1982). He presented charts to permit rapid design checks for a variety of load types and end conditions. These graphs were based on solutions tabulated by Heins and Seaburg (1963). Moore (1986) studied the stability of Z-sections subjected to bending and torsion. All the above studies were in the elastic range. Not much work has been done towards bending and torsion in the inelastic range.

The aim of this study is to provide an analysis in the elastic and inelastic ranges for beams of thin walled open cross section subjected to biaxial bending together with concentrated torsion applied at mid span, and also to use the results in a preliminary study of possible simplified interactive approaches to design.

# 8.2 Analysis

The formulation presented in chapter 3 and used as the basis for the inelastic ultimate strength analysis procedure described in chapters 4 and 5 has been applied to a series of problems involving various combinations of compression, bending about the principal axes and direct torsional loading. Problems of this type, involving the interaction of flexural-torsional buckling effects and applied torsion in the inelastic range, have not, to the author's's knowledge, previously been studied. In making these studies some attention has been given to the different possibilities - in terms of degree of nonlinearity included - presented by the formulation of chapter 3. 1- Linear analysis

- 192 -
In this analysis the usual linear stiffness matrices (tangential  $[K_t^L]$  and geometrical  $[K_G^L]$ ) and the second order straindisplacement relations ( $\Sigma = W_{,x} - U_{,xx}Z - V_{,xx}Y + \Phi_{,xx}\overline{w}_n$ ) are used.

#### 2- Nonlinear Analysis

The higher order terms of axial strain  $(\frac{1}{2}(U_{,x}^2+V_{,x}^2+\rho_{,x}^2\phi_{,x}^2))$ , and the bending strains about the Y and Z axes  $(U_{,xx}\Phi-V_{,xx}\frac{\Phi^2}{2})$  and  $V_{,xx}\Phi+U_{,xx}\frac{\Phi^2}{2})$  are added to the second order strain as:

$$\Sigma = W_{,x} + \frac{1}{2}(U_{,x}^{2} + V_{,x}^{2} + \rho \Phi^{2})^{x}$$

$$-(U, xx^{+}V, xx^{\Phi}-U, xx^{\frac{\Phi}{2}})Z - (V, xx^{+}U, xx^{\Phi}+V, xx^{\frac{\Phi}{2}})Y - \Phi, xx^{\frac{1}{\omega}}n$$

and the same stiffness matrices stated in 1 are employed.

#### 3- Full nonlinear analysis

Both the nonlinear strains described in 2 and the nonlinear geometrical  $[K_G^{NL}]$  and nonlinear tangential  $[K_t^{NL}]$  stiffness matrices are employed. Further details are given in Chapter 4.

The differences between these three types of analyses are demonstrated by several examples for both the elastic and the inelastic ranges.

#### 8.3 Wumerical Results

Numerical results for a simply supported beam of W12x14 section subjected to biaxial bending and torsion in the form of equal or unequal end moments about the strong axis together with a concentrated torque at midspan are presented. The boundary conditions are  $U = V = \Phi^{n} = 0$  at both ends with warping permitted. The effects of

- 193 -

residual stress and initial deflections are not considered in the elastic analysis, although these imperfections are considered in the inelastic cases.

Several examples are presented in this study, including some comparisons with results obtained previously by other workers.

### 8.3.1 Comparison with Previous Analysis

Pastor and DeWolf (1979) presented an elastic analysis for I-beams under equal end moments and constant torsion. Based on a small deflection theory, so that the terms other than those of first order were neglected, they employed the following differential equations for a beam under bending and torsion:

 $EI_V V'' = M_V - M_V U'$ 

 $EI_{z}U'' = M_{v}\Phi - M_{x}V'$ 

$$\mathsf{GK}\Phi^{\dagger} - \mathsf{EI}_{\omega}\Phi^{\dagger} = \mathsf{M}_{\chi} - \mathsf{M}_{\chi}\mathsf{U}_{\cdot}^{\dagger}$$

in which E is the modulus of elasticity,  $I_y$  and  $I_z$  are the moments of inertia about the major and minor axes. G is the elastic shear modulus, K is the torsion constant, (GK is the torsional rigidity) and  $I_{\omega} = I_z \frac{H^2}{\mu}$ .

They provided general solutions (complementary and particular) to the above equations in terms of  $\Phi$ . Their study was directed at evaluating the bending ( $\sigma_y$  and  $\sigma_z$ ) and warping ( $\sigma_\omega$ ) stresses and the torsional rotations at mid-span. They considered several slenderness ratios with the torsional moment M<sub>t</sub> equal to  $\frac{M_cr}{100}$  fixed whilst the end moments were varied in steps of M<sub>p</sub>/17. Tabulated

- 194 -

results for stresses and rotations were provided. By imposing some limitations on the present approach such as neglecting the higher order terms in the strain-displacement relations  $(U_{,xx}^{\Phi}, V_{,xx}^{\Phi}, \text{etc.})$  and the coupling between the displacements which are presented in the nonlinear tangential and geometrical stiffness matrices (chapter 4) it is possible to match the basis of this approximate solution. An example is chosen for the purpose of comparison so as to check the validity of the present formulation. In this comparison the load increment was taken similar to those used by Pastor and DeWolf. At each load step the convergence to the correct result is based on the out-of-balance force and displacement increments respectively. Further details of this method were given in Chapter 5.

Table 8.1 summarizes the comparison between the Author's results and those of Pastor and DeWolf for the bending stresses  $\sigma_y$  and  $\sigma_x$  about the Y and Z axes respectively, and the warping stresses  $\sigma_w$  at midspan. It is clear that in the early stages of loading the differences between the stresses are small, being between 2 and 6 per cent. As soon as the applied load approaches the failure load, however, differences between the two sets of results increase, reaching 20% in some instances. Table 8.1 presents a comparison between the total stresses and the rotations in which the percentage differences are not more than 13.7 and 10 per cent for total stress ( $\sigma_t$ ) and rotations at mid-span respectively. Fig. 8.1 presents the comparison of the three stress components in graphical form, whilst Fig. 8.2 shows the variation of the total peak stress and twist.

It is believed that the differences between the two sets of results at high load levels are attributable to the choice of unsuitable approximations for the degree of twisting by Pastor and

- 195 -

۰.

DeWolf. The author's solution makes no approximations and places no constraints on the deflected form of the member.

#### 8.3.2 Elastic Analysis

This section presents some quantitative indication of the variations obtained by adopting the three types of analysis stated in section 8.2. Several examples have been considered in which the biaxial flexural loading is assumed to act in conjunction with a constant applied mid-span torsional moment ( $M_t$ ). The load path assumed is full torsional load (up to the selected value) followed by flexural loading to failure. Results have been obtained for a simply supported beam-column of W12\*14 section, free to warp at both ends with or without an initial lack of straightness  $U_o = \frac{L}{1000}$  at mid-span.

#### 8.3.2.1 Biaxial Bending and Torsion

A simply supported I-beam subjected to an axial load applied with eccentricities at both ends  $\frac{e_{y1}}{B} = \frac{e_{y2}}{B} = 0.025$  and  $\frac{e_{z1}}{D} = \frac{e_{z2}}{D} = 0.25$ has been considered. Fig. 8.3 and Table 8.2 summarise the findings for the full range of  $\frac{M_t}{M_y}$  values in the form of interactive combinations of P and M<sub>t</sub> at failure. As the problem approaches the case of pure torsional loading (P--> 0) so the differences between the results obtained from the three approaches become less. In this region the degree of coupling between bending and torsional action is smaller with the result that those contributions to the stiffness matrices that depend on the product of derivatives of displacements and rotations become less significant. Conversely geometrical nonlinearity is of more significance when there is greater scope for the various interactive effects produced by the compressive load acting through the various

- 196 -

deformations.

Figs. 8.4a and 8.4b present a load-rotation and loaddeflection response for the case in which  $M_t$  is zero. The maximum load obtained from the linear analysis overestimates that given by the full nonlinear analysis by 25 per cent, whilst partial nonlinear analysis overestimates the failure load by 15 per cent as compared with the full nonlinear analysis. Load-rotation responses for  $\frac{M_t}{M_p}$ =0.035 and 0.069 are shown in Figs. 8.5a and 8.5b. In the first case the failure load given by the linear analysis exceeds those of the partial and full nonlinear analyses by 10% and 17% respectively.

Load rotation curves and load deflection responses for the case of  $\frac{M_t}{M_p}$  kept constant and equal to 0.138, while the axial load is applied with equal eccentricities  $(\frac{e_{y1}}{B} - \frac{e_{y2}}{B} - 0.125)$  and  $\frac{e_{z1}}{D} - \frac{e_{z2}}{D} - 0.257)$  at both ends for the three types of analyses (linear, nonlinear analysis, and full nonlinear) are presented in Fig. 8.6. In each type of analysis the eccentric load is increased gradually by a percentage of the squash load  $(\overline{P}_y)$ . Its increment is made smaller as soon as the deflections rapidly increased, sometimes reaching  $\frac{\overline{P}_y}{1000}$  or less near failure.

Fig. 8.6a represents load twisting curves in non-dimensional form as P/  $\overline{P}_y$  vs  $\phi/\phi_{max}$ , in which  $\overline{P}_y$  and  $\phi_{max}$  are 203 KiP and 38.8° respectively. The load-deflection response is displayed in Fig. 8.6b. which describes the interaction of P/ $\overline{P}_y$  and V/V<sub>max</sub> where V<sub>max</sub> is the maximum in-plane displacement at failure (V<sub>max</sub>=22.6 mm). In both Figs the torsional moment is kept unchanged ( $\frac{M_t}{M_p}$ =0.138) followed by increasing the axial load until collapse. It is clear from Figs 8.6a and 8.6b at the early stages of loading, that the differences between the linear, nonlinear, and full nonlinear analysis are negligible. As

- 197 -

the eccentric load is increased the variations of the three analyses can be easily recognised. These discrepancies in the results occured due to taking into account the nonlinear strain-displacement relations with the linear stiffnesses in the nonlinear analysis and also due to the coupling between the displacements and rotations in the nonlinear tangential and geometrical stiffness matrices together with the nonlinear strains in the full nonlinear analysis. The ratios between the maximum load ( $P_{max}$ ) to the yield load ( $\overline{P}_y = Ax\overline{\sigma}_y$ ) for the linear, nonlinear, and full nonlinear analysis at failure are 0.0395, 0.0347, and 0.033 respectively.

In each load increment the equilibrium condition of the convergence criteria must be satisfied, in other words the square root of the change of the displacement ( $\Delta R$ ) and out-of-balance force ( $\Delta P$ ) should be less than the tolerance error which was chosen equal to 0.001; this was explained in detail in Chapter 5.

### 8.3.3 Inelastic Analysis

In the inelastic range, the critical loads decrease from the elastic values because the stiffnesses of the member are reduced by spread of yield. Several illustrative examples with different slendernesses covering bending and torsion and compression have been considered.

### 8.3.3.1 Torsional moment (M,) applied at mid-span

A simply supported beam having a length given by the slenderness ratio  $\overline{\lambda}_1 = \sqrt{M_p/M_{cr}} = .4$  and initial imperfections under torsional moment applied at mid-span is investigated. The imperfections are out-of-plane displacement (U\_) and residual stress. The lack of

- 198-

straightness is represented by a sine wave and its maximum amplitude is equal to L/1000 at mid-span. The residual stress  $(\sigma_r)$  distribution is given in Fig. 8.7, in which its values at the flange tips  $(\sigma_f)$ , flangeweb junctionsns  $(\sigma_{fW})$ , and along the web  $(\sigma_W)$  take the value  $0.3\sigma_y$ , where  $\sigma_v$  is the yield stress.

This example contains two cases; in the first the effect of the residual stress is incorporated, in the second it is ignored. The two cases are solved by linear analysis, in which the influence of both the higher order terms of the strain-displacement relations and the nonlinear stiffness matrices (tangential and geometrical) are not accounted. In both cases the torsional moment  $(M_t)$  is increased monotonically until failure. The results, in the form of a moment-twist relation, are given in Fig. 8.7. It shows in the early stages of loading (elastic range) that both approaches give very similar results. But as soon as yield occurs the differences become noticeable as seen in the figure. Fig. 8.8 presents spread of yield at the initiation, part way to failure and at the final stages. Use of the more rigorous approach causes a reduction in the load at which yield starts of 25 per cent and a decrease in the collapse load of 9 per cent.

#### 8.3.3.2 Bending and Torsion

Three values of non-dimensional slenderness  $\int_{M_cr}^{M_p} M_{cr}$  given by 0.4, 0.7, and 1.2 are considered herein. In each case the applied loads are equal end moments followed by torsional moment ( $M_t$ ) at mid-span vary from zero up to failure.

The applied end moments are equal  $(M_{y1}=M_{y2})$  and are set; at the same time for every value the torsional moment  $M_t$  is varied from zero to the collapse load. Results obtained by linear analysis are provided in Fig. 8.9, which summarizes the interaction between the ultimate torsional

- 199 -

and bending moments in the form of  $\frac{M_t}{M_p} vs \frac{M_y}{M_y}$ . These curves illustrate the real response of the structural member subjected to this type of loading. Fig 8.9 shows that in each example the values of the ultimate torsional moment depends on the values of the applied end moments. Its value decreases when the end moments are increasing because of early yield in the cross-section

### 8.3.3.3 Nonuniform Bending and Torsion

A beam-column of I-section (W12x14) subjected to unequal end moments and torsion ( $M_t$ ) at mid-span has been investigated. The beam length is give by slenderness ratio ( $\overline{\lambda}_3$ ) equal to 1.2. The value of  $M_{y2}$ which represents the moment at end 2 is kept equal to zero, while  $M_{y1}$ adopts a series of values. For every value of  $M_{y1}$  the torsional moment is increased from zero to maximum (failure) load. A full set of results is presented in Fig. 8.10.

The results of these analyses are compared with the equivalent set from the previous study using equal end moments in the form of interaction curves of  $M_y$  and  $M_t$  in Fig. 8.10. It shows that as bending increases the differences between the ultimate torsional moments is decreased. Traces of the spread of yield through the cross section for both cases are presented in Figs 8.11 for unequal end moments and 8.12 for equal end moments.

Figs 8.11a, 8.11b, and 8.11c show the yield spread for 3 different values of end moments at 3 different levels of applied torque representing first yield, an intermediate stage and failure. The ultimate torsional moments are greater than those at first yield by 7, 9, and 9.2 per cent.

Figs. 8.12a, 8.12b, and 8.12c present the yield spread for

- 200 -

the case of equal end moments and torsion applied at mid-span. The values of the torsional moments at first yield are less than those at collapse by 8, 8, and 6.4 per cent. The differences between the torsional moments at first yield for the case of applied equal and unequal end moments are 63, 49, and 21 per cent, and at the ultimate load are 63, 49, and 22 per cent.

#### 8.3.3.4 Biaxial Bending and Torsion

Several examples of the response of a simply supported I-Beam (W12x14) subjected to biaxial bending and torsion are presented herein. The boundary conditions employed are; end displacements (in-plane and out-of-plane) restrained, together with twist prevented at the two ends, while warping is permitted. The influence of residual stresses, assuming a Lehigh distribution, is considered in the case of torsional moment only.

The linear and nonlinear analysis of a beam-column subjected to eccentric compressive loads together with a torsional moment at midspan are investigated for two illustrative examples. In both the torsional moment is kept constant, while the eccentric axial load is varied from zero up to collapse. The compressive load is applied with eccentricities  $\frac{e_{y1}}{B}=0.125$  /  $\frac{e_{z1}}{D}=0.686$  at node A and  $\frac{e_{y2}}{B}=0.0$  /  $\frac{e_{z2}}{d}=0.686$  at node B with the torsional moment being held at 3.0 Kip.in for the first example, while for the second case the axial load has eccentricities  $\frac{e_{y1}}{B}=0.063$  /  $\frac{e_{z1}}{D}=0.172$  and  $\frac{e_{y2}}{B}=0.063$  /  $\frac{e_{z2}}{D}=0.127$  at nodes A and B respectively and M<sub>t</sub>= 9.0 Kip.in. The slenderness ratio ( $\frac{L}{r_y}$ ) for both cases is equal to 25.

Figures 8.13a and 8.13b give load-rotation curves and load-

- 201 -

deflection (in-plane and out-of-plane) curves for the results obtained from linear and nonlinear analysis of the first example. The loss of stiffnesses due to yield spread is demonstrated in Fig. 8.14 The differences in the first yield loads for the two analyses is 6 per cent. The value of  $\frac{P}{P_y}$  at failure is 0.098 for linear analysis and 0.090 for the second solution; thus the simpler linear analysis overestimates the failure load by some 9 per cent.

The load-deflection responses for the second example are given in Fig. 8.15. Loss of stiffness due to spread of yield is presented in Fig. 8.16. First yield occured at loads corresponding to  $\frac{P}{P_y}$ =0.078 and  $\frac{P}{P_y}$ =0.08 respectively, while the difference at failure is 6 y per cent.

# 8.4 Examination of the Failure State of Members under Bending and Torsion

For a simply supported I-beam subjected to a twisting moment  $(M_t)$  at mid-span the total twisting moment is the sum of the St. Venant torsion  $(T_{sv})$  and warping torsion  $(T_w)$ ; the torque diagram is given in Fig. 8.17 and the overall differential equation describing the twisting is:

$$M_{t} = T_{sv} + T_{w}$$
(8.1)

where

$$T_{w} = GK\Phi', \quad T_{sv} = EI_{\omega}\Phi'' \quad (8.2)$$

so equation 8.1 can be written as

$$M_{t} = GK\Phi' - EI_{\omega}\Phi''$$
(8.3)

- 202 -

equation 8.3 can be written in another form as

$$A = \lambda^2 \Phi' - \Phi'' \tag{8.4}$$

.

where

$$A = \frac{M_t}{EI_{\omega}} \quad \text{and} \quad \lambda^2 = \frac{GK}{EI_{\omega}}$$

Assume the general solution of equation 8.4 is

$$\Phi = A_1 + A_2 SINH\lambda X + A_3 COSH\lambda X + \frac{A}{\lambda^2} X$$
(8.5)

where  $A_1$ ,  $A_2$ , and  $A_3$  are constants of integration; these constants are evaluated to comply with the torsional boundary conditions. X is the distance along the longitudinal axis. The boundary conditions for a simply supported I-beam used herein are stated below:

$$\Phi \Big|_{x=0,1} = 0$$
 (8.6a)  
 $\Phi' \Big|_{x=\frac{1}{2}} = 0$  (8.6b)

Substituting equations 8.6a and 8.6b in equation 8.5 and making some manipulation yields

$$\Phi = \frac{M_{t}}{EI_{\omega}\lambda^{3}} \left[ -\frac{1}{\cosh\frac{\lambda}{21}} SINH\lambda X + \lambda X \right]$$
(8.7)  
$$\Phi^{"} = \frac{M_{t}}{EI_{\omega}\lambda} \left[ -\frac{1}{\cosh\frac{\lambda}{21}} SINH\lambda X \right]$$
(8.8)

The equations presented by Matthey (1984) and Chen and Atsuta (1977) for calculating the first yield moment and plastic moment about the Y and Z axes are employed as:

a) At first yield

$$M_{yy} = \frac{I_{y\sigma}}{\frac{D}{2}} \sigma_{y} \sigma_{y} Z_{z}$$

$$M_{yz} = \frac{I_z}{\frac{B}{2}} \sigma_y^{z}, \sigma_y, Z_y$$

where  $Z_y$  and  $Z_z$  are the section modulii about the Y and Z axes b) At full plasticity

$$M_{py} = S_{y} \cdot \sigma_{y}$$

$$M_{pz} = S_z \cdot \sigma_y$$

where

$$S_{y} = [BT(h-t) + T_{w}(\frac{h}{2}T)^{2}]$$

$$S_{z} = [B \frac{T_{2}}{2} + 1/4T_{w}^{2}(h-2T)]$$

For Section WF12x14

 $S_y = 17.2285in^3$   $I_y = 87.177in^4$  K = 0.0629in<sup>4</sup>  $S_z = 1.95415in^3$   $I_z = 2.453in^4$   $r_y = 4.5728in$ 

 $\sigma_{v} = 50$ KSI A = 4.169in<sup>2</sup>  $r_{z} = 0.7910$ in

- 204 -

$$M_{pz} = 97.7075 kip.in$$
  $M_{yz} = 62.325 kip.in$ 

# 8.4.1 Stability of I-Section under Flexural and Torsional Loading

based on a simple design check; Pastor and DeWolf (1979) suggested that the summation of the stresses caused by flexural and torsional loads on an I-section (W12x14) be limited to the yield stress  $(\sigma_y)$ . This can be illustrated as:

$$\sigma_{by} + \sigma_{bz} + \sigma_{\omega} \geqslant \sigma_{y} \tag{8.9}$$

in which  $\sigma_{by}$  and  $\sigma_{bz}$  are the bending stresses about the Y and Z axes, and  $\sigma_{c}$  is the warping stress. If we substitute for bending stress by

$$\sigma_{by} = \frac{M_y}{I_y} Z = \frac{M_y}{Z_y}$$
(8.10a)

$$\sigma_{bz} = \frac{M_y}{I_z} Y = \frac{M_z}{Z_z}$$
(8.10b)

Substituting equations 8.10a and 8.10b into equation 8.9 yields

$$\frac{M_{y}}{Z_{y}\sigma_{y}} + \frac{M_{z}}{Z_{z}\sigma_{y}} + \frac{\sigma_{\omega}}{\sigma_{y}} > 1$$
(8.11)

or

$$\frac{M_y}{M_{yy}} + \frac{M_z}{M_{yz}} + \frac{\sigma_\omega}{\sigma_y} > 1$$
(8.12)

where

$$M_{yy} = Z_z \cdot \sigma_y$$
 and  $M_{yz} = Z_z \cdot \sigma_y$ 

in which M and M are the first yield moments about the Y and Z axes.

The value of  $M_z$  is taken equal to the twist times the applied end moments in the form of  $M_z = \Phi M_y$ , where  $\Phi$  is the twist calculated by equation 8.7. The warping stress  $(\sigma_{\omega})$  in terms of the normalized function  $(\overline{\omega}_n)$ , and the second derivative of  $\Phi$  with respect to the X-axis  $(\Phi'')$  is given by equation 8.8 and it can be written as

$$\sigma_{\omega} = E \overline{\omega}_{n} \phi^{n}$$
(8.13)

Substitute for  $\sigma_{\rm w}$  and M<sub>z</sub> in equation 8.12, which yields

$$\frac{M_y}{M_{yy}} + \frac{M_z}{M_{yz}} + \frac{E\overline{\omega}_n \Phi''}{\sigma_y} \ge 1$$
(8.14)

An investigation was conducted of three cases which represent simply supported beams. Their end displacements (U and V) together with the twist ( $\overline{\Phi}$ ) are restrained, whilst the warping is permitted at the two ends. The beam lengths are given by the slendernesses  $\overline{\lambda}_1$ ,  $\overline{\lambda}_2$ , and  $\overline{\lambda}_3$  equal to 0.4, 0.7, and 1.2 respectively.

Substituting for warping flexural rigidity (EI<sub> $\omega$ </sub>),  $\lambda$  which is equal to square root of the torsional rigidity divided by warping flexural rigidity ( $\lambda = \sqrt{GK/EI_{\omega}}$ ), and for the beam length (L) in equations 8.7 and 8.8, this yields:

Case A  $\overline{\lambda}_1 = 0.4$ 

$$\Phi = 0.0001799 M_{t}$$
 (8.15a)

$$\Phi'' = 0.2753\% \times 10^{-5} M_{t}$$
 (8.15b)

Case B  $X_{1=0.7}$ 

$$\Phi = 0.0010132 M_{+}$$
 (8.16a)

 $\Phi'' = 0.480 \times 10^{-6} M_{+}$  (8.16b)

Case C  $\overline{X}_{1=1.2}$ 

$$\Phi'' = 0.744934 \times 10^{-5} M_{t}$$
 (8.17b)

# 8.4.1.1 Stability of Beam-Column of I-Section under Bending and Torsion Based on Pastor and DeWolf (1979) Suggestions

Stresses caused by flexural and torsional loads based on the Author's full finite element (f.e.) analysis and the use of the equations 8.15 to 8.17 to calculate the interaction stresses are presented in Tables 8.3, 8.4, and 8.5 for each load combination in each case. The end moments are applied first incrementally to a set limit followed by incrementing the torsional moment ( $M_t$ ) to failure. For zero torsional moment the end moments are simply increased to collapse and a similar procedure is used when the uniform moments are zero. The interaction between torsion and bending at the ultimate condition has been discussed in full in section 8.3.3.2. Equation 8.14 is used to calculate the stresses at first yield. The ultimate values of the

bending moments about the strong axis and the corresponding torsional moment at mid-span, the maximum twists at mid-span obtained from the analysis and from equation 8.7 and 8.8 are presented in Tables 8.3, 8.4, and 8.5 for case A, case B, and case C respectively.

1.-

Table 8.3 contains the results for a stocky beam-column of slenderness  $\overline{\lambda}_1$ . The summation of stresses calculated by equation 8.14 are less than 1 for  $\frac{M_y}{M_p}$ >0.35 and the corresponding  $\frac{M_t}{M_y}$ >0.043. When  $\frac{M_y}{M_p}$ <br/>(0.35 the summation of  $\frac{M_y}{M_y}$ ,  $\frac{M_z}{M_y}$ , and  $\frac{\sigma_\omega}{\sigma_y}$  are greater than 1. So according to the assumptions of Pastor and DeWolf (1979) the results are not safe. Table 8.4 presents the interaction of case B in which the summation of the stresses are less than 1, except when the  $\frac{M_t}{M_y}$  and  $\frac{M_y}{M_p}$  are equal to 0.070/.070 and 0.115/0./72 respectively. So the last two examples are not safe according to Pastor and DeWolf. For the slender member the summation of the interaction of stresses is fully presented in Table 8.5, in which  $A_{fy}$  is less than 1 so the examples investgated in this case are considered to be safe.

Comparison of the results provided in Tables 8.3 - 8.5 shows that the assumption made by Pastor and DeWolf are applicable for some problems and not good for others. For a slender member the application of their assumptions is in good agreement, but for both the intermediate and more stocky members, it seems the results of some examples are not acceptable. In the interaction equation employed by Pastor and DeWolf, the torsional moment is not incorporated and instead they used warping stress. The inclusion of  $\sigma_{\omega}$  in their analysis it seems to be not applicable in the inelastic range because the warping stress increases rapidly in the yielded regions, so the interaction equation which used the ratio of  $\sigma_{\omega} / \sigma_{y}$  must be less than one is not valid. In the next section an attempt to incorporate this effect in

- 208 -

another more realistic form is made.

8.4.1.2 Stability of Beam-Column of Thin-Walled I-Section Subjected to Combined Bending and Torsion Based on Author's Suggestions

In this section equation 8.12 has been used with some modification for two separate methods as:

i- Based on Initial Yielding

ii- Based on Full Plasticity

#### i- Based on Initial Yielding

In this case the interaction equation which represent the summation of stresses calculated after first yield is given by

$$A_{fy} = \frac{M_y}{M_{yy}} + \frac{M_z}{M_{yz}} + \frac{M_t}{M_{y\omega}}$$
(8.18)

in which  $A_{fy}$  is the summation of the ratio of moments at initial yield.  $M_{y\omega}$  is the first yield warping moment (Chen and Atsuta 1977) and is given by

 $M_{y\omega} = \frac{I_{\omega}}{\frac{BD}{4}} \cdot \sigma_{y}$ (8.19)

where I is the warping moment of inertia, B and D are cross-section width and depth respectively.

#### ii- Based of Full Plasticity

The contribution to fully plastic interaction is given in the form:

$$A_{fp} = \frac{M_y}{M_{py}} + \frac{M_z}{M_{pz}} + \frac{M_t}{M_{p\omega}}$$
(8.20)

 $A_{fp}$  is equal to the summations of  $M_y/M_{py}$ ,  $M_z/M_{pz}$ , and  $M_t/M_{p\omega}$  at full plasticity.  $M_{py}$ ,  $M_{pz}$ , and  $M_{p\omega}$  are the plastic moments about Y and Z axes are calculated using Matthey (1984) equations and plastic warping torque. The last term  $(M_{p\omega})$  is that torque required to cause full plastification of flanges due to tensile or compressive stresses. These stresses are illustrated in Fig. 8.17 previously presented by (Salmon and Johnson 1980), in which the warping moment can be written as:

$$M_{p\omega} = \frac{B^2 T_f (D - T_f)}{L} \sigma_y = \frac{2185}{L} (W12x14)$$
 (8.21)

where  $T_{f}$  is the flange thickness. The warping moments calculated by the above formula for the three cases described in section 8.4.1 ( $\overline{\lambda}_{1}$ =0.4,  $\overline{\lambda}_{2}$ =0.7, and  $\overline{\lambda}_{2}$ =1.2) are 43.2, 24.8, and 19.2 kip.in respectively.

Tables 8.6, 8.7, and 8.8 contains the results of the whole series based on the use of equations 8.18 and 8.20 for the first yield and full plasticity respectively. The interaction of stresses  $A_{fy}$  and  $A_{py}$  are fully listed. It shows that the summation of warping and bending stress according to the first yield criterion or the plastic criterion are less than 1. The form of interaction gives a safe prediction of failure. Tables 8.6, 8.7, and 8.8 show that  $A_{fy}$  is increased by increasing the uniform end moments.

The square root of the sum of the terms in the interaction formulae of equations 8.18 and 8.20 squared can be written as:

$$C_{1} = \sqrt{(\frac{M_{y}}{M_{yy}})^{2} + (\frac{M_{z}}{M_{yz}})^{2} + (\frac{M_{t}}{M_{y\omega}})^{2}}$$
(8.22a)

$$C_{2} = \sqrt{\left(\frac{M_{y}}{M_{py}}\right)^{2} + \left(\frac{M_{z}}{M_{pz}}\right)^{2} + \left(\frac{M_{t}}{M_{p\omega}}\right)^{2}}$$
(8.22b)

The above two equations are based on initial and full plastic moments and are presented in a parabolic form. The reasons for incorporating them in this fashion, are first to investigate their interaction with the ratio of torsional moment with respect to either initial warpin moment or to full plastic warping moment  $\left(\frac{M_{t}}{M_{t}}\right)$ ,  $\frac{M_{t}}{M_{t}}$ , and second to compare the results obtained against those of equations 8.18 and 8.20. Tabulated numerical calculations using equations 8.18, 8.20, 8.22a, and 8.22b involving series of slendernesses are fully presented in Table 8.9. It shows that for  $\frac{L}{r_{e}}$ =35 the values of C<sub>1</sub> and y  $A_{fy}$  are increasing when the ultimate torsional moment decreases, and similarly for  $A_{fp}$ . The term  $C_2$  is gradually decreasing when  $\frac{M_t}{M_p}$  is  $p_W$  the beteen 0.15 and 0.085, and then it starts increasing. The value of the two terms  $A_{fp}$  and  $C_2$  are equal when  $\frac{M_y}{M_{pw}} = 0$ ,  $\frac{M_t}{M_{pw}} = 0.154$  and  $\frac{M_y}{M_{py}} = 0.836/\frac{M_y}{M_{py}} = 0$ , because twisting is a function of  $M_t$  and  $M_z = \phi M_y$ . The value of  $\phi$ according to equation 8.15 is equal to zero when the applied torque is zero and  $M_z$  is equal to zero when the uniform end moments is zero. Similar behaviour occurd for the intermediate and the more slender member. The ratio of  $\frac{C_2}{A_{fD}}$  ranges from 0.75 to 1.0, 0.7 to 1.0, and 64 to 1 for the three slendernesses  $(\overline{\lambda}_1, \overline{\lambda}_2, \text{ and } \overline{\lambda}_3)$  respectively. Fig. 8.19 illustrates the interaction curves for three slendernesses  $\overline{\lambda}_1$  ,  $\overline{\lambda}_2$  , and  $\overline{\lambda}_3$  of beam-column subjected to flexural and torsional loads. The

uniform end moments are applied about the strong axis and at the same time as the torsional moment at the mid-span. Details of load incrementation have been given in section 8.4.1.1. These curves indicate that it is possible to provide a practical prediction formula which can then be used to determine the ultimate loads. Further explanations will be provided in section 8.5.

8.4.1.3 Stability of Beam-Column under Biaxial Bending and Torsion

In this case selected values of the mid-span torsional moment  $(M_t)$  are applied followed by an axial load with eccentricities  $\frac{e_{y1}}{D}$ ,  $\frac{e_{y2}}{D} = 0.167$  and  $\frac{e_{z1}}{B}$ ,  $\frac{e_{z2}}{B} = 0.257$  about both ends. At every value of  $M_t$  the end load is increased up to failure. The interaction of the compressive force P and torsional moment  $M_t$  is given in Table 8.8 where it is compared with the results generated from the following equation.

$$AA = \frac{P}{P_{pl}} + \frac{M_t}{M_{wp}} + \frac{M_y}{M_{pyl}} + \frac{M_z}{M_{pzl}}$$

where  $P_{pl} = \sigma_y A$ , and  $M_t$ ,  $M_y$ ,  $M_z$ ,  $M_{wp}$ ,  $M_{pyl}$ , and  $M_{pzl}$  are defined in section 8.4.2. The results of the above equation shows that the term AA is less than 1.

# 8.5 <u>Prediction of Stability of Beam-Column under Flexural</u> Bending and Torsion Using a Regression Analysis

The ultimate bending and torsion values presented in Table 8.9 and illustrated in the interaction curves of Fig. 8.19 have been used herein to develop a practical formula based on the results previously discussed concerning beams subjected to combined bending and torsion. Three equations are developed based on:

- 1- Interaction of  $M_t/M_{pw}$  and  $A_{fy}$
- 2- Interaction of  $M_t/M_{pw}$  and  $C_1$

3- Combination of 1 and 2

where  $A_{fy}$  and  $C_1$  have been defined in equations 8.18 and 8.22a respectively.

The development of these equations are based on the application of regression techniques (Draper and Smith, 1961, Neville and Kennedy ,1964) A statistical package SPSS-X21, available at USCC (University of Sheffield Computer Centre), is employed to calculate different parameters such as, mean value, maximum/ minimum values, standard deviation, correlation, mean square, residual standard errors, etc. and then these variables are used in deriving regression equations. The equations conducted for the three slendernesses  $\overline{\lambda}_1=0.4$ ,  $\overline{\lambda}_2=0.7$ , and  $\overline{\lambda}_3=1.2$  are summarized in the following three cases:

Case i $M_t/M_{pw}$  is function of  $A_{fy}$  and LCase ii $M_t/M_{pw}$  is function of  $C_1$  and L

**Case iii**  $M_t/M_{pw}$  is function of  $A_{fy}$ ,  $C_1$ , and L in which  $\frac{M_t}{M_{pw}}$ ,  $A_{fy}$ ,  $C_1$ , and L can represent any values of moments and slendernesses. In other words the equations can be applied to any case previously explained.

The prediction equation of case i in which  $\frac{M_t}{M_{pw}}$  F(A<sub>fy</sub>, L) can be written as:

$$\frac{M_t}{M_{pw}} = 0.167937 - 0.1217A_{fy} - 0.001037L \qquad (8.23)$$

the other equation which based on case ii is

- 213 -

$$\frac{M_t}{M_{pw}} = 0.164857 - 0.119876C_1 - 0.001118L$$
(8.24)

The last equation which employs  $\frac{M_t}{M_p}$  as a function of  $A_{fy}$ ,  $C_1$ , L can be written as:

$$\frac{M_t}{M_{pw}} = 0.164835 - 0.083002C_1 - 0.03828A_{fy} - 0.001094L \quad (8.25)$$

Equations 8.23 - 8.25 can be used to determine the ultimate moment for any set limit of torsion or vice versa. They can also be used for slendernesses between 0.4  $<\sqrt{\lambda} <$ 1.2. These equations can play an important part in the design of members having thin-walled open crosssection in the design of steel structures and leads the way for further work in this area.

## 8.6 General Features of the Analysis

Different cases of loading and slenderness have been addressed to provide some idea of the interaction of these variable on the failure condition. Before a fully validate ultimate strength approach to the design of slender members under any form of combined torsional/ flexural/ compressive loading can be produced a considerable amount of further work is required. This should cover:

- 1- Different slenderness ratio.
- 2- Different loading conditions.
- 3- Different boundary conditions.
- 4- Variable compressive load eccentricities.

- 214 -

- 5- Including initial imperfections ( such as residual stress and initial deflections).
- 6 Interaction equation on the form

$$\frac{P}{P_{pl}} + \frac{M_t}{M_{\omega p}} + \frac{M_y}{M_{pyl}} + \frac{M_z}{M_{pzl}} \leqslant 1$$

NOTE An extension to some of this work is now in progress and the results are encouraging.

#### 8.7 Conclusions

The elastic and inelastic 3-dimensional behaviour of beamcolumns under bending and torsional loading has been studied. The presence of initial imperfections such as residual stresses and initial deflections together with monitoring of the yield spread through the cross-section have been considered. Several example problems, which have not been previously solved, have been investigated.

The solutions for these examples were obtained by three types of analysis; 'Linear', 'Nonlinear', and Full Nonlinear. The results show the importance of including of the so call 'higher order terms' of the strain displacement relations and the nonlinear geometrical and tangential stiffness matrices in the analysis. From these problems, several important observations are possible:

- 1- The ability of the formulation presented in chapter 3 to cover the full range of analysis of beam-columns in 3-dimensions.
- 2- The need to include higher order terms in the straindisplacement relations and stiffness matrices in order to avoid overestimates of load carrying capacity.

- 215 -

Table 8.1 Comparison of Author's result against those obtained by Pastor and de Wolf (1979), for a beam subjected to constant torsional moment and equal end moments, ( $\overline{\sigma}_y = 50$  ksi, w12x43)

	σγ	σ <sub>y</sub>	% age	σ <sub>z</sub> /	σ <sub>y</sub>	% age	σ	σ <sub>y</sub>	% age	σ <sub>t</sub> ,	νσ <sub>y</sub>	% age	Rotatio "rad" 1	ons 10 <sup>-2</sup>	% age
$\frac{M}{M}$ cr	Author	Pastor	difference	Author	Pastor	error	Author	Pastor	error	Author	Pastor	error	Author	Pastor	error
0.0	0.0	0.0	-	0.0	0.0	-	0.284	0.282	0.7	0.284	0.282	0.7	2.95	2.96	0.3
0.1	0.025	0.026	3.8	0.067	0.068	1.5	0.292	0.284	2.7	0.386	0.378	2.1	3.04	3.00	1.3
0.2	0.050	0.052	3.8	0.134	0.136	1.5	0.300	0.290	3.3	0.484	0.478	1.2	3.12	3.09	1.0
0.3	0.078	0.082	4.8	0.201	0.204	1.5	0.313	0.304	3.0	0.592	0.588	0.7	3.29	3.25	1.2
0.4	0.118	0.120	1.7	0.271	0.270	0.4	0.335	0.322	4.0	0.718	0.712	0.8	3.57	3.51	1.7
0.5	0.156	0.166	6.0	0.345	0.338	2.1	0.369	0.354	4.2	0.870	0.858	1.4	4.01	3.93	2.0
0.6	0.218	0.232	0.6	0.430	0.406	6.0	0.426	0.402	6.0	1.076	1.040	3.3	4.70	4.57	2.8
0.7	0.346	0.338	2.4	0.552	0.474	14.1	0.547	0.486	12.5	1.444	1.298	10.1	5.91	5,71	3.4
0.8	0.580	0.542	7.0	0.707	0.554	21.6	0.749	0.656	14.2	2.016	1.74	13.7	8.98	8.00	10.1

<u>\_\_\_</u>

	F	۶/P <sub>y</sub> ×10 <sup>-2</sup>		Differe	ences bet	ween	% age error between		
M M cr		Type of anal	ysis 10 <sup>-2 '</sup>	- -		x 10 <sup>-2</sup>	1&2	1&3	2&3
10 <sup>-2</sup>	Linear I	Non-linear 2	Full non-linear 3	1 & 2 A	1 & 3 B	2 and 3 C			
0.0	19.30	16.40	14.50	2.90	4.80	1.90	15.0	24.9	13.1
2.30	13.98	12.55	11.26	1.43	2.72	1.29	10.2	19.5	10.3
3.45	11.88	10.66	9.90	1.22	1.98	0.76	10.3	16.7	7.1
4.60	10.23	9.26	8.34	0.97	1.89	0.92	9.5	18.5	9.9
6.90	7.58	6.82	6.06	0.76	1.52	0.76	10.0	20.1	11.1
9.20	6.06	5.12	4.55	0.94	1.51	0.57	15.5	29,5	11.1
11.50	4.90	4.11	3.47	0.79	1.43	0.64	16.1	29.2	15.6
13.79	3.95	3.47	3.32	0.48	0.63	0.15	12.15	15.9	4.3
16.09	2.33	2.21	2.05	0.12	0.25	0.16	5.2	10.7	7.2
18.22	0.0	0.0	0.0	0	0	0	0	0	0

Table 8.2 Results obtained for three analyses of beam-column under biaxial bending and torsional moment at mid-span

φ x10 <sup>-3</sup>	M y	M <sub>t</sub>	$\phi_{c} \times 10^{-3}$	$\phi " \times 10^{-4}$	$M_{z} = \oint_{c} M_{y}$	σ <sub>w</sub> =	I	First Yie	eld	$A_{fy} =$
'rad <sup>''</sup>	KID.IN	кір.іп	'rad'	edu. 0.90	kip₊in	ב <b>יי</b> ה ס	A	В	С	A+D+C
9.017	0.0	53.785	9.676	1.481	0.0	52.759	0.0	0.0	1.055	1.055
8.264	100.0	45.117	8.117	1.243	0.812	44.256	0.136	0.013	0.885	1.034
7.461	150.0	40.889	7.356	1.126	1.103	40.109	0.204	0.018	0.802	1.024
6.908	200.0	37.332	6.715	1.028	1.343	36.620	0.272	0,022	0.732	1.026
5.564	300.0	29,697	5.342	0.818	1.603	29.131	0.409	0.026	0.583	1.018
4.162	400.0	21.841	3.929	0.601	1.572	21.424	0.545	0.025	0.429	0.999
3.121	500.0	15.14	2.724	0.417	1.362	14.851	0.681	0.022	0.297	1.00
0.409	720.8	0.0	0.0	0.0	0.0	0.0	0.981	0.0	0.0	0.981

Table 8.3 Presentation of Calculation of Stresses Caused by Bending and Torsion of Case A

 $A = M_y / M_{yy} \qquad B = M_z / M_{yz} \qquad C = \sigma_w / \sigma_y$ 

-218-

٠

$\phi \times 10^{-3}$ f.e.	M klp.in	M <sub>t</sub> kip.in	$\phi_{c} x \ 10^{-3}$ egn. 8.9a	$\phi$ " x 10 <sup>-6</sup> eqn. 8.9b	$M_z = \phi_C M_z$	σ =	F	irst Yie	ld	$A_{fy} =$
'rad'	•		'rad'	-	⇒ kip.in	Ενφ"	А	В	<b>C</b> .	A+B+C
30.166	0.0	29.521	29.91	141.701	0.0	49.50	0.0	0.0	0.990	0.990
27.615	50 <b>.</b> 0	26.789	27.139	128.568	1.357	44.915	0.068	0.022	0.898	0.989
25.743	100.0	24.635	24.960	118.248	2.496	41.309	0.136	0,041	0.826	1.003
23.625	150.0	22.110	22.401	106.128	3.360	37.076	0.204	0.055	0.742	1.002
21.294	200.0	19.742	20.003	94.762	4.001	33.105	0.273	0.065	0.662	0.999
13.240	400.0	10.754	10.896	51.619	4.358	18.033	0.545	0.072	0.361	0.977
10.19	500.0	7.330	7.427	35.184	3.713	12.292	0.681	0.061	0.256	0.988
3.147	630.0	0.0	0.0	0.0	0.0	0.778	0.731	0.0	0.0	0.731

Table 8.4 Presentation of Calculation of Stresses, Caused by Bending and Torsion of Case B

 $A = M_y / M_{yy} \qquad B = M_z / M_{yz} \qquad C = \sigma_w / \sigma_y$ 

$\phi \times 10^{-2}$ f.e.	M y	Mt	$\phi_{c} \times 10^{-2}$	¢" × 10 <sup>-5</sup>	$M_{z} = \oint_{C} M_{y}$	σ = w	F	irst Yie]	lđ	$A_{fy} = A + B + C$
"rad"	K1p.1n	K1p.1n	eqn. 8.9a	eqn. 8.9b	kip∙in	Ewqu	A	В	С	A+B+C
9.381	0.0	19.30	8.962	14.377	0.0	37.415	0.0	0.0	0.748	0.748
7.018	100.0	13.96	6.479	10.399	6.479	36.329	0.136	0.106	0.727	0.969
6.199	150.0	11.64	5.400	8.696	8.100	30.380	0.204	0.132	0.608	0.944
5.450	200.0	9,55	4.43	7.114	8.860	24.853	0.273	0.145	0.497	0.914
4.866	250.0	7.77	3.606	5.788	9.015	20.220	0.341	0.147	0.404	0.892
4.318	300.0	6.19	2.873	4.615	8.619	16.109	0,409	0.141	0.322	0.871
3.835	350.0	4.70	2.181	3.501	7.635	12.231	0.477	0.125	0.245	0.846
1.779	562.0	0.0	0.0	0.0	0.0	0.0	0.766	0.0	0.0	0.766

Table 8.5 Presentation of Calculation of Stresses Caused by Bending and Torsion of Case C

 $A = M_y / M_{yy} \qquad B = M_z / M_{yz} \qquad C = \sigma_w / \sigma_y$ 

¢ ×10 <sup>-3</sup>	M y	Mt	¢ <sup>x10<sup>-3</sup></sup>	<sub>φ</sub> ,"x10 <sup>-4</sup>	M <sub>z</sub> =∲M <sub>z</sub> cy	First Yield A <sub>fy</sub>			A <sub>fy</sub> =	A <sub>fy</sub> = Full Plasticity			$A_{fp} = $
f.e.	kip.in	kip.in.	eqn. 8.9a 'rad'	eqn. 8.9b	kip. <sub>in</sub>	A	В	С	(A+B+C)	Ā	B	ē	(A+B+C)
9.017	0.0	53.785	9.676	1.481	0.0	0.0	0.0	0.154	0.154	0.0	0.0	0.689	0,689
8.264	100.0	45.117	8.117	1.243	0.812	0.136	0.013	0.129	0.278	0.117	0.008	0.578	Ó.703
7.461	150.0	40.889	7.356	1.126	1.103	0.204	0.018	0.117	0.339	0.174	0.011	0.524	0.709
6.908	200.0	37.332	6.715	1.028	1.343	0.272	0.022	0.107	0.401	0.232	0.014	0.478	0.724
5.564	300.0	29.697	5.342	0.818	1.603	0.409	0.026	0.085	0.52	0.348	0.016	0.381	0.745
4.162	400.0	21.841	3.929	0.601	1.572	0.545	0.025	0.062	0.632	0.464	0.016	0.280	Ó.760
3.121	500.0	15.14	2.724	0.417	1.362	0.681	0.022	0.043	0.746	0.580	0.014	0.194	0.783
0.409	720.8	0.0	0.0	0.0	0.0	0.981	0.0	0.0	0.981	0.882	0.0	0.0	0.882

Table 8.6 Presentation of Calculation of Stress Caused by Torsion of Case A

 $A = M_y / M_{yy} \qquad B = M_z / M_{yz} \qquad C = M_t / M_{pw} \qquad \overline{A} = M_y / M_{py} \qquad \overline{B} = M_z / M_{pz} \qquad \overline{C} = M_t / M_{pw}$ 

 $A_{fy} = \frac{M_y}{M_y} + \frac{M_z}{M_y} + \frac{M_t}{M_y} + \frac{M_t}{M_y} + \frac{M_t}{M_p} + \frac{M_z}{M_p} + \frac{M_t}{M_p} + \frac{M_t$ 

$\phi \times 10^{-3}$	M y kin in	M <sub>t</sub> kin in	$\phi_{c} \times 10^{-3}$	$\phi_{c}^{"x10}^{-4}$	$\begin{array}{c c} A \\ M_z = \phi, M_z \\ F \end{array}$		irst Yie	1đ	$A_{fy} = (A + B + C)$	Ful	l Plastic	city	$A_{fp} = (\bar{A} + \bar{B} + \bar{C})$
'rad'	KIP.III	KIP.III	rad'	eqii. 0.30	ki₽∙in	A	В	С	(ATDTO)	Ā	B	Ē	(ATDTC)
30.166	0.0	29.521	29.91	141.701	0.0	0.0	0.0	0.084	0.084	0.0	0.0	0.681	0.681
27.615	50.0	26.789	27.139	128.568	1,357	0.068	0.022	0.077	0.167	0.058	0.014	0.619	0.691
25.743	100.0	24.635	24.960	118.245	2.496	0.136	0.041	0.070	0.247	0.116	0.026	0.569	0.711
23.625	150.0	22.110	22.401	106.128	3.360	0.204	0.055	0.064	0.323	0.174	0.035	0.511	0.720
21.294	200.0	19.742	20.003	94.762	4.001	0.273	0.065	0.056	0.394	0.232	0.041	0.456	0.729
13.24	400.0	10.754	10.896	51.619	4.359	0.595	0.071	0.031	0.647	0.464	0.045	0.248	0.757
10.19	500.0	7.330	7.427	35.184	3.713	0.681	0.061	0.021	0.763	0.580	0.038	0.169	0.787
3.147	<b>630.0</b> ·	0.0	0.0	0.0	0.0	0.858	0.0	0.0	0.858	0.731	0.0	0.0	0.731

.

Table 8.7 Presentation of Calculation of Stress Caused by Bending and Torsion of Case B

I. 222 1

Table 8.8 Presentation of Calculation of Stress Caused by Bending and Torsion of Case C

$$A = M_y/M_y \qquad B = M_z/M_y \qquad C = M_t/M_y \qquad \overline{A} = M_y/M_p \qquad \overline{B} = M_z/M_p \qquad \overline{C} = M_z/M_w \qquad \overline{C} = M_z/M_$$

φx10 <sup>-2</sup>	M y	.M <sub>t</sub>	φx10 <sup>-2</sup> c	φ"x10 <sup>-5</sup> c	M_=∲M z c y	First Yield			(A+B+C)	Full Plasticity		city	(Ā+B+C)
f.e. ''rad'	kip.in	kip.in	egn.8.9a 'rad'	egn. 8.9b	kip.in	A	В	с		Ā	Ē	Ē	
9.381	0.0	19.30	8.962	14.377	0.0	0.0	0.0	0.055	0.055	0.0	0.0	0.773	0.773
7.018	100.0	13.96	6.479	10.394	6.479	0.136	0.106	0.040	0,282	0.116	0.066	0.562	0.744
6.200	150.0	11.64	5.400	8,696	8.100	0.204	0.132	0.033	0.369	0.174	0.083	0.469	0.726
5.450	200.0	9.55	4.430	7.114	8.860	0.273	0.145	0.027	0.445	0.232	0.091	0.385	0.708
4.866	250.0	7.77	3.606	5.788	9.015	0.341	0.147	0.022	0.510	0.290	0,092	0.313	0.695
4.318	300.0	6.19	2.873	4.615	8.619	0.409	0.141	0.018	0,568	0.348	0.088	0.249	0.685
3.835	350.0	4.70	2.181	3.501	7.635	0.477	0.125	0.013	0.615	0.468	0.078	0.189	0.735
1.779	562.0	0.0	0.0	0.0	0.0	0.766	0.0	0.0	0.766	0.652	0.0	0.0	0.652

$$A_{fy} = \frac{M_y}{M_y} + \frac{M_z}{M_yz} + \frac{M_t}{M_{yw}} \qquad A_{fp} = \frac{M_y}{M_py} + \frac{M_z}{M_pz} + \frac{M_t}{M_{pw}}$$

.

L/r <sub>z</sub>	M <sub>t</sub> /M <sub>pw</sub>	<sup>м</sup> у∕м у ру	x10 <sup>-2</sup> f.e.	<sup>A</sup> fy	A <sub>fp</sub>	c1	с <sub>2</sub>
35.0	0.154	0.0	0.902	0.154	0.689	0.154	0.689
	0.129	0.116	0.826	0.278	0.703	0.187	0.590
	0.117	0.174	0.746	0.339	0.709	0.235	0.552
	0.107	0.232	0.691	0.401	0.724	0.292	0.531
	0.085	0.348	0.556	0.42	0.745	0.418	0.516
	0.062	0.464	0.416	0.632	0.760	0.549	0.542
	0.043	0.580	0.312	0.746	0.783	0.683	0.612
	0.0	0.836	0.041	0981	0.881	0.981	0.882
64.0	0.084 0.076 0.070 0.063 0.056 0.031 0.021 0.0	0.0 0.058 0.116 0.174 0.232 0.464 0.580 0.731	3.017 2.762 2.574 2.363 2.129 1.324 1.019 0.315	0.084 0.167 0.247 0.323 0.394 0.647 0.763 0.858	0.681 0.691 0.711 0.720 0.729 0.757 0.757 0.787 0.731	0.084 0.105 0.158 0.221 0.286 0.551 0.684 0.858	0.681 0.622 0.581 0.794 0.513 0.528 0.605 0.731
111.0	0.055	0.0	9.381	0.055	0.773	0.055	0.773
	0.040	0.116	7.018	0.282	0.744	0.177	0.578
	0.033	0.174	6.200	0.369	0.726	0.245	0.507
	0.027	0.232	5.45	0.445	0.708	0.310	0.459
	0.022	0.290	4.866	0.510	0.695	0.372	0.437
	0.018	0.348	4.318	0.568	0.685	0.433	0.437
	0.013	0.406	3.835	0.615	0.735	0.493	0.511
	0.0	0.652	1.779	0.766	0.652	0.766	0.652

Table 8.9 Interaction of Stresses of the Three Cases (A, B and C)

.

ł

Table 8.10 Ultimate Strength of Beam-Column under Biaxial Bending and Torsion

$\frac{e_{y1}}{B} = \frac{e_{y2}}{B} = 0.167$	-	$\frac{e_{z1}}{D} = \frac{e_{z2}}{D}$	=0.257	$\frac{L}{r_y} = 64$
---	---	---------------------------------------	--------	----------------------

Mt	P	M Y	Mz		First	Yield		A+B+C+D	A <sup>2</sup>	B²	C <sup>2</sup>	D <sup>2</sup>	$\sqrt{A^2 + B^2 + C^2 + D^2}$
kip.in	kip	kip.in	kip.in	A	В	С	D						
0.0	36.05	72.09	9.01	0.173	0.098	0.145	0.0	0.416	0.020	0.010	0.021	0.0	0.247
5.0	29.12	58.24	7.28	0.140	0.079	0.117	0.264	0.60	0.020	0.006	0.014	0.070	0.332
9.0	22.64	45.28	5.66	0.109	0.062	0.091	0.476	0.738	0.012	0.004	0.008	0.227	0.501
14.0	8.32	16.64	2.08	0.040	0.023	0.033	0.74	0.836	0.002	0.001	0.001	0.548	0.743
15.925	0.0	0.0	0.0	0.0	0.0	0.0	0.842	0.842	0.0	0.0	0.0	0.709	0.842

• •

A = P/PL

 $B = M_y / M_{yy}$  $C = M_z / M_{yz}$  $D = M_t / M_{yw}$ 



FIG. 8-1 COMPARISON BETWEEN THE BENDING AND WARPING STRESSES OBTAINED BY PASTOR AND DeWOLF (1979) AGAINST THOSE CONDUCTED BY AUTHOR'S APPROACH

-226-





BY ELASTIC ANALYSIS

-228-




COMPARISON BETWEEN THREE ANALYSES OF A BEAM -FIG. 8.4 COLUMN UNDER BIAXIAL BENDING AND INITIAL  $U_{\bullet} = \frac{L}{1000}$  (ELASTIC ANALYSIS ) DEFLECTION







FIG.8.7 LOAD ROTATION CURVE OF I-BEAM UNDER TORSIONAL MOMENT AT MIDSPAN, WITH OR WITHOUT INCLUSION OF RESIDUAL STRESS

-232-



(a) Inclusion of residual stresses Lehigh distribution



FIG. 8.8 TYPICAL SPREAD OF YIELD OF BEAM-COLUMN UNDER CONSTANT TORSION AT MID-SPAN

-233-



-234-









M<sub>+</sub> = 9-188 Kip.in



a =  $M_{y_1} = 400 \text{ Kip.in}, M_{y_2} = 0, \nabla_{\Gamma} = 0$ 









b\_  $M_{y_1} = 300 \text{ Kip.in}, M_{y_2} = 0$ ,  $\overline{\nabla_r} = 0$ 







c =  $M_{y_1} = 100 \text{ Kip. In}, M_{y_2} = 0, \overline{\nabla}_{\Gamma} = 0$ 

FIG. 8-11 TYPICAL SPREAD OF YIELD AT MID-SPAN 0**F** BEAM - COLUMN UNDER UNEQUAL END MOMENT TORSION, NO RESIDUAL STRESS INCLUDED AND







 $M_t = 3.48$  Kip.in

 $a - M_{y_1} = M_{y_2} = 400 \text{ Kip.in}, \nabla_{\Gamma} = 0$ 







b -  $M_{y_1} = M_{y_2} = 300 \text{ Kip. in } \nabla_{\Gamma} = 0$ 



 $c - M_{y_1} = M_{y_2} = 100 \text{ Kip.in}, \nabla_{\Gamma} = 0$ 

TYPICAL SPREAD OF YIELD (MID-SPAN) OF FIG. 8.12 BEAM-COLUMN SUBJECTED TO EQUAL CONSTANT END MOMENTS AND TORSIONAL MOMENT AT MID - SPAN





FIG. 8.13 COMPARISON OF AUTHOR'S RESULTS FOR LINEAR AND NONLINEAR ANALYSIS OF BEAM-COLUMN SUBJECTED TO COMPRESSIVE LOAD WITH ECCENTRICITY AND CONSTANT TORSION APPLIED AT MIDSPAN





FIG 8J5 COMPARISON BETWEEN LINEAR AND NON-LINEAR ANALYSIS OF BEAM-COLUMN UNDER BIAXIAL BENDING AND CONSTANT TORSION AT MIDSPAN



FIG. 8.16 TYPICAL SPREAD OF YIELD OF BEAM- COLUMN UNDER BIAXIAL BENDING AND TORSION

-



FIG. 8-17 TORQUE DIAGRAMS FOR PIN-ENDED BEAM (SALMON AND JOHNSON, 1980)



FIG. 8.18 INTERACTION CURVES FOR BEAM - COLUMNS OF I-BEAM SECTION (W12×14) UNDER FLEXURAL BENDING AND TORSION AT MID-SPAN

-243-

#### Chapter 9

## Conclusions and Further Work

## 9.1 Introduction

The aim of this thesis has been to study the full range behaviour of beam-columns in 3-dimensions. This has been achieved by developing a formulation and computer program based on the finite element method for analysing the behaviour of members of different types of cross-section under a variety of loading and boundary conditions.

This chapter contains a summary of the more important conclusions drawn from the findings presented at the end of each chapter and an indication of areas of future work which would augment the analytical study presented in the thesis.

# 9.2 General Formulation

A general formulation for beam-columns in 3-dimensions acted upon by a wide range of loading and provided with very general support conditions applicable to many kinds of cross-section has been derived. The derivation was obtained using two methods: the principle of virtual work and total potential energy. In both methods, the effect of several factors on the ultimate strength behaviour have been included. The initial crookedness is represented by either a sinusoidal form or more exactly by polynomial functions. The residual stress distributions were incorporated in the form of parabolic, triangular, Lehigh or any other patterns. These patterns were implemented with some modification in

- 244 -

order to satisfy the equilibrium conditions for three-dimensional response.

The main reason for formulating the governing equations by two separate methods was to provide a check on the large number of complicated terms which resulted due to coupling between the displacements and rotations.

## 9.3 Derivation of Stiffness Matrices

The element stiffness matrices and the strain-displacement matrices for a beam-column of thin-walled open cross-section in 3-dimensions were derived. The linear and nonlinear tangential and geometrical stiffness matrices correspond to.- an element with 14 D.O.F.(seven degrees of freedom at each node: axial, in-plane, and outof-plane displacements, three rotations about X, Y, and Z axes and warping). The interpolation functions employed were linear polynomials for the axial displacements, cubic polynomial for lateral displacements and twists, and quadratic polynomial for rotations and warping.

The terms in both the linear and nonlinear stiffness matrices were calculated by two different methods. The first is based on the average of the values of the sectorial/ sectional properties and internal forces for the two nodes, while the second uses a linear function to evaluate these values between the two nodes. The transformation matrix for the element displacements with respect to local coordinate and the global coordinate systems was developed in a sophisticated yet straightforward fashion.

- 245 -

## 9.4 Analysis Type Options

The inclusion of the higher terms in the formulations have been arranged to form at least three types of analyses; 'Linear' uses the linear tangential and geometrical stiffnesses together with linear strains, 'Nonlinear' employs linear tangential and geometrical stiffness matrices and the higher order strain-displacement relationships, and ' Full Nonlinear" which incorporates the contributions of both higher order strain terms and nonlinear stiffness matrices.

#### 9.5 Development of Computer Program (TDCP)

A three dimensional computer program (TDCP) which implements the formulation has been developed. TDCP is based on finite element techniques and is written in the Fortran 77 Language. The program contains many subroutines, the functions of which can be summarized as:

- 1- Calculating the sectional and sectorial properties in the elastic and inelastic ranges for complex cross-sections.
- 2- Calculating the internal strains and stress resultants.
- 3- Controlling convergence to the correct result based on the incremental displacements and out-of-balance forces.
- 4- Tracing the full load-deflection response and loss of stiffness due to yield spread.

#### 9.6 <u>Comparison with Previous Work</u>

The validity and versatility of both the rigorous formulation and TDCP have been tested against previously obtained experimental and theoretical results. Several examples covering flexure, flexuraltorsional action, biaxial bending, and bending and torsion have been

- 246 -

checked in the elastic and inelastic ranges. They contain a wide range of parameters e.g. different types of cross-section, loading, boundary conditions and initial geometrical imperfections. In all cases excellent agreement has been achieved when some limitations were imposed on the present formulation in order to match the more restricted approaches used elsewhere.

## 9.7 Ultimate Strength Behaviour of Members under Bending and Torsion

Preliminary results from a new investigation of the ultimate strength behaviour of beam-columns subjected to biaxial bending and torsion have been reported, problems in which direct torsional loading is present have received little attention previously.

Several problems of bending and torsion have been solved. The solutions were obtained by three types of analyses; 'Linear', 'Nonlinear', and Full Nonlinear. The results show the significance of the inclusion of the so called 'higher order terms' in the straindisplacement relations and the nonlinear geometrical and tangential stiffness matrices in the analysis.

Several important observations are possible:

- 1- The ability of the formulation presented to cover the full range of analysis of beam-columns in 3-dimensions.
- 2- The need to include higher order terms in the straindisplacement relations and stiffness matrices in order to avoid overestimates of load carrying capacity.
- 3- This analysis represents a powerful tool for the further study of problems in each of the classes previously identified.

- 247 -

#### 9.8 Future Work

The work undertaken in this thesis has established a general formulation and a program TDCP which can be used for determining the ultimate strength of beam-columns. This work can be easily extended to incorporate many aspects such as:

1- Semi-Rigid Connections

This work is now under investigation and the preliminary results are encouraging (Wang et al. 1987).

- 2. Other types of material such as Aluminium can be included. In such case the characteristic of the material is nonlinear; this extension is also in progress.
- 3\_ Analysis of Frames in Three-dimensions; with rigid and semi-
- 4- Considerations of requirements for members deforming in space to permit more rational strength/ stiffness requirements to be determined.

- Abedel-Sayed, G. and Aglan, A.A. (1973), "Inelastic Lateral Torsional Buckling of Beam-Columns", Publications, IABSE, Vol. 23-II, pp. 1-16.
- 2. Anslijn, R. (1983), "Tests on Steel I-Beam-Columns Subjected to Thrust and Biaxial Bending", Report MT 157, Centre de Recherches Scientifiques et Techniques de L'industrie des Fabrications Metalliques, Bruxelles, August.
- 3. Anderson, J.M., and Trahair, N.S., (1972), "Stability of Monosymmetric Beams and Cantilevers," Journal of the Structural division, ASCE, Vol. 98, No. ST1, Proc. Paper 8846, Jan. pp. 269-286.
- 4. Attard, M.M., (1985) "Lateral Buckling Analysis of Beams by the FEM", Computer and Structures Vol. 23, No. 2, Pergamon Press Ltd. Printed in U.K., April.
- Attard, M.M. (1986), "Nonlinear Theory of Non-uniform Torsion of Thin-Walled Beams", International Journal of Thin-Walled Structures, Vol. 4, No. 2, pp. 101-134.
- Barsoum, R.S., and Gauligher, R.H. (1970), "Finite Element Analysis of Torsional and Torsional-flexural Stability Problems", International Journal of Numerical Method in Engineering, Vol. 2, No. 1, pp. 335-352.
- 7. Bathe, K.J., and Wilson, E. L., (1976), "Numerical Methods in Finite Element Analysis", Prentice-Hall, Inc., Englewood Cliffs, New Jersey.
- 8. Bergan, G.P. and Clough, W.R., (1972), "Convergence Criteria for

- 249 -

Iterative Processes", AIAA Journal, Vol. 10, No. 8, pp. 1107-1108.

- Birnstiel, C. (1968), "Experiments on H-Columns under Biaxial Bending", Journal of the Structural Division, ASCE, Vol. 94, No. ST 10, Proc. Paper 6186, Oct., pp. 2429-2449.
- 10. Black, M.M., (1967a) "Nonlinear Behaviour of Thin-Walled Unsymmetrical Beam-Sections Subjected to Bending and Torsion", In. Thin-Walled Structures, Chatto and Winds, London.
- 11. Black, M.M. (1967b), "Nonlinear Behaviour of Thin-Walled Unsymmetrical Beam-Sections Subjected to Bending and Torsion", Thin-Walled Structures a Collection of Papers on the Stability and Strength of Thin- alled Structural Members and Frames, Chalto & Windus Ltd., London, U.K.
- Bleich, F. (1933), "Stahlhochbouten", Vol. II, Julius Springer, Berlin, P. 925.
- Bleich, F. and Bleich, H. (1936), "Bending Torsion and Buckling of Bars Composed of Thin-Walls", Preliminary Publication, 2nd Congress, IABSE, English edition, Berlin, P. 871.
- 14. Bleich, F., (1952), "Bucking Strength of Metal Structures ," McGraw-Hill Book Co. Inc., New YORK, N.Y.
- 15. Boresi, P. and Lynn, P.P., (1974), "Elasticity in Engineering Mechanics", Prentice-Hall, Inc., Englewood Cliffs, New Jersey, USA.
- 16. Brown, T.G., (1981) "lateral Torsional Buckling of Tapered I-beams," J. of the Strutrural division, ASCE, Vol. 107, No. ST4, Proc. Paper 16211, April, pp. 689-697.
- 17. Cescotto, S., Gilson, S., Plumir, A., and Rondal, J. (1983), " Etudedeu Du Fiambement Plan De Poutres Colonnes en Aciek a Section Nonsymetrique", J. of CRSTIFM, MT153, Bruxelles, May.

- 250 -

- Chaudhary, A.B., (1982), "Generalized Stiffness Matrix for Thin Walled Beams ", Journal of the Strutural Division, ASCE, VOL. 108, No. ST3, Proc. Paper 16938, March.
- 19. Chan, S.L. and Kitipornchai, S. (1986), "Geometric Nonlinear Thin-Walled Members", Research Report No. CE67, Dept. Civil Eng. Univ. of Queensland Australia, July.
- 20. Chen, W.F., (1977), "Theory of Beam-Columns The State-of-Art Review", Proceedings, International Colloquium on Stability of Structures Under Static and Dynamic Loads", SSRC/ ASCE, Washington D.C., March 17-19, pp. 631-648.
- 21. Chen, W.F., (1979), "Influence of End-Restrained Column Stability", Civil Engineering, Report No. CE - STr - 79-2, Purdue University, West Lafayette, Inc., August, (Preprint, ASCE Atlanta Convention, Oct., 22-26,1979).
- 22. Chen, W.F., (1981), "Recent Advances on Analysis and Design of steel Beam-Columns in USA", Inelastic Instability of Steel Strutrures and Strustural Elements, U. S. Japan, Seminar, Tokyo, edited by Fujita, Y. and Galambos, T.V., pp. 224-292.
- 23. Chen, W.F. and Atsuta, T. (1976), "Theory of Beam-Columns- Volume I: "In-Plane Behaviour and Design", McGraw-Hill, N.Y.
- 24. Chen, W.F., and Atsusta ,T. (1977), "Theory of Beam-Column", Volume 2 Space Behaviour and Design", Mc Graw-hill, N.Y.
- 25. Chen, W.F. and Cheong-Siat-Moy, F. (1980), "Limit States Design of Steel Beam-Columns - A state-of-Art Review", Journal of Solid Mechanics Archives, Noordhoff, Leyden, The Netherlands, March.
- 26. Chen, A.M. and Santathadaporn, S. (1968), "Review of Column Behaviour under Biaxial Loading", Journal of the Structural Division, ASCE, Proc. Paper 6316, Vol. 94, No ST12, Dec., pp.

- 251 -

2999-3017.

- 27. Chu, K. and Johnston R.B (1971), "Torsion in Beams with Open Sections", Journal of the Structural Division, ASCE, Vol. 105, No. St11, pp. 2163-2185.
- 28. Clark, J.W. and Hill, H.N. (1960), "Lateral Buckling of Beams", Journal of the Structural Division, ASCE, Vol. 86, No. ST7, Proc. Paper 2559, July, pp. 175-196.
- 29. Crisfield, (1981), "A Fast Incremental/ Iterative Solution Procedure that Handless Snap-Through", Journal of Computer & Structures, Vol. 13, Pergamon Press Ltd., U.K., pp. 55-65.
- 30. Cuk, P.E., (1984), "Flexural-Torsional Buckling in Frame Structutres", Phd Thesis Submitted for Fullfillment of Degree of Doctor of Philosophy, University of Syedney, School of Civil and Mining Engineering, Sydney, Feb.
- 31. Culver, C.G. (1966a), "Exact Solution of The Biaxially Bending Equations", Journal of the Structural Division, ASCE, Vol. 92, No. ST2, Proc. Paper 4772, April, pp. 63-83.
- 32. Culver. C.G (1966b), " Initial Imperfection in Biaxial Bending", Journal of the Structural Division, ASCE, Vol. 92, NO. ST3, Proc.
  Paper 4846, June, pp. 119-135.
- 33. Dinno,K.S. and Merchant, W. (1965), " A Procedure for Calculating the Plastic Collapse of I-Sections under Bending and Torsion", The Structural Engineer, Vol. 43, No. 7, pp. 219-221.
- 34. Draper and Smith (1967), "Applied Regression Techniques", John Wily and Son 2nd, Ed., London.
- 35. Dux ,P.F., and Kitipornchai, B.E. (1983), "Inelastic Beam Buckling Experiments", Journal of Constructional steel Research, Vol. 3, No. 1, pp. 3-9.

- 252 -

- 36. El-Khenfas, M.A., (1987a), "TDFE-Computer Program for Analysis of Beams in Space in the Elastic and Inelastic Ranges", Report, No. 1CE, Civil and Structural Engineering Department, Sheffield University.
- 37. El-Khenfas, M.A., (1987b), "Elastic Analysis of Beam-Columns in Space", Report, No. 2CE, Civil and Structural Engineering department, Sheffield University.
- 38. El-Khenfas, M.A., (1987c), "Inelastic Analysis of Beam-Column in 3-Dimensions", Report, No. 3CE, Civil and Structural Engineering Department, Sheffield University.
- 39. El-Khenfas, M.A., and Nethercot, D.A., (1987a), " A General Formulation for Three Dimensional Analysis of Beam-columns", (submitted for publication).
- 40. El-Khenfas, M.A. and Nethercot, D.A., (1987b), "Elastic Analysis of Beam-Columns in Space", (submitted for publication).
- 41. El-Khenfas, M.A. and Nethercot, D.A., (1987c)"Inelastic Analysis of Beam-Columns in Space", (submitted for publication).
- 42. Epstein, M. and Murray, D.W., (1976), "Three Dimensional Large Deformation Analysis of Thin-Walled Beams", Int. Journal of Solids Structures, Vol. 12, Pregamon Press, Printed in U. K.
- 43. Epstein, M. ,Nixon, D., and Murray, D. W., (1978), "Large Displacements Inelastic Analysis of Beam-Columns", Journal of the Strutural division, ASCE, Vol 104, No. ST5, Proc. Paper 13768, May, pp. 841-853.
- 44. Erwin, K. (1976), "Advanced Engineering Mathemathics", John Wiley& Sons, Fifth Edition, N.Y. (Printed in USA).
- 45. Fujita, Y and Yoshida, K. (1972), "Compressive Strength of Columns with Initial Deflections", International Colloquium on Columns

- 253 -

Strength, Paris, France, 23-24, November, pp. 108- 120.

- 46. Fukumoto, Y. and Kubo, M., (1977), "An Experimental Review of Lateral Buckling of Beams and Girders", Proceeding of the International Colloquium on Stability of Structure under Static and Dynamic Loads, SSRC-ASCE, Washington, D.C., Macrh, pp. 541-562.
- 47. Fukumoto, Y., Kubo, M. and Yokoyama, A., (1971), "Inelastic Lateral Buckling Strength of Monosymmetric I-Beams", Preprint,, Annual Meeting of JSCE, october, (in Japanese), pp. 161-162.
- 48. Fukumoto, Y. and Nishida ,S., (1981), "Ultimate Load Behaviour of Curved I-beams", Journal of Engineerimg Mechanics Division, ASCE , Vol. 107, April, pp. 367-385.
- 49. Galambos, T.V., (1963) "Inelastic Buckling of Beams", Journal of the Structural Duvision, Vol. 89, No. ST5, ASCE, Proc. Paper 3683, Oct., pp. 217-292.
- 50. Galambos, T.V., (1968), "Structural Members and Frames", Prentice-Hall Inc., England Cliffs, New Jersey.
- 51. Goodier, J.N., (1942), "Torsional and Flexural Buckling of Bars of Thin Walled Open Section Under Compressive and Bending Loads", Journal of AppliedMechanics, Transactions, ASME, Vol. 64, Sept., pp. A-103.
- 52. Goodier, J.N., (1956), "Flexural-Torsional Buckling of Bars of Open -Section Under Compressive and Bending Loads", Journal of Applied Mechanics, Transaction, ASCE, Vol. 64, Sept.
- 53. Hancock, K.J. (1983), "The Behaviour and Design Cold Formed Channels in Compression", Journal of the Australian Instute of teel construction, Vol. 17 No. 3, Atissin 004-2205.
- 54. Hancock, G.J. and Trahair. N. S. (1978), "Finite Element Analysis

- 254 -

of the Lateral Buckling of Continously Restrained Beam-Column", Civil Engineering Transactions, Intitution of Engineers, Australia, Vol. 20, No. 2, pp. 120-127.

- 55. Hancock, G.J. and Trahair, N.S.(1979), "Lateral Buckling of Roof Purlins with Diaphram Restraints", Civil Engineering Transaction Institution of Engineerrs, Australia, Vol. CE21, No. 1, pp. 10-15.
- 56. Hasegawa, A., Liyanage, K., Ikeda, T., and Nishino, F. (1985), "A Concise and Explicit Formulation of Out-Of-Plane Instability of Thin-Walled Members", Proc. of Japan Society of Civil Engineers, Div. of Structural Engineering/ Earthquake Engineering, Vol. 2, No. 1, April, pp. 81-89.
- 57. Heins, C.P. and Seaburg, P.A. (1963), "Torsional Analysis of Rolled Steel Sections", Bethlehem Steel Corporation, Bethlehem, Pennsylivania.
- 58. Hill, H.N. and Clark, J.W. (1951a), "Lateral Buckling of Eccentrically Loaded I-Section Columns", Transaction, ASCE, Vol. 116, pp. 1179-1191.
- 59. Hill, H.N. and Clark, J.W. (1951b), "Lateral Buckling of Eccentrically Loaded I and H-Sections columns ", Proceedings ,1st U.S, National Congress of Applied Mechanics, pp. 407-413.
- 60. Horne, M.R., (1954), "The Flexural-Torsion Buckling of Members of Symmetrical I-Section Under Combined Thrust and Unequal Terminal Moments", Quarterly Journal of Mechanics and Applined Mathematics, Vol. 7, Part 4, pp. 410-426.
- 61. Johnston , B.G., (1941), "Lateral Buckling of I-Section Columns with Eccentric End Loads in the Plane of the Web", Transactions, ASME, Vol. 63 .P.A-176.
- 62. Johnston, B.G. (ed.), (1976), "Guide to Design Criteria for

Structures", 3rd. edn., Strutural Stability Research Council, John Wiley and Sons, Inc., New Yourk, N.Y., 1976.

- 63. Johnston, B.G., (1982), " Design of W-shapes for Combined Bending and Torsion", Engineering Journal of the AISC, Vol. 19, 2nd quarter, pp. 65-85.
- 64. Johnston, B.G., Lin, F., and Galambos, T.V. (1980), "Basic Steel Design", Printice-Hall, Englewood Cliffs, New Jersy.
- Kappus, R. (1937), "Drillhnicken Zentrisch Gedruckter Stabe mit offenem Profile in Elastischen Bereich", Luftfahrt-Forschung, Vol. 14, No. 9, Translated in NACA Technical Memoir No. 851, 1938.
- 66. Kanok-Nukulchai, W. Hasegawa, A. and Nishino, F. (1986), "Generic Formulation Procedure for Large Deformation Analysis of Structural Elements", Structural Eng./ Earthquake Eng., Proc. JSCE, No. 368/ I-5, Vol. 3., No. 1, April, pp. 65-73.
- 67. Kennedy, J.B. and Madugula, M.K.S., (1982), "Buckling of Angles: State of Arte", Journal of the Structural Div., ASCE, Proc. Paper 17323, Vol. 108, No. ST9, Sept., pp. 1967-1980.
- Kennedy. J.B. and Murty, M.K.S., (1972), "Buckling of Steel Angle and Tee Struts", Journal of the Strutural Div., ACSE, Proc. Paper 9348, Vol. 98, No. St1, November, pp. 2507-2521.
- 69. Kerensky, A.O., Flint, A.R., and Brown, W.C. (1956), "The Basis for Design of Beams and Plate Girders in the Revised British Standard 153", Proceedings, ICE, Part 3, Vol. 5, Aug. pp. 396-444.
- 70. Kitipornchai, S. and Chan, S.L., (1987), "Non-Linear Finite ELement Analysis of Angles and Tee Beam-Columns", Journal of the Structural Engineering, ASCE, Vol. 113, No. 4, Proc. Paper 21416, April, 1987., pp. 721-739.
- 71. Kitipornchai, S., E., and Lee, H.W., (1984), "Inelastic Buckling of

- 256 -

Singe Angle, Tee, and Double Angle Struts", Research Report No. CE55, Univ. of Queensland, Civil Engineering Dept., Sept.

- 72. Kitipornchai, B., E., and Lee, H.W. (1986a), "Inelastic Buckling of Singe Angle, Tee, and Double Angle Strct.", Journal of Constructional SteeL Reseach, June, pp. 1-18.
- 73. Kitipornchai, S. and Lee, H.W., (1986b), "Inelastic Experiments on Angles and Tee Struts, Double-Angle Struts", Journal of Constructional Steel Research, 6, pp. 219-235.
- 74. Kitipornchi, S. and Trahair, N.S., (1975), "Inelastic Buckling of Simply Supported Steel I-Beams", Journal of the Structural Engineering, ASCE Vol. 101, No. ST7, Proc. Paper. 11419, July. pp. 133-1347.
- 75. Kitipornchi, S. and Trahair, N.S., (1980), "Buckling Problems of Monosymmetric I-Beams", Journal of the Structural Division, Vol. 106, No.ST5, ASCE, Proc. Paper No. 15406, May.
- 76. Kitipornchi, S. and Wang-Chung, M. (1985), "Stability of Monosymmetric Cantilevers", Research Report, Department of Civil Engineering, University of Queensland, No. C65, Dec.
- 77. Kitipornchai, S.E, Wang, C.M, and Trahair, N.S. (1985), "Buckling of Monosymmetric I-Beams under Moment Gradient", University of Queensland, Dept. of Civil Engineering, Research Report No. CE67, October.
- 78. Kollbrunner, C.F., Hajdin, N., and Branislavc,C., (1978), "Elastic-Plastic Thin-Walled I-section Beam Subjected to Bending and Warping Torsion," Institute for Engineering Research, Report 43, Zurich, May.
- 79. Kollbrunner, C.F., Hajdin, N., and Obradovic, P, (1979), "Elastic-Plastic Fixed Ended Beams of I-Section Subjected to Bending and

- 257 -

Warping Torsion", Institute for Engineering Research, Report 46, Zurich, December.

- 80. Lay, M.G. and Ward, R.F. (1969), "Residual stresses in Steel Structures", Australian Institute of Steel Construction, Vol. 3, pp. 2-21.
- 81. Lee, G.C. and Galambos, T.V. (1962), "The Post-Buckling Strength of Wide-Flange Beams", Journal of the Engineering Mechanics Division, ASCE, Vol. 88, No. EM1, Proc. Paper 3059, Feb., pp. 59-75.
- 82. Lindner, J., (1972), "Theoretical Investigation of Columns under Biaxial Bending", IABSE - CRC - ECCS, Colloquium on Column Strength, November.
- 83. Lindner, J., (1981), "BIEGEDRILLKNICKEN HOCHFESTER STAHLE", Technische Universitat Berlin, Institut fur Baukoustruktionen und Festigkeit, Bericht Nr. VR 2025, Sept.
- 84. Love, A.E.H., (1944), "Atreatise on the Mathematical Theory of Tlasticty", 4th ed. Dover Publications, New York.
- 85. Massonnet, C., (1947) "Le Flamblage des Barres a Section Ouverte et a Parios Minces ", Hommage Fac. SCI .Appl. a L'Assoc. des Ingeniews Sortis de L'Ecole de Liege ,Liege , pp. 35-146.
- 86. Massonet, C.E. (1976) ,"Forty Years of Research on Beam-Columns in Steel ",SM Archives, Vol. 1, Issue 1, Noordhoff International leyden, pp. 27-157.
- 87. Matthey, P. (1984), "Simulation Numerique du Comportement des Poutres-Columnnes Metalliques", Rapport ICOM137, Ecole Polytechnique Federal de Lausanne, Department de Genie Civil, November.
- 88. Medwodowski, S.J., (1985), "Warping Moment Distribution", J. of the

- 258 -

Struc. Eng., Vol. 111, No. 2, ASCE, Proc. Paper No. 19507, Februray.

- 89. Moore, D.B, (1986) "A Nonlinear Theory for the Behaviour of Thin-Walled Sections Subjected to Combined Bending and Torsion", Journal od Thin-Walled Structures, Vol. 4, No. 6, U.K., pp. 449-466.
- 90. NaKasheima, M., Nakumura, T. and Wakabayashi, M., (1983), "Post-Buckling Instability of Steel Beam-Columns", Journal of the Structural Division, ASCE, Vol. 109, No. ST6, Proc. Paper 18068, pp. 1414-1430.
- 91. Nethercot, D.A. (1972), "The Effect of Load Position on the Lateral Stability of Beams", Proceedings, Conference on Finite Element Methods in Civil Engineering Institute of Canada, Monteral, June, p. 347.
- 92. Nethercot, D.A. (1973a), "Inelastic Buckling of Monosymmetric I-Beams", Journal of the Structural Division, ASCE, Vol. 99, No. ST 7, Technical Note, July, pp. 1696-1701.
- 93. Nethercot, D.A. (1973b), "The Solution of the Inelastic Lateral Stability Problems by Finite Element Methods", Proceedings, 4th Australian Conference on Mechanics of Structures and Materials, Brisbane, Australia, pp. 183-190.
- 94. Nethercot, D.A., (1974), "Residual Stresses and Their Influence Upon the Lateral Buckling of Rolled Steel Beams", The Structural
  - Engineer, Vol. 52, No. 3, March, pp. 89-96.
- 95. Nethercot, D.A. (1983), "Evaluation of Interaction for use in Design Specification in Western Europe", Proceedings 3rd. International Colloquium on Stability of Metal Structures, Canada, pp. 293-314.

- 96. Nethercot, D.A., (1986), "Lateral Buckling", 2nd Regional Colloquium on Stability of Steel Structures, Hungary, Sept. 25-26, pp. 99-117.
- 97. Nethercot, D.A. and Rockey, K.C. (1971), "Finite Element Solution for the Buckling of Columns and Beams", International Journal of Mechanical Sciences, Vol. 13, pp. 945-949.
- 98. Nethercot. D.A. and Trahair, N.S., (1983), "Design of Latterally unspported Beams", Beams and Beam-Columns, Stability and Strength, edited by R. Narayanan, Applied Science Publishers, London, pp. 71-94.
- 99. Neville, A.M. and Kennedy, J.B.(1964), "Basic Statistical Methods for Engineers and Scientists", Scranton, International Textbook Co., London, U.K.
- 100. Nishida, S. and Fukumoto ,Y., (1985), "Ultimate load Behaviour of Beams with Initial Imperfection", Journal of Structural Engineering, ASCE, Vol. 111, No. 6, Proc. Paper 19811, June, pp. 1288-1305.
- 101. Nylander, H. (1949), "Torsional and Lateral Buckling of Eccentrically Compressed I and T columns", Transactions, Royal Institute of Technology, Stockholm, Sweden.
- 102. O'Connor, C. (1964), "The Buckling of A Monosymmetric Beam Loaded in the Plane of Symmetry", Australian Journal of Applied Science, East Melbourne, Australia, Vol. 15, No. 4, Dec., pp 191-203.
- 103. Ohga, M. and Hara, T. (1986) ,"Analysis of Thin-Walled Members by Finite Element-Transfer Matrix Method," Structural Engineering/ Earthquake Eng. Vol. 3, No. 1, Japan Society of Civil Eng., Proc. of JSCE No. 368/ I-5, April.
- 104. Opperman, G. H. P., (1983) "Refined Beam-Column Analysis",

- 260 -

Proceedings, 3rd. International Colloquium on Stability of Metal Structures, Paris, Nov, pp. 39-47.

- 105. Pastor, P.T. and DeWolf, T.J, (1979), "Beams with Torsional and Flexural Loads", Journal of the Structural Division, ASCE, Vol. 105. No. ST 3, Proc. paper 14461, March, pp. 527-535.
- 106. Peano and Riccioni, R. (1978), "Automated Discerisation Error Gontrol in Finite Element Analysis in the Comrical Environment", Edited by J. Robinson, Robinson & Verwood, England, pp. 368-387.
- 107. Pekoz, T. and Winter, G. (1966), "Torsional Flexural Buckling of Thin-Walled Open Sections under Eccentric Axial Loading", Paper Submitted to ASCE Annual Meeting and Trans-porttation Engineering Conference, Philadelphi, Pennsylvania.
- 108. Petterrsson, O. (1952), "Combined Bending and Torsion of I-Beams of Monosymmetical Cross-section", Bulletin No. 10, Division of Building Statics and Structural Engineering, Royal Institute of Technology, Stockholm.
- 109. Premieniecki, J.S. (1968), "Theory of Matrix Structural Analysis", McGraw-Hill, N.Y.
- 110. Sakimoto, T., Yamo, T., Kikuchi, R. and Sakata, T., (1985), "Nonlinear Analysis of Thin-Walled Frames and Members with Arbitrary Open Cross-Section", Structural Engineering/ Earthquake Engineering, Vol. 2, No. 2, PP 139-147.
- 111. Salmon, C.G. and Johnson, J.E., (1980), "Steel Structures" 2nd edition, Harper and Row, Publishers, N.Y. 946 p.
- 112. Salvadori, M.G., (1955), "Lateral Buckling of I-Beams", Transaction ASCE, Vol. 120, Paper 2773.
- 113. Salvadori, M.G., (1956), "Lateral Buckling of Eccentrically Loaded I-Columns", Transactions, ASCE, Vol. 121, Paper 2836, pp. 1163-

- 261 -

1178.

- 114. Soltis, A.L. and Christiano, P. (1972), "Finite Deformation of Biaxially Loaded Columns", Journal of the Structural Division, ASCE, Proc. Paper 9407, Vol. 98, No. ST12, Dec., pp. 2647-2661.
- 115. "Steel Structures for Buildings", CSA Standard S16, Canadian Standards Association, Rexdale, Ontario, Canada, 1969.
- 116. "Specification for the Design, and Fabrication and Erection of Strutural Stell for Buildings", American Institute of Steel Constrction, New Yourk, N. Y., 1969.
- 117. Sourchnikoff, B. (1951), "Strength of I-Beams in Combined Bending and Torsion", ASCE, Transactions, Vol. 116, pp. 1319-1342.
- 118. Syal, C.I. and Sharma, S.S., (1971), "Biaxially Loaded Beam-Column Analysis", Journal of the Structural Division, ASCE, Proc. paper. 8384, Vol. 97., No. ST 9, Sept., pp. 2245-2259.
- 119. Rajasekaran , S. (1971) ,"Finite Element Analysis of Thin Walled Members of Open Sections," Structural Engineering Report No.34, Department of Civil Engineering ,University of Alberta Edmonton, Alberta, Canada, Sept.
- 120. Rajasekaran, S. and Murray ,D.W. (1973), "Finite Element Solution of Inelasstic Beam Equations", Journal of the Structural Divison, ASCE, Vol. 99, No. ST6, Proc. paper 9773, June, pp. 1024-1042.
- 121. Ramm, E. and Osterrieder, P. (1983), "Ultimate Load Analysis of Three- Dimensional Beam Structures with Thin-Walled Cross-Section Using Finite Elements", Stability of Metal Structures, Paris 16-17 Nov., primaray report.
- 122. Rao, S.S., (1982), "The Finite Element Method in Engineering", Pergamon Press Ltd. U.K.
- 123. Razzaq, Z. (1974), "Theoretical and Experimental Study of

- 262 -

Biaxially Loaded Thin-Walled Beams of Open Section with and without Torsion", A Dissertation Presented to the Sever Insitute of Wasington Univ. Inpartial Fullfillment of D. Sc..

- 124. Razzaq, Z. and Galambos, T.V. (1979), "Biaxial Bending Tests with or without Torsion", Journal of the Structural Division, ASCE, Vol. 105, No. St11, pp. 2163-2185.
- 125. Richards, T.H. (1977), "Energy Methods in Stress Analysis", Ellis Horwood limited, coll House, Westergate, Chichester, Sussex, England.
- 126. Roberts, T.M. (1981), "Second Order Strains and Instablity of Thin -Walled Bars of Open Cross-Section", Int. Journal Mech. Sciences, Vol. 23, pp 297-306.
- 127. Roberts, T.M., and Azizian, Z.G., (1983a), "Nonlinear Analysis of Thin-Walled Bars", International Journal of Mechanical Sciences, Vol. 25, No. 8, pp. 543-612.
- 128. Roberts, T.M. and Azizian, Z.G., (1983b), "Influence of Pre-Buckling Displacements on the Elastic Critical Loads of Thin-Walled Bars of Open Cross-section", International Journal of Mechanical Sciences, Vol. 25, No. 2, pp. 93-104.
- 129. Tebedge, N. and Tall, L. (1973), "Linear Stability Analysis of Beam-Columns", Journal of the Structural Division, ASCE, Vol. 99, No. ST12, Proc. paper 10232, Dec., pp. 2439-2457.
- 130. Tezcan, S.S. and Mahapatra, B.C., (1969), "Tangent Stiffness Matrix for Space Frame Mebers", Journal of the Structural Division, ASCE, Vol. 95, No. ST6, Proc. Paper. 6627, pp. 1257-1270.
- 131. Terrington, J.S., (1968) ,"Comined Bending and Torsion of Beams", BCSA Publication No. 31, (First part).

- 263 -

- 132. Terrington, J.S., (1970), "Comined Bending and Torsion of Beams", BCSA Publication No. 31, (Second part).
- 133. Thomas, F.B., Lehigh, and Lay, G.M., (1972), "The Behaviour of Laterally Unsupported Angles", The Institution of Engineers, Civil Engineering Transaction, Australia, pp. 103-110.
- 134. Timoshenko, S.P. (1910), "Einige Stabilitats- Probleme der Elasticitats Theories", Zeitscrift fur Mathematic und Physik, Vol. 58, pp. 337-385,. Also in "Collected papers of Stephen P. Timoshenko", McGraw-Hill, N.Y. 1953.
- 135. Timoshenko, S., (1945), "Theory of Bending of Torsion and Buckling of Thin Walled of Open Cross Section ," Journal ,Franklin Inst., Philadelphia, March, April, May.
- 136. Timoshenko, S.P. (1983), "History of Strength of Materials", Dover Publications, New York, 452 p.
- 137. Timoshenko, S.P. and Gere, J.M. (1961), "Theory of Elastic Stability", 2nd ed., McGraw-Hill, New York.
- 138. Trahair, N.S., (1961), "The Ultimate Load Carrying Capacity of Slender Stanchions Bent About the Major Axis", Proceedings, ICE, Vol. 19, August, pp. 479-507.
- 139. Trahair, N.S., (1966), "Elastic Stability of I-Beam Elements in Rigid Jointed Structures", Journal of the Institution of Engineers, Australia, Vol. 38, No. 7-8, July-Aug., pp. 171-180.
- 140. Trahair, N.S., (1969), "Restrained Elastic Beam-Column", Journal of Structural Division, ASCE, Vol. 95, No. ST12, Dec., pp. 2641-2664.

1\_\_\_\_

141. Trahair, S.N. and Kitipornchai, S. (1972), "Buckling of Inelastic I-beams under Uniform Moment", Journal of the Structural Division, ASCE, Vol. 98, No. ST11, Proc. Paper 9339, November, pp. 2551-

- 264 -
2565.

- 142. Vinnakota, N. S. (1977a), "Inelastic Stability of Laterally Unsupported I-Beams", Computer and Structures, Vol. 7, No. 3, June, pp. 377-389.
- 143. Vinnakota, S., (1977b), "Finite Difference Method for Plastic Beam-Column", Theory of Beam Column in Space Behaviour and Design, Theory of Beam-Columns, Vol. 2, Chap. 10, W.F., Chen and T., Atsuta. eds, McGraw-hill, N.Y...
- 144. Vlasov, V. Z., (1961), "Thin Walled Elastic Beams", 2nd Ed., National Science Foundation Washington, D.C.
- 145. Ugural, A.C. and Fensher, S.K., (1979), "Advanced Strength and Applied Elasticity", Elsevier North Holland Publishing Co., New York, 433. pp.
- 146. Wagner, H. (1929), "Verarehung und Knickung von offenen Profilen", 25th. Anniversary Publication, Technische Hochschule, Danzig, 1904-1929. Translated in N.A.C.A., Technical Memoir, No. 807, 1936.
- 147. Wang, Y.C., El-Khenfas, M.A., And Nethercot, D.A. (1987), "Lateral- Torsional Buckling of Ren-Restrained Beams", Journal of Construction Steel Research, (in press).
- 148. Warnick, W.L. and Walston, W.H. Jr. (1980), "Secondary Deflections and Lateral Stability of Beams", Journal of the Engineering Mechanics Division, Proceedings of the American Society of Civil Engineering, Vol. 106, No. EM6, Proc. paper 15927, December. pp. 1307 -1325.
- 149. Weaver, W., Jr., (1967), " Computer Programs for Structural Analysis", D. Van Nostrand Company, Inc., Canada.

150. Wekezer, J.W., (1985), "Nonlinear Torsion of Thin-walled Bars of

- 265 -

Variable Cross-section ", Int. J. Mech., Sci Vol. 27, No. 10, Printed in Great Britain, pp. 631-641.

- 151. Wilhoite, G., Zandonini, R., and Zavelalani, A. (1984a), "Behavior and Strength of Angkes in Compression: An Experimental Inevestigation", ASCE Annual Convention and Structures Congress, San Francisico, U.S., October.
- 152. Wilhoite, G., Zandonini, R., and Zavelalani, A. (1984b), " Behavior and Strength of T-Shapes in Compression: An Experimental Inevestigation", ASCE Annuaal Convention and Structures Congress, San Francisico, U.S., October.
- 153. Winter, G., (1941), "Lateral Stability of Unsymmetrical I-Beams and Trusses in bending", Proceedings, ASCE, Vol. 67, No. 10, Dec., pp 1851-1864.
- 154. Yang Y.B., and Mc Guire, W., (1986), "Stiffness Matrix for Geometric Nonlinear Analysis", Journal of Structural Engineering, ASCE, Vol. 112, No. 4, Proc. Paper No. 20561, April, pp. 853-877.
- 155. Young, B.W., (1975) " Residual Stress in Hot-Rolled Section", Proceedings, International Colloquium on Column Strength, IABSE, Vol. 23, pp. 25-38.
- 156. Young, B. W., (1971), "Residual Stresses in Hot-rolled Sections", Department of Engineering, University of Cambridge, CUED/C-Struct. /TR. 8.
- 157. Yoshida, H. and Maegawa, K., (1984), "Lateral Instability of I-Beams with Imperfections", Journal of Structural Engineering, Vol.110, No. 8, ASCE, Proc. Paper 19073, August.
- 158. Zbirohowski-koscia, K.(1967), "Thin Walled Beams from Theory to Practice", Crosby Lockwood & Son LTD, U. K.
- 159. Zienkiewciez, O. C. (1977) ," The finite Element Method ", Third Ed., McGraw-Hill Book Company, U. K.

## Appendix A

The equilibrium equation for a member having an arbitrary cross-section can be obtained by substituting equations 3.1, 3.2, 3.24, and 3.25 into equation 3.38. It can be converted to a finite element procedure by integrating over the area of the section and making use of the definition of the stress resultants.

Recall again equation 3.38

$$\int_{V} [E \epsilon_{xx} \delta W_{,x} + \frac{1}{2} \sigma_{xx} \delta [W_{,x}^{2} + V_{,x}^{2} + U_{,x}^{2}] dV$$

$$\int_{V} [G(W_{,y} + U_{,x}) \delta (W_{y} + U_{,x}) + (W_{,z} + V_{,x}) \delta (W_{,z} + V_{,x})] dV$$

$$+ \frac{1}{2} \int_{V} \sigma_{xy} \delta [W_{,x} W_{,y} + V_{,x} V_{,y} + U_{,x} U_{y}] dV$$

$$+ \frac{1}{2} \int_{V} \sigma_{xz} \delta [W_{,x} W_{,z} + V_{,x} V_{,z} + U_{x} U_{z}] dV = \int_{S} F_{1} \delta U_{1} ds$$

$$V$$

$$which can also be written as$$

$$I_{1} + I_{2} + I_{3} + I_{4} + I_{5} + - I_{6} = 0.$$

$$(A-1)$$

$$where$$

$$I_{1} = \int_{V} [E \epsilon_{xx} \delta W_{,x} dV$$

$$(A-2a)$$

$$I_{2} = \int [G\{W, y^{+}U, x\}\delta(W_{y}^{+}U, x) + \{W, z^{+}V, x\}\delta(W, z^{+}V, x)]dV + (A_{2b})$$

$$V$$

$$I_{3} = +\frac{1}{2} \int \sigma_{xx} \delta \{W_{,x}^{2} + V_{,x}^{2} + U_{,x}^{2}\} dV \qquad (A_{2}c)$$

•

•

$$I_{4} = \frac{1}{2} \int_{V} \sigma_{xy} \delta[W_{,x}W_{,y} + V_{,x}V_{,y} + U_{,x}U_{,y}] dV \qquad (A_2d)$$

$$I_{5} = \frac{1}{2} \int \sigma_{xz} \delta[W, x^{W}, z^{+V}, x^{V}, z^{+U}, x^{U}, z] dV \qquad (A_{2e})$$

$$V$$

$$I_{6} = \frac{1}{2} \int F_{i} \delta U_{i} ds \qquad (A_{2}f)$$

v Substitute equations 3.1, 3.2 and 3.24 into equation A\_2a to yield

$$\int_{0}^{L} \left[ [EA\{W_{,x} + \frac{1}{2}(U_{,x}^{2} + V_{,x}^{2} + \rho^{2}\phi_{,x}^{2}) - \overline{Y}(V_{,xx}\phi + U_{,xx}) - \overline{Z}(V_{,xx}-U_{,xx}\phi) - \overline{Y}(V_{,xx}\phi + U_{,xx}) - \overline{Z}(V_{,xx}-U_{,xx}\phi) - Y_{s}\phi_{,x} + Z_{s}V_{,x}\phi_{,x} - \frac{ES_{\omega}}{EA}\}\delta W_{,x}$$

$$- Y_{s}\phi_{,x} + Z_{s}V_{,x}\phi_{,x} - \frac{ES_{\omega}}{EA}\}\delta W_{,x}$$

$$- EA\overline{Y}[W_{,x}\delta(V_{,xx}\phi + U_{,xx}) + \{\frac{1}{2}(U_{,x}^{2} + V_{,x}^{2}) + Y_{s}\phi_{,x}V_{,x} - Z_{s}\phi_{,x}U_{,x}]]\delta U_{,xx}$$

$$+ W_{,x}\delta(V_{,x}\phi_{,x}) + \frac{1}{2}(Y_{s}^{2} + Z_{s}^{2})\phi_{,x}^{2}\delta U_{,xx}]$$

$$- EA\overline{Z}[W_{,x}\delta(V_{,xx}-U_{,xx}\phi) + \{\frac{1}{2}(U_{,x}^{2} + V_{,x}^{2}) + Y_{s}\phi_{,x}V_{,x} - Z_{s}\phi_{,x}U_{,x}]\}\delta V_{,xx}$$

$$+ Y_{s}\phi_{,x}V_{,x} - Z_{s}\phi_{,x}U_{,x}]\delta V_{,xx}$$

$$- W_{,x}\delta(U_{,x}\phi_{,x}) + \frac{1}{2}(Y_{s}^{2} + Z_{s}^{2})\phi_{,x}^{2}\delta V_{,xx}]$$

$$+ EI_{\omega}\phi_{,xx}\delta\phi_{,xx} + EI_{y}[V_{,xx}\delta(V_{,xx}-U_{,x}\phi_{,x})]\delta V_{,xx}$$

•

$$- \delta^{(U}_{,xx}V_{,xx}\phi) - Z_{g}\phi_{,x}^{2}\delta^{V}_{,xx}]$$

$$+ EI_{z}[U_{,xx}\delta^{(U}_{,xx}+V_{,x}\phi_{,x}) - \delta^{(U}_{,xx}V_{,xx}\phi) - Y_{g}\phi_{,x}^{2}\delta^{U}_{,xx}]$$

$$+ EI_{yz}[\delta^{(U}_{,xx}V_{,xx} - U_{,xx}\phi_{,xx} + V_{,xx}\phi_{,xx})$$

$$+ V_{,xx}\delta^{(\phi}_{,x}V_{,x})$$

$$- U_{,xx}\delta^{(\phi}_{,x}U_{,x}) - \phi_{,x}^{2}(Z_{g}\delta^{U}_{,xx} + Y_{g}\delta^{V}_{,xx})]$$

$$+ EI_{wy}[\delta^{(\phi}_{,xx}(V_{,xx}\phi + U_{,xx})]$$

$$+ \phi_{,xx}\delta^{(\phi}_{,x}V_{,x}) - Y_{g}\phi_{,x}^{2}\delta^{\phi}_{,xx}]$$

$$+ EI_{wz}[\delta^{(\phi}_{,xx}(V_{,xx} - U_{,xx}\phi)]$$

$$- \phi_{,xx}\delta^{(\phi}_{,x}U_{,x}) - Z_{g}\phi_{,x}^{2}\delta^{\phi}_{,xx}]$$

$$- ES_{\omega}[W_{,x} + \frac{1}{2}[U_{,x}^{2} + V_{,x}^{2}] + Y_{g}\phi_{,x}V_{,x} - Z_{g}\phi_{,x}U_{,x}$$

$$+ \frac{1}{2}(Y_{g}^{2} + Z_{g}^{2})]\delta\phi_{,xx}$$

$$(A_{3})$$

where

.

$$\rho^{2} = (Y - Y_{s})^{2} + (Z - Z_{s})^{2} \qquad (A_{4a})$$

$$H_{y} = \frac{1}{2} \int_{A}^{A} Y(Y^{2} + Z^{2}) dA$$
 (A\_4b)

$$H_{z} = \frac{1}{2} \int_{A} Z(Y^{2} + Z^{2}) dA$$
 (A\_4c)

.

•

$$H_{y} = \frac{1}{2A} \int_{A} Y_{\omega}(Y^{2}+Z^{2}) dA \qquad (A_{4}d)$$

which may be written as:

$$I_{2} = \int_{0}^{1} Gk \Phi_{,x} \delta \Phi_{,x} dx \qquad (A_{5})$$

$$K = \int_{A} \left[ (Z-Z_s)^2 + (Y-Y_s)^2 - 2(Y-Y_s) \frac{\partial \overline{\omega}}{\partial Z} - 2(Z-Z_s) \frac{\partial \overline{\omega}}{\partial y} + (\frac{\partial \overline{\omega}}{\partial y})^2 + (\frac{\partial \omega}{\partial z})^2 \right]$$
(A\_6)

in which K has been defined (Love 1944 and Nishida & Fukumoto 1985) as the torsional rigidity

$$I_{3} = \int_{0}^{1} F_{x}[U_{,x}\delta U_{,x}+V_{,x}\delta V_{,x}+\rho^{2}\Phi_{,x}\delta\Phi_{,x} + \int_{0}^{1} [Z_{s}\delta\{\Phi_{,x}(\Phi V_{,x}+U_{,x})\} - Y_{s}\delta\{\Phi_{,x}(V_{,x}-U_{,x}\Phi)\}]dx$$

$$= \int_{0}^{1} [M_{y}\delta\{\Phi_{,x}(V_{,x}\Phi+U_{,x})\} + M_{z}\delta\{\Phi_{,x}(V_{,x}-\Phi U_{,x})\}]dx \qquad (A_7)$$
In the same manner  $I_{4} + I_{5}$  can be written as
$$I_{4}+I_{5} = \int_{0}^{1} Q_{y}[\delta\{V_{,x}\Phi-W_{,x}(V_{,x}\Phi+U_{,x})]dx + I_{5}C(\Phi_{,x}(V_{,x}\Phi+U_{,x})]dx + I_{5}C(\Phi_{,x}(V_{,x}\Phi+U_{,x})]dx$$

•

•

$$\int_{0}^{1} [M_{x} \delta \{V_{xx} (V_{x} \Phi + U_{x}) + (U_{x} \Phi_{x} - U_{x}) U_{x} \}] dx$$

$$\int_{0}^{1} \int_{0}^{1} [M_{x} \delta \{V_{x} (V_{xx} \Phi + V_{x} \Phi_{x}) + (U_{x} \Phi + V_{x}) U_{xx} \}] dx + A$$

$$(A_8)$$

$$0$$
where
$$1$$

$$A = \int_{0}^{1} [\sigma_{xz} \frac{d\overline{\omega}_{n}}{dy} + \sigma_{xy} \frac{d\overline{\omega}_{n}}{dy}] \Phi_{,x} \overline{X} dx \qquad (A_9)$$

The above equation may be shown, after some manipulation to be equal to zero.

where

$$\overline{X} = \delta[\Phi_{,xx}(-W_{,x}+(Y-Y_{s}) \{V_{,xx}\Phi+U_{,xx}+\Phi_{,x}V_{,x}\} + (Z-Z_{s})\{V_{,xx}-U_{,xx}\Phi-\Phi_{,x}U_{,x}+\overline{\omega}\Phi_{,xx}\}]$$

# Appendix B

The equilibrium equation for a member having an arbitrary cross-section can be obtained by substituting equations 3.1, 3.2, 3.24, and 3.25 into equation 3.38 It can be converted to a finite element procedure by integrating over the area of the section, making use of the definition of the stress resultants. Neglecting  $W_{,x}W_{,y}$ ,  $W_{,x}W_{,z}$ , and  $W_{,x}^2$  from equation 3.38 yields:

$$\int_{V} \sigma_{xx} \delta[W_{,x} + \frac{1}{2}(U_{,x}^{2} + V_{,x}^{2})] dV$$

$$\frac{1}{2} \int_{V} \sigma_{xy} \delta[(W_{,y} + U_{,x}) + (W_{,z} + V_{,x})] dV$$

$$\frac{1}{2} \int_{V} [\sigma_{xz} \delta \{U_{,x}U_{,y} + V_{,x}V_{,y}\} + \sigma_{xy} \delta \{U_{,x}U_{,z} + V_{,x}V_{,z}\}] dV$$

$$+ \int_{S} F_{i} \delta U_{i} ds = 0 \qquad (B_{1})$$
Equation B\_1 can be written as :

$$I_1 + I_3 + I_3 + I_4 = 0$$
 (B\_2)

where

$$I_{1} = \int_{V} \sigma_{xx} \delta \left[ U_{,x} + \frac{1}{2} \left( U_{,2}^{2} + V_{,x}^{2} \right) \right] dV \qquad (B_{3a})$$

$$I_{2} = \frac{1}{2} \int [\sigma_{xy} \delta \{ W_{,x} + U_{,x} \} + \{ W_{,z} + V_{,x} \} ] dV \qquad (B_{3b})$$

$$I_{3} = \frac{1}{2} \int_{V} [\sigma_{xy} \delta \{U, x^{U}, y\}^{+}$$

$$+ \sigma_{xy} \delta \{U, x^{U}, z + V, x^{V}, z\}] dV \qquad (B_{3}c)$$

$$I_{4} = - \int F_{1} U_{1} ds \qquad (B_{3}d)$$

•

•

s Substituting equation 3.1, 3.2 and 3.24 into equation B\_3a yields:

$$I_{1} = \int_{V} \sigma_{xx} [\delta\{W_{,x} + \frac{1}{2}(U_{,x}^{2} + V_{,x}^{2} + \rho^{2}\phi_{,x}^{2})]$$

$$-Y(V_{,xx}\phi + U_{,xx}\frac{\phi^{2}}{2}) + Z(U_{,xx}\phi + V_{,xx}\frac{\phi^{2}}{2})]$$

$$-Y\delta U_{,xx} - ZV_{,xx} - \overline{\omega}\delta\phi_{,xx}$$

$$+\delta\{Z_{s}\phi_{,x}(V_{,x}\phi + U_{,x}) - Y_{s}\phi_{,x}(V_{,x} - U_{,x}\phi)\}]dV$$

$$= \int_{V} \sigma_{,xx}[\delta W_{,x} - Y\delta U_{,xx} - Z\delta V_{,xx} - \overline{\omega}\delta\phi_{,xx}]dV$$

$$+ \int_{0} F[U_{,x}\delta U_{,x} + V_{,x}\delta V_{,x} - Y_{s}\delta\{\phi_{,x}(V_{,x} - U_{,x}\phi)\}]$$

$$+ Z_{s}\delta\{\phi_{,x}(V_{,x}\phi + U_{,x})\} + M_{p}\phi_{,x}\delta\phi_{,x}$$

$$- M_{z}\delta\{V_{,xx}\phi - U_{,xx}\frac{\phi^{2}}{2}\} + M_{y}\delta\{U_{,xx} + V_{,xx}\frac{\phi^{2}}{2}\}]dx$$
The above equation can be written after substituting for  $\sigma_{,xx}$  as:

$$I_{1} = \int_{0}^{L} \left[ EA[\{W_{,x} + \frac{1}{2}(U_{,x}^{2} + V_{,x}^{2} + \rho^{2} \Phi_{,x}^{2}\} \right]$$

$$-EA\overline{Y}(V_{,xx}\phi+U_{,xx})+EA\overline{Z}(U_{,xx}\phi-V_{,xx})$$

$$-Y_{s}EA\phi_{,x}V_{,x} + Z_{s}\phi_{,x}U_{,x}-ES_{\omega}\phi_{,x}]\deltaW_{,x}$$

$$= [EA\overline{Y}\{W_{,x}+\frac{1}{2}(U_{,x}^{2}+V_{,x}^{2})-Y_{s}\phi_{,x}V_{,x}+Z_{z}\phi_{,x}U_{,x}]\deltaU_{,xx}$$

$$= [EA\overline{Z}\{W_{,x}+\frac{1}{2}(U_{,x}^{2}+V_{,x}^{2})-Y_{s}\phi_{,x}V_{,x}+Z_{z}\phi_{,x}U_{,x}]\deltaV_{,xx}$$

$$= [ES_{\omega}\{W_{,x}+\frac{1}{2}(U_{,x}^{2}+V_{,x}^{2})-Y_{s}\phi_{,x}V_{,x}+Z_{z}\phi_{,x}U_{,x}]\delta\phi_{,xx}$$

$$+EI_{z}\{V_{,xx}\phi+U_{,xx}\}\deltaU_{,xx} + EI_{y}\{U_{,xx}\phi-V_{,xx}\}\deltaV_{,xx}$$

$$+EI_{\omega}\phi_{,xx}\delta\phi_{,xx} + EI_{\omegay}\{\phi_{,xx}\deltaU_{,xx}+(V_{,xx}\phi+U_{,xx})\delta\phi_{,xx}\}$$

$$+ EI_{\omega z}\{\phi_{,xx}\deltaV_{,xx}+(U_{,xx}\phi+V_{,xx})\delta\phi_{,xx}\}$$

$$+ EI_{\omega z}\{\phi_{,xx}\deltaV_{,xx}+(U_{,xx}\phi+V_{,xx})\delta\phi_{,xx}\}$$

$$+ [EAH_{y}+EA\frac{\overline{Y}}{2}[Z_{s}^{2}+Y_{s}^{2}]-Y_{s}EI_{z}-Z_{s}EI_{yz}]\phi_{,x}^{2}\deltaU_{,xx}$$

$$+ [EAH_{\omega}+\frac{ES}{2}\{Z_{s}^{2}+Y_{s}^{2}]-Y_{s}EI_{y\omega}-Z_{s}EI_{\omega}]\phi_{,x}^{2}\delta\phi_{,xx}$$

$$F[U_{,x}\deltaU_{,x}+V_{,x}\deltaV_{,x}-Y_{s}\delta\{\phi_{,x}(V_{,x}-U_{,x}\phi)\}$$

$$Z_{s}\delta\{\phi_{,x}(V_{,x}\phi+U_{,x})\}]+M_{p}\phi_{,x}\delta\phi_{,x}$$

 $-M_{z}\delta\{V, xx^{\Phi} - U, \frac{\Phi^{2}}{xx^{2}}\} + M_{y}\delta\{U, xx^{\Phi} + V, \frac{\Phi^{2}}{xx^{2}}\} dx \qquad (B_{5})$ Substituting equations 3.1 and 3.2 in equation B\_3c yields the following

$$I_{3} = \int_{0}^{1} Q_{y} \delta[V_{,x} \Phi - U_{,x} \frac{\Phi^{2}}{2}] dx - \int_{0}^{1} Q_{z} \delta[U_{,x} \Phi + V_{,x} \frac{\Phi^{2}}{2}] dx$$

$$\int_{V} \{\sigma_{xz}(Y-Y_{s}) - \sigma_{xy}(Z-Z_{s})\} \delta(\Phi_{,x} \frac{\Phi^{2}}{2}) dx$$

$$\int_{V} \{\sigma_{xz}(Z-Z_{s}) = \sigma_{xy}(Y-Y_{s})\} \delta(\Phi_{,x} \Phi) dx$$

$$\int_{V}^{M} x^{\delta(\Phi_{x}, \frac{\Phi^{2}}{2}) dV}$$
(B\_6)

Substituting equations 3.1 & 3.2 and equation 3.24 into equation. B\_3b yields :

$$I_{2} = -\int_{0}^{1} [Q_{y}\{V_{,x}\phi - U_{,x}\frac{\phi^{2}}{2}\} + Q_{z}\{U_{,x}\phi - V_{,x}\frac{\phi^{2}}{2}\}]dx$$

$$- \int_{V} [\{\sigma_{xy}(Z-Z_s) - \sigma_{xz}(Y-Y_s)\}\phi_{,x}]$$

$$- \{\sigma_{xy}(Z-Z_s) - \sigma_{xz}(Y-Y_s)\}\phi_{,x}\phi dV]]dv$$

$$- \int_{0}^{1} M_x \delta\{\frac{\phi^2}{2}\phi_{,x}\}dx$$
(B\_7)

Adding equations B\_6 and B-7 yields :

$$I_2 + I_3 = KG\Phi_{,x}\delta\Phi_{,x}$$
 (B\_8)

·

The last equation has previously been derived by several authors e.g. Vlasov (1961), Bleich (1952), Timoshenko and Gere (1961), and Rajasekaran (1971). The general equilibrium condition can be written as;

 $I_1 + I_2 + I_3 + I_4 = 0$ 

(B\_9)

# Appendix C

The term CC in equation 3.63 is explained in full detail in . the following procedure viz:

$$CC = \frac{1}{2} \int_{V} \left[ \sigma_{xx} \{Y^2 \{U_{,xx}^2 + 2U_{,xx}(V_{,xx}^0 + V_{,x}^0, x)\} + \overline{\omega}_n^2 \phi_{,x}^2 + \{Z^2 \{V_{,xx}^2 - 2V_{,xx}(U_{,xx}^0 + U_{,x}^0, x)\} + \overline{\omega}_n^2 \phi_{,x}^2 + YZ \{(V_{,xx}^0 + U_{,xx} + V_{,x}^0, x)V_{,xx} - U_{,xx}(U_{,xx}^0 + U_{,x}^0, x)\} + ZY \{(-U_{,xx}^0 + V_{,xx} + U_{,x}^0, x)U_{,xx} + V_{,xx}(V_{,xx}^0 + V_{,x}^0, x)\} + 2Y \overline{\omega}_n (V_{,xx}^0 + U_{,xx} + V_{,x}^0, x)\Phi_{,xx} + V_{,xx}(V_{,xx}^0 + V_{,x}^0, x)\} + 2Z \overline{\omega}_n (V_{,xx}^0 - U_{,x}^0, x)\Phi_{,xx}] dV \qquad (cc-1)$$
  
Rearranging equation cc-1 as  

$$CC = \frac{1}{2} \int_{V} \left[ \sigma_{,xx} \{Y^2 U_{,xx}^2 + Z^2 V_{,xx}^2 + \overline{\omega}_n^2 \Phi_{,xx}^2 + V_{,xx}^0, xx + YZ U_{,xx}^0, xx + ZY V_{,xx}^0, xx + Y \overline{\omega}_n \Phi_{,xx}^0, xx + H.0.T \right] dV \qquad (cc-2)$$
The longitudinal stress can be represented by using interaction stress formulae plus the higher order terms (H.0.T) as:

$$\sigma_{xx} = \frac{P}{A} + \frac{M_y}{I_y} + \frac{M_z}{I_z} + \frac{M_{\omega}}{I_{\omega}} + H.0.T. \qquad (cc-3)$$

Substituting equation cc-3 into equation cc-2 yields

-277-

$$\frac{1}{2} \int_{0}^{1} \left[ \prod_{x} U_{xxx}^{*} + I_{y} V_{xxx}^{*} + I_{w} \Phi_{xxx}^{2} \right]$$

$$I_{yz} U_{xxx} V_{xxx} + I_{zy} V_{xxx} U_{xxx} + I_{yw} \Phi_{xx} U_{xxx}$$

$$I_{yw} U_{xx} \Phi_{xxx} + I_{zw} \Phi_{xx} V_{xxx} + I_{zw} V_{xxx} \Phi_{xxx} \right]$$

$$+ \frac{W_{y}}{I_{y}} [a_{1} U_{xxx}^{2} + b_{1} V_{xxx}^{2} + a_{4} \Phi_{xx}^{2}$$

$$+ b_{2} U_{xxx} V_{xxx} + a_{1} V_{xx} U_{xxx} + 2a_{3} \Phi_{xx} U_{xxx} + 2d_{3} V_{xx} \Phi_{xx} \right]$$

$$+ \frac{M_{z}}{I_{z}} [a_{2} U_{xxx}^{2} + b_{2} V_{xxx}^{2} + c_{2} \Phi_{xx}^{2}$$

$$+ b_{2} U_{xx} V_{xxx} + a_{1} V_{xx} U_{xxx} + 2b_{2} \Phi_{xx} U_{xxx} + 2b_{3} V_{xx} \Phi_{xx} \right]$$

$$+ \frac{M_{w}}{I_{w}} [a_{3} U_{xxx}^{2} + b_{3} V_{xxx}^{2} + c_{3} \Phi_{xx}^{2}$$

$$+ b_{2} U_{xx} V_{xxx} + b_{3} V_{xxx}^{2} + c_{3} \Phi_{xx}^{2}$$

$$+ b_{2} U_{xx} V_{xxx} + b_{2} V_{xx} U_{xxx} + 2a_{4} \Phi_{xx} U_{xxx} + 2b_{3} V_{xx} \Phi_{xx} \right] dx \qquad (cc.4)$$

$$variations of equation cc.4 as$$

$$\delta(CC) = \int_{0}^{1} [(D_{1} U_{xxx} + D_{5} V_{xxx} + D_{6} \Phi_{xxx}) \delta U_{xxx}$$

$$(D_{2} U_{xxx} + D_{4} V_{xxx} + D_{7} \Phi_{xx}) \delta \Phi_{xxx} dx$$

$$(D_{3} U_{xxx} + D_{6} V_{xxx} + D_{7} \Phi_{xxx}) \delta \Phi_{xxx} dx$$

$$(cc.4)$$

where

 $D_1 = \left[\frac{P}{A}I_z + a_1\frac{M_y}{I_y} + a_2\frac{M_z}{I_z} + a_3\frac{M_\omega}{I_\omega}\right]$ 

$$D_{2} = \left[\frac{P}{A}I_{y} + c_{4}\frac{M_{y}}{I_{y}} + b_{2}\frac{M_{z}}{I_{z}} + b_{3}\frac{M_{\omega}}{I_{\omega}}\right]$$

$$D_{3} = \left[\frac{P}{A}I_{\omega} + c_{1}\frac{M_{y}}{I_{y}} + c_{2}\frac{M_{z}}{I_{z}} + c_{3}\frac{M_{\omega}}{I_{\omega}}\right]$$

$$D_{4} = \left[\frac{P}{A}I_{yz} + a_{1}\frac{M_{y}}{I_{y}} + b_{2}\frac{M_{z}}{I_{z}} + d_{3}\frac{M_{\omega}}{I_{\omega}}\right]$$

$$D_{5} = \left[\frac{P}{A}I_{zy} + b_{2}\frac{M_{y}}{I_{y}} + a_{1}\frac{M_{z}}{I_{z}} + d_{3}\frac{M_{\omega}}{I_{\omega}}\right]$$

$$D_{6} = \left[\frac{P}{A}I_{\omega y} + a_{3}\frac{M_{y}}{I_{y}} + d_{3}\frac{M_{z}}{I_{z}} + a_{4}\frac{M_{\omega}}{I_{\omega}}\right]$$

$$D_{7} = \left[\frac{P}{A}I_{\omega z} + d_{3}\frac{M_{y}}{I_{y}} + b_{3}\frac{M_{z}}{I_{z}} + a_{4}\frac{M_{\omega}}{I_{\omega}}\right]$$
and

$$a_{1} = \int_{A} Y^{2} Z dA \quad a_{2} = \int_{A} Y^{3} dA \quad a_{3} = \int_{\overline{w}} \overline{w} Y^{2} dA \quad a_{4} = \int_{\overline{w}} \overline{w}^{2} Y dA$$

$$b_{1} = \int_{A} Z^{3} dA \quad b_{2} = \int_{A} Y Z^{2} dA \quad b_{3} = \int_{\overline{w}} \overline{w} Z^{2} dA$$

$$c_{2} = \int_{A} \overline{w}^{2} Z dA \quad c_{3} = \int_{A} \overline{w}^{3} dA \quad c_{3} = \int_{A} \overline{w}^{3} dA$$

•

-

·

.

P011= 
$$\frac{.6}{1}\Phi_1 + \frac{9}{70}\chi_1 + \frac{.6}{1}\Phi_2 - \frac{9}{70}\chi_2$$
  
P012=  $\frac{1}{70}\Phi_1 + \frac{1}{140}\chi_1 + \frac{4}{35}\Phi_2 - \frac{3}{140}\chi_2$   
P014=  $\frac{4}{35}\Phi_1 + \frac{31}{140}\chi_1 - \frac{1}{70}\Phi_2 - \frac{1}{140}\chi_2$   
P022=-  $\frac{43}{420}\Phi_1 + \frac{1^2}{120}\chi_1 + \frac{131}{420}\Phi_2 - \frac{1^2}{168}\chi_2$   
P024=-  $\frac{1}{60}\Phi_1 - \frac{1^2}{840}\chi_1 - \frac{1}{60}\Phi_2 + \frac{1^2}{840}\chi_2$   
P044=-  $\frac{131}{40}\Phi_1 + \frac{1^2}{168}\chi_1 + \frac{43}{420}\Phi_2 - \frac{1^2}{120}\chi_2$   
P013= -P011 P023= -P012  
P034= -P013 P034= -P014

•

$$S011 = \frac{6}{13} \phi_1 + \frac{6}{12} \chi_1 + \frac{6}{13} \phi_2 - \frac{6}{12} \chi_2$$

$$S012 = \frac{4 \cdot 2}{12} \phi_1 + \frac{4}{1} \chi_1 + \frac{1 \cdot 8}{12} \phi_2 - \frac{2}{1} \chi_2$$

$$S014 = \frac{1 \cdot 8}{12} \phi_1 + \frac{2}{1} \chi_1 + \frac{4 \cdot 2}{12} \phi_2 - \frac{4}{1} \chi_2$$

$$S024 = \frac{\phi_1}{1} + \frac{1}{15} \chi_1 + \frac{\phi_2}{1} - \frac{1}{15} \chi_2$$

$$S044 = \frac{8}{1} \phi_1 + \frac{2}{15} \chi_1 + \frac{3 \cdot 2}{1} \phi_2 - \frac{1}{3} \chi_2$$

$$S013 = -S012 \qquad S023 = -S012$$

$$S034 = -S014$$

•

$$\begin{aligned} & \text{YU11} = \frac{6}{1} \text{U}_{1} + \frac{4}{12} \frac{2}{12} \theta_{\text{U1}} = \frac{6}{13} \text{U}_{2} + \frac{1}{12} \frac{8}{12} \theta_{\text{U2}} \\ & \text{YU12} = \frac{6}{12} \text{U}_{1} + \frac{4}{10} \theta_{\text{U1}} + \frac{6}{12} \text{U}_{2} + \frac{4}{10} \theta_{\text{U2}} \\ & \text{YU13} = \frac{6}{13} \text{U}_{1} + \frac{1}{12} \frac{8}{12} \theta_{\text{U1}} = \frac{6}{13} \text{U}_{2} + \frac{4}{12} \frac{2}{12} \theta_{2} \\ & \text{YU14} = -\frac{6}{12} \text{U}_{1} - \frac{2}{10} \theta_{\text{U1}} + \frac{6}{12} \text{U}_{2} - \frac{4}{10} \theta_{\text{U2}} \\ & \text{YU21} = \frac{4}{12} \frac{2}{12} \text{U}_{1} + \frac{3}{2} \frac{2}{10} \theta_{\text{U1}} - \frac{4}{12} \frac{2}{12} \text{U}_{2} + \frac{1}{10} \theta_{\text{U2}} \\ & \text{YU22} = \frac{4}{10} \text{U}_{1} + \frac{1}{3} \theta_{\text{U1}} - \frac{4}{10} \text{U}_{2} + \frac{1}{15} \theta_{\text{U2}} \\ & \text{YU23} = \frac{1}{12} \frac{8}{10} + \frac{4}{10} \theta_{\text{U1}} - \frac{1}{12} \frac{8}{12} \text{U}_{2} + \frac{1}{10} \theta_{\text{U2}} \\ & \text{YU24} = -\frac{2}{10} \text{U}_{1} - \frac{2}{15} \theta_{\text{U1}} - \frac{1}{12} \frac{8}{12} \text{U}_{2} + \frac{1}{10} \theta_{\text{U2}} \\ & \text{YU41} = \frac{1}{12} \frac{8}{10} + \frac{1}{10} \theta_{\text{U1}} - \frac{1}{12} \frac{8}{12} \text{U}_{2} + \frac{1}{10} \theta_{\text{U2}} \\ & \text{YU42} = \frac{2}{10} \text{U}_{1} + \frac{1}{10} \theta_{\text{U1}} - \frac{4}{12} \text{U}_{2} + \frac{2}{15} \theta_{\text{U2}} \\ & \text{YU43} = \frac{4}{12} \frac{2}{10} + \frac{1}{10} \theta_{\text{U1}} - \frac{4}{10} \text{U}_{2} + \frac{3}{10} \theta_{\text{U2}} \\ & \text{YU44} = -\frac{4}{10} \text{U}_{1} - \frac{1}{15} \theta_{\text{U1}} - \frac{4}{10} \text{U}_{2} + \frac{1}{10} \theta_{\text{U2}} \\ & \text{YU31} = -\text{YU11} \quad \text{YU32} = -\text{YU12} \\ & \text{YU31} = -\text{YU11} \quad \text{YU32} = -\text{YU14} \\ & \text{ZU11} = -\frac{1}{12} \theta_{\text{U1}} + \frac{1}{10} \theta_{\text{U1}} - \frac{6}{12} \text{U2} + \frac{4}{10} \theta_{\text{U2}} \\ & \text{ZU12} = \frac{6}{12} \text{U}_{1} + \frac{2}{10} \theta_{\text{U1}} - \frac{6}{12} \text{U2} + \frac{4}{10} \theta_{\text{U2}} \\ & \text{ZU14} = -\frac{6}{12} \text{U1} + \frac{2}{10} \theta_{\text{U1}} + \frac{6}{12} \text{U2} - \frac{2}{10} \theta_{\text{U2}} \\ & \text{ZU22} = -\frac{4}{10} \text{U1} + \frac{1}{3} \theta_{\text{U1}} + \frac{4}{10} \theta_{\text{U1}} + \frac{1}{10} \theta_{\text{U2}} - \frac{1}{15} \theta_{\text{U2}} \\ & \text{ZU22} = -\frac{4}{10} \text{U1} + \frac{1}{3} \theta_{\text{U1}} + \frac{4}{10} \text{U2} - \frac{1}{15} \theta_{\text{U2}} \\ & \text{ZU22} = -\frac{4}{10} \text{U1} + \frac{1}{3} \theta_{\text{U1}} + \frac{4}{10} \text{U2} - \frac{1}{15} \theta_{\text{U2}} \\ & \text{ZU22} = -\frac{4}{10} \text{U1} + \frac{1}{3} \theta_{\text{U1}} + \frac{4}{10} \text{U2} - \frac{1}{15} \theta_{\text{U2}} \\ & \text{U14} = -\frac{1}{10} \theta_{\text{U1}} + \frac{1}{10} \theta_{\text{U1}} + \frac{1}{10} \theta_{\text{U1}} + \frac{1}{10} \theta_{\text{U$$

-281-

.

$$2U24 = \frac{1}{30} u_1 - \frac{1}{30} u_2$$

$$2U44 = \frac{4}{1}U_1 + \frac{1}{15}\theta_{u1} - \frac{4}{1}U_2 - \frac{1}{3}\theta_{u2}$$

$$2U13 = -2U11 \qquad 2U23 = -2U12$$

$$2U33 = 2U11 \qquad 2U34 = -2U14$$

$$AU1 = \frac{6}{51}U_1 + \frac{1}{10}\theta_{u1} - \frac{6}{51}U_2 + \frac{1}{10}\theta_{u2}$$

$$AU2 = \frac{1}{10}U_1 + \frac{21}{15}\theta_{u1} - \frac{1}{10}U_2 - \frac{1}{30}\theta_{u2}$$

$$AU4 = \frac{1}{10}U_1 - \frac{1}{15}\theta_{u1} - \frac{1}{10}U_2 - \frac{21}{15}\theta_{u2}$$

AU13=-AU1

For YV11, YV12, etc. change U to V,  $\vartheta_{\rm u}$  to  $\vartheta_{\rm v}$  etc. in the expression for Yu11, Yu12, etc.

$$W01 = -.5\phi_1 - .11\phi_2 + .1\chi_2$$

$$W02 = .11\phi_1 - .11\phi_2 + \frac{1^2}{60}\chi_2$$

$$W04 = -\frac{1}{10}\phi_1 - \frac{1^2}{60}\chi_1 + \phi_2$$

$$W03 = -W01$$

$$CX11 = \frac{.6}{12}\phi_1 - \frac{1}{70}\chi_1 - \frac{.6}{12}\phi_2 + \frac{.4}{35}\chi_2$$

$$CX12 = \frac{9}{70}\phi_1 + \frac{1}{140}\chi_1 - \frac{9}{70}\phi_2 + \frac{.3}{140}\chi_2$$

$$CX13 = \frac{.6}{12}\phi_1 + \frac{.3}{35}\chi_1 - \frac{.6}{12}\phi_2 - \frac{.1}{70}\chi_2$$

$$CX14 = -\frac{.9}{70}\phi_1 - \frac{.3}{140}\chi_1 + \frac{.9}{70}\phi_2 - \frac{.1}{140}\chi_2$$

$$CX21 = -\frac{.1}{70}\phi_1 - \frac{.43}{70}\chi_1 + \frac{.1}{70}\chi_2 - \frac{.1}{60}\chi_2$$

$$CX22 = \frac{.1}{140}\phi_1 + \frac{.1^2}{120}\chi_1 - \frac{.1}{140}\phi_2 - \frac{.1^2}{.60}\chi_2$$

$$CX24 = -\frac{.3}{.740}\chi_1 + \frac{.1}{.768}\chi_1 - \frac{.3}{.740}\chi_2 + \frac{.1^3}{.840}\chi_2$$

$$CX44 = -\frac{.1}{.740}\phi_1 - \frac{.1^2}{.840}\chi_1 + \frac{.3}{.740}\chi_2 - \frac{.43}{.120}\chi_2$$

$$CX44 = -\frac{.1}{.140}\phi_1 + \frac{.1^2}{.840}\chi_1 + \frac{.1}{.70}\phi_2 - \frac{.43}{.120}\chi_2$$

$$CX31 = -CX11 \qquad CX32 = -Cx12$$

$$CX33 = -CX13 \qquad CX34 = CX14$$

$$B011 = \frac{.6}{1} \Phi_1 + \frac{9}{70} X_1 + \frac{.6}{1} \Phi_2 - \frac{9}{70} X_2$$

$$B012 = \frac{-1}{70} \Phi_1 + \frac{1}{140} X_1 + \frac{4}{35} \Phi_2 - \frac{3}{140} X_2$$

$$B014 = \frac{4}{35} \Phi_1 + \frac{3}{140} 1X_1 + \frac{1}{70} \Phi_2 - \frac{1}{140} X_2$$

$$B022 = \frac{43}{420} 1\Phi_1 + \frac{1^2}{120} 1X_1 + \frac{13}{420} 1\Phi_2 - \frac{1^2}{168} X_2$$

$$B024 = -\frac{1}{60} 1\Phi_1 - \frac{1^2}{840} 1X_1 - \frac{1}{60} \Phi_2 - \frac{1^2}{840} X_2$$

$$B044 = \frac{13}{40} 1\Phi_1 + \frac{1^2}{168} 1X_1 + \frac{43}{420} 1\Phi_2 - \frac{1^2}{120} X_2$$

$$B013 = -B011 \qquad B023 = -B012 \qquad B033 = -B013 \qquad B034 = -B014$$

 $\begin{aligned} xu 11 &= -\frac{1 \cdot 542858}{1^2} u_1 - \frac{6}{351} u_1 + \frac{1 \cdot 542858}{1^2} u_2 - \frac{6}{351} u_2 \\ xu 12 &= -\frac{6}{351} u_1 - \frac{3}{35} u_1 + \frac{6}{351} u_2 + \frac{1}{70} u_2 \\ xu 14 &= -\frac{6}{351} u_1 + \frac{1}{70} u_1 + \frac{6}{351} u_2 + \frac{13}{735} u_2 \\ xu 22 &= -\frac{3}{35} u_1 + \frac{21}{35} u_1 + \frac{3}{35} u_1 + \frac{1}{105} u_2 \\ xu 24 &= \frac{1}{70} u_1 - \frac{1}{105} u_1 - \frac{1}{70} u_2 + \frac{1}{105} u_2 \\ xu 44 &= -\frac{3}{35} u_1 - \frac{1}{105} u_1 + \frac{3}{35} u_2 + \frac{21}{35} u_2 \\ xu 34 &= -\frac{3}{35} u_1 - \frac{1}{105} u_1 + \frac{3}{35} u_2 + \frac{21}{35} u_2 \\ xu 35 &= -xu 11 \\ \end{aligned}$ 

For XV11, XV12, etc. change  $u_1$ ,  $\theta_{u1}$  etc. to be either  $v_1$ ,  $\theta_{v1}$  etc. in the expression for XU11, XU12, etc., or  $\phi_1$ ,  $\chi_1$  in the expression for X011, X012, etc.

#### Appendix G

The coefficients of the tangential geometrical stiffness matrix of a beam-column of thin-walled open cross-section under different loading conditions are listed in Table 4.7 and presented herein as:

 $a = a_{u1} - \overline{z}(b_{\phi1} - a_{\phi1}) - Y_s a_{\phi1} - Z_s a_{\phi1}$  $b = a_{u2} - Z(b_{\phi 2} - a_{\phi 2}) - Y_s a_{\phi 2} - Z_s a_{\phi 2}$  $j = a_{u3} - \overline{Z}(b_{\phi3} - a_{\phi3}) - Y_s a_{\phi3} - Z_s a_{\phi3}$  $g = a_{114} - \overline{Z}(b_{\phi 4} - a_{\phi 4}) - Y_s a_{\phi 4} - Z_s a_{\phi 4}$  $c = a_{v1} - Y(b_{\phi1} - a_{\phi1}) + Z_s a_{\phi1} - Y_s a_{\phi1}$  $d = a_{v2} - Y(b_{\phi2} - a_{\phi2}) + Z_s a_{\phi2} - Y_s a_{\phi2}$  $h = a_{v3} - Y(b_{\phi3} - a_{\phi3}) + Z_s a_{\phi3} - Y_s a_{\phi3}$  $i = a_{v4} - \tau(b_{\phi4} - a_{\phi4}) + Z_s a_{\phi4} - Y_s a_{\phi4}$  $e = -\Upsilon(X_{v1} - a_{v1}) + \Upsilon(X_{v1} - a_{v1}) - \Upsilon_{s}a_{v1} + Z_{s}a_{v1}$  $f = -T(X_{v2} - a_{v2}) + T(X_{u2} - a_{u2}) - Y_s a_{u2} + Z_s a_{v2}$  $o = -Y(X_{v3} - a_{v3}) + Z(X_{u3} - a_{u3}) - Y_s a_{u3} + Z_s a_{v3}$  $n = -\overline{Y}(X_{v4} - a_{v4}) + \overline{Z}(X_{u4} - a_{u4}) - Y_s a_{u4} + Z_s a_{v4}$ a1 =  $EI_{\tau}(C_{b11} + C_{b11}) + EI_{\tau v}(f_{b11} + C_{b11} + C_{b11})$   $(\xi \pi \phi_{0} + \pi \xi \phi_{0} + \pi \xi \phi_{1}) + EI_{\chi} (f_{\phi} g_{\mu} + \xi \phi_{0}) + EI_{\chi} (f_{\phi} g_{\mu} + \xi \phi_{0}) + \xi \phi_{1} = e^{-2}$  $p8 = -EI^{\Lambda}(C^{\phi 33} + C^{\phi 33}) + EI^{x\Lambda}(L^{\phi 33} + C^{\phi 33} + C^{\phi 33})$  $p^{1} = -EI^{\lambda}(C^{\phi}\pi^{5} + C^{\phi}S\pi) + EI^{\lambda}(L^{\phi}S\pi + C^{\phi}S\pi + C^{\phi}\pi^{5})$  $pe = -EI^{\lambda}(C^{\phi_{35}} + C^{\phi_{53}}) + EI^{x\lambda}(L^{\phi_{53}} + C^{\phi_{35}} + C^{\phi_{53}})$  $(\pi_{1}\phi^{0}) + \pi_{1}\phi^{0} + \pi_{1}\phi^{1} + (\pi_{1}\phi^{0}) + \pi_{2}C_{0} + \pi_{1}\phi^{0} + \sigma_{1}\phi^{0} +$  $p^{t} = -EI^{\lambda}(C^{\phi_{31}} + C^{\phi_{13}}) + EI^{2\lambda}(L^{\phi_{13}} + C^{\phi_{31}} + C^{\phi_{13}})$  $p^{3} = -EI^{\lambda}(C^{\phi_{55}} + C^{\phi_{55}}) + EI^{x\lambda}(I^{\phi_{55}} + C^{\phi_{55}} + C^{\phi_{55}})$  $p_{2} = -EI_{y}(C_{\phi}_{12} + C_{\phi}_{21}) + EI_{zy}(F_{\phi}_{12} + C_{\phi}_{21} + C_{\phi}_{12})$  $p_{1} = -EI_{y}(c_{\phi_{11}} + c_{\phi_{11}}) + EI_{y}(c_{\phi_{11}} + c_{\phi_{11}}) + c_{\phi_{11}}$ side =  $-EI^{\lambda}(c^{\phi}\pi\pi + c^{\phi}\pi\pi) + EI^{z\lambda}(t^{\phi}\pi\pi + c^{\phi}\pi\pi) + c^{\phi}\pi\pi$  $(\xi_{\mu}\phi^{0}) + \mu_{\xi}\phi^{0} + \mu$  $s^{g} = EI^{x}(C^{\phi_{33}} + C^{\phi_{33}}) + EI^{x\lambda}(L^{\phi_{33}} + C^{\phi_{33}} + C^{\phi_{33}})$  $a^{1} = EI^{2}(C^{\phi} \pi^{5} + C^{\phi} S^{\pi}) + EI^{2\lambda}(U^{\phi} S^{\pi} + C^{\phi} S^{\pi} + C^{\phi} \pi^{5})$  $\mathbf{s}^{e} = \mathbf{EI}^{\mathbf{x}}(\mathbf{C}^{\phi_{35}} + \mathbf{C}^{\phi_{53}}) + \mathbf{EI}^{\mathbf{x}}(\mathbf{L}^{\phi_{53}} + \mathbf{C}^{\phi_{35}} + \mathbf{C}^{\phi_{53}})$  $(\pi_{1}\phi^{0}) + (\pi_{1}\phi^{0}) + (\pi_{1}\phi^{0}) + (\pi_{1}\phi^{0}) = e^{2\pi}$  $g^{\#} = EI^{x}(c^{\phi_{31}} + c^{\phi_{13}}) + EI^{x\lambda}(t^{\phi_{13}} + c^{\phi_{31}} + c^{\phi_{13}})$  $B_{3} = EI_{z}(C_{\phi22} + C_{\phi22}) + EI_{zy}(f_{\phi22} + C_{\phi22} + C_{\phi22})$  $s^{z} = EI^{z}(c^{\phi_{15}} + c^{\phi_{51}}) + EI^{z\lambda}(t^{\phi_{15}} + c^{\phi_{51}} + c^{\phi_{15}})$ 

$$b_{10} = -EI_y(C_{\phi 44} + C_{\phi 44}) + EI_{zy}(f_{\phi 44} + C_{\phi 44} + C_{\phi 44})$$

$$c_{1} = EI_{\omega}^{F}_{\omega}(C_{\varphi 11} + 2C_{\varphi 11}) + EI_{\omega y}(e_{v_{11}}+C_{v_{11}} - Y_{s}C_{\varphi 11} - 2Y_{s}C_{\varphi 11})$$

$$= EI_{\omega z}(-e_{u_{11}} + C_{u_{11}} + Z_{s}C_{\varphi 11} + 2Z_{s}C_{\varphi 11})$$

$$c_{2} = EI_{\omega}^{T}_{\omega}(C_{\varphi 21} + 2C_{\varphi 12}) + EI_{\omega y}(e_{v_{12}}+C_{v_{21}} - Y_{s}C_{\varphi 21} - 2Y_{s}C_{\varphi 12})$$

$$= EI_{\omega z}(-e_{u_{21}} + C_{u_{21}} + Z_{s}C_{\varphi 21} + 2Z_{s}C_{\varphi 12})$$

$$c_{3} = EI_{\omega}^{T}_{\omega}(C_{\varphi 22} + 2C_{\varphi 22}) + EI_{\omega y}(e_{v_{22}}+C_{v_{22}} - Y_{s}C_{\varphi 22} - 2Y_{s}C_{\varphi 22})$$

$$= EI_{\omega z}(-e_{u_{22}} + C_{u_{22}} + Z_{s}C_{\varphi 22} + 2Z_{s}C_{\varphi 22})$$

$$c_{4} = EI_{\omega}^{T}_{\omega}(c_{\varphi 31} + 2C_{\varphi 13}) + EI_{\omega y}(e_{v_{31}}+C_{v_{31}} - Y_{s}C_{\varphi 31} - 2Y_{s}C_{\varphi 13})$$

$$c_{5} = EI_{\omega}^{T}_{\omega}(c_{\varphi 41} + 2C_{\varphi 14}) + EI_{\omega y}(e_{v_{41}}+C_{v_{41}} - Y_{s}C_{\varphi 41} - 2Y_{s}C_{\varphi 14})$$

$$c_{6} = EI_{\omega}^{T}_{\omega}(c_{\varphi 32} + 2C_{\varphi 23}) + EI_{\omega y}(e_{v_{23}}+C_{v_{32}} - Y_{s}C_{\varphi 32} - 2Y_{s}C_{\varphi 23})$$

$$c_{7} = EI_{\omega}^{T}_{\omega}(C_{\varphi 42} + 2C_{\varphi 24}) + EI_{\omega y}(e_{v_{42}}+C_{v_{42}} - Y_{s}C_{\varphi 42} - 2Y_{s}C_{\varphi 24})$$

$$c_{8} = EI_{\omega}^{T}_{\omega}(C_{\varphi 33} + 2C_{\varphi 33}) + EI_{\omega y}(e_{v_{33}}+C_{v_{33}} - Y_{s}C_{\varphi 33} - 2Y_{s}C_{\varphi 33})$$

$$c_{9} = EI_{\omega}^{T}_{\omega}(c_{\varphi 43} + 2C_{\varphi 34}) + EI_{\omega y}(e_{v_{43}}+C_{v_{43}} - Y_{s}C_{\varphi 43} - 2Y_{s}C_{\varphi 34})$$

•

$$e_{I} = -EI^{\lambda_{m}}(D^{\phi_{IJ}} + U^{\phi_{IJ}}) + EI^{\lambda}(e^{\lambda_{IJ}} + C^{\lambda_{IJ}}e^{\phi_{IJ}} + SR^{\lambda}C^{\phi_{IJ}})$$

$$+ EI^{\lambda_{Z}}(e^{n_{IJ}} + C^{\lambda_{IJ}}) + EI^{m_{Z}}(U^{\phi_{II}} + SR^{\lambda}e^{\phi_{II}}) + SR^{\lambda}C^{\phi_{II}})$$

$$= -EI^{\lambda_{m}}(D^{\phi_{IJ}} + U^{\phi_{IJ}}) + EI^{\lambda}(e^{\lambda_{IJ}} + C^{\lambda_{IJ}}e^{\phi_{II}}) + SR^{\lambda}C^{\phi_{II}})$$

 $q_{IS} = -EI^{\lambda}(\xi^{\phi}\pi^{\mu} + c^{\phi}\pi^{\mu}) - EI^{\Sigma}(\xi^{\phi}\pi^{\mu} - c^{\phi}\pi^{\mu}) - EI^{\lambda\Sigma}(c^{\phi}\pi^{\mu} + c^{\phi}\pi^{\mu})$  $q_{11} = -EI_{\gamma}(E_{\phi_{3}} + E_{\phi_{3}}) - EI_{\gamma}(E_{\phi_{3}} - E_{\phi_{3}}) - EI_{\gamma}(E_{\phi_{3}} + E_{\phi_{3}})$  $(\pi \epsilon \phi_{J} + \epsilon \pi \phi_{J})^{z_{IJ}} - (\pi \epsilon \phi_{J} - \pi \epsilon \phi_{J})^{z_{IJ}} - (\epsilon \pi \phi_{J} + \pi \epsilon \phi_{J})^{z_{IJ}} = 0 \epsilon_{p}$  $(e^{\phi_{33}} + e^{\phi_{33}}) - e^{\phi_{33}} - e^{\phi_{33}} - e^{\phi_{33}} - e^{\phi_{33}} - e^{\phi_{33}} - e^{\phi_{33}} + e$  $q_{8} = -EI^{\lambda}(L^{\phi_{5}\eta} + C^{\phi_{\eta5}}) - EI^{z}(L^{\phi_{5}\eta} - C^{\phi_{5}\eta}) - EI^{\lambda z}(C^{\phi_{\eta5}} + L^{\phi_{5}\eta})$  $q_{7} = -EI_{y}(r_{\phi}23 + C_{\phi}32) - EI_{z}(r_{\phi}23 - C_{\phi}32) - EI_{y}(c_{\phi}32 + r_{\phi}23)$  $q^{\mu} = -EI^{\lambda}(L^{\phi} J^{\mu} + C^{\phi} J^{\mu}) - EI^{\Sigma}(L^{\phi} J^{\mu} - C^{\phi} J^{\mu}) - EI^{\lambda}(C^{\phi} J^{\mu} + L^{\phi} J^{\mu}) = 9$  $(E^{1}\phi_{J} + E^{0}\phi_{J})^{2} = -(E^{1}\phi_{J})^{2} = -(E^{0}\phi_{J})^{2} = -(E^{0}\phi_{J}$  $q^{\dagger} = -EI^{\lambda}(L^{\phi_{55}} + C^{\phi_{55}}) - EI^{\Sigma}(L^{\phi_{55}} - C^{\phi_{55}}) - EI^{\lambda_{\Sigma}}(C^{\phi_{55}} + L^{\phi_{55}})$  $q_{3} = -EI_{y}(f_{\phi_{12}} + C_{\phi_{21}}) - EI_{z}(f_{\phi_{12}} - C_{\phi_{21}}) - EI_{yz}(C_{\phi_{12}} + f_{\phi_{12}})$  $q_{2} = -EI_{y}(f_{\phi_{12}} + C_{\phi_{21}}) - EI_{z}(f_{\phi_{12}} - C_{\phi_{21}}) - EI_{yz}(C_{\phi_{22}} + f_{\phi_{12}})$  $(^{\downarrow\downarrow\phi}_{J} + ^{\downarrow\downarrow\phi}_{C})^{\lambda}_{II} - (^{\downarrow\downarrow\phi}_{C} - ^{\downarrow\downarrow\phi}_{C})^{\lambda}_{II} - (^{\downarrow\downarrow\phi}_{C} + ^{\downarrow\downarrow\phi}_{C})^{\lambda}_{II} = I_{p}$ 

 $- EI^{mz}(-e^{n\#\pi} + C^{n\#\pi} + Z^{2}C^{\phi\#\pi} + SZ^{2}C^{\phi}\#\pi)$   $e^{I0} = EI^{m}_{0}(C^{\phi\#\pi} + SC^{\phi\#\pi}) + EI^{m\lambda}(e^{A\#\pi} + C^{A\#\pi} - X^{2}C^{\phi\#\pi} - SX^{2}C^{\phi}\#\pi)$   $- EI^{mz}(-e^{n\#3} + C^{n\#3} + Z^{2}C^{\phi\#3} + SZ^{2}C^{\phi}\pi^{3})$ 

$$EI_{yz}(e_{u12} + C_{v12}) + EI_{wz}(f_{\phi 12} + \frac{1}{2}C_{\phi 12} - e_{\phi 12})$$

$$e_{3} = -EI_{yw}(D_{\phi 12} + f_{\phi 12}) + EI_{y}(e_{v21} + C_{v12}+C_{u12} + 2E_{y}C_{\phi 12})$$

$$e_{4} = -EI_{yw}(D_{\phi 22} + f_{\phi 22}) + EI_{y}(e_{v22} + C_{v22}+C_{u22} + 2E_{y}C_{\phi 22})$$

$$e_{4} = -EI_{yw}(D_{\phi 22} + f_{\phi 22}) + EI_{y}(e_{v22} + C_{v22}+C_{u22} + 2E_{y}C_{\phi 22})$$

$$e_{5} = -EI_{yw}(D_{\phi 13} + f_{\phi 13}) + EI_{y}(e_{v31} + C_{v13}+C_{u13} + 2E_{y}C_{\phi 13})$$

$$e_{6} = -EI_{yw}(D_{\phi 23} + f_{\phi 23}) + EI_{w}(e_{v32} + C_{v23}+C_{u23} + 2E_{y}C_{\phi 32})$$

$$e_{7} = -EI_{yw}(D_{\phi 14} + f_{\phi 14}) + EI_{y}(e_{v41} + C_{v14}+C_{u14} + 2E_{y}C_{\phi 14})$$

$$e_{8} = -EI_{yw}(D_{\phi 24} + f_{\phi 24}) + EI_{y}(e_{v42} + C_{v24}+C_{u24} + 2E_{y}C_{\phi 24})$$

$$e_{9} = -EI_{yw}(D_{\phi 33} + f_{\phi 33}) + EI_{y}(e_{v33} + C_{v33}+C_{u33} + 2E_{y}C_{\phi 33})$$

$$e_{10} = -EI_{yw}(D_{\phi 44} + f_{\phi 34}) + EI_{y}(e_{v34} + C_{v43}+C_{u43} + 2E_{y}C_{\phi 44})$$

$$e_{11} = -EI_{yw}(D_{\phi 44} + f_{\phi 34}) + EI_{y}(e_{v34} + C_{v34}+C_{u44} + 2E_{y}C_{\phi 44})$$

$$e_{11} = -EI_{yw}(D_{\phi 44} + f_{\phi 34}) + EI_{y}(e_{v34} + C_{v34}+C_{u44} + 2E_{y}C_{\phi 44})$$

$$e_{11} = -EI_{yw}(D_{\phi 44} + f_{\phi 34}) + EI_{y}(e_{v34} + C_{v34}+C_{u44} + 2E_{y}C_{\phi 44})$$

$$e_{11} = -EI_{yw}(D_{\phi 44} + f_{\phi 34}) + EI_{y}(e_{v34} + C_{v34}+C_{u44} + 2E_{y}C_{\phi 44})$$

$$e_{11} = -EI_{yw}(D_{\phi 44} + f_{\phi 34}) + EI_{y}(e_{v34} + C_{v34}+C_{u44} + 2E_{y}C_{\phi 44})$$

$$e_{11} = -EI_{yw}(D_{\phi 44} + f_{\phi 34}) + EI_{y}(e_{v34} + C_{v44}+C_{u44} + 2E_{y}C_{\phi 44})$$

$$e_{11} = -EI_{yw}(D_{\phi 44} + f_{\phi 44}) + EI_{y}(e_{v34} + C_{v44}+C_{v44} + 2E_{y}C_{\phi 44})$$

$$e_{11} = -EI_{yw}(D_{\phi 44} + f_{\phi 44}) + EI_{y}(e_{v34} + C_{v44}+C_{v44} + 2E_{y}C_{\phi 44})$$

$$e_{11} = -EI_{yw}(D_{\phi 44} + f_{\phi 44}) + EI_{y}(e_{v34} + C_{v44}+C_{v44} + 2E_{y}C_{\phi 44})$$

$$e_{11} = -EI_{yw}(D_{\phi 44} + f_{\phi 44}) + EI_{y}(e_{v34} + C_{v44}+C_{v44}) + 2E_{y}C_{\phi 44})$$

$$e_{11} = -EI_{yw}(D_{\phi 44} + f_{\phi 44}) + EI_{y}(e_{v44} + C_{v44}) + EI_{y}(e_{v44} + C_{v44}) + EI_{y}(e_{v44} + C_{v44}) + EI_{y}(e_{v44} + C_{v$$

•

$$+ EI_{yz}(e_{u3u} + C_{vu3}) + EI_{wz}(f_{\psi u3} + \frac{1}{2}C_{\psi 3u} - e_{\psi 3u})$$

$$= 11 = -EI_{yw}(D_{\psi uu} + f_{\psi uu}) + EI_{y}(e_{vuu} + C_{vuu} + C_{uuu} + 2E_{y}C_{\psi uu})$$

$$+ EI_{yz}(e_{uuu} + C_{vuu}) + EI_{wz}(f_{\psi uu} + \frac{1}{2}C_{\psi uu} - e_{\psi uu})$$

$$f_{1} = -EI_{yw}(D_{\psi 11} + f_{\psi 11}) + EI_{y}(e_{v11} + C_{v11} + C_{u11} + 2E_{y}C_{\psi 11})$$

$$+ EI_{yz}(e_{u11} + C_{v11}) + EI_{wz}(f_{\psi 11} - \frac{1}{2}C_{\psi 11} + e_{\psi 11})$$

$$f_{2} = -EI_{yw}(D_{\psi 12} + f_{\psi 12}) + EI_{y}(e_{v21} + C_{v12} + C_{u21} + 2E_{y}C_{\psi 12})$$

$$+ EI_{yz}(e_{u12} + C_{v12}) + EI_{wz}(f_{\psi 12} - \frac{1}{2}C_{\psi 12} + e_{\psi 12})$$

$$f_{3} = -EI_{yw}(D_{\psi 12} + f_{\psi 12}) + EI_{y}(e_{v21} + C_{v12} + C_{u12} + 2E_{y}C_{\psi 12})$$

$$+ EI_{yz}(e_{u21} + C_{v12}) + EI_{wz}(f_{\psi 12} - \frac{1}{2}C_{\psi 21} + e_{\psi 21})$$

$$f_{4} = -EI_{yw}(D_{\psi 22} + f_{\psi 22}) + EI_{y}(e_{v22} + C_{v22} + C_{u22} + 2E_{y}C_{\psi 22})$$

$$+ EI_{yz}(e_{u22} + C_{v22}) + EI_{wz}(f_{\psi 22} - \frac{1}{2}C_{\psi 22} + e_{\psi 22})$$

$$f_{5} = -EI_{yw}(D_{\psi 13} + f_{\psi 13}) + EI_{y}(e_{v31} + C_{v13} + C_{u13} + 2E_{y}C_{\psi 13})$$

$$+ EI_{yz}(e_{u31} + C_{v13}) + EI_{wz}(f_{\psi 13} - \frac{1}{2}C_{\psi 31} + e_{\psi 31})$$

$$f_{6} = -EI_{yw}(D_{\psi 23} + f_{\psi 23}) + EI_{y}(e_{v21} + C_{v22} + C_{v23} + C_{u23} + 2E_{y}C_{\psi 32})$$

$$+ EI_{yz}(e_{u32} + C_{v23}) + EI_{wz}(f_{\psi 23} - \frac{1}{2}C_{\psi 32} + e_{\psi 32})$$

$$f_{7} = -EI_{yw}(D_{\psi 14} + f_{\psi 14}) + EI_{y}(e_{v41} + C_{v14} + C_{u14} + 2E_{y}C_{\psi 14})$$

$$+ EI_{yz}(e_{uu1} + C_{v14}) + EI_{wz}(f_{\psi 14} - \frac{1}{2}C_{\psi 41} + e_{\psi 41})$$

$$f_{8} = -EI_{yw}(D_{\psi 24} + f_{\psi 24}) + EI_{y}(e_{v42} + C_{v24} + C_{u24} + 2E_{y}C_{\psi 24})$$

$$+ EI_{yz}(e_{u42} + C_{v24}) + EI_{wz}(f_{\phi24} - \frac{1}{2}C_{\phi42} + e_{\phi42})$$

$$f_{9} = -EI_{yw}(D_{\phi33} + f_{\phi33}) + EI_{y}(e_{v33} + C_{v33} + C_{u33} + 2\overline{B}_{y}C_{\phi33})$$

$$+ EI_{yz}(e_{u33} + C_{v33}) + EI_{wz}(f_{\phi33} - \frac{1}{2}C_{\phi33} + e_{\phi33})$$

$$f_{10} = -EI_{yw}(D_{\phi43} + f_{\phi34}) + EI_{y}(e_{v34} + C_{v43} + C_{u43} + 2\overline{B}_{y}C_{\phi43})$$

$$+ EI_{yz}(e_{u43} + C_{v34}) + EI_{wz}(f_{\phi34} - \frac{1}{2}C_{\phi43} + e_{\phi43})$$

$$f_{11} = -EI_{yw}(D_{\phi34} + f_{\phi43}) + EI_{y}(e_{v43} + C_{v34} + C_{u34} + 2\overline{B}_{y}C_{\phi34})$$

$$+ EI_{yz}(e_{u34} + C_{v43}) + EI_{wz}(f_{\phi43} - \frac{1}{2}C_{\phi34} + e_{\phi34})$$

$$e_{12} = -EI_{yw}(D_{\phi44} + f_{\phi44}) + EI_{y}(e_{v44} + C_{v44} + C_{u44} + 2\overline{B}_{y}C_{\phi44})$$

$$+ EI_{yz}(e_{u44} + C_{v44}) + EI_{wz}(f_{\phi44} - \frac{1}{2}C_{\phi44} + e_{\phi44})$$

#### Appendix H

## Nonlinear Geometrical Stiffness Matrix

coefficients of the nonlinear geometrical stiffness The matrix given in Table 4.8 are explained in detail as:  $A_{1} = -\frac{1}{2}M_{x}(V_{\phi 11} + V_{\phi 11}) \qquad A_{2} = -\frac{1}{2}M_{x}(V_{\phi 12} + V_{\phi 21})$  $A_{3} = -\frac{1}{2}M_{x}(V_{\phi 22} + V_{\phi 22}) \qquad A_{4} = -\frac{1}{2}M_{x}(V_{\phi 13} + V_{\phi 31})$  $A_5 = -\frac{1}{2}M_x(V_{14} + V_{41}) \qquad A_6 = -\frac{1}{2}M_x(V_{432} + V_{423})$  $A_7 = -\frac{1}{2}M_x(V_{\phi 24} + V_{\phi 42}) \qquad A_8 = -\frac{1}{2}M_x(V_{\phi 33} + V_{\phi 33})$  $A_{9} = -\frac{1}{2}M_{x}(V_{b34} + V_{b43}) \qquad A_{10} = -\frac{1}{2}M_{x}(V_{b44} + V_{b44})$  $B_{1} = FY_{s}(n_{\phi 11} + m_{\phi 11}) - Q_{z}U_{\phi 11} + M_{z}(n_{\phi 11} + m_{\phi 11}) + \frac{1}{2}M_{x}(W_{\phi 11} - T_{\phi 11})$  $B_{2} = FY_{s}(n_{d21}+m_{d12})-Q_{z}U_{d21} + M_{z}(n_{d21}+m_{d12}) + \frac{1}{2}M_{x}(W_{d12}-T_{d21})$  $B_{3} = FY_{s}(n_{d_{12}} + m_{d_{21}}) - Q_{z}U_{d_{12}} + M_{z}(n_{d_{21}} + m_{d_{12}}) + \frac{1}{2}M_{x}(W_{d_{12}} - T_{d_{21}})$  $B_{4} = FY_{s}(n_{\phi 22} + m_{\phi 22}) - Q_{z}U_{\phi 22} + M_{z}(n_{\phi 22} + m_{\phi 22}) + \frac{1}{2}M_{x}(W_{\phi 22} - T_{\phi 22})$  $B_{5} = FY_{s}(n_{d31}+m_{d13})-Q_{z}U_{d31} + M_{z}(n_{d13}+m_{d31}) + \frac{1}{2}M_{x}(W_{d13}-T_{d31})$ 

 $B_{7} = FY_{s}(n_{\phi32}+m_{\phi23})-Q_{z}U_{\phi32} +M_{z}(n_{\phi23}+m_{\phi32})+\frac{1}{2}M_{x}(W_{\phi23}-T_{\phi32})$   $B_{8} = FY_{s}(n_{\phi42}+m_{\phi24})-Q_{z}U_{\phi42} +M_{z}(n_{\phi24}+m_{\phi42})+\frac{1}{2}M_{x}(W_{\phi24}-T_{\phi42})$   $B_{9} = FY_{s}(n_{\phi33}+m_{\phi33})-Q_{z}U_{\phi33} +M_{z}(n_{\phi33}+m_{\phi33})+\frac{1}{2}M_{x}(W_{\phi33}-T_{\phi33})$   $B_{10} = FY_{s}(n_{\phi11}+m_{\phi11})-Q_{z}U_{\phi11} +M_{z}(n_{\phi11}+m_{\phi11})+\frac{1}{2}M_{x}(W_{\phi11}-T_{\phi11})$ 

$$B_{11} = FY_{s}(n_{\phi34} + m_{\phi43}) - Q_{z}U_{\phi34} + M_{z}(n_{\phi43} + m_{\phi34}) + \frac{1}{2}M_{x}(W_{\phi43} - T_{\phi34})$$

$$B_{12} = FY_{s}(n_{\phi44} + m_{\phi44}) - Q_{z}U_{\phi44} + M_{z}(n_{\phi44} + m_{\phi44}) + \frac{1}{2}M_{x}(W_{\phi44} - T_{\phi44})$$

$$B_{13} = FY_{s}(n_{\phi13} + m_{\phi31}) - Q_{z}U_{\phi13} + M_{z}(n_{\phi31} + m_{\phi13}) + \frac{1}{2}M_{x}(W_{\phi31} - T_{\phi13})$$

$$B_{14} = FY_{s}(n_{\phi23} + m_{\phi32}) - Q_{z}U_{\phi23} + M_{z}(n_{\phi32} + m_{\phi23}) + \frac{1}{2}M_{x}(W_{\phi32} - T_{\phi23})$$

$$B_{15} = FY_{s}(n_{\phi14} + m_{\phi41}) - Q_{z}U_{\phi14} + M_{z}(n_{\phi41} + m_{\phi14}) + \frac{1}{2}M_{x}(W_{\phi41} - T_{\phi14})$$

$$B_{16} = FY_{s}(n_{\phi24} + m_{\phi42}) - Q_{z}U_{\phi24} + M_{z}(n_{\phi42} + m_{\phi24}) + \frac{1}{2}M_{x}(W_{\phi42} - T_{\phi24})$$

 $C_{1} = FZ_{s}^{(n_{\phi 11} + m_{\phi 11}) - Q_{y}U_{\phi 11}} - M_{y}^{(n_{\phi 11} + m_{\phi 11}) - \frac{1}{2}M_{x}^{(W_{\phi 11} - T_{\phi 11})}$  $C_{2} = FZ_{s}^{(n_{\phi21}+m_{\phi12})-Q_{y}U_{\phi21}} - M_{y}^{(n_{\phi21}+m_{\phi12})-\frac{1}{2}M_{x}^{(W_{\phi12}-T_{\phi21})}}$  $C_{3} = FZ_{s}^{(n_{\phi 12}+m_{\phi 21})-Q_{y}U_{\phi 12}} - M_{y}^{(n_{\phi 21}+m_{\phi 12})-\frac{1}{2}M_{x}^{(W_{\phi 12}-T_{\phi 21})}$  $C_{4} = FZ_{s}(n_{\phi 22}+m_{\phi 22})-Q_{y}U_{\phi 22} -M_{y}(n_{\phi 22}+m_{\phi 22})-\frac{1}{2}M_{x}(W_{\phi 22}-T_{\phi 22})$  $C_5 = FZ_s(n_{\phi31}+m_{\phi13})-Q_yU_{\phi31} -M_y(n_{\phi13}+m_{\phi31})-\frac{1}{2}M_x(W_{\phi13}-T_{\phi31})$  $C_7 = FZ_s(n_{\phi 32}+m_{\phi 23})-Q_yU_{\phi 32} -M_y(n_{\phi 23}+m_{\phi 32})-\frac{1}{2}M_x(W_{\phi 23}-T_{\phi 32})$  $C_8 = FZ_s(n_{\phi 42} + m_{\phi 24}) - Q_y U_{\phi 42} - M_y(n_{\phi 24} + m_{\phi 42}) - \frac{1}{2}M_x(W_{\phi 24} - T_{\phi 42})$  $C_9 = FZ_s(n_{\phi 33} + m_{\phi 33}) - Q_y U_{\phi 33} - M_y(n_{\phi 33} + m_{\phi 33}) - \frac{1}{2}M_x(W_{\phi 33} - T_{\phi 33})$  $C_{10} = FZ_{s}^{(n_{\phi_{11}}+m_{\phi_{11}})-Q_{y}U_{\phi_{11}} - M_{y}^{(n_{\phi_{11}}+m_{\phi_{11}})-\frac{1}{2}M_{x}^{(W_{\phi_{11}}-T_{\phi_{11}})}$  $C_{11} = FZ_{s}(n_{\phi34}+m_{\phi43}) - Q_{y}U_{\phi34} - M_{y}(n_{\phi43}+m_{\phi34}) - \frac{1}{2}M_{x}(W_{\phi43}-T_{\phi34})$  $C_{12} = FZ_{s}(n_{\phi 44} + m_{\phi 44}) - Q_{y}U_{\phi 44} - M_{y}(n_{\phi 44} + m_{\phi 44}) - \frac{1}{2}M_{x}(W_{\phi 44} - T_{\phi 44})$  $C_{13} = FZ_{s}(n_{\phi 13}+m_{\phi 31}) - Q_{y}U_{\phi 13} - M_{y}(n_{\phi 31}+m_{\phi 13}) - \frac{1}{2}M_{x}(W_{\phi 31}-T_{\phi 13})$ 

$$\begin{split} p_{\theta} &= +\frac{5}{7} o^{\Lambda} z^{2} [-\frac{5}{7} n^{\theta} 3^{3} + g^{\theta} 3^{3} +$$

۰.,

••

$$E_{9} = -W_{2}(a_{9}3^{2}-p_{3}^{2}) + \frac{1}{2}Q_{2}C_{9}^{2}A_{3}^{2} \left[ -\frac{1}{2}Q_{9}A_{3}^{2}+Q_{9}^{2}Q_{9}^{2} \right] + \frac{1}{2}Q_{2}C_{9}^{2}A_{3}^{2} \right]$$

$$E_{9} = W_{2}(a_{9}4^{2}-p_{9}4) - \frac{1}{2}Q_{2}C_{9}A_{3}^{2}$$

$$E_{1} = W_{2}(x_{1}u_{1}+a_{9}u_{1}) - \frac{1}{2}Q_{2}V_{1}u_{1}^{2} - W_{2}(x_{1}u_{1}-a_{9}u_{1}) - \frac{1}{2}Q_{2}B_{1}u_{3}^{2}$$

$$E_{1} = W_{2}(x_{1}u_{1}+a_{1}v_{1}) - \frac{1}{2}Q_{2}V_{1}u_{3}^{2} - W_{2}(x_{1}u_{2}-a_{1}^{2}) - \frac{1}{2}Q_{2}B_{1}u_{3}^{2}$$

$$E_{1} = W_{2}(a_{1}-b_{1}) - \frac{1}{2}Q_{2}V_{1}u_{3}^{2} - W_{2}(x_{1}u_{2}-a_{1}^{2}) - \frac{1}{2}Q_{2}B_{1}u_{3}^{2}$$

$$E_{1} = W_{2}(a_{1}-b_{1}) - \frac{1}{2}Q_{2}V_{1}u_{3}^{2} - W_{2}(x_{1}u_{2}-a_{1}^{2}) - \frac{1}{2}Q_{2}B_{2}u_{3}^{2}$$

$$E_{2} = W_{2}(a_{1}-b_{1}) - \frac{1}{2}Q_{2}V_{1}u_{3}^{2} - W_{2}(x_{1}u_{2}-a_{1}^{2}) - \frac{1}{2}Q_{2}B_{2}u_{3}^{2}$$

$$E_{2} = W_{2}(a_{1}-b_{1}) - \frac{1}{2}Q_{2}V_{1}u_{3}^{2} - \frac{1}{2}Q_{2}V_{2}u_{3}^{2}$$

$$E_{3} = -W_{2}(a_{1}-b_{1}) - \frac{1}{2}Q_{2}V_{2}u_{3}^{2} - \frac{1}{2}Q_{2}V_{2}u_{3}^{2}$$

$$E_{4} = W_{2}(a_{1}-b_{1}) - \frac{1}{2}Q_{2}V_{2}u_{3}^{2} - \frac{1}{2}Q_{2}V_{2}u_{3}^{2}$$

$$E_{5} = W_{2}(a_{1}-b_{1}) - \frac{1}{2}Q_{2}V_{2}u_{3}^{2} - \frac{1}{2}Q_{2}V_{2}u_{3}^{2}$$

$$E_{5} = W_{2}(a_{1}-b_{1}) - \frac{1}{2}Q_{2}V_{2}u_{3}^{2} - \frac{1}{2}Q_{2}V_{2}u_{3}^{2}$$

$$E_{5} = W_{2}(a_{1}-b_{1}) - \frac{1}{2}Q_{2}V_{2}u_{3}^{2} - \frac{1}{2}Q_{2}V_{2}u_{3}^{2} - \frac{1}{2}Q_{2}V_{2}u_{3}^{2} - \frac{1}{2}Q_{2}V_{2}u_{3}^{2} - \frac{1}{2}Q_{2}U_{2}u_{3}^{2} -$$

 $\mathsf{E}^{\mathsf{IO}} = -\mathsf{W}^{\mathsf{Z}}(\mathsf{s}^{\phi}{}^{\dagger}-\mathsf{p}^{\phi}{}^{\dagger}) + \frac{\mathsf{I}}{\mathsf{S}}\mathsf{O}^{\mathsf{Z}}\mathsf{O}^{\phi}{}^{\dagger}$ 

A .

•

••

An illustration of the interaction between the coefficients  $V_{\phi 11}, C_{\phi 11}, \text{etc.}$ , with either the section properties or the internal loads is presented. Two examples are given herein to show the procedure, which are:

example 1

$$A_1 = -\frac{1}{2}M_x(V_{\phi 11} + V_{\phi 11})$$

Substituting for the term  $V_{\begin{subarray}{c} \varphi 11 \end{subarray}}$  from appendix J in the above equation yields

$$A_{1} = -\left[\left(-\frac{1}{351^{2}}M_{x 1} - \frac{6}{151^{2}}\Delta M_{x}\right)\phi_{1} + \left(\frac{3}{351}\right)\chi_{1}\right]$$

+
$$(\frac{27}{351^2}M_{x1}-\frac{24}{351^2}\Delta M_{x})\Phi_2+(\frac{3}{351}M_{x1}+\frac{3}{351}\Delta M_{x})\chi_2$$

example 2

$$D = EI_{y}[U_{\phi 21} + \frac{1}{2}U_{12}]$$

The above equation can be written, after substituting for the terms  $U_{\phi\,21}$  and  $U_{\phi\,12}$  from Appendix N as:

$$D = \left[ \left( \frac{5}{42} \right) - \frac{5}{168} \right] EI_{y1} + \left( \frac{1}{229} \right) - \frac{3}{280} \right] \Delta EI_{y} ] \Phi_{1}$$

+ 
$$\left[\left(\frac{1}{168}1^2 - \frac{1}{168}1^2\right)EI_{y1} + \left(-\frac{1}{1260}1^2 - \frac{1}{420}1^2\right)\Delta EI_{y}\right]\chi_1$$

+ 
$$\left[\left(\frac{2}{105}1-\frac{17}{840}1\right)EI_{y1}+\left(-\frac{1}{70}1-\frac{3}{280}1\right)\Delta EI_{y}\right]\Phi_{2}$$

+ 
$$\left[\left(\frac{13}{840}\right)^2 - \frac{13}{840}\right]^2\right) EI_{y1} + \left(\frac{1}{90}\right)^2 + \frac{1}{120}\right]^2\right) \Delta EI_{y1}x_2$$

The same procedure can be applied for the rest of the terms and their interaction with the forces or the section properties.

## Appendix J

Details of the coefficients stated in Tables 4.7 and 4.8 are presented in this appendix. The terms can be employed in the nonlinear geometrical and tangential stiffness matrices by interchanging some parameters as described in appendix 1. Those coefficient are

$$\begin{aligned} a_{u1} &= \left(-\frac{6}{51^2} - \frac{.6}{1^2}\right) U_1 + \left(-\frac{1}{10} - \frac{A}{10}\right) \theta_{u1} \\ &+ \left(\frac{6}{51^2} - \frac{.6}{1^2}A\right) U_2 + \left(0 - \frac{1}{101}\right) \theta_{u2} \\ a_{u2} &= \left(-\frac{1}{101} - \frac{1}{101}\right) U_1 + \left(-\frac{2}{15} - \frac{1}{30}\right) \theta_{u1} \\ &+ \left(-\frac{1}{101} - \frac{1}{101}\right) U_2 + \left(\frac{1}{30} - \frac{A}{60}\right) \theta_{u2} \\ a_{u3} &= \left(\frac{6}{51^2} - \frac{.6}{1^2}A\right) U_1 + \left(-\frac{1}{10} - \frac{1}{10}\right) \theta_{u1} \\ &- \left(\frac{6}{51^2} - \frac{.6}{1^2}A\right) U_2 + \left(\frac{1}{101} - 0\right) \theta_{u2} \\ a_{u4} &= \left(-\frac{1}{101} - 0\right) U_1 + \left(\frac{1}{30} - \frac{1}{60}\right) \theta_{u1} - \left(\frac{1}{101} - 0\right) U_2 + \left(-\frac{2}{15} - \frac{1}{10}\right) \theta_{u2} \\ b_{u1} &= \left(\frac{1.2}{1^2} - \frac{A}{101^2}\right) U_1 + \left(0 - \frac{1}{101}\right) \theta_{u1} \\ &+ \left(-\frac{1.2}{1^2} - \frac{1.1}{101^2}\right) U_1 + \left(\frac{1}{101} - 0\right) \theta_{u2} \\ b_{u2} &= \left(\frac{1.1}{1^2} - \frac{2}{101}\right) U_1 + \left(\frac{2}{15} - \frac{1}{30}\right) \theta_{u1} \\ &+ \left(-\frac{1}{101} - \frac{2}{10}\right) U_2 + \left(0 - \frac{1}{30}\right) \theta_{u2} \\ b_{u3} &= \left(-\frac{1.2}{1^2} - \frac{1}{101^2}\right) U_1 + \left(0 - \frac{1}{101}\right) \theta_{u1} \end{aligned}$$

$$+ (\frac{1 \cdot 2}{12} \quad \frac{1 \cdot 1}{101^2}) U_2 - (\frac{1}{101} \quad \circ) \theta_{u2}$$

$$b_{u4} = (-\frac{\cdot 4}{1} \quad \frac{2}{101}) U_1 + (-\frac{1}{30} \quad \frac{1}{30}) \theta_{u1}$$

$$+ (\frac{1 \cdot 1}{1} \quad \frac{\cdot 4}{1}) U_2 + (\frac{2}{15} \quad \frac{1}{10}) \theta_{u2}$$

$$Change \quad a_{u1}, a_{u2}, a_{u3}, a_{u4}, U_1, \theta_{u1}, U_2, \theta_{u2} \text{ into either } a_{v1},$$

 $a_{v2}$ ,  $a_{v3}$ ,  $a_{v4}$ ,  $V_1$ ,  $\theta_{v1}$ ,  $V_2$ ,  $\theta_{v2}$  or  $a_{\phi 1}$ ,  $a_{\phi 2}$ ,  $a_{\phi 3}$ ,  $a_{\phi 4}$ ,  $\phi_1$ ,  $\chi_1$ ,  $\phi_2$ ,  $\chi_2$ respectively. A similar procedure can be done for  $b_{u1}$ , etc..

$$\begin{aligned} \operatorname{cu}_{11} &= (0 \quad \frac{18}{351^3}) \operatorname{U}_1 + (-\frac{1\cdot 2}{1^2} \quad -\frac{1\cdot 286}{1^2}) \operatorname{\theta}_{u1} \\ &+ (0 \quad -\frac{18}{351^3}) \operatorname{U}_2 + (-\frac{\cdot 6}{1^2} \quad -\frac{12}{351^2}) \operatorname{\theta}_{u2} \\ \operatorname{cu}_{21} &= (-\frac{1\cdot 2}{1^2} \quad -\frac{1\cdot 286}{1^2}) \operatorname{U}_1 + (-\frac{4}{101} \quad \frac{1}{351}) \operatorname{\theta}_{u1} \\ &+ (-\frac{6}{1^2} \quad -\frac{9}{351^2}) \operatorname{U}_2 \quad - (0 \quad -\frac{1}{141}) \operatorname{\theta}_{u2} \\ \operatorname{cu}_{31} &= (0 \quad -\frac{18}{351^3}) \operatorname{U}_1 + (-\frac{\cdot 6}{1^2} \quad -\frac{9}{351^2}) \operatorname{\theta}_{u1} \\ &+ (0 \quad -\frac{18}{351^3}) \operatorname{U}_2 + (\frac{12}{351^2} \quad -\frac{12}{351^2}) \operatorname{\theta}_{u2} \\ \operatorname{cu}_{41} &= (-\frac{6}{1^2} \quad -\frac{12}{351^2}) \operatorname{U}_1 \quad - (0 \quad -\frac{1}{141}) \operatorname{\theta}_{u1} \\ &+ (\frac{12}{351^2} \quad -\frac{12}{351^2}) \operatorname{U}_2 \quad + (-\frac{4}{1} \quad -\frac{5}{281}) \operatorname{\theta}_{u2} \\ \operatorname{cu}_{12} &= (\frac{1\cdot 2}{1^2} \quad -\frac{12}{351^2}) \operatorname{U}_2 \quad + (-\frac{4}{1} \quad -\frac{6}{351}) \operatorname{\theta}_{u2} \\ \operatorname{cu}_{22} &= (-\frac{2}{101} \quad -\frac{1}{71}) \operatorname{U}_1 \quad + (-\frac{1}{3} \quad -\frac{2}{205} \operatorname{A}) \operatorname{\theta}_{u1} \\ &+ (-\frac{2}{101} \quad -\frac{1}{351}) \operatorname{U}_2 \quad + (\frac{1}{101} \quad -\frac{1}{71}) \operatorname{\theta}_{u1} \\ &+ (-\frac{2}{101} \quad -\frac{1}{351}) \operatorname{U}_2 \quad + (-\frac{1}{101} \quad -\frac{2}{101} \operatorname{A}) \operatorname{\theta}_{u2} \\ \operatorname{cu}_{23} &= (\frac{1\cdot 2}{1^2} \quad -\frac{12}{351^2}) \operatorname{U}_1 \quad - (2 \quad -\frac{1}{101} \quad -\frac{1}{71}) \operatorname{\theta}_{u1} \end{aligned}$$

$$-\left(\frac{1.2}{12},\frac{12}{3512}\right)U_{2} + \left(\frac{4}{1},\frac{5}{351}\right)\theta_{u2}$$

$$cu_{24} = \left(-\frac{4}{1},-6/351\right)U_{1} + \left(\frac{1}{30},-\frac{2}{205}\right)\theta_{u1}$$

$$+ \left(\frac{4}{1},\frac{6}{351}\right)U_{2} + \left(\frac{1}{15},\frac{4}{35}\right)\theta_{u2}$$

$$cu_{13} = \left(0,-\frac{18}{351^{3}}\right)U_{1} + \left(\frac{1.2}{12},\frac{1.286}{12}\right)\theta_{u1}$$

$$+ \left(0,\frac{18}{351^{3}}\right)U_{2} + \left(\frac{6}{12},\frac{12}{351^{2}}\right)\theta_{u2}$$

$$cu_{23} = \left(\frac{1.2}{1^{2}},\frac{1.286}{1^{2}}\right)U_{1} + \left(\frac{4}{101},-\frac{1}{351}\right)\theta_{u1}$$

$$+ \left(\frac{6}{12},\frac{9}{351^{2}}\right)U_{2} + \left(0,-\frac{1}{141}\right)\theta_{2}$$

$$cu_{33} = \left(0,-\frac{18}{351^{3}}\right)U_{1} + \left(\frac{6}{1^{2}},\frac{9}{351^{2}}\right)\theta_{u1}$$

$$- \left(0,-\frac{18}{351^{3}}\right)U_{2} + \left(-\frac{12}{351^{2}},-\frac{12}{351^{2}}\right)\theta_{u2}$$

$$cu_{43} = \left(\frac{16}{12},-\frac{12}{351^{2}}\right)U_{1} + \left(0,-\frac{1}{141}\right)\theta_{u1}$$

$$- \left(\frac{12}{351^{2}},-\frac{12}{351^{2}}\right)U_{2} - \left(\frac{4}{1},-\frac{5}{351^{2}}\right)\theta_{u2}$$

$$cu_{14} = \left(\frac{1.2}{1^{2}},-\frac{6}{71^{2}}\right)U_{2} - \left(\frac{4}{1},-\frac{5}{351}\right)\theta_{u2}$$

$$cu_{24} = \left(-\frac{1.4}{1},-\frac{1.14286}{1}\right)U_{1} + \left(-\frac{1}{15},-\frac{1}{21}\right)\theta_{u1}$$

$$+ \left(-\frac{4}{101},-\frac{8}{351}\right)U_{2} + \left(-\frac{1}{30},-\frac{11}{210}\right)\theta_{u2}$$

$$cu_{34} = \left(-\frac{1.2}{1^{2}},-\frac{6}{71^{2}}\right)U_{1} + \left(-\frac{1}{30},-\frac{11}{210}\right)\theta_{u2}$$

$$cu_{44} = \left(-\frac{2}{1},-\frac{6}{351}\right)U_{1} + \left(-\frac{1}{30},-\frac{11}{210}\right)\theta_{u1}$$

- 300-

· ·

۰.

o-

۰.

1
+ 
$$(\frac{.2}{1} + \frac{6}{351})U_2$$
 +  $(\frac{1}{3} + \frac{11}{35})\theta_{u21}$   
Change  $cu_{11}$ ,  $cu_{12}$ , ...,  $U_1$ ,  $\theta_{u1}$ , etc. to the form  $cv_{11}$ ,  
 $cv_{12}$ , ...,  $V_1$ ,  $\theta_{v1}$ , etc. . Same procedure for the terms such as  $\phi_1$ ,  $\chi_1$ .

$$\begin{split} \mathbf{e}_{u11} &= \left(\frac{6}{1^3} + \frac{33}{351^3}\right) \mathbb{U}_1 + \left(\frac{4\cdot2}{1^2} + \frac{4}{71^2}\right) \mathbb{e}_{u1} \\ &= \left(\frac{6}{1^3} + \frac{33}{351^3}\right) \mathbb{U}_2 + \left(\frac{1\cdot8}{1^2} + \frac{13}{71^2}\right) \mathbb{e}_{u2} \\ \mathbf{e}_{u13} &= \left(\frac{6}{1^3} + \frac{5\cdot057}{1^3}\right) \mathbb{U}_1 + \left(\frac{1\cdot8}{1^2} + \frac{1\cdot42857}{1^2}\right) \mathbb{e}_{u1} \\ &= \left(\frac{6}{1^3} + \frac{5\cdot057}{1^3}\right) \mathbb{U}_2 + \left(\frac{4\cdot2}{1^2} + \frac{3\cdot6285}{1^2}\right) \mathbb{e}_{u1} \\ \mathbf{e}_{u14} &= \left(-\frac{6}{1^2} - \frac{3}{71^2}\right) \mathbb{U}_1 + \left(\frac{2}{1^2} - \frac{4}{351}\right) \mathbb{e}_{u2} \\ \mathbf{e}_{u21} &= \left(\frac{4\cdot2}{1^2} + \frac{4}{1^2}\right) \mathbb{U}_1 + \left(\frac{2\cdot2}{1} - \frac{11}{351}\right) \mathbb{e}_{u2} \\ \mathbf{e}_{u21} &= \left(\frac{4\cdot2}{1^2} - \frac{4}{1^2}\right) \mathbb{U}_1 + \left(\frac{2\cdot2}{1} - \frac{11}{351}\right) \mathbb{e}_{u2} \\ \mathbf{e}_{u22} &= \left(-\frac{1}{101} - \frac{11}{351}\right) \mathbb{U}_1 + \left(-9\cdot666 - 7\cdot9236\right) \mathbb{e}_{u1} \\ &+ \left(\frac{1}{101} + \frac{11}{351}\right) \mathbb{U}_2 + \left(\frac{1}{15} - \frac{1}{105}\right) \mathbb{e}_{u2} \\ \mathbf{e}_{u23} &= \left(\frac{1\cdot8}{1^2} + \frac{1\cdot42857}{1^2}\right) \mathbb{U}_1 + \left(\frac{4\cdot8}{1} + \frac{18}{351}\right) \mathbb{e}_{u1} \\ &= \left(\frac{1\cdot8}{1^2} + \frac{1\cdot42857}{1^2}\right) \mathbb{U}_2 + \left(\frac{1}{1} + \frac{32}{351}\right) \mathbb{e}_{u2} \\ \mathbf{e}_{u24} &+ \left(-\frac{2}{1} - \frac{4}{351}\right) \mathbb{U}_1 + \left(-\frac{2}{15} - \frac{2}{35}\right) \mathbb{e}_{u1} \\ &+ \left(\frac{4\cdot2}{1} + \frac{45}{351}\right) \mathbb{U}_1 + \left(-\frac{1}{15} - \frac{2}{35}\right) \mathbb{e}_{u2} \\ \mathbf{e}_{u31} &= \left(-\frac{6}{1^3} - \frac{33}{351^3}\right) \mathbb{U}_1 + \left(-\frac{1.8}{1^2} - \frac{13}{351^2}\right) \mathbb{e}_{u2} \end{split}$$

$$\begin{split} \mathbf{e}_{u32} &= (\frac{-6}{12} - \frac{6}{351^2}) \mathbf{U}_1 + (\frac{1}{101} + \frac{11}{351}) \mathbf{\theta}_{u1} \\ &+ (\frac{16}{12} + \frac{6}{351^2}) \mathbf{U}_2 + (\frac{-2}{11} - \frac{3}{351}) \mathbf{\theta}_{u2} \\ \mathbf{e}_{u33} &= -(\frac{6}{1^3} + \frac{5.057}{1^3}) \mathbf{U}_1 - (\frac{1.8}{1^2} + \frac{1.42857}{1^2}) \mathbf{\theta}_{u1} \\ &+ (\frac{6}{1^3} + \frac{5.057}{1^3}) \mathbf{U}_2 - (\frac{4.2}{1^2} + \frac{3.6285}{1^2}) \mathbf{\theta}_{u1} \\ \mathbf{e}_{u34} &= (\frac{16}{1^2} + \frac{3}{71^2}) \mathbf{U}_1 + (\frac{-2}{1} + \frac{4}{351}) \mathbf{\theta}_{u2} \\ \mathbf{e}_{u34} &= (\frac{1.6}{1^2} + \frac{3}{71}) \mathbf{U}_2 + (\frac{4}{101} + \frac{11}{351}) \mathbf{\theta}_{u2} \\ \mathbf{e}_{u41} &= (\frac{1.8}{1^2} + \frac{13}{351^2}) \mathbf{U}_1 + (\frac{1}{1} + \frac{3}{351}) \mathbf{\theta}_{u1} \\ &- (\frac{1.8}{1^2} + \frac{13}{351^2}) \mathbf{U}_2 + (\frac{18}{1} + \frac{2}{71}) \mathbf{\theta}_{u2} \\ \mathbf{e}_{u42} + (\frac{42}{1} + \frac{3}{351}) \mathbf{U}_1 + (\frac{1}{2} + \frac{3}{351}) \mathbf{\theta}_{u1} \\ &- (\frac{1.2}{1} + \frac{3}{351}) \mathbf{U}_2 + (\frac{2}{15} + \frac{8}{105}) \mathbf{\theta}_{u2} \\ \mathbf{e}_{u43} &= (\frac{4.2}{1^2} + \frac{3.6285}{1^2}) \mathbf{U}_1 + (\frac{1}{1} + \frac{32}{351}) \mathbf{\theta}_{u1} \\ &- (\frac{4.2}{1} + \frac{3}{351}) \mathbf{U}_2 + (\frac{2}{15} + \frac{8}{105}) \mathbf{\theta}_{u2} \\ \mathbf{e}_{u43} &= (\frac{4.2}{1^2} + \frac{3.6285}{1^2}) \mathbf{U}_1 + (\frac{1}{1} + \frac{32}{351}) \mathbf{\theta}_{u1} \\ &- (\frac{4.2}{1^2} + \frac{3.6285}{1^2}) \mathbf{U}_2 + (\frac{3.2}{1} + \frac{2.714}{1}) \mathbf{\theta}_{u2} \\ \mathbf{e}_{u44} &= (-\frac{4}{101} - \frac{11}{351}) \mathbf{U}_1 + (-\frac{1}{15} - \frac{2}{35}) \mathbf{\theta}_{u1} \\ &+ (\frac{4}{101} + \frac{11}{351}) \mathbf{U}_2 + (-\frac{1}{3} - \frac{9}{35}) \mathbf{\theta}_{u2} \\ \mathbf{Change} \mathbf{e}_{u11}, \mathbf{e}_{u12}, \mathbf{U}_1, \mathbf{\theta}_{u1}$$

$$R_{\phi 11} = \left(-\frac{1}{3} - \frac{43}{420}\right)\phi_1 + \left(\frac{5}{42}1 - \frac{1}{229}1\right)\chi_1$$
$$+ \left(-\frac{1}{3} - \frac{43}{420}\right)\phi_2 + \left(\frac{5}{42}1 - \frac{1}{229}1\right)\chi_2$$

•

$$\begin{split} & \mathsf{R}_{\varphi 21} = \mathsf{R}_{\varphi 12} = (\frac{5}{841} - \frac{3}{140} 1) \phi_1 + (\frac{1}{168} 1^2 - \frac{1}{1260} 1^2) \chi_1 \\ & + (\frac{5}{84} 1 + \frac{3}{140} 1) \phi_2 + (-\frac{11}{840} 1^2 - \frac{1}{252} 1^2) \chi_2 \\ & \mathsf{R}_{\varphi 22} = (-\frac{1}{84} 1^2 - \frac{1}{210} 1^2) \phi_1 + (0 - \frac{1}{2520} 1^3) \chi_1 \\ & + (\frac{1}{84} 1^2 + \frac{1}{210} 1^2) \phi_2 + (-\frac{1}{420} 1^3 - \frac{1}{1260} 1^2) \chi_2 \\ & \mathsf{R}_{\varphi 32} = \mathsf{R}_{\varphi 23} = (-\frac{17}{420} 1 - \frac{3}{140} 1) \phi_1 + (-\frac{1}{168} 1^2 - \frac{1}{252} 1^2) \chi_1 \\ & + (\frac{17}{420} 1 + \frac{3}{140} 1) \phi_2 + (-\frac{1}{280} 1^2 - \frac{1}{1260} 1^2) \chi_2 \\ & \mathsf{R}_{\varphi 42} = \mathsf{R}_{\varphi 24} = (\frac{13}{420} 1^2 + \frac{1}{60} 1^2) \phi_1 + (\frac{1}{210} 1^3 + \frac{1}{315} 1^3) \chi_1 \\ & + (-\frac{13}{420} 1^2 - \frac{1}{60} 1^2) \phi_2 + (-\frac{1}{420} 1^3 - \frac{1}{1260} 1^3) \chi_2 \\ & \mathsf{R}_{\varphi 33} = (-\frac{1}{3} - \frac{97}{420}) \phi_1 + (-\frac{17}{210} 1 - \frac{10}{167} 1) \chi_1 \\ & + (\frac{1}{3} + \frac{47}{420}) \phi_2 + (\frac{5}{42} 1 - \frac{25}{218} 1) \chi_2 \\ & \mathsf{R}_{\varphi 43} = \mathsf{R}_{\varphi 34} = (\frac{23}{84} 1 + \frac{27}{140} 1) \phi_1 + (\frac{19}{286} 1^2 + \frac{7}{138} 1^2) \chi_1 \\ & + (\frac{23}{84} 1 - \frac{27}{140} 1) \phi_2 + (-\frac{1}{168} 1^2 - \frac{1}{149} 1^2) \chi_2 \\ & \mathsf{R}_{\varphi 444} = (\frac{11}{840} 1^2 + \frac{1}{315} 1^2) \phi_1 + (-\frac{3}{280} 1^3 - \frac{1}{135} 1^3) \chi_1 \\ & + (\frac{11}{840} 1^2 + \frac{1}{315} 1^2) \phi_2 + (0 + \frac{1}{2520} 1^3) \chi_2 \\ & \mathsf{R}_{\varphi 144} = (\frac{11}{840} 1^2 + \frac{1}{315} 1^2) \phi_2 + (0 + \frac{1}{2520} 1^3) \chi_2 \\ & \mathsf{Change} \ \mathsf{R}_{\varphi 111} , \ \mathsf{R}_{\varphi 12} , \{\mbox{ Change}} \ \mathsf{R}_{\varphi 111} , \ \mathsf{R}_{\varphi 12} , \{\mbox{ Change}} \ \mathsf{R}_{\varphi 111} , \ \mathsf{R}_{\varphi 12} , \{\mbox{ Change}} \ \mathsf{R}_{\varphi 111} , \ \mathsf{R}_{\varphi 12} , \{\mbox{ Change}} \ \mathsf{R}_{\varphi 111} , \ \mathsf{R}_{\varphi 12} , \{\mbox{ Change}} \ \mathsf{R}_{\varphi 111} , \ \mathsf{R}_{\varphi 12} , \{\mbox{ Change}} \ \mathsf{R}_{\varphi 111} , \ \mathsf{R}_{\varphi 12} , \{\mbox{ Change}} \ \mathsf{R}_{\varphi 111} , \ \mathsf{R}_{\varphi 12} , \{\mbox{ Change}} \ \mathsf{R}_{\varphi 111} , \ \mathsf{R}_{\varphi 12} , \{\mbox{ Change}} \ \mathsf{R}_{\varphi 111} , \ \mathsf{R}_{\varphi 12} , \{\mbox{ Change}} \ \mathsf{R}_{\varphi 111} , \ \mathsf{R}_{\varphi 12} , \{\mbox{ Change}} \ \mathsf{R}_{\varphi 111} , \ \mathsf{R}_{\varphi 12} , \{\mbox{ Change}} \ \mathsf{R}_{\varphi 111} , \ \mathsf{R}_{\varphi 12} , \{\mbox{ Change}} \ \mathsf{R}_{\varphi 111} , \ \mathsf{R}_{\varphi 12} , \{\mbox{ Change}} \ \mathsf{R}_{\varphi 12} , \{\mbox$$

$$U_{v11} = \left(-\frac{1}{3} - \frac{43}{420}\right)V_1 + \left(-\frac{5}{84} - \frac{3}{140}\right)\theta_{v1}$$
$$+ \left(-\frac{1}{6} - \frac{1}{12}\right)V_2 + \left(\frac{53}{840} - \frac{9}{140}\right)\theta_{v2}$$

$$\begin{split} & \mathsf{U}_{\mathbf{v}12} = (-\frac{5}{84} - \frac{3}{140})\mathsf{V}_1 + (-1/841^2 - \frac{1}{210}\mathsf{1}^2)\,\mathsf{\theta}_{\mathbf{v}1} \\ & + (-\frac{17}{420}\mathbf{1} - \frac{3}{140})\mathsf{V}_2 + (\frac{13}{420}\mathbf{1}^2 - \frac{1}{60}\mathsf{1}^2)\,\mathsf{\theta}_{\mathbf{v}2} \\ & \mathsf{U}_{\mathbf{v}13} = (-\frac{1}{6} - \frac{1}{12})\mathsf{V}_1 + (-\frac{17}{420}\mathbf{1} - \frac{3}{140}\mathsf{1})\,\mathsf{\theta}_{\mathbf{v}1} \\ & + (-\frac{1}{3} - \frac{47}{420})\mathsf{V}_2 + (\frac{23}{84}\mathbf{1} - \frac{27}{140}\mathsf{1})\,\mathsf{\theta}_{\mathbf{v}2} \\ & \mathsf{U}_{\mathbf{v}14} = (\frac{53}{420}\mathbf{1} - \frac{9}{140}\mathsf{1})\mathsf{V}_1 + (\frac{13}{420}\mathsf{1}^2 - \frac{1}{60}\mathsf{1}^2)\,\mathsf{\theta}_{\mathbf{v}1} \\ & + (\frac{23}{84}\mathbf{1} - \frac{27}{140}\mathsf{1})\mathsf{V}_2 + (\frac{11}{840}\mathsf{1}^2 - \frac{1}{315}\mathsf{1}^2)\,\mathsf{\theta}_{\mathbf{v}2} \\ & \mathsf{U}_{\mathbf{v}21} = (\frac{5}{42}\mathbf{1} - \frac{1}{229}\mathsf{1})\mathsf{V}_1 + (\frac{1}{168}\mathsf{1}^2 - \frac{1}{1260}\mathsf{1}^2)\,\mathsf{\theta}_{\mathbf{v}1} \\ & + (\frac{2}{105}\mathbf{1} - \frac{1}{72}\mathsf{1})\mathsf{V}_2 + (\frac{13}{840}\mathsf{1}^2 - \frac{1}{90}\mathsf{1}^2)\,\mathsf{\theta}_{\mathbf{v}2} \\ & \mathsf{U}_{\mathbf{v}22} = (\frac{1}{168}\mathsf{1}^2 - \frac{1}{1260}\mathsf{1}^2)\mathsf{V}_1 + (0 - \frac{1}{2520}\mathsf{1}^3)\,\mathsf{\theta}_{\mathbf{v}1} \\ & + (-\frac{1}{168}\mathsf{1}^2 - \frac{1}{1250}\mathsf{1}^2)\mathsf{V}_2 + (\frac{13}{210}\mathsf{1}^3 - \frac{1}{315}\mathsf{1}^3)\,\mathsf{\theta}_{\mathbf{v}2} \\ & \mathsf{U}_{\mathbf{v}23} = (-\frac{2}{105}\mathsf{1} - \frac{1}{72}\mathsf{1})\mathsf{V}_1 + (-1/168\mathsf{1}^2 - \frac{1}{2520}\mathsf{1}^2)\,\mathsf{\theta}_{\mathbf{v}1} \\ & + (\frac{17}{210}\mathsf{1} - \frac{10}{167}\mathsf{1})\mathsf{V}_2 + (\frac{19}{280}\mathsf{1}^2 - \frac{7}{138}\mathsf{1}^2)\,\mathsf{\theta}_{\mathbf{v}2} \\ & \mathsf{U}_{\mathbf{v}24} = (\frac{13}{840}\mathsf{1}^2 - \frac{1}{90}\mathsf{1}^2)\mathsf{V}_1 + (\frac{1}{210}\mathsf{1}^3 - \frac{1}{315}\mathsf{1}^3)\,\mathsf{\theta}_{\mathbf{v}1} \\ & + (\frac{19}{280}\mathsf{1}^2 - \frac{7}{138}\mathsf{1}^2)\mathsf{V}_2 + (-\frac{3}{280}\mathsf{1}^3 - \frac{1}{135}\mathsf{1}^3)\,\mathsf{\theta}_{\mathbf{v}2} \\ & \mathsf{U}_{\mathbf{v}41} = (-\frac{17}{210} - \frac{9}{428}\mathsf{1})\mathsf{V}_1 + (-\frac{11}{840}\mathsf{1}^2 - \frac{1}{315}\mathsf{1}^3)\,\mathsf{\theta}_{\mathbf{v}2} \\ & \mathsf{U}_{\mathbf{v}42} = (-\frac{11}{840}\mathsf{1}^2 - \frac{1}{252}\mathsf{1}^2)\mathsf{V}_1 + (-\frac{1}{420}\mathsf{1}^3 - \frac{1}{315}\mathsf{1}^3)\,\mathsf{\theta}_{\mathbf{v}1} \\ & + (-\frac{2}{105}\mathsf{1} - \frac{1}{194}\mathsf{1})\mathsf{V}_2 + (\frac{11}{840}\mathsf{1}^2 - \frac{1}{315}\mathsf{1}^2)\,\mathsf{\theta}_{\mathbf{v}2} \\ & \mathsf{U}_{\mathbf{v}42} = (-\frac{11}{840}\mathsf{1}^2 - \frac{1}{252}\mathsf{1}^2)\mathsf{V}_1 + (-\frac{1}{420}\mathsf{1}^3 - \frac{1}{1260}\mathsf{1}^3)\,\mathsf{\theta}_{\mathbf{v}1} \\ & + (-\frac{1}{280}\mathsf{1}^2 - \frac{1}{1260}\mathsf{1}^2)\mathsf{V}_2 + (-\frac{1}{420}\mathsf{1}^3 - \frac{1}{1260}\mathsf{1}^3)\,\mathsf{\theta}_{\mathbf{v}2} \\ \end{split}$$

$$U_{v43} = \left(-\frac{2}{105} 1 + \frac{1}{194} 1\right) V_1 + \left(-\frac{1}{280} 1^2 + \frac{1}{1260} 1^2\right) \theta_{v1}$$

$$+ \left(\frac{5}{42} 1 + \frac{25}{218} 1\right) V_2 + \left(-\frac{1}{168} 1^2 + \frac{1}{149} 1^2\right) \theta_{v2}$$

$$U_{v44} = \left(\frac{11}{840} 1^2 + \frac{1}{315} 1\right) V_1 + \left(-\frac{1}{420} 1^3 + \frac{1}{1260} 1^3\right) \theta_{v1}$$

$$+ \left(-\frac{1}{168} 1^2 + \frac{1}{149} 1^2\right) V_2 + \left(0 + \frac{1}{2520} 1^3\right) \theta_{v2}$$

$$U_{v11} = -U_{v11} + U_{v32} = U_{v12} + U_{v33} = -U_{v13} + U_{v34} = -U_{v41}$$
Change terms such as  $U_{u11}$ ,  $U_{u12}$ ,  $U_1$ , and  $\theta_{u1}$  to either  $U_{\phi11}$ ,  $U_{\phi12}$ ,  $\Phi_1$ , and  $\chi_1$  or to  $U_{v11}$ ,  $U_{v12}$ ,  $V_1$ , and  $\theta_{v1}$ .

$$\begin{split} & X_{u1} = (-\frac{1\cdot2}{1^2} - \frac{1}{101^2})U_1 + (\frac{1\cdot1}{1} - \frac{2}{101})\theta_{u1} \\ & + (-\frac{1\cdot2}{1^2} - \frac{1}{101^2})U_2 + (-\frac{4}{101} - \frac{2}{101})\theta_{u2} \\ & X_{u2} = (\frac{1}{101} - 0)U_1 + (\frac{2}{15} - \frac{1}{30})\theta_{u1} \\ & + (-\frac{1}{101} - 0)U_2 + (-\frac{1}{30} - \frac{1}{30})\theta_{u2} \\ & X_{u3} = (-\frac{1\cdot2}{1^2} - \frac{11}{101^2})U_1 + (\frac{1}{101} - \frac{2}{101})\theta_{u1} \\ & + (\frac{1\cdot2}{1^2} - \frac{11}{101^2})U_2 + (-\frac{11}{101} - \frac{9}{701})\theta_{u2} \\ & X_{u3} = (\frac{1}{101} - \frac{1}{101}U_1 + (-\frac{1}{30} - 0)\theta_{u1} \\ & + (-\frac{1}{101} - \frac{1}{101})U_2 + (-\frac{1}{30} - \frac{1}{30})\theta_{u2} \\ & X_{u3} = (\frac{1}{101} - \frac{1}{101})U_2 + (-\frac{1}{30} - \frac{1}{30})\theta_{u2} \\ & Change X_{u1}, X_{u2} to X_{v1} and X_{v2}, similar procedure for X_{\phi1} and \\ & X_{\phi2} \text{ by changing } V_1, \theta_{v1}, \Phi_1, \chi_1 \text{ instead of } U_1 \text{ and } \theta_{u1} \text{ respectively.} \end{split}$$

$$\begin{split} \mathsf{B}_{u1} &= (.5 \quad \frac{13}{70})\mathsf{U}_1 + (-\frac{1}{10}\mathsf{l} \quad -\frac{103}{210}\mathsf{l})_{\theta_{u1}} \\ &+ (-.5 \quad -\frac{13}{70})\mathsf{U}_2 + (\frac{1}{10}\mathsf{l} \quad \frac{11}{420})_{\theta_{u2}} \\ \mathsf{B}_{u2} &= (\frac{1}{10}\mathsf{l} \quad \frac{3}{70}\mathsf{l})\mathsf{U}_1 + (-.5\mathsf{l}^2 \quad -\frac{17}{105}\mathsf{l}^2)_{\theta_{u1}} \\ &+ (-\frac{1}{10}\mathsf{l} \quad \frac{3}{70}\mathsf{l})\mathsf{U}_2 + (\frac{1}{60}\mathsf{l} \quad \frac{3}{140}\mathsf{l})_{\theta_{u2}} \\ \mathsf{B}_{u3} &= (.5 \quad \frac{11}{35})\mathsf{U}_1 + (\frac{1}{10}\mathsf{l} \quad \frac{3}{420}\mathsf{l})_{\theta_{u1}} \\ &+ (-.5 \quad -\frac{11}{35})\mathsf{U}_2 + (-\frac{1}{10}\mathsf{l} \quad \frac{23}{210})_{\theta_{u2}} \\ \mathsf{B}_{u4} &= (-\frac{1}{10}\mathsf{l} \quad -\frac{2}{35}\mathsf{l})\mathsf{U}_1 + (-\frac{1}{60}\mathsf{l}^2 \quad -\frac{1}{84}\mathsf{l}^2)_{\theta_{u1}} \\ &+ (\frac{1}{10}\mathsf{l} \quad \frac{2}{35}\mathsf{l})\mathsf{U}_2 + (0 \quad \frac{1}{210}\mathsf{l})_{\theta_{u2}} \\ \mathsf{Change} \quad \mathsf{B}_{u1}, \quad \mathsf{B}_{u2} \quad \mathsf{to} \quad \mathsf{B}_{v1}, \quad \mathsf{B}_{v2}, \quad \mathsf{B}_{\phi1}, \quad \mathsf{B}_{\phi2} \text{ and at the same time} \\ \mathsf{replace} \quad \mathsf{U}_1, \quad \theta_{u1} \text{ by either } \mathsf{V}_1, \quad \theta_{v1} \text{ or by } \Phi_1, \quad \mathsf{X}_{u1} \text{ respectively.} \end{split}$$

$$\begin{aligned} C_{u1} &= (.5 \quad \frac{13}{70})U_1 + (\frac{1}{10}1 \quad \frac{3}{70}1)\theta_{u1} \\ &+ (.5 \quad \frac{11}{35})U_2 + (-\frac{1}{10}1 \quad -\frac{2}{35})\theta_{u2} \\ C_{u2} &= (-\frac{1}{10}1 \quad -\frac{103}{210}1)U_1 + (-.51^2 \quad -\frac{17}{105}1^2)\theta_{u1} \\ &+ (-\frac{1}{10}1 \quad \frac{3}{420}1)U_2 + (-\frac{1}{60}1 \quad \frac{1}{84}1)\theta_{u2} \\ C_{u3} &= (.5 \quad \frac{11}{70})U_1 + (-\frac{1}{10}1 \quad -\frac{3}{70}1)\theta_{u1} \\ &+ (-.5 \quad -\frac{11}{35})U_2 + (\frac{1}{10}1 \quad \frac{2}{35})\theta_{u2} \\ C_{u4} &= (\frac{1}{70}1 \quad -\frac{11}{420}1)U_1 + (\frac{1}{60}1^2 \quad \frac{3}{140}1^2)\theta_{u1} \\ &+ (-\frac{1}{10}1 \quad -\frac{23}{210}1)U_2 + (0 \quad \frac{1}{210}1)\theta_{u2} \\ & \text{Change B}_{u1}, B_{u2} \text{ to } B_{v1}, B_{v2}, B_{\phi1}, B_{\phi2} \text{ and at the same time} \end{aligned}$$

١,

replace  $U_1$ ,  $\theta_{u1}$  by either  $V_1$ ,  $\theta_{v1}$  or by  $\phi_1$ ,  $\chi_{u1}$  respectively.

۰.

•

$$\begin{split} \mathbf{v}_{\phi 11} &= (-\frac{1}{351^2} - \frac{61}{1051^2}) \mathbf{v}_1 + (\frac{3}{351} - 0) \mathbf{x}_1 \\ &+ (\frac{27}{351^2} - \frac{24}{351^2}) \mathbf{v}_2 + (\frac{3}{351} - \frac{3}{351}) \mathbf{x}_2 \\ \mathbf{v}_{\phi 12} &= (-\frac{5}{71} - \frac{3}{351}) \mathbf{v}_1 + (\frac{-2}{35} - 1/10) \mathbf{x}_1 \\ &+ (\frac{5}{71} - \frac{22}{351}) \mathbf{v}_2 + (-\frac{2}{35} - \frac{3}{35}) \mathbf{x}_2 \\ \mathbf{v}_{\phi 21} &= (\frac{31}{351} - \frac{8}{351}) \mathbf{v}_1 + (\frac{1}{7} - \frac{3}{70}) \mathbf{x}_1 \\ &+ (\frac{4}{351} - \frac{1}{351}) \mathbf{v}_2 + (-\frac{2}{35} - \frac{1}{70}) \mathbf{x}_2 \\ \mathbf{v}_{\phi 14} &= (\frac{2}{141} - \frac{1}{141}) \mathbf{v}_1 + (\frac{3}{10} - \frac{1}{70}) \mathbf{x}_1 \\ &+ (\frac{5}{71} - \frac{22}{351}) \mathbf{v}_2 + (-\frac{2}{35} - \frac{1}{70}) \mathbf{x}_2 \\ \mathbf{v}_{\phi 14} &= (\frac{2}{141} - \frac{1}{141}) \mathbf{v}_1 + (\frac{3}{10} - \frac{1}{70}) \mathbf{x}_1 \\ &+ (\frac{5}{71} - \frac{22}{351}) \mathbf{v}_2 + (-\frac{2}{35} - \frac{3}{70}) \mathbf{x}_2 \\ \mathbf{v}_{\phi 22} &= (-\frac{16}{35} - \frac{1}{30}) \mathbf{v}_1 + (-\frac{1}{351} - \frac{1}{140}) \mathbf{x}_1 \\ &+ (-\frac{3}{70} - \frac{1}{30}) \mathbf{v}_2 + (\frac{1}{210} - \frac{1}{420}) \mathbf{x}_2 \\ \mathbf{v}_{\phi 24} &= (\frac{17}{35} - \frac{13}{210}) \mathbf{v}_1 + (\frac{4}{1051} - \frac{1}{644}) \mathbf{x}_1 \\ &+ (\frac{9}{35} - \frac{43}{210}) \mathbf{v}_2 + (-\frac{1}{3} - \frac{1}{60}) \mathbf{x}_2 \\ \mathbf{v}_{\phi 44} &= (\frac{3}{70} - \frac{1}{105}) \mathbf{v}_1 + (\frac{1}{2101} - \frac{1}{420}) \mathbf{x}_1 \\ &+ (\frac{16}{35} - \frac{49}{210}) \mathbf{v}_2 + (-\frac{1}{351} - \frac{11}{1420}) \mathbf{x}_2 \\ \mathbf{v}_{\phi 13} \mathbf{z} - \mathbf{v}_{\phi 11} - \mathbf{v}_{\phi 23} - \mathbf{v}_{\phi 34} - \mathbf{v}_{\phi 34} - \mathbf{v}_{\phi 34} \\ \mathbf{v}_{\phi 13} \mathbf{z} - \mathbf{v}_{\phi 11} - \mathbf{v}_{\phi 34} - \mathbf{v}_{\phi 34} - \mathbf{v}_{\phi 34} \\ \mathbf{v}_{\phi 13} \mathbf{z} - \mathbf{v}_{\phi 11} - \mathbf{v}_{\phi 34} - \mathbf{v}_{\phi 34} - \mathbf{v}_{\phi 34} - \mathbf{v}_{\phi 14} \\ \mathbf{v}_{hange} - \mathbf{v}_{\phi 11}, \mathbf{v}_{\phi 12}$$
 to  $\mathbf{v}_{v 11}, \mathbf{v}_{v 12}, \mathbf{v}_{u 11}, \mathbf{v}_{u 12}$ , and at the same time replace  $\mathbf{v}_1, \mathbf{x}_{u 1}$  by either  $\mathbf{v}_1, \mathbf{v}_{v 1}$  or  $\mathbf{u}_1, \mathbf{e}_{u 1}$  respectively

$$\begin{split} m_{u11} &= (\frac{.6}{1} - \frac{17}{701})U_1 + (\frac{9}{70} - \frac{2}{35})\theta_{u1} \\ &+ (\frac{.6}{1} - \frac{5}{141})U_2 + (-\frac{9}{70}1 - \frac{1}{14}1)\theta_{u2} \\ m_{\phi 12} &= (-\frac{1}{10} - \frac{3}{140})U_1 + (\frac{1}{14}1 - \frac{3}{70}1)\theta_{u1} \\ &+ (\frac{4}{35} - \frac{11}{140})U_2(-\frac{3}{140}1 - \frac{-1}{70}1)\theta_{u2} \\ m_{\phi 14} &= (\frac{4}{35} - \frac{1}{28})U_1 + (\frac{3}{140}1 - \frac{1}{140}1)\theta_{u1} \\ &+ (-\frac{1}{70} - \frac{1}{28})U_2(-\frac{-1}{140}1 - \frac{3}{4}1)\theta_{u2} \\ m_{\phi 22} &= (\frac{43}{420}1 - \frac{11}{840}1)U_1 + (\frac{1}{120}1^2 - \frac{1}{488}1^2)\theta_{u1} \\ &+ (\frac{2}{65}1 - \frac{17}{840}1)U_2 + (-\frac{1}{169}1^2 - \frac{-1}{280}1^2)\theta_{u2} \\ m_{\phi 24} &= (-\frac{1}{60}1 - \frac{1}{840}1)U_1 + (-\frac{1}{840}1^2 - 0)\theta_{u1} \\ &+ (-\frac{1}{60}1 - \frac{13}{840}1)U_2 + (\frac{1}{120}1^2 - \frac{1}{840}1^2)\theta_{u2} \\ m_{\phi 44} &= (\frac{43}{420}1 - \frac{9}{840}1)U_1 + (\frac{1}{168}1^2 - \frac{1}{420}1^2)\theta_{u2} \\ m_{\psi 13} &= -m_{u11} - \frac{m_{u22} - m_{u21}}{m_{u21} - m_{u23}} - \frac{m_{u31} - m_{u14}}{m_{u41} - m_{u14}} \end{split}$$

Change  $m_{u11}$ ,  $m_{u12}$  to  $m_{v11}$ ,  $v_{v12}$ ,  $m_{\phi11}$ ,  $m_{\phi12}$ , and at the same time replace  $U_1$ ,  $\theta_{u1}$  by either  $V_1$ ,  $\theta_{v1}$  or  $\Phi_1$ ,  $\chi_{u1}$  respectively

$$n_{\phi 11} = \left(\frac{.6}{1} \quad \frac{17}{701}\right)\phi_1 + \left(-\frac{1}{70} \quad \frac{3}{140}\right)\chi_1$$

 $m_{u42} = m_{u24} = m_{u43} = m_{u34}$ 

$$\begin{split} u^{\phi \pi 3} &= \left(-\frac{L_0}{1} - \frac{59}{10}\right) \phi^T + \left(-\frac{90}{1} - \frac{8\pi 0}{13}\right) \chi^T \\ &+ \left(-\frac{1\pi 0}{3} - \frac{1\pi 0}{1}\right) \frac{1\pi 0}{1}\right) \phi^T + \left(-\frac{19}{10} + \frac{\pi 0}{1}\right) \chi^T - \frac{\pi 0}{3} \chi^T \chi^T \\ &+ \left(-\frac{32}{3} - \frac{58}{3}\right) \phi^T + \left(-\frac{90}{1}\right) - \frac{8\pi 0}{3}\right) \chi^T \\ &+ \left(-\frac{32}{\pi} - \frac{58}{3}\right) \phi^T + \left(-\frac{90}{1}\right) - \frac{8\pi 0}{3}\right) \chi^T \\ &+ \left(-\frac{32}{\pi} - \frac{58}{1}\right) \phi^T + \left(-\frac{90}{1}\right) - \frac{8\pi 0}{3}\right) \chi^T \\ &+ \left(-\frac{32}{\pi} - \frac{1\pi 0}{1}\right) + \frac{1\pi 0}{1}\right) \phi^T + \left(-\frac{90}{1}\right) - \frac{8\pi 0}{3}\right) \chi^T \\ &+ \left(-\frac{32}{\pi} - \frac{1\pi 0}{1}\right) \phi^T + \left(-\frac{90}{1}\right) - \frac{8\pi 0}{1}\right) \chi^T \\ &+ \left(-\frac{32}{\pi} - \frac{1\pi 0}{1}\right) \phi^T + \left(-\frac{90}{1}\right) - \frac{8\pi 0}{1}\right) \chi^T \\ &+ \left(-\frac{32}{\pi} - \frac{1\pi 0}{1}\right) \phi^T + \left(-\frac{90}{1}\right) - \frac{8\pi 0}{1}\right) \chi^T \\ &+ \left(-\frac{32}{\pi} - \frac{1\pi 0}{1}\right) \phi^T + \left(-\frac{90}{1}\right) - \frac{8\pi 0}{1}\right) \chi^T \\ &+ \left(-\frac{32}{\pi} - \frac{1\pi 0}{1}\right) \phi^T + \left(-\frac{1}{\pi}\right) - \frac{8\pi 0}{1}\right) \chi^T \\ &+ \left(-\frac{1}{\pi} - \frac{1}{2}\right) \phi^T + \left(-\frac{1}{\pi}\right) - \frac{8\pi 0}{1}\right) \chi^T \\ &+ \left(-\frac{1}{\pi} - \frac{1}{2}\right) \phi^T + \left(-\frac{1}{\pi}\right) - \frac{1}{\pi^2}\right) \chi^T \\ &+ \left(-\frac{1}{\pi} - \frac{1}{\pi^2}\right) \phi^T + \left(-\frac{1}{\pi^0}\right) - \frac{\pi 0}{1}\right) \chi^T \\ &+ \left(-\frac{1}{\pi} - \frac{1}{\pi^2}\right) \phi^T + \left(-\frac{1\pi 0}{1}\right) - \frac{\pi 0}{1}\right) \chi^T \\ &+ \left(-\frac{1}{\pi} - \frac{1}{\pi^2}\right) \phi^T + \left(-\frac{1\pi 0}{\pi^2}\right) - \frac{\pi 0}{\pi^2}\right) \chi^T \\ &+ \left(-\frac{1}{\pi} - \frac{1}{\pi^2}\right) \phi^T + \left(-\frac{1\pi 0}{\pi^2}\right) - \frac{\pi 0}{\pi^2}\right) \chi^T \\ &+ \left(-\frac{1}{\pi} - \frac{1}{\pi^2}\right) \phi^T + \left(-\frac{1\pi 0}{\pi^2}\right) - \frac{\pi 0}{\pi^2}\right) \chi^T \\ &+ \left(-\frac{1}{\pi} - \frac{1}{\pi^2}\right) \phi^T + \left(-\frac{1\pi 0}{\pi^2}\right) - \frac{\pi 0}{\pi^2}\right) \chi^T \\ &+ \left(-\frac{1}{\pi^2} - \frac{1\pi 0}{\pi^2}\right) \phi^T + \left(-\frac{1\pi 0}{\pi^2}\right) - \frac{\pi 0}{\pi^2}\right) \chi^T \\ &+ \left(-\frac{1}{\pi^2} - \frac{\pi 0}{\pi^2}\right) \psi^T + \left(-\frac{1\pi 0}{\pi^2}\right) - \frac{\pi 0}{\pi^2}\right) \chi^T \\ &+ \left(-\frac{\pi 0}{\pi^2} - \frac{\pi 0}{\pi^2}\right) \psi^T + \left(-\frac{\pi 0}{\pi^2}\right) - \frac{\pi 0}{\pi^2}\right) \chi^T \\ &+ \left(-\frac{\pi 0}{\pi^2} - \frac{\pi 0}{\pi^2}\right) \psi^T + \left(-\frac{\pi 0}{\pi^2}\right) - \frac{\pi 0}{\pi^2}\right) \chi^T \\ &+ \left(-\frac{\pi 0}{\pi^2} - \frac{\pi 0}{\pi^2}\right) \psi^T + \left(-\frac{\pi 0}{\pi^2}\right) - \frac{\pi 0}{\pi^2}\right) \chi^T \\ &+ \left(-\frac{\pi 0}{\pi^2} - \frac{\pi 0}{\pi^2}\right) \psi^T + \left(-\frac{\pi 0}{\pi^2}\right) - \frac{\pi 0}{\pi^2}\right) \chi^T \\ &+ \left(-\frac{\pi 0}{\pi^2} - \frac{\pi 0}{\pi^2}\right) \psi^T + \left(-\frac{\pi 0}{\pi^2}\right) - \frac{\pi 0}{\pi^2}\right) \chi^T \\ &+ \left(-\frac{\pi 0}{\pi^2} - \frac{\pi 0}{\pi^2}\right) \psi^T + \left(-\frac{\pi 0}{\pi^2}\right) - \frac{\pi 0}{\pi^2}\right) \chi^T \\ &+ \left(-\frac{\pi 0}{\pi$$

•

$$+ (\frac{1}{70} - \frac{1}{28})\phi_{2} + (\frac{61}{130}1 - \frac{3}{8}1)\chi_{2}$$

$$n_{\phi 4 4} = (-\frac{1}{140}1 - \frac{3}{4}1)\phi_{1} + (\frac{1}{840}1^{2} - \frac{1}{840}1^{2})\chi_{1}$$

$$+ (\frac{1}{140}1 - \frac{3}{4}1)\phi_{2} + (-\frac{1}{120}1^{2} - \frac{1}{168}1^{2})\chi_{2}$$

$$\cdot$$

$$n_{\phi 31} = -n_{\phi 11} - n_{\phi 32} = -n_{\phi 12} - n_{\phi 33} = -n_{\phi 13} - n_{\phi 34} = -n_{\phi 14}$$

Change  $n_{\phi 11}$ ,  $n_{\phi 12}$  to  $n_{v11}$ ,  $n_{v12}$ ,  $n_{u11}$ ,  $n_{u12}$ , and at the same time replace  $\Phi_1$ ,  $\chi_{u1}$  by either  $V_1$ ,  $\theta_{v1}$  or  $U_1$ ,  $\theta_{u1}$  respectively

$$\begin{aligned} f_{u11} &= (\frac{6}{1^3} - \frac{33}{351^3})U_1 + (\frac{.6}{1^2} - \frac{6}{351^2})\theta_{u1} \\ &+ (\frac{6}{1^3} - \frac{5.057}{1^3})U_2 + (-\frac{.6}{1^2} - \frac{3}{71^2})\theta_{u2} \\ f_{u12} &= (\frac{4.2}{1^2} - \frac{4}{71^2})U_1 + (-\frac{1}{101} - \frac{11}{351})\theta_{u1} \\ &+ (\frac{1.8}{1^2} - \frac{1.42857}{1^2})U_2 + (-\frac{.2}{1} - \frac{4}{351})\theta_{u2} \\ f_{u13} &= (\frac{1.8}{1^2} - \frac{13}{351^2})U_1 + (\frac{.2}{1} - \frac{3}{351})\theta_{u1} \\ &+ (\frac{4.2}{1^2} - \frac{3.62857}{1^2})U_2 + (-\frac{.4}{1} - \frac{11}{351})\theta_{u2} \\ f_{u22} &= (\frac{2.2}{1} - \frac{11}{351})U_1 + (\frac{1}{3} - \frac{8}{105})\theta_{u1} \\ &+ (\frac{.8}{1} - \frac{1}{351})U_2 + (-\frac{2}{15} - \frac{2}{35})\theta_2 \\ f_{u24} &= (\frac{1}{1} - \frac{3}{351})U_1 + (\frac{1}{15} - \frac{1}{105})\theta_{u1} \\ &+ (\frac{1}{1} - \frac{32}{351})U_2 + (-\frac{1}{15} - \frac{2}{35})\theta_2 \\ f_{u33} &= (\frac{6}{1^3} - \frac{33}{751^3})U_1 + (\frac{.6}{1^2} - \frac{6}{351^2})\theta_{u1} \\ &+ (\frac{.6}{1^3} - \frac{5.057}{1^3})U_2 + (-\frac{.6}{1^2} - \frac{3}{71^2})\theta_{u2} \\ f_{u12} &= (-\frac{1.8}{1^2} - \frac{13}{351^2})U_1 + (-\frac{2}{101} - \frac{3}{351})\theta_{u1} \end{aligned}$$

$$\begin{aligned} &+(\frac{4\cdot2}{12}-\frac{3\cdot62857}{12})U_2+(\frac{4}{1}-\frac{11}{351})\theta_{u2} \\ f_{u44} &=(\frac{4\cdot8}{1}-\frac{2}{71})U_1+(\frac{2}{15}-\frac{8}{105})\theta_{u1} \\ &+(\frac{3\cdot2}{1}-\frac{2\cdot714}{1})U_2+(-\frac{3}{15}-\frac{9}{35})\theta_2 \\ f_{u21} &= f_{u12} - f_{u31} &= f_{u13} - f_{u14} \\ f_{u32} &= f_{u23} - f_{u42} &= f_{u24} - f_{u43} &= f_{u34} - f_{u32} \\ f_{u32} &= f_{u23} - f_{u12} + f_{u14} - f_{u43} &= f_{u34} - f_{u32} \\ &- f_{u11} - f_{u12} - f_{u12} - f_{u12} + f_{u14} - f_{u23} \\ &- f_{u12} - f_{u13} - f_{u14} - f_{u13} - f_{u14} - f_{u13} - f_{u12} \\ &- f_{u32} - f_{u12} - f_{u14} - f_{u14} - f_{u14} - f_{u23} \\ &- f_{u32} - f_{u14} \\ &- f_{u32} - f_{u14} -$$

$$W_{v11} = \left(-\frac{1 \cdot 543}{1^2} - \frac{27}{351^2}\right) V_1 + \left(-\frac{6}{351} - \frac{9}{701}\right) \theta_1$$

$$+ \left(\frac{1 \cdot 543}{1^2} - \frac{27}{351^2}\right) V_2 + \left(-\frac{6}{351} - \frac{9}{2101}\right) \theta_2$$

$$W_{v12} = \left(-\frac{6}{351} - \frac{9}{701}\right) V_1 + \left(\frac{3}{35} - \frac{1}{28}\right) \theta_{v1}$$

$$+ \left(\frac{6}{351} - \frac{9}{701}\right) V_2 + \left(\frac{1}{70} - \frac{1}{140}\right) \theta_{v2}$$

$$W_{v13} = \left(\frac{1 \cdot 543}{1^2} - \frac{27}{351^2}\right) V_1 + \left(\frac{6}{351} - \frac{9}{701}\right) \theta_1$$

$$+ \left(-\frac{1 \cdot 543}{1^2} - \frac{27}{351^2}\right) V_2 + \left(\frac{6}{351} - \frac{7}{301}\right) \theta_2$$

$$W_{v14} = \left(-\frac{6}{351} - \frac{9}{2101}\right) V_1 + \left(\frac{6}{35} - \frac{1}{70}9\theta_{v1}\right)$$

$$+ \left(\frac{6}{351} - \frac{9}{701}\right) V_2 + \left(-\frac{3}{35} - \frac{1}{20}\right) \theta_{v2}$$

$$W_{v22} = \left(-\frac{3}{35} - \frac{1}{28}\right) V_1 + \left(-\frac{2}{351} - \frac{1}{280}1\right) \theta_{v1}$$

$$+ \left(\frac{3}{35} - \frac{1}{28}\right) V_1 + \left(-\frac{1}{1051} - \frac{1}{840}1\right) \theta_{v1}$$

$$W_{v23} = \left(\frac{6}{351} - \frac{9}{701}\right) V_1 + \left(\frac{3}{35} - \frac{1}{28}\right) \theta_{v1}$$

۰,

$$\begin{aligned} &+(\frac{6}{351} - \frac{9}{701})V_2 + (-\frac{1}{70} - \frac{1}{140})\theta_{V2} \\ W_{V24} &= (\frac{1}{35} - \frac{1}{140})V_1 + (-\frac{1}{105}1 - \frac{1}{840}1)\theta_{V1} \\ &+ (-\frac{1}{70} - \frac{1}{140})V_1 + (-\frac{1}{105}1 - \frac{9}{840}1)\theta_{V1} \\ W_{V33} &= (-\frac{1.543}{1^2} - \frac{27}{351^2})V_1 + (-\frac{6}{351} - \frac{9}{701})\theta_1 \\ &+ (\frac{1.543}{1^2} - \frac{27}{351^2})V_2 + (-\frac{6}{351} - \frac{3}{701})\theta_2 \\ W_{V34} &= (\frac{6}{351} - \frac{9}{701})V_1 + (-\frac{1}{70} - \frac{2}{70})\theta_{V1} \\ &+ (-\frac{6}{351} - \frac{3}{701})V_2 + (\frac{3}{35} - \frac{1}{20})\theta_{V2} \\ W_{V44} &= (-\frac{3}{35} - \frac{1}{20})V_1 + (-\frac{1}{105}1 - \frac{9}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{2}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{2}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{2}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{2}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{2}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{2}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{2}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{2}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{2}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{2}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{2}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{2}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{2}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{2}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{3}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{3}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{35} - \frac{1}{20})V_1 + (\frac{3}{351} - \frac{51}{840}1)\theta_{V1} \\ &+ (\frac{3}{351} - \frac{1}{840})V_1 \\ &+ (\frac{3}{351} - \frac{1}{351} - \frac{51}{840})V_1 \\ &+ (\frac{3}{351} - \frac{1}{351} - \frac{51}{840})V_1 \\ &+ (\frac{3}{351} - \frac{1}{351} - \frac{51}{840})V_1 \\ &+ (\frac{3}{351} - \frac{1}{351} - \frac{1}{351} - \frac{51}{351} - \frac{51}{$$

$$D_{u11} = (0 \quad \frac{18}{351^3})U_1 + (\frac{1 \cdot 2}{1^2} \quad \frac{12}{351^2})\theta_{u1}$$

$$+ (0 \quad \frac{18}{351^3})U_2 + (\frac{1 \cdot 2}{1^2} \quad \frac{6}{71^2})\theta_{u2}$$

$$D_{u12} = (\frac{1 \cdot 2}{1^2} \quad \frac{1 \cdot 286}{1^2})U_1 + (\frac{2}{101} \quad \frac{1}{71})\theta_{u1}$$

$$+ (\frac{1 \cdot 2}{1^2} \quad \frac{1 \cdot 286}{1^2})U_2 + (\frac{1 \cdot 4}{1} \quad \frac{1 \cdot 1428}{1})\theta_{u2}$$

$$D_{14} = (\frac{\cdot 6}{1^2} \quad \frac{12}{351^2})U_1 + (\frac{\cdot 4}{1} \quad \frac{6}{351})\theta_{u1}$$

$$+ (\frac{1 \cdot 6}{1^2} \quad \frac{12}{351^2})U_2 + (\frac{-2}{1} \quad \frac{6}{351})\theta_{u2}$$

-312-

. .

:

`,

.

$$\begin{split} & \mathsf{D}_{u22} = (\frac{4}{101} - \frac{1}{351})\mathsf{U}_1 + (-\frac{1}{3} - \frac{2}{205})\mathsf{\theta}_{u1} \\ & + (\frac{4}{101} - \frac{1}{351})\mathsf{U}_2 + (-\frac{1}{15} - \frac{1}{21})\mathsf{\theta}_{u2} \\ & \mathsf{D}_{u23} = (-\frac{16}{1^2} - \frac{9}{351^2})\mathsf{U}_1 + (-\frac{2}{101} - \frac{1}{351})\mathsf{\theta}_{u1} \\ & + (\frac{16}{1^2} - \frac{9}{351^2})\mathsf{U}_2 + (-\frac{4}{101} - \frac{8}{351})\mathsf{\theta}_{u2} \\ & \mathsf{D}_{u24} = (0 - \frac{1}{141})\mathsf{U}_1 + (\frac{1}{30} - \frac{2}{205})\mathsf{\theta}_{u1} \\ & + (0 - \frac{1}{141})\mathsf{U}_2 + (-\frac{1}{30} - \frac{11}{210})\mathsf{\theta}_{u2} \\ & \mathsf{D}_{33} = (0 - \frac{18}{351^3})\mathsf{U}_1 + (-\frac{12}{1^2} - \frac{12}{351^2})\mathsf{\theta}_{u1} \\ & + (0 - \frac{18}{351^3})\mathsf{U}_2 + (\frac{1.2}{1^2} - \frac{6}{71^2})\mathsf{\theta}_{u2} \\ & \mathsf{D}_{34} = (\frac{12}{351^2} - \frac{1.2}{351^2})\mathsf{U}_1 + (\frac{4}{1} - \frac{6}{351})\mathsf{\theta}_{u1} \\ & + (-\frac{12}{351^2} - \frac{6}{71} - \frac{18}{351})\mathsf{U}_2 + (\frac{1.2}{1^2} - \frac{6}{71^2})\mathsf{\theta}_{u2} \\ & \mathsf{D}_{44} = (\frac{4}{101} - \frac{5}{281})\mathsf{U}_1 + (\frac{1}{3} - \frac{4}{35})\mathsf{\theta}_{u1} \\ & + (-\frac{4}{101} - \frac{5}{281})\mathsf{U}_2 + (\frac{1}{3} - \frac{11}{35})\mathsf{\theta}_{u2} \\ & \mathsf{D}_{u13}=\mathsf{D}_{u11} - \mathsf{D}_{u21}=\mathsf{D}_{u12} - \mathsf{D}_{u31}=\mathsf{D}_{u13} - \mathsf{D}_{u32}=\mathsf{D}_{u23} \\ & \mathsf{D}_{u41}=\mathsf{D}_{u14} - \mathsf{D}_{u42}=\mathsf{D}_{u24} - \mathsf{D}_{u43}=\mathsf{D}_{u34} \\ & \mathsf{Change the terms }\mathsf{u}_1, \, \mathsf{\theta}_{u1}, \, \mathsf{U}_2, \, \mathsf{\theta}_{u2} \text{ with } \mathsf{D}_{u11}, \, \mathsf{D}_{u12}, \text{ etc. to} \\ \end{split}$$

 $V_1$ ,  $\theta_{v1}$ ,  $V_2$ ,  $\theta_{v2}$ ,  $D_{v11}$ , etc. Similar procedure for  $D_{\phi11}$ , etc.

$$T_{u11} = \left(-\frac{1}{351^2} - \frac{61}{1051^2}\right) U_1 + \left(\frac{31}{351} - \frac{8}{351}\right)^{\theta} u_1 \\ + \left(\frac{1}{351^2} - \frac{61}{1051^2}\right) U_1 + \left(-\frac{4}{351} - \frac{1}{71}\right)^{\theta} u_1$$

1

$$\begin{split} & T_{u} 12 = (\frac{3}{351} \quad 0) U_{1} + (\frac{1}{7} \quad \frac{3}{70}) \theta_{1} \\ & + (\frac{3}{351} \quad 0) U_{2} + (-\frac{2}{35} \quad -\frac{3}{70}) \theta_{u2} \\ & T_{u13} = (-\frac{27}{351^{2}} \quad -\frac{24}{351^{2}}) U_{1} + (\frac{4}{351} \quad -\frac{1}{351}) \theta_{u1} \\ & + (\frac{27}{351^{2}} \quad \frac{24}{351^{2}}) U_{1} + (-\frac{31}{351} \quad -\frac{23}{351}) \theta_{u1} \\ & T_{u} 14 = (\frac{3}{351} \quad \frac{3}{351}) U_{1} + (-\frac{2}{35} \quad -\frac{1}{70}) \theta_{1} \\ & + (\frac{3}{351} \quad -\frac{3}{351}) U_{2} + (\frac{1}{7} \quad -\frac{1}{7}) \theta_{u2} \\ & T_{u21} = (-\frac{5}{71} \quad \frac{3}{351}) U_{2} + (-\frac{1}{7} \quad -\frac{1}{7}) \theta_{u2} \\ & T_{u22} = (-\frac{2}{35} \quad -\frac{1}{70}) U_{1} + (-\frac{16}{35} \quad -\frac{1}{30}) \theta_{1} \\ & + (\frac{5}{71} \quad -\frac{3}{351}) U_{2} + (-\frac{9}{35} \quad -\frac{11}{210}) \theta_{u2} \\ & T_{u22} = (-\frac{2}{35} \quad -\frac{1}{70}) U_{1} + (-\frac{1}{351} \quad -\frac{1}{420}1) \theta_{u1} \\ & + (\frac{2}{35} \quad \frac{1}{70}) U_{u2} + (-\frac{1}{351} \quad -\frac{1}{420}1) \theta_{u2} \\ & T_{u23} = (-\frac{2}{71} \quad -\frac{3}{141}) U_{1} + (-\frac{3}{70} \quad -\frac{1}{30}) \theta_{1} \\ & + (\frac{2}{71} \quad -\frac{3}{141}) U_{2} + (-\frac{17}{35} \quad -\frac{19}{105}) \theta_{u2} \\ & T_{u24} = (\frac{3}{70} \quad \frac{1}{35}) U_{1} + (\frac{1}{2101} \quad -\frac{1}{424}1) \theta_{u1} \\ & + (-\frac{3}{70} \quad -\frac{1}{35}) U_{u2} + (\frac{4}{1051} \quad -\frac{11}{420}1) \theta_{u2} \\ & T_{u31} = (\frac{1}{351^{2}} \quad -\frac{1}{1051^{2}}) U_{1} + (-\frac{31}{351} \quad -\frac{8}{351}) \theta_{u1} \\ & + (\frac{1}{351^{2}} \quad -\frac{61}{1051^{2}}) U_{1} + (-\frac{3}{70} \quad -\frac{1}{70}) \theta_{1} \\ & + (\frac{3}{351} \quad 0) U_{1} + (-\frac{1}{7} \quad -\frac{3}{70}) \theta_{1} \\ & + (\frac{3}{351} \quad 0) U_{2} + (\frac{2}{35} \quad \frac{3}{70}) \theta_{u2} \\ \end{split}$$

-314-

$$\begin{split} \mathsf{P}_{33} &= (\frac{27}{351^2} - \frac{24}{351^2}) \mathsf{U}_1 + (-\frac{4}{351} - \frac{1}{351}) \mathsf{\theta}_{u1} \\ &+ (-\frac{27}{351^2} - \frac{24}{351^2}) \mathsf{U}_2 + (\frac{31}{351} - \frac{23}{351}) \mathsf{\theta}_{u2} \\ \mathsf{T}_{u34} &= (-\frac{3}{35} - \frac{3}{35}) \mathsf{U}_1 + (\frac{2}{351} - \frac{1}{70}) \mathsf{d}_{u1} \\ &+ (\frac{3}{35} - \frac{6}{70}) \mathsf{U}_{u2} + (-\frac{1}{71} - \frac{1}{10}) \mathsf{\theta}_{u2} \\ \mathsf{T}_{u41} &= (\frac{2}{71} - \frac{1}{141}) \mathsf{U}_1 + (\frac{17}{35} - \frac{13}{210}) \mathsf{\theta}_{u1} \\ &+ (-\frac{2}{71} - \frac{1}{141}) \mathsf{U}_2 + (\frac{3}{70} - \frac{1}{105}) \mathsf{\theta}_{u2} \\ \mathsf{T}_{u42} &= (\frac{3}{70} - \frac{1}{70}) \mathsf{U}_1 + (\frac{4}{1051} - \frac{1}{841}) \mathsf{\theta}_{u1} \\ &+ (-\frac{3}{70} - \frac{1}{70}) \mathsf{U}_{u2} + (\frac{1}{2101} - \frac{1}{420}) \mathsf{\theta}_{u2} \\ \mathsf{T}_{u43} &= (\frac{5}{71} - \frac{22}{351}) \mathsf{U}_2 + (\frac{16}{35} - \frac{43}{210}) \mathsf{\theta}_{1} \\ &+ (-\frac{5}{71} - \frac{22}{351}) \mathsf{U}_2 + (\frac{16}{35} - \frac{49}{210}) \mathsf{\theta}_{u2} \\ \mathsf{T}_{u44} &= (-\frac{2}{35} - \frac{3}{70}) \mathsf{U}_1 + (-\frac{1}{351} - \frac{1}{60}) \mathsf{\theta}_{u1} \\ &+ (\frac{2}{35} - \frac{3}{70}) \mathsf{U}_{u2} + (-\frac{1}{351} - \frac{11}{420}) \mathsf{\theta}_{u2} \\ \mathsf{T}_{u44} &= (-\frac{2}{35} - \frac{3}{70}) \mathsf{U}_{u2} + (-\frac{1}{351} - \frac{11}{420}) \mathsf{\theta}_{u2} \\ \mathsf{T}_{u44} &= (-\frac{2}{35} - \frac{3}{70}) \mathsf{U}_{u2} + (-\frac{1}{351} - \frac{1}{60}) \mathsf{\theta}_{u1} \\ &+ (\frac{2}{35} - \frac{3}{70}) \mathsf{U}_{u2} + (-\frac{1}{351} - \frac{11}{420}) \mathsf{\theta}_{u2} \\ \mathsf{Change the terms u}_{1}, \mathfrak{\theta}_{u1}, \mathfrak{U}_{2}, \mathfrak{\theta}_{u2} \text{ with } \mathsf{T}_{u11}, \mathsf{T}_{u12}, \text{ etc. to} \\ \mathsf{V}_{1}, \mathfrak{\theta}_{v1}, \mathsf{V}_{2}, \mathfrak{\theta}_{v2}, \mathsf{D}_{v11}, \text{ etc. Similar procedure for } \mathsf{T}_{\phi11}, \text{ etc.} \end{split}$$

· . .